

Di Campli San Vito, Patrizia (2021) *Investigation of thermal feedback for incar applications*. PhD thesis.

https://theses.gla.ac.uk/82021/

Copyright and moral rights for this work are retained by the author

A copy can be downloaded for personal non-commercial research or study, without prior permission or charge

This work cannot be reproduced or quoted extensively from without first obtaining permission from the author

The content must not be changed in any way or sold commercially in any format or medium without the formal permission of the author

When referring to this work, full bibliographic details including the author, title, awarding institution and date of the thesis must be given

Enlighten: Theses <u>https://theses.gla.ac.uk/</u> research-enlighten@glasgow.ac.uk

INVESTIGATION OF THERMAL FEEDBACK FOR IN-CAR APPLICATIONS

PATRIZIA DI CAMPLI SAN VITO

SUBMITTED IN FULFILMENT OF THE REQUIREMENTS FOR THE DEGREE OF Doctor of Philosophy

SCHOOL OF COMPUTING SCIENCE

College of Science and Engineering University of Glasgow



February 18, 2021

Abstract

Driving is a highly demanding task and modern cars employ a multitude of sensors and features to aid the driver. Safety can be increased by minimising visual distraction during driving and tactile feedback is often introduced as an alternative to, or an enhancement of, visual icons. Research on tactile in-car feedback is highly focused on vibrotactile feedback, restricting the design space by ignoring other tactile modalities with the potential to increase driving safety. Thermal feedback has been tested for mobile environments and shown high recognition accuracy, with the added advantage of causing potentially strong emotional associations with concepts, such as danger or urgency and familiarity. This thesis, therefore, explores the effectiveness of thermal feedback for in-car applications. The novelty of the feedback within this environment dictates the need to not only investigate the perceptability of this modality as a secondary task during driving, but also the impact of thermal feedback on driving behaviour and workload in a safe environment. Seven driving simulator studies tested different applications and aspects of thermal cues, such as directional cues, binary and in combination with spatial information, and different types of notifications. Results show the challenges and advantages of thermal cue design for presentation during driving and the effectiveness of the modality for navigation. Binary directional cues have high recognition rates, but face the challenge of the return to a neutral base temperature being misinterpreted as new cues. The number of these false positives was especially high for long thermal cues, which had the highest recognition rates. Design choices will have to be made in consideration of this fact. Spatial directional cues were effective, but the simultaneous presentation of cues with opposing direction of temperature change on each hand confused rather than aided the driver. In addition, the perceived urgency of thermal cues was compared to vibration and the two modalities were investigated together for informational notifications. Thermal cues were consistently rated as less urgent than both bimodal and vibrotactile cues. The addition of thermal feedback to urgent vibration cues led to longer reaction times, which renders bimodal tactile stimuli unsuited for urgent warnings. However, they could very accurately convey information to classify messages. Both thermal only and bimodal tactile stimuli had high recognition rates. While thermal feedback could not replace visual information during the transfer of control in a semi-autonomous car, the addition of bimodal tactile feedback led to an improvement of driving behaviour and was preferred by participants. These findings show the potential for thermal feedback within the driving environment.

Acknowledgements

There are quite a few people I want to thank for their support throughout my PhD journey. First and foremost, I want to thank my supervisors Prof. Stephen Brewster and Prof. Frank Pollick not only for providing me with this opportunity, but also for their constant and unwavering support and understanding.

I also want to thank my examiners Prof. Jan van Erp and Dr. Helen Purchase for ignoring my nervousness during the viva and providing constructive and helpful feedback.

This research was partly funded by Jaguar Land Rover and I want to thank them for their support and for welcoming me to their labs and offices during my internship. In particular, I want to thank my supervisors Stuart White, Edward Brown and Simon Thompson for their valuable discussions and for tackling numerable administrative tasks with me.

I would also like to thank my dear colleagues from the Multimodal Interaction Group, past and present. You are an amazing group and made me feel comfortable and welcome from the beginning. A special thank you to Dr. Euan Freeman, Dr. Mark McGill and Dr. Graham Wilson for answering my many, many questions with unwavering patience and good will.

And of course a very heartily thank you to Dr. Gözel Shakeri, who I was not only lucky enough to have as my office neighbour during most of my PhD, but who is my rock in Glasgow and has been from the first time we spoke.

The School of Computing Science at the University of Glasgow is full of amazing and supportive people and I cannot thank all of them, but I want to thank those who have helped me set up and run the PhD Support Group, in particular Fatma Elsafoury, Iulia Paun and Tom Wallis.

And last but not least I want to thank my family all over Europe and my friends, both in the UK and abroad, who have supported me from afar and believed in me. A special thanks to Dr. Monica Pereira for her constant stream of encouragement and of course to Dr. Johannes Georg Klotz for... well, everything. Including, but not limited to: listening to my rants, discussing all kinds of topics with me, finding encouraging words (even when the other kind came easier...) and just basically being there, only a call away.

Declaration and Contributing Papers

The research presented in this thesis is entirely the author's own work. This thesis exploits only the parts of these papers that are directly attributable to the author:

Experiments 1 and 2 in Chapter 3 have been published at AutomotiveUI 2018 and a Special Issue of the International Journal of Mobile Human Computer Interaction:

Di Campli San Vito P., Pollick F., White S., Skrypchuk L., Mouzakitis A., Brewster S.: **Investigation of Thermal Stimuli for Lane Changes**. In *Proceedings of the 10th International ACM Conference on Automotive User Interfaces and Interactive Vehicular Applications - AutoUI '18*, ACM Press, 2018. https://doi.org/10.1145/3239060.3239062

Di Campli San Vito P., Pollick F., White S., Skrypchuk L., Mouzakitis A., Brewster S.: **Thermal Feedback for Simulated Lane Change Scenarios**. In *Special Issue of the International Journal of Mobile Human Computer Interaction (IJMHCI): Recent Advances in Automotive User Interfaces and Interactive Vehicular Applications Research* 11(2), IGI Global, pp. 39-57, 2019. https://doi.org/10.4018/IJMHCI.2019040103

A Demonstration of Experiment 3 in Chapter 4 has been presented and published at ICMI 2017:

Di Campli San Vito P., Pollick F., White S., Brewster S.: **Thermal In-Car Interaction for Navigation**. In *Proceedings of the 19th ACM International Conference on Multimodal Interaction – ICMI '17*, ACM Press, 2017. https://doi.org/10.1145/3136755.3143029

Experiments 3 and 4 in Chapter 4 have been demonstrated, presented and published at CHI 2019:

Di Campli San Vito P., Pollick F., Brown E., Skypchuk L., Mouzakitis A., Brewster S.: **Haptic Navigation Cues on the Steering Wheel**. In *Proceedings of the 37th Annual ACM Conference on Human Factors in Computing Systems – CHI* '19, ACM Press, 2019. https://doi.org/10.1145/3290605.3300440 Experiment 5 and 6 in Chapter 5 have been accepted for publication and will be presented at ICMI 2020:

Di Campli San Vito P., Pollick F., Thompson S., Skypchuk L., Mouzakitis A., Brewster S.: **Purring Wheel: Thermal and Vibrotactile Notifications on the Steering Wheel**. In *Proceedings of the 19th ACM International Conference on Multimodal Interaction – ICMI '20*, ACM Press, 2020. https://doi.org/10.1145/3382507.3418825

Experiment 7 in Chapter 5 has been published and presented as Work in Progress at AutomotiveUI 2020:

Di Campli San Vito P., Brown E., Pollick F., Thompson S., Skypchuk L., Mouzakitis A., Brewster S.: **Haptic Feedback for the Transfer of Control in Autonomous Vehicles**. In *Adjunct Proceedings of the 12th International ACM Conference on Automotive User Interfaces and Interactive Vehicular Applications - AutoUI '20*, ACM Press, 2020. https://doi.org/10.1145/3409251.3411717

Table of Contents

Al	ostrac	:t		i
A	cknow	ledgme	ents	ii
De	eclara	tion and	d Contributing Papers	iv
Li	st of]	Fables		XV
Li	st of I	Figures		xviii
1	Intr	oductio	n	1
	1.1	Motiva	ation	1
	1.2	Thesis	Statement	4
	1.3	Resear	ch Questions	4
	1.4	Thesis	Outline	4
2	Lite	rature l	Review	7
	2.1	Therm	al Perception in Humans	7
		2.1.1	Thermal Receptors and Physiological Effects	8
		2.1.2	Spatial and Temporal Aspects of Thermal Perception	11
		2.1.3	Thermal Phenomena	11
		2.1.4	Summary	12
	2.2	Therm	al Changes as Feedback	13
		2.2.1	Thermal Displays: Characteristics and Design Recommendations .	13
		2.2.2	Affective Computing	17
		2.2.3	Virtual Reality	22

		2.2.4	Stationary and Mobile Application	22
		2.2.5	Thermal-Vibrotactile Feedback	24
		2.2.6	Temperature in Driving	25
		2.2.7	Summary	25
	2.3	Drivin	g Background	26
		2.3.1	Driver Distraction	26
		2.3.2	Autonomous Vehicles	29
		2.3.3	Summary	30
	2.4	Haptic	Feedback for In-Car Applications	31
		2.4.1	Navigation	32
		2.4.2	Notifications	34
		2.4.3	Transfer of Control	36
		2.4.4	Summary	38
	2.5	Literat	cure Review Conclusions	38
3	The	rmal Ri	nary Directional Cues	41
5	3 1	Introdu		 1
	3.1	Appar		-11
	3.2	Experi	ment 1. Thermal I and Change	-12 43
	5.5	2 2 1	Methodology	т <i>э</i> 43
		222	Driving Task and Faadhaak	43
		222	Driving Task and Procedure	44
		224		43
		3.3.4		47
		226	Experiment 1 Discussion	40 54
	2.4	5.5.0	mant 2: Different Stimuli Designs	56
	5.4	2 4 1	Methodology	56
		5.4.1 2.4.2	Driving Tools and Foodbools	50
		5.4.2	Driving Task and Frequence	51
		5.4.5		58
		5.4.4		58
		3.4.5	Results	- 59

		3.4.6	Experiment 2 Discussion	65
	3.5	Chapte	er 3 Discussion	66
	3.6	Conclu	usions and Research Question 1	67
4	The	rmal Di	rectional Cues with Multiple Devices on the Steering Wheel	69
	4.1	Introdu	uction	69
	4.2	Experi	ment 3: Thermal Navigation	69
		4.2.1	Methodology	70
		4.2.2	Apparatus	70
		4.2.3	Driving Task and Feedback	71
		4.2.4	Participants and Procedure	74
		4.2.5	Measures	75
		4.2.6	Results	76
		4.2.7	Experiment 3 Discussion	79
	4.3	Experi	ment 4: Haptic Navigation	81
		4.3.1	Methodology	81
		4.3.2	Apparatus	82
		4.3.3	Driving Task and Feedback	83
		4.3.4	Participants and Procedure	84
		4.3.5	Measures	85
		4.3.6	Results	86
		4.3.7	Experiment 4 Discussion	91
	4.4	Chapte	er 4 Discussion	92
	4.5	Conclu	usions and Research Question 2	93
5	Uni	nodal a	nd Multimodal Notifications with Thermal Feedback	94
	5.1	Introdu	uction	94
	5.2	Experi	ment 5: Uni- and Bimodal Tactile Notifications with Different Levels	
		of Urg	ency	95
		5.2.1	Methodology	95
		5.2.2	Apparatus	96

		5.2.3	Driving Task and Feedback	97
		5.2.4	Participants and Procedure	99
		5.2.5	Measures	101
		5.2.6	Results	101
		5.2.7	Experiment 5 Discussion	116
	5.3	Experi	ment 6: Uni- and Bimodal Tactile Notifications with Different Types	
		of Info	ormation	117
		5.3.1	Methodology	117
		5.3.2	Apparatus	118
		5.3.3	Driving Task and Feedback	118
		5.3.4	Participants and Procedure	120
		5.3.5	Measures	121
		5.3.6	Results	121
		5.3.7	Experiment 6 Discussion	128
	5.4	Experi	iment 7: Multimodal Feedback for Transfer of Control in Semi-Autonom	nous
		Cars .		130
		5.4.1	Methodology	130
		5.4.2	Apparatus	131
		5.4.3	Driving Task and Feedback	133
		5.4.4	Participants and Procedure	135
		5.4.5	Measures	135
		5.4.6	Results	137
		5.4.7	Experiment 7 Discussion	142
	5.5	Chapte	er 5 Discussion	144
	5.6	Conclu	usions and Research Question 3	146
6	Con	alucion		147
U		Desse	s	147
	0.1	Reseat	reh Question 2	14/
	0.2	Resear		148
	6.3	Kesear	Con Question 3	149
	6.4	Contri	butions and Recommendations	150

	6.5	Limita	tions and Future Work	152
		6.5.1	Simulator Studies	152
		6.5.2	Simple Driving Tasks	152
		6.5.3	Selection of Use Cases	152
		6.5.4	Pre-Set Thermal Cues	153
		6.5.5	Different Thermal Devices	153
		6.5.6	Influence of Hand Placement	153
		6.5.7	Limited Location	154
		6.5.8	Short Duration Studies	154
		6.5.9	Participant Demographic and Number	154
	6.6	Conclu	usions	154
A	Арр	endix		156
	A.1	Hardw	are Manuals	156
		A.1.1	Thermal Bluetooth Board Manual	156
		A.1.2	Thermal USB Board Manual	172
	A.2	Experi	ment 1	188
		A.2.1	Experiment 1: Information Sheet	188
		A.2.2	Experiment 1: Consent Form	191
		A.2.3	Experiment 1: Questionnaires	193
	A.3	Experi	ment 2	198
		A.3.1	Experiment 2: Information Sheet	198
		A.3.2	Experiment 2: Consent Form	201
		A.3.3	Experiment 2: Questionnaires	203
	A.4	Experi	ment 3	205
		A.4.1	Experiment 3: Information Sheet	205
		A.4.2	Experiment 3: Consent Form	208
		A.4.3	Experiment 3: Questionnaires	210
	A.5	Experi	ment 4	215
		A.5.1	Experiment 4: Information Sheet	215
		A.5.2	Experiment 4: Consent Form	218

	A.5.3	Experiment 4: Questionnaires	220
A.6	Experi	ment 5	228
	A.6.1	Experiment 5: Information Sheet	228
	A.6.2	Experiment 5: Consent Form	231
	A.6.3	Experiment 5: Questionnaires	233
	A.6.4	Experiment 5: Statistics of Subjective Ratings with 0 Ratings Include	d236
	A.6.5	Experiment 5: Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings without 0 ratings; significant results and discussion can be found in section 5.2.6	240
	A.6.6	Experiment 5: Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for for Missed Stimuli (<i>0</i> Ratings) of the Subjective Rating; significant results and discussion can be found in section 5.2.6	243
	A.6.7	Experiment 5: Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Recognition Rate; significant results and discussion can be found in section 5.2.6	246
	A.6.8	Experiment 5: Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Recognition Time; significant results and discussion can be found in section 5.2.6	249
A.7	Experi	ment 6	252
	A.7.1	Experiment 6: Information Sheet	252
	A.7.2	Experiment 6: Consent Form	255
	A.7.3	Experiment 6: Questionnaires	257
A.8	Experi	ment 7	262
	A.8.1	Experiment 7: Information Sheet	262
	A.8.2	Experiment 7: Consent Form	265
	A.8.3	Experiment 7: Questionnaires	267

Bibliography

List of Tables

1.1	Experiment Overview: Describing the modalities (in italic), goals and re- search questions posed for each experiment presented in this thesis.	6
3.1	Experiment 1: Detailed participant data, showing range (R) , mean (M) and standard deviation (SD) .	46
3.2	Experiment 1: Statistics for Comparison of Lane Deviation, captured during the first part of the experiment.	49
3.3	Experiment 1: Statistics for NASA TLX rating of workload; significant re- sults are bold and marked with an asterisk.	51
3.4	Experiment 1: Statistics for Subjective Additional Ratings; significant results are bold and marked with an asterisk.	54
3.5	Experiment 2: Detailed participant data, showing range (R) , mean (M) and standard deviation (SD) .	58
3.6	Experiment 2: Statistics for Recognition Rates for all participants; significant results are bold and marked with an asterisk.	60
3.7	Experiment 2: Statistics for Recognition Rates with Gender as between sub- ject factor; significant results are bold and marked with an asterisk	61
3.8	Experiment 2: Statistics for Recognition Rate <i>post hoc</i> tests for the Interac- tion between LEN and EXT for warm stimuli of all participants; significant results are bold and marked with an asterisk.	61
3.9	Experiment 2: Statistics for False Positives for all participants; significant results are bold and marked with an asterisk.	63
3.10	Experiment 2: Statistics for False Positives with Gender as between subject factor; significant results are bold and marked with an asterisk.	63
3.11	Experiment 2: Statistics for Time to Complete Lane Change for all participants; significant results are bold and marked with an asterisk.	65

2 1 2	Experiment 2. Statistics for Time to Complete Long Change with Conder or	
3.12	between subject factor; significant results are bold and marked with an asterisk.	65
4.1	Experiment 3: Participant data, showing range (R) , mean (M) and standard deviation (SD) .	74
4.2	Experiment 3: Statistics for NASA TLX rating of workload; significant re- sults are bold and marked with an asterisk.	77
4.3	Experiment 3: Statistics for Subjective Additional Ratings; significant re- sults are bold and marked with an asterisk.	78
4.4	Experiment 3: Statistics for Deviation from Ideal Path; significant results are bold and marked with an asterisk.	79
4.5	Experiment 4: Participant data, showing range (R) , mean (M) and standard deviation (SD) .	85
4.6	Experiment 4: Statistics for NASA TLX rating of workload; significant re- sults are bold and marked with an asterisk.	86
4.7	Experiment 4: Statistics for Subjective Additional Ratings; significant re- sults are bold and marked with an asterisk.	87
4.8	Experiment 4: Statistics for Deviation from Ideal Path; significant results are bold and marked with an asterisk.	89
4.9	Experiment 3 and 4: Statistics for the Comparison of NASA TLX rating of workload; significant results are bold and marked with an asterisk.	90
4.10	Experiment 3 and 4: Statistics for Comparison of Subjective Additional Rat- ings; significant results are bold and marked with an asterisk.	90
5.1	Experiment 5: Participant data, showing range (R) , mean (M) and standard deviation (SD) .	100
5.2	Experiment 5: Statistics of <i>post hoc</i> test results for Modality for Subjective Ratings without 0 ratings; significant results are bold and marked with an	
5.3	asterisk	103 104
5.4	Experiment 5: Statistics of <i>post hoc</i> test results for Modality for <i>0</i> Ratings; significant results are bold and marked with an asterisk	106

5.5	Experiment 5: Heat Map of <i>post hoc</i> tests for the Interaction between Ur- gency and Modality for Missed Stimuli (<i>0</i> Ratings) during the Subjective Rating, all results can be found in Appendix A.6.6; <u>Low</u> , <u>Medium and Urgent</u> notifications for <u>B</u> imodal, <u>T</u> hermal, <u>C</u> old and <u>W</u> arm, and <u>V</u> ibration are coloure	ed
	yellow ($p < 0.05$), orange ($p < 0.001$) and red ($p < 0.0001$)	107
5.6	Experiment 5: Statistics of <i>post hoc</i> test results for Modality for Recognition Rate; significant results are bold and marked with an asterisk	109
5.7	Experiment 5: Heat Map of <i>post hoc</i> tests for the Interaction between Ur- gency and Modality for Recognition Rate, all results can be found in Ap- pendix A.6.7; <u>Low</u> , <u>M</u> edium and <u>U</u> rgent notifications for <u>B</u> imodal, <u>T</u> hermal, <u>C</u> old and <u>W</u> arm, and <u>V</u> ibration are coloured yellow ($p < 0.05$), orange ($p < 0.001$) and red ($p < 0.0001$)	110
5.8	Experiment 5: Statistics of <i>post hoc</i> test results for Modality for Recognition Time; significant results are bold and marked with an asterisk.	111
5.9	Experiment 5: Significant results of <i>post hoc</i> tests for the Interaction be- tween Urgency and Modality for Recognition Time, all results can be found in Appendix A.6.8.	113
5.10	Experiment 6: Participant data, showing range (R) , mean (M) and standard deviation (SD) .	121
5.11	Experiment 6: Statistics for Recognition Rate; significant results are bold and marked with an asterisk.	122
5.12	Experiment 6: Statistics of <i>post hoc</i> test results of Recognition Rate for the interaction between Modality and Importance; significant results are bold and marked with an asterisk.	123
5.13	Experiment 6: Statistics of <i>post hoc</i> test results of Recognition Rate for the interaction between Modality and Nature; significant results are bold and marked with an asterisk.	123
5.14	Experiment 6: Statistics of <i>post hoc</i> test results of Recognition Rate for the interaction between Importance and Nature; significant results are bold and marked with an asterisk.	124
5.15	Experiment 6: Statistics of <i>post hoc</i> test results of Recognition Rate for the interaction between Modality, Importance and Nature; significant results are bold and marked with an asterisk.	124
5.16	Experiment 6: Statistics for Lane Deviation; significant results are bold and marked with an asterisk.	125

Experiment 7: Participant data, showing range (R) , mean (M) and standard deviation (SD) .	135
Experiment 7: Statistics for Time Observations; significant results are bold and marked with an asterisk.	137
Experiment 7: Statistics for the Comparison of Time Observations between Buttons and Gear Paddles; significant results are bold and marked with an asterisk.	142
Experiment 7: Statistics for the Comparison of Subjective Data between But- tons and Gear Paddles; significant results are bold and marked with an asterisk	k. 142
Design recommendations overview collected from the experimental findings of this thesis.	151
Experiment 5: Statistics for the Subjective Ratings including <i>0</i> Ratings; significant results are bold and marked with an asterisk.	236
Experiment 5: Statistics of <i>post hoc</i> test results for Urgency for Subjective Ratings with 0; significant results are bold and marked with an asterisk	236
Experiment 5: Statistics of <i>post hoc</i> test results for Modality for Subjective Ratings with 0; significant results are bold and marked with an asterisk	236
Experiment 5: Part 1 Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings with <i>0</i> ratings included; significant results are bold and marked with an asterisk.	237
Experiment 5: Part 2 of Statistics of <i>post hoc</i> tests for the Interaction be- tween Urgency and Modality for Subjective Ratings with 0 ratings included; significant results are bold and marked with an asterisk.	238
Experiment 5: Part 3 of Statistics of <i>post hoc</i> tests for the Interaction be- tween Urgency and Modality for Subjective Ratings with 0 ratings included; significant results are bold and marked with an asterisk.	239
Experiment 5: Part 1 Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings without <i>0</i> ratings included; significant results are bold and marked with an asterisk.	240
Experiment 5: Part 2 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings without <i>0</i> ratings included; significant results are bold and marked with an asterisk.	241
	 Experiment 7: Participant data, showing range (R), mean (M) and standard deviation (SD). Experiment 7: Statistics for Time Observations; significant results are bold and marked with an asterisk. Experiment 7: Statistics for the Comparison of Time Observations between Buttons and Gear Paddles; significant results are bold and marked with an asterisk. Experiment 7: Statistics for the Comparison of Subjective Data between Buttons and Gear Paddles; significant results are bold and marked with an asterisk. Experiment 7: Statistics for the Comparison of Subjective Data between Buttons and Gear Paddles; significant results are bold and marked with an asterisk Design recommendations overview collected from the experimental findings of this thesis. Experiment 5: Statistics for the Subjective Ratings including <i>0</i> Ratings; significant results are bold and marked with an asterisk. Experiment 5: Statistics of <i>post hoc</i> test results for Urgency for Subjective Ratings with 0; significant results are bold and marked with an asterisk. Experiment 5: Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings with 0 ratings included; significant results are bold and marked with an asterisk. Experiment 5: Part 2 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings with 0 ratings included; significant results are bold and marked with an asterisk. Experiment 5: Part 3 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings with 0 ratings included; significant results are bold and marked with an asterisk. Experiment 5: Part 1 Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings with 0 ratings included; significant results are bold and marked with an asterisk. Experiment 5: Part 1 Statistics of <i>post hoc</i> tests for the Intera

A.9 Experiment 5: Part 3 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Subjective Ratings without 0 ratings included; significant results are bold and marked with an asterisk.	242
A.10 Experiment 5: Part 1 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Missed Stimuli (0 Ratings) of the Subjective Rating; significant results are bold and marked with an asterisk.	243
A.11 Experiment 5: Part 2 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Missed Stimuli (0 Ratings) of the Subjective Rating; significant results are bold and marked with an asterisk.	244
A.12 Experiment 5: Part 3 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for for Missed Stimuli (0 Ratings) of the Subjective Rating; significant results are bold and marked with an asterisk	245
A.13 Experiment 5: Part 1 Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Recognition Rate; significant results are bold and marked with an asterisk.	246
A.14 Experiment 5: Part 2 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Recognition Rate; significant results are bold and marked with an asterisk.	247
A.15 Experiment 5: Part 3 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Recognition Rate; significant results are bold and marked with an asterisk.	248
A.16 Experiment 5: Part 1 Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Recognition Time; significant results are bold and marked with an asterisk.	249
A.17 Experiment 5: Part 2 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Recognition Time; significant results are bold and marked with an asterisk.	250
A.18 Experiment 5: Part 3 of Statistics of <i>post hoc</i> tests for the Interaction between Urgency and Modality for Recognition Time; significant results are bold and marked with an asterisk.	251

List of Figures

2.1	Thermal sensitivity over the body: warm (upper section) and cold (bottom section); smaller bars indicate higher sensibility (extracted from [1])	9
2.2	Thenar eminence (red) of the hand, the most temperature sensitive part of the inner hand; finger tips (light blue) are less sensitive	10
2.3	Peltier device (white) attached to heat-sink (black).	16
2.4	SAE Levels of Autonomy.	29
3.1	Experiment 1 and 2: Hardware and experimental setup	42
3.2	Empty five lane motorway, implemented in OpenDS	43
3.3	Experiment 1: Thermal feedback.	45
3.4	Experiment 1: Time to Complete Lane Change; error bars always show the standard error.	50
3.5	Experiment 1: Recognition Rate; significant results are in bold font	51
3.6	Experiment 1: Perceived workload; significant results are in bold font	52
3.7	Experiment 1: Subjective Additional Ratings; significant results are in bold	
	font	53
3.8	Experiment 1: Final rating for both feedback types	53
3.9	Experiment 2: Thermal feedback.	57
3.10	Experiment 2: Recognition Rate for all Cold Stimuli.	60
3.11	Experiment 2: Recognition Rate for all Warm Stimuli.	60
3.12	Experiment 2: False Positives for all Cold Stimuli.	62
3.13	Experiment 2: False Positives for all Warm Stimuli.	62
3.14	Experiment 2: Time to Complete Lane Change for all Cold Stimuli	64
3.15	Experiment 2: Time to Complete Lane Change for all Warm Stimuli	64
4.1	Experiment 3: Hardware.	71

4.2	Experiment 3: Steering Wheel and experimental setup.	71
4.3	Experiment 3: Simulator environment, implemented in OpenDS	72
4.4	Experiment 3: Thermal feedback.	74
4.5	Experiment 3: Perceived workload; significant results are in bold font	77
4.6	Experiment 3: Subjective Additional Ratings; significant results are in bold	
	font	78
4.7	Experiment 3: Deviation from Ideal Path; significant results are in bold font.	80
4.8	Experiment 4: Steering Wheel.	82
4.9	Experiment 4: Solenoid embedded into the Steering Wheel for Cutaneous	
	Push Feedback.	83
4.10	Experiment 4: Peltier device and setup.	83
4.11	Experiment 4: Thermal feedback.	84
4.12	Experiment 4: Perceived workload; significant results are in bold font	87
4.13	Experiment 4: Subjective Additional Ratings; significant results are in bold font.	88
	E-main and A. Daristian from Ideal Dath significant models and in hold fort	00
4.14	Experiment 4: Deviation from Ideal Path; significant results are in bold font.	90
4.145.1	Experiment 4: Deviation from Ideal Path; significant results are in bold font. Experiment 5: Hardware and Steering Wheel.	90 96
4.145.15.2	Experiment 4: Deviation from Ideal Path; significant results are in bold font. Experiment 5: Hardware and Steering Wheel	90 96 96
4.145.15.25.3	Experiment 4: Deviation from Ideal Path; significant results are in bold font. Experiment 5: Hardware and Steering Wheel	90 96 96 97
4.14 5.1 5.2 5.3 5.4	Experiment 4: Deviation from Ideal Path; significant results are in bold font. Experiment 5: Hardware and Steering Wheel. Experiment 5: Interface for Subjective Rating in Part 1. Experiment 5: Experimental setup. Experiment 5: Simulator Environment, arrow on gantry indicating the desti-	90 96 96 97
4.14 5.1 5.2 5.3 5.4	Experiment 4: Deviation from Ideal Path; significant results are in bold font. Experiment 5: Hardware and Steering Wheel. Experiment 5: Interface for Subjective Rating in Part 1. Experiment 5: Experimental setup. Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change.	90 96 96 97 98
 4.14 5.1 5.2 5.3 5.4 5.5 	Experiment 4: Deviation from Ideal Path; significant results are in bold font.Experiment 5: Hardware and Steering Wheel.Experiment 5: Interface for Subjective Rating in Part 1.Experiment 5: Experimental setup.Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change.Experiment 5: Vibration patterns.	90 96 96 97 98 98
 4.14 5.1 5.2 5.3 5.4 5.5 5.6 	Experiment 4: Deviation from Ideal Path; significant results are in bold font.Experiment 5: Hardware and Steering Wheel.Experiment 5: Interface for Subjective Rating in Part 1.Experiment 5: Experimental setup.Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change.Experiment 5: Vibration patterns.Experiment 5: Thermal feedback.	90 96 97 98 99 100
4.14 5.1 5.2 5.3 5.4 5.5 5.6 5.7	Experiment 4: Deviation from Ideal Path; significant results are in bold ront.Experiment 5: Hardware and Steering Wheel.Experiment 5: Interface for Subjective Rating in Part 1.Experiment 5: Experimental setup.Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change.Experiment 5: Vibration patterns.Experiment 5: Thermal feedback.Experiment 5: Subjective Rating with and without 0 Ratings included; significant results are in bold font.	90 96 97 98 99 100
 4.14 5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 	Experiment 4: Deviation from Ideal Path; significant results are in bold font.Experiment 5: Hardware and Steering Wheel.Experiment 5: Interface for Subjective Rating in Part 1.Experiment 5: Experimental setup.Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change.Experiment 5: Vibration patterns.Experiment 5: Number of Missed Stimuli (0 Ratings) during the Subjective Rating.	90 96 97 98 99 100 102 105
 4.14 5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 	Experiment 4: Deviation from Ideal Path; significant results are in bold font.Experiment 5: Hardware and Steering Wheel.Experiment 5: Interface for Subjective Rating in Part 1.Experiment 5: Experimental setup.Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change.Experiment 5: Vibration patterns.Experiment 5: Thermal feedback.Experiment 5: Subjective Rating with and without 0 Ratings included; significant results are in bold font.Experiment 5: Number of Missed Stimuli (0 Ratings) during the Subjective Rating.Experiment 5: Recognition Rate.	90 96 97 98 99 100 102 105 108
4.14 5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10	Experiment 4: Deviation from Ideal Path; significant results are in bold ront.Experiment 5: Hardware and Steering Wheel.Experiment 5: Interface for Subjective Rating in Part 1.Experiment 5: Experimental setup.Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change.Experiment 5: Vibration patterns.Experiment 5: Thermal feedback.Experiment 5: Subjective Rating with and without 0 Ratings included; significant results are in bold font.Experiment 5: Number of Missed Stimuli (0 Ratings) during the Subjective Rating.Experiment 5: Recognition Rate.Experiment 5: Recognition: All Rating in Numbers.	 90 96 96 97 98 99 100 102 105 108 111
 4.14 5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 	Experiment 4: Deviation from Ideal Path; significant results are in bold font. Experiment 5: Hardware and Steering Wheel. Experiment 5: Interface for Subjective Rating in Part 1. Experiment 5: Experimental setup. Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change. Experiment 5: Vibration patterns. Experiment 5: Nubjective Rating with and without 0 Ratings included; significant results are in bold font. Experiment 5: Number of Missed Stimuli (0 Ratings) during the Subjective Rating. Experiment 5: Recognition Rate. Experiment 5: Recognition Time.	 90 96 96 97 98 99 100 102 105 108 111 112
 4.14 5.1 5.2 5.3 5.4 5.5 5.6 5.7 5.8 5.9 5.10 5.11 5.12 	Experiment 4: Deviation from Ideal Path; significant results are in bold ront. Experiment 5: Hardware and Steering Wheel. Experiment 5: Interface for Subjective Rating in Part 1. Experiment 5: Experimental setup. Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change. Experiment 5: Vibration patterns. Experiment 5: Number of Meedback. Experiment 5: Number of Missed Stimuli (<i>O</i> Ratings) during the Subjective Rating. Experiment 5: Recognition Rate. Experiment 5: Recognition Time.	 90 96 96 97 98 99 100 102 105 108 111 112 114

5.13	Experiment 6: Steering Wheel.	118
5.14	Experiment 6: High and Low Importance Vibration Patterns	119
5.15	Experiment 6: Thermal feedback.	120
5.16	Experiment 6: Recognition Rate	122
5.17	Experiment 6: Lane Deviation.	125
5.18	Experiment 6: All Subjective Ratings.	126
5.19	Experiment 6: Ranking of Stimuli	127
5.20	Experiment 7: Hardware and experimental setup	132
5.21	Experiment 7: Driving simulator environment as captured in the video log. The side mirror content was presented to drivers on screens at the side of the driving cabin.	133
5.22	Experiment 7: Thermal feedback: Handover (left) and Takeover (right)	134
5.23	Experiment 7: Time Observations (including Buttons)	138
5.24	Experiment 7: Perceived workload; significant results are in bold font	139
5.25	Experiment 7: Additional Subjective and Trust Ratings; significant results are in bold font.	140
5.26	Experiment 7: Ranking of Conditions.	141

Chapter 1

Introduction

1.1 Motivation

Modern cars utilise a plethora of sensors to improve driving safety and driving experience. Some of these sensors provide information on the state of the world and the car either presents this information to the driver, for example by turning on an icon on the dashboard when the temperature outside sinks under 4°C, or takes immediate action by for example turning on the windscreen wipers when it rains or the lights when it gets dark. Others collect and report data about the state of the car, such as air pressure of the tyres or fuel and oil levels. More recently, car manufacturers have started to embed sensors to observe driving related information, mostly to improve driving safety or efficiency, as for example through lane departure warnings and gear changing instructions. In addition, passengers and drivers alike regularly use the car's integrated entertainment system and even connect personal devices with it, resulting in more communication between the car and its occupants.

The state of the car is typically conveyed through predominantly visual notifications: icons on the dashboard, on the screen in the centre console or on head up displays. However, driving is a highly visual task and the eyes should be fixed on the road ahead, with eyesoff-the-road time considered to be one of the main contributors to crashes and incidents on the road and especially long glances over 2s being safety critical [2]. For many occasions, auditory feedback, as abstract alerts or language-based, are used to lessen the visual and mental workload by providing information in a modality different to that used for the primary task, as suggested in the Multiple Resources Theory [3]. Examples for abstract auditory feedback are pings when the seat-belt is not fastened or when warning icons appear for the first time. Language-based feedback might be navigation instructions. However, audio feedback is not desirable or applicable in all situations: audio feedback can be disruptive when drivers are engaged in conversation or listen to radio programmes, and deaf drivers cannot benefit from this feedback at all. These issues can be overcome by using feedback engaging the sense of touch: haptic feedback. The most common haptic modality used in cars is vibrotactile feedback. Some car manufacturers, such as Honda¹, have integrated vibration on the steering wheel to warn drivers about an imminent lane departure. Research has shown that this feedback can be interpreted without any prior training [4] and can reduce the number of road crashes [5].

Other kinds of in-car applications have also been explored utilising vibrotactile feedback. Navigational cues, for example, have been given through vibration on the seat [6, 7], seat belt [8, 9] or steering wheel [10, 11, 12]. Different haptic modalities such as shear or skin stretching feedback [13, 14] have also been shown to be effective in providing directional information. In addition to lane departure warnings [4, 15, 16], haptic feedback was also investigated for collision warnings [17, 18, 19] and for providing information after a transfer of control back to the driver for autonomously driving cars [20, 21, 22]. In those cases, information is mostly given in a multimodal setting, adding vibrotactile feedback to audio and/or visual cues, as they are usually highly urgent and drivers have to react to them quickly and accurately. Multimodal feedback, especially in combination with vibration, can decrease reaction times [23, 24, 25] and increase the feeling of urgency [26].

This research shows that vibration as a feedback type has many advantages in the driving environment, but there are caveats. Most of the research has been conducted in labs, simulating the car's environment. The influence of real life driving on differing road conditions has not been thoroughly investigated. Naturalistic road situations can include speed bumps, potholes and raised profile edge lines on motorways, which can influence the vibration of the car substantially. In addition, the choice of car and its behaviour to those differing road conditions could have further significant influence on the perceptability of vibrotactile cues during driving. Multimodal feedback could help overcome these issues. Therefore, the investigation of other tactile feedback that could enhance or even replace vibrotactile feedback in a car is advisable.

One such tactile feedback, which has been shown to be effective in bumpy and noisy environments [27], but not yet been investigated for in-car use, is thermal feedback. There has been a rise in research of temperature changes as feedback in human computer interaction over the last few decades. While research has been, and is continually, scoping properties of thermal displays and defining guidelines [28, 29, 30], thermal feedback has found its most popular use case in virtual reality settings. The thermal sense plays an important role in the human's ability to differentiate textures [31, 32] and has been engaged to enhance the feeling of touching and exploring objects in virtual reality [33, 34]. Additionally, temperature has been used to increase immersion into the virtual world by assigning appropriate thermal properties to surroundings [35, 36, 37] and objects [38, 39].

¹https://www.silkohonda.com/manufacturer-information/honda-sensing/ (accessed 13/12/2019)

Outside of virtual reality, thermal feedback was tested for mobile [40, 29, 41] and stationary [42, 43, 27] use. The investigations ranged from exploring perceptability and influences of different aspects of the thermal stimuli in specific environments, over affective associations with temperatures and use cases for those, to using thermal icons to convey specific (or abstract) information, as well as combinations of those.

Thermal icons have shown high recognition rates for mobile interaction, especially for direction of temperature change (warm/cool), but also for subjective intensity, combining the extent and rate of temperature change [44, 45]. In addition, emotional, or affective, associations with temperatures have been found [46, 47, 48]. While some findings seem contradictory [49], showing that the mapping of temperatures to emotions can be complex and dependent on the context, the shared associations between individuals can be quite strong.

The effect of ambient temperature and clothing on temperature perception has been studied as well [50, 51]. In the well-defined interior of a vehicle, environmental influences can be easily captured and the thermal stimuli potentially adapted. During manual driving, the cues could be presented on the hands, unobstructed by any material. Even in the case of material obstructing direct touch (by wearing gloves, for example), sensors could be embedded into the steering wheel to determine what kind of material is in direct contact with the wheel. Steering wheels already have warming abilities in many modern cars, providing comfort to drivers and passengers alike and adding some familiarity to the concept. The concept of cooling steering wheels, however, is novel.

Before thermal feedback can be used during driving, its safety and effectiveness in this new environment have to be tested. Previous research investigated thermal perception as a main task, with participants' main focus being on the thermal cues. But driving is a highly demanding primary task, which pushes the perception of other cues into the domain of a secondary task. How well will thermal cues be recognised under those conditions? Additionally, distractions from the driving task can have grave and dangerous consequences. Before introducing new stimuli types to this environment, it has to be ensured that they do not negatively influence the driving behaviour. The influence of new modalities, therefore, needs to be tested in a safe environment.

Thermal feedback has the potential to increase driving safety, by minimising the visual load through tactile presentation of information. Haptic feedback in the car is already used to decrease the pressure of visual feedback, but the design space is limited. Thermal feedback can present binary information, such as direction of temperature change, or could be presented on different locations and add spatial information. It could also be combined with vibration and used as bimodal tactile feedback. Temperature has a strong connection to emotion and could enrich in-car feedback with a modality that can be used to design easily understand-able notifications with affective associations. The influence of thermal feedback on driver

and driving behaviour is, however, unknown and there is a pressing need to understand how best to utilise this modality in the car and understand possible consequences during driving. This thesis, therefore, investigates the effectiveness of thermal feedback for use in vehicular environments.

1.2 Thesis Statement

Haptic feedback within cars is used to reduce visual distraction during driving, one of the main contributors to crashes. This research explores aspects and applications of thermal feedback for in-car use to convey information to the driver non-visually. Results show that thermal cues can be used to convey information to the driver accurately and non-urgently and its effectiveness can be enhanced through multimodal combination with vibrotactile feedback and offers a rich, new modality to broaden the design space of tactile in-car feedback.

1.3 Research Questions

This thesis aims to answer following research questions:

1. Research Question 1:

How accurately can thermal feedback give binary direction cues in an automotive setting (with one thermal device)?

2. Research Question 2:

How accurately can thermal feedback give direction cues in an automotive setting with the spatial information from multiple thermal devices?

3. Research Question 3:

How effectively can thermal feedback convey notifications in a car in a a. unimodal setting?

b. bimodal tactile setting?

1.4 Thesis Outline

Chapter 2, *Literature Review*, reviews literature on thermal perception and thermal cues in human computer interaction. Furthermore, basic concepts needed to investigate interaction in the driving context are discussed, as well as research on the use of haptic feedback within

1.4. Thesis Outline

the driving environment. The following chapters discuss the experimental work, an experiment overview is provided in Table 1.1.

In Chapter 3, *Thermal Binary Directional Cues*, Research Question 1 is investigated with two experiments. Both studies presented thermal directional cues on one device, where the direction of temperature change prompted the direction of a lane change. The first experiment, Experiment 1, compared thermal to audio feedback. Experiment 2 explored several factors of the thermal feedback design.

Chapter 4, *Thermal Directional Cues with Multiple Devices on the Steering Wheel*, expands on ideas from the previous chapter and presents navigation information on multiple thermal devices. Experiment 3 compares thermal to audio navigation, Experiment 4 to cutaneous push feedback. These two studies aimed to answer Research Question 2.

Chapter 5, *Unimodal and Multimodal Notifications with Thermal Feedback*, presents three experiments looking into thermal and bimodal tactile notifications. Experiments 5 and 6 answered both parts of Research Question 3 (a. and b.), while Experiment 7 only investigated bimodal aspects (3 b.). While in Experiment 5, the same information (urgency) was presented in the different modalities, Experiment 6 looked at different types of information (Nature and Importance). Experiment 7 explored the benefits of bimodal tactile feedback in a multimodal setting for transfer of control in semi-autonomous cars.

Chapter 6, *Conclusions*, will end the thesis with a summary and discussion of all findings. Additionally, limitations and ideas for future work are proposed. Finally, conclusions and recommendations are presented.

Experiment	Chapter	Research	Modalities / Goals
		Question	
Experiment 1	3	1	thermal vs. audio
			Compare recognition, driving behaviour, time and
			subjective data between thermal and audio cues
Experiment 2	3	1	thermal
			Explore influence of several factors of stimuli
			design on recognition, false positives and time
Experiment 3	4	2	thermal vs. audio
			Compare effectiveness, subjective and driving data of
			thermal and audio cues for turn-by-turn navigation
Experiment 4	4	2	thermal vs. cutaneous push
			Compare effectiveness, subjective and driving data of
			thermal and cutaneous push cues for turn-by-turn
			navigation
			Compare subjective data with Experiment 3
Experiment 5	5	3 a./b.	thermal and/or vibrotactile
			Compare thermal, vibrotactile and bimodal stimuli
			and their influence on perception of notification
			urgency and recognition rate and time
Experiment 6	5	3 a./b.	thermal vs. thermal and vibrotactile
			Compare recognition and preferences of thermal
			and bimodal stimuli for notifications with
			two types of information
Experiment 7	5	3 b.	thermal and vibrotactile,
			accompanied by audio and visual
			Compare effect of bimodal feedback on gaze, trust
			and preferences for control transfer of autonomous
			cars

Table 1.1: Experiment Overview: Describing the modalities (in italic), goals and research questions posed for each experiment presented in this thesis.

Chapter 2

Literature Review

This chapter will describe research needed as basis to answer the questions posed in the previous chapter. In order to use thermal cues in interaction with humans, temperature perception and characteristics of the human skin have to be understood. Section 2.1 will, therefore, provide insights into basic human thermal sensing, followed by an overview of thermal feedback use in human computer interaction in Section 2.2. The second half of the chapter will look into driving related issues. Section 2.3 with provide basics on driving related research and Section 2.4 will concentrate on haptic feedback within the driving environment, to inspire how and for what thermal feedback might be most suitable and effectively used.

2.1 Thermal Perception in Humans

The human body depends on homeostasis, meaning that the core temperature of the body needs to be kept at a relatively stable temperature of around 37°C. Variations in skin temperature, however, occur by simply touching an object [31] and can vary up to 12°C [32]. Thermal comfort describes how satisfied an individual is within their thermal environment. Research has shown that gender can have a significant influence on the hedonistic experience of surrounding temperatures [52, 53] and on the sensitivity of warm temperatures [54], but when temperature was used for interaction, gender effects were found to be negligible [55]. The feeling of temperature comfort is influenced by the core temperature of a person, however, the perception of temperature changes was found to be independent of it [56, 57]. In human computer interaction, thermal feedback is presented peripherally on the skin. The following literature discussion, therefore, will focus on cutaneous thermal perception, temperature perception of the skin. First, basic human thermal sensing is described, followed by specific aspects of thermal perception.

2.1.1 Thermal Receptors and Physiological Effects

The human skin perceives temperature through two types of thermal receptors on free nerve endings: C-fibres, with unmyelinated axons, react to warm temperature changes, while myelinated axons of A δ -fibres react to cold temperatures [58, 59]. The density of thermal receptors in the skin varies on body location and type of receptor. Small locations of the body that can detect warm or cold temperatures changes are called warm or cold spots, respectively, and encompass only a few millimetres. These are dispersed over the body independent of each other and in differing numbers. More cold than warm spots have been identified [60], with, for example, seven cold spots per 100m² on the forearm, but only 0.24 warm spots [32]. In addition to the higher number of cold spots in the skin, cold receptors react faster to temperature changes, with a conduction velocity of 12-30m/s, while warm receptors show a velocity of 0.5-2m/s [59]. In comparison, vibration, detected by Pacinian corpuscles embedded in the subcutaneous layers of the skin, has a conduction velocity of 30-70m/s and can be detected more rapidly [59]. These two factors lead to a difference of thermal sensing of both the direction of temperature change (warm/cold) and on different locations. Over the body, the sensitivity to temperature varies greatly, up to 100-fold, and declines with age [1, 61]. So is, for example, the face very sensitive to thermal changes, while the extremities show poor sensitivity, see Figure 2.1. Cold could be detected better over the whole body than warm, with thermal sensitive spots being better at recognising both directions of change when compared to other regions. The less sensitive regions also decline faster during ageing than the more sensitive regions [1].

Observations of thermal sensitivity on the hands showed temperature changes were easiest to detect on the thenar eminence [62], the glabrous (hairless) skin under the thumb, see Figure 2.2. Temperature changes (from a base temperature of 33°C) of 0.11°C for warm and 0.07°C for cold changes could be detected on the thenar region, while only changes of 0.16°C for warm and 0.12°C for cold could be felt on the less sensitive fingertips [63, 1].

Thermal receptors react to temperature changes on a range of 5°C and 43°C, but cold receptors are most active at around 25°C, while warm receptors peak at 45°C [32]. Additional receptors charge at noxious (harmful) temperatures: if there is a risk of tissue damage, so-called thermal nociceptors convey the feeling of pain. The skin rests at around 32°C to 34°C within a neutral zone of 30°C to 36°C, in which slow temperature changes are hardly detected, as the skin adapts to these [62, 32, 58]. Nociceptors charge when temperature decreases below 12°C or increases above 45°C [58]. Ambient temperature was found to have an influence on the perception of temperature changes: in cold rooms stimulus intensity was rated lower than in neutral environment for both warm and cold temperature changes, but there was no change found between neutral and warm environments [50].

Thermal receptors are underpinned by transient receptor potential (TRP) channels, which



Figure 2.1: Thermal sensitivity over the body: warm (upper section) and cold (bottom section); smaller bars indicate higher sensibility (extracted from [1]).



Figure 2.2: Thenar eminence (red) of the hand, the most temperature sensitive part of the inner hand; finger tips (light blue) are less sensitive.

react to different, sometimes overlapping temperature ranges, but can also be activated by *hot* or *cool* food products, such as chilli peppers or mint [32, 58]. Thermal fibres and TRP channels do not operate completely independently [62, 64], influencing each other invoking interactions that could explain some phenomena observed in thermal perception (see Section 2.1.3).

The thermal sense is aiding the identification and discrimination of different types of objects [31, 65, 32]. Thermal properties of materials are especially important, when visual identification is inhibited. The skin temperature changes when touching an object depend on, for example, the object's conductivity and heat capacity, the temperature difference between skin and object and thermal contact resistance between the two [32]. The thermal properties of the object materials have to be sufficiently different for the thermal sense to contribute significantly to object identification [31].

To Summarise: warm and cold temperature sensitivity differs over the surface of the body and perception changes with age. In addition, cold temperatures are perceived faster and over a larger range. The neutral temperature zone of the skin is between 30°C and 36°C and within this range the skin adapts to small temperature changes fast. Temperatures under 12°C and over 45°C lead to painful sensation and potentially tissue damage.

2.1.2 Spatial and Temporal Aspects of Thermal Perception

Thermal perception is dependent on spatial and temporal aspects of the temperature change. Thermal sensation on the skin depends on the size of the area at which a thermal change is presented: the temperature change is perceived as being stronger when it is presented on a larger area of the skin [66, 67, 68]. This is called *spatial summation* and can be observed for both warm [69, 66] and cold changes [67]. For example, when presenting thermal stimuli on the fingertips, it has been found that presenting them on different fingers improved the identification [70, 71]. Even thermal changes on bilateral body parts, such as both hands, are summed up, suggesting that the summation effect itself manifests in the nervous system rather than the skin [68]. The influence of spatial summation on the perception of a thermal stimulus is almost equal to the extent of temperature change itself, i.e. how much the temperature is changed [69, 67]. This means that the same sensation can be simulated by doubling the area and halving the extent of temperature change. For warm temperature changes, this effect of spatial summation decreases when the temperature approaches the pain threshold, but not for cold changes [67]. Spatial summation can, however, inhibit the distinction of two differing thermal stimuli in close proximity [72, 73].

Temporal aspects also play an important role in thermal perception. Human skin adapts to the temperature it is exposed to over time. This temporal adaptation is fast for temperatures close to skin temperature, adapting a temperature difference of 1°C within one minute [32]. For higher temperature differences between 28°C and 37.5°C, complete adaptation can be reached within 25min [74].

Another important temporal factor influences the perceived intensity of the thermal change: how fast the temperature changes. The higher the rate of change (ROC), the faster the skin reacts to the temperature change [75, 76]. For example, a change in ROC from 0.5°C/s to 4°C/s doubled the perceived intensity of the stimulus [77]. In addition, nociceptive (painful) sensations increased for faster temperature changes, more noticeably for warm than cold [75, 77].

2.1.3 Thermal Phenomena

Based on the physiological features of thermal sensing, some interesting interactions of warm and cold cues can be observed. Green [78] observed a couple of thermal phenomena in experiments with three thermal devices, presenting stimuli to three fingers of the same hand. *Referral* described the phenomenon occurring when the outer two fingers were warmed (or cooled), but the middle finger kept thermally neutral: participants reported a warm (or cold) sensation on the middle finger. Prolonged presentation of these stimuli led to adaptation and the temperature differences could be distinguished [79]. *Enhancement* described the observation of the middle finger feeling warmer (or colder) than the outer two fingers, when the same temperature was presented on all three, while *Domination* described the effect that if the direction of change of the middle finger differed from the outer finger, the same sensation was reported for the middle finger than the outer fingers. The most surprising effect observed was coined *Synthetic Heat*, or *Thermal Grill Illusion*, and occurred sometimes, but not always, when the outer fingers were both warmed and the middle finger was cooled. Not only was the sensation on the middle finger warm, as it would because of Domination, but it was described as *very warm* or *hot*, to the point of mild hot burning sensation [80]. Perceived intensity of this effect was higher for more pronounced temperatures differences [81] and diminished after 10s [80].

These phenomena closely relate to the effect of spatial summation (as described in 2.1.2) and the underlying shared TRP channels (discussed in 2.1.1).

Singhal and Jones [82, 83] investigated if the *cutaneous rabbit* illusion, or *sensory saltation* [84], could also be observed for thermal feedback. When several mechanical stimuli spaced evenly on the skin were being activated in progression, participants reported feeling one smooth movement *as if a tiny rabbit was hopping*. The time intervals between the activation of the stimulators influenced the perception of the location it was presented on. The same effect was found for thermal feedback: for short intervals participants perceived the position of the second pulse to be closer to the location of the third pulse [82, 83].

2.1.4 Summary

Human skin does not perceive temperature equally on all locations. Some areas are more sensitive, such as the face, while others, such as the extremities, have poor sensitivity. On the hands, the thenar region, the palm directly under the thumb, is most sensitive to temperature changes, but thermal changes are still perceivable on the fingers. In addition, the skin more easily perceives cold temperatures and reacts faster to them. The non-painful range of temperature lies within 12°C and 45°C, with the skin's neutral state between 30°C and 36°C, in which it adapts to temperatures fast.

Temperature on the skin feels warmer (or colder), when the area of stimulation is bigger. This spatial summation not only occurs when the areas are adjacent to each other, but can still be observed over several fingers and even bilateral body parts, such as both hands. The speed or rate at which temperature changes occur has a profound influence on the perceived intensity of the change. Faster rates lead to an increased feeling of intensity.

These characteristics of the human thermal sensing constrain and define the possible parameters of thermal displays and cue design. In addition to these, the occurrence of thermal phenomena, especially when presenting warm and cold stimuli in adjacent areas, have to be kept in mind when designing cues, but can be purposefully used to enhance sensations. This basic knowledge on thermal sensing in humans informed any design of thermal displays used within this thesis. The next section will explore research on thermal feedback within human computer interaction, and discuss design and use of thermal cues.

2.2 Thermal Changes as Feedback

Thermal displays have become more popular in human computer interaction over the last two decades and have been applied to many different kinds of use cases. More abstract observations have brought insights into the design space and limits of thermal displays, discussed in Section 2.2.1, while applied settings have added to the specific requirements needed in defined areas, like Affective Computing (Section 2.2.2), and environments, such as Virtual Reality (Section 2.2.3), and Stationary and Mobile Applications (Section 2.2.4). The bimodal combination of thermal and vibrotactile feedback and their interaction will be discussed separately in Section 2.2.5. Finally, temperature in the driving context will be briefly discussed. This section concentrates on feedback presented on the hands, but discusses feedback on other locations to some degree. Kappers & Plaisier [85] have conducted an extended review of thermal feedback presented on body parts other than hand and face.

2.2.1 Thermal Displays: Characteristics and Design Recommendations

Thermal Display Requirements

Based on human thermal perception, Jones & Berris [28] have proposed properties that should be provided by a thermal display. They identified that devices should be able to present temperatures between 22°C and 42°C. They observed that skin temperatures in a room of approximately 20-22°C stayed between 32°C and 35°C for most subjects, leading to the conclusion that thermal displays do not require constant measuring of skin temperature. The rate of change used in displays should be at least 0.3°C/s to ensure detection. As perceived intensity of a temperature change depends on several factors, such as rate of change, area of presentation and extent of temperature change, the use of discrete temperatures as cue was found to be ineffective.

In addition, thermal perception is sluggish compared to other tactile modalities, such as vibration. But recognition time can be decreased by presenting warm and cold simultaneously on two small, adjacent devices before presenting the stimulus, benefiting from the poor thermal resolution of the skin and the fast adaption to temperature changes close to skin temperature. This method increased the difference of skin temperature and stimulus and led to faster recognition [86, 87].

Thermal Feedback Design

The design of thermal cues was evaluated by Wilson *et al.* [29]. They tested different extents (1°C, 3°C and 6°C) and rates of change (1°C/s and 3°C/s) in both directions on different body locations, starting from a neutral temperature of 32° C, which was presented before and between stimuli. They tested these factors stationary at a table and while walking indoors and summarised their findings as recommendations. They found that stimuli were harder to detect while walking and detection took longer. The thenar eminence and the palm were found to be most sensitive, reaffirming previous findings discussed in Section 2.1.1. In terms of perceptability and comfort, they found that both 1°C/s and 3°C/s were detected with similar accuracy, however, the faster rate of change was detected faster, but rated as less comfortable. Cold temperatures were easier and faster detected than warm cues. The lowest extent of temperature change had the lowest recognition rate, the other two could be identified and even differentiated efficiently.

In further experiments, Wilson *et al.* [44] also investigated thermal icons for the mobile environment. Thermal icons were structured thermal feedback, presenting two types of information: *source* (personal/work) and *importance* (standard/important). Two factors were tested: direction of temperature (source) and subjective intensity (importance), a combination of extent and rate of temperature change. Full icons were identified with 83% accuracy, the single factors direction of change and subjective intensity with 97% and 85% accuracy, respectively.

Singhal & Jones [45] investigated different patterns of thermal cues and how well they could be perceived. Their cues described linear in- and decrease of temperature, almost quadratic pulses at two different temperature ranges and a step-like increase in temperature at two different temperature ranges. The neutral temperature of the device was 30°C and the cues were presented on the tips of two fingers. The patterns were identified correctly between 75% and 85%, with the linear decrease (only cooling) reaching the highest and the linear increase (only warming) the lowest recognition accuracy.

This requirements described the frame in which thermal in-car displays should be operating. Design cue decisions in the research presented in this thesis were informed by these basic observations.

Environmental Influences

The environment can have an influence on the effectiveness of thermal displays. Investigating the influence of clothing on perception and comfort, Halvey *et al.* [51] found that, while the thenar eminence produced the best results, the waist was also identified as a suitable location for thermal feedback with slightly lower detection rates, but similar detection time, comfort and intensity ratings. Materials with lower thermal conductivity needed higher temperature changes, but the detection time did not differ and higher thermal conductivity led to a higher number of errors. They also found that higher extent of temperature change was perceived as more comfortable with clothing between the device and the skin and the higher perception rate of cool temperature changes found in previous studies could not be observed with clothing.

In further studies, Halvey *et al.* [88] explored the influence of ambient temperature and humidity on detection and perception of thermal cues. They tested stimuli at outdoor locations in varying weather conditions over five months. The best results in terms of detection rate (84%), detection time (3.03s) and comfort ratings was achieved for temperatures between 15°C and 20°C. However, results at other ambient temperatures still indicated suitability. Humidity only had a significant influence on detection rate, time and comfort ratings for extreme conditions. Wilson *et al.* [89] found that ambient temperatures significantly effected identification accuracy and time. They pointed out that there were high individual differences, but on average the identification worsened when the ambient temperature was warmer (between 25°C to 26°C).

Ketna & Leelanupab [27] investigated thermal feedback in noisy and bumpy environments. Audio and vibrotactile cues are often sought as alternatives to visual feedback, but have been shown to perform poorly in noisy and bumpy environments, respectively. They tested unimodal thermal and multimodal cues including thermal and found that the detection of thermal cues was not influenced by either noise or environmental vibration. In multimodal cues, thermal feedback enhanced detection.

The in-car environment presents many possibilities for locations on which to present thermal feedback and can be bumpy and noisy by nature. Other environmental parameters can easily be controlled within the well-defined area of the vehicle cabin and pose no concern for the use of thermal feedback within the car. This previous research showed that thermal feedback through clothing could be detected and its detection was not influenced by noise and a bumpy environment, showing its suitability for the in-car environment. However, as detection through closing needs some more research into suitable adaptation, this thesis will focus on the presentation of temperature changes directly on the skin.

Thermal Hardware

Some characteristics of thermal displays are particular to the method used to present temperature changes. Peltier devices are the most often used thermal devices. They are thermoelectric devices, warming one side of a plate and cooling the other, depending on the direction of the current. Cooling one side of the device, therefore, leads to the need of dissipating hot temperatures on the opposite side. This is usually achieved by adding a heat-sink (see Fig-



Figure 2.3: Peltier device (white) attached to heat-sink (black).

ure 2.3), water-cooling system or alternatives. Extent and rate of temperature change depend on the direction and magnitude of the current [90].

Kratz & Dunnigan [91] used liquid cooling and electro-resistive heating for a grid device, instead of Peltiers. This setup was only practical for at least semi-stationary use and further work would have to be done to achieve precise temperature output. However, the setup is more easily scalable than grids with Peltier devices would be, as those need a lot more energy.

Nakajima *et al.* [92] used light and mist beams to produce mid-air thermal sensations, specifically to reproduce the Thermal Grill Illusion. This method needs a larger setup than Peltier elements and precise temperatures are hard to generate.

Hirai & Miki [93] designed a thermal device which changes the thermal conductivity of its surface to evoke different tactile sensations, rather than the present temperatures directly. The device varies by filling or emptying cavities under the surface with liquid metal. This method closely simulates the thermal sensation when touching and identifying an object and could enhance the experience in virtual environments.

Brooks *et al.* [94] used a different approach to illicit the feeling of warm and cold in VR: they presented smells associated with thermal sensation, namely eucalyptus (mint) for cold and chilli peppers for warm sensations, and presented them directly under the nose. They managed to produce simple warm and cool sensations with this setup, without the need to warm or cool participants or the environment and with comparably low energy demand. In addition to smell, taste sensations can be associated with temperature: presenting thermal feedback on the tongue can elicit taste sensation [95].

The most used and fastest thermal devices in previous research are thermo-electric Peltier devices. Temperatures can be presented precisely with appropriate control mechanisms and
were, therefore, chosen to be used for the research presented in this thesis.

Summary of Thermal Displays

Research of thermal displays and thermal cue design suggests that temperature changes of 1°C/s and 3°C/s could be well recognised. Base temperatures between 30°C and 32°C were suitable for temperature changes up to 6°C. Temperature changes, rather than discrete temperatures, were detected and the thenar region was confirmed as the most suitable location, but finger tips were also successfully presented with thermal cues. Direction of temperature change was identified as the best recognised factor of thermal design, but rate of temperature change and extent of temperature change in combination had promising recognition rates. Thermal feedback was well recognised in a noisy and bumpy environment, but ambient temperature influenced recognition.

2.2.2 Affective Computing

Affective computing describes the interface of computing systems and emotions. For thermal feedback, affective computing is achieved by enhancing a system with thermal capabilities to present or manipulate emotions.

Research has shown that temperatures can influence or represent social presence, where colder temperatures are linked to exclusion [96] and warmer temperatures to interpersonal *warmer* judgement of personality, such as being more generous and caring [97] and increased social proximity [98]. This relationship can be easily observed in descriptive language, where *icy stare* and *cold shoulder* have negative connotations, while *warm smile* and *hot date* tend to be interpreted positively.

Feedback in the car could potentially benefit vastly from affective associations, as it could create more easily understandable cues, conveying complex messages non-visually. The following section will discuss several areas in which affective associations have been investigated. The mapping of emotion to cue is dependent on the use case, so several different areas could potentially provide interesting mapping contexts for in-car communication. First, more abstract mappings will be discussed, followed by a number of areas in which thermal cues have been used to provide emotion-based feedback.

Abstract Mapping to Arousal-Dominance-Valence

Emotional responses to thermal cues have been tested and it was found that warm temperature changes increased dominance and arousal [99] and were rated as pleasant for a change of 4°C, but unpleasant for 6°C [100]. Emotional valence (positive/negative) was found to mirror the use in language: warm was linked to positive and cold to negative [101]. A study investigating arousal and valence found that they were linked: ratings were found either in the low arousal/high valence quadrant (calm/pleasant) or in the high arousal/low valence quadrant (excited/unpleasant) [43]. Warm temperatures were rated more pleasant and their arousal increased with higher extent and rate of temperature change.

Tewell *et al.* [102] tested an array of thermal devices and influenced the arousal of an incoming text message by presenting temperature on a differing number of devices, valence was influenced by the content of the message. They found that warm temperatures were perceived as more arousing, but did by itself not give any valence information.

Wilson *et al.* [103] mapped emotional responses to multimodal cues, with thermal, vibrotactile and visual modalities. They generated several *lookup tables*, showing which combinations of modalities could potentially convey specific emotional sensations and defined guidelines based on their findings. The combination of several modalities was helpful to increase the range of emotions and they found that the overpowering effect of vibration could be tempered by the addition of other modalities. Presenting three cues together could, however, be perceptually taxing on the senses. Vibration mostly influenced perceived arousal, while thermal or visual feedback influenced valence.

These findings inform on general reactions of temperature changes on abstract emotions, independent of use cases. They were used to form expectations for affective associations within specific use cases.

Providing Affect for Communication

More specified thermal feedback was used to enhance communication with emotional meaning. Iwasaki *et al.* built the AffectPhone [104], a prototype attached to a mobile phone, which measured the galvanic skin response (representing the emotional state) of one communication partner and presented the corresponding temperature to the other.

A prototype, sending thermal messages remotely between either mother and daughter or two co-workers, was tested by Lee & Lim [48]: one partner could send thermal messages to the other by pressing buttons. As before, warm temperatures were linked to more positive and cold to more negative feelings, with varying degrees depending on the extent of temperature change, but the interpretation of the temperature was context dependent. The feedback was experienced as non-obtrusive and playful.

To increase the feeling of social presence during messaging, Gooch *et al.* [105] built a thermal harness to simulate a thermal hug. They found that the added thermal cue showed an increased sense of presence, but could not prove that the increase could be attributed to the feedback or the context of use.

Another study by El Ali et al. [106] investigated thermal stimuli on the chest, enhancing

(neutrally-spoken) voice messages with affect. They found that the stimuli could be well perceived through the fabric and warm and cool cues increased arousal and increased (warm) or decreased (cold) valence of the message.

Suhonen *et al.* [107] investigated three different haptic modalities and their influence on interpersonal communication: thermal, vibrotactile and squeeze feedback. Squeeze feedback was achieved by tightening a wristband in varying patterns. In contrast to other previously discussed research, participants experienced cold temperature changes as more pleasant. Squeeze feedback was received positively and both squeeze and thermal feedback were connected to everyday notifications, for which vibration is already used. Haptic feedback was described as efficient and intimate, adding a rich, new dimension to communication. A similar setup tested by Song *et al.* [108] looked at identification of squeeze and thermal cues and compared them to vibration. They found that squeeze feedback was poorer.

Osawa & Katsura [109] discussed a thermal glove set that mirrors the thermal sensation of one glove onto the other, with potential use cases in remote care or robot interaction. Peña & Tanaka [110] went further and prototyped a robot with thermal capabilities in the skin. Investigating the effect of thermal skin on a robot, Park & Lee [111] found that warm skin temperature increased social presence, perceived friendship and perceived emotional stability.

Within the vehicle, feedback is given to communicate information to the driver. These findings on enhancing communication with emotions could be used to add an additional, easy to understand layer to the communication between car and driver. In this thesis, thermal feedback was used for notifications, looking into how the thermal feedback can increase the feeling of urgency, which was believed to coincide with the increase of arousal found in this previous research.

Enhancing Media with Emotions

The emotional interpretation of media, such as pictures and videos, can be very subjective. Artists could potentially use thermal feedback to guide the observer and convey their intended emotions.

Or they could assign specific temperatures to locations within pictures, as did Nakashige *et al.* [112]. They designed a thermal trackball, which changed the temperature when the cursor touched parts of a picture that were linked, such as a bowl of warm soup or cold orange juice. Participants rated pictures of dishes as more delicious, when the matching temperature was presented and reported emotional associations such as a *loving home* with warm soup.

When exploring different aspects of thermal augmentation of pictures, Akazue *et al.* [46] found that the timing of thermal presentation can either create anticipation or increase the

emotion of the presented picture. Furthermore, they found that the presented temperatures increased valence and arousal in pictures with low valence and arousal and decreased the two factors for pictures with high valence/arousal.

Looking at both pictures and music, Halvey *et al.* [42] found that valence and arousal were influenced by thermal cues for both visual and auditory media, where thermal feedback generally enhanced arousal. Their results suggested that thermal feedback adjusted to the content of the media, might have a more pronounced influence, which should be investigated further.

This research found very precise mappings of emotional associations and temperatures. Warm temperatures invoked a feeling of *home* and could increase the emotion of presented content. This specific associations informed the use case design of thermal notifications within the thesis.

Other Use Cases

As affective interpretations are context dependent [48], the design of temperature feedback has to be adapted and tested for specific use cases. Wilson *et al.* [113] explored a number of different application areas of thermal feedback and captured reactions and interpretations. In their experiment, they presented temperatures as indication of online activity of phone contacts, physical presence and availability of a person with thermal cues on the door knob, content use before deletion and restaurant ratings. They found that participants generally agreed on the meaning of the presented stimuli. They interpreted warm temperatures as a representation of presence and activity and presented negative emotions. Up to seven different levels of temperature were labelled within each application. In the case of the door knob, warmer temperatures were interpreted as the person being in the office, with increasing temperature indicating a higher level of business. In the content deleting task, participants were shown the potential risk of deleting data by showing thermally how much it was used beforehand. Warm temperatures again were identified as being used frequently and, therefore, posing a higher risk when being deleted.

This idea of using temperature to present risk or danger was picked up and used for studies investigating the benefits of using thermal feedback to inform of the security of web pages [49]. The idea was first explored in an online questionnaire, where associations were collected. Participants mostly correlated warm temperatures with secure web pages and cold with insecure web pages. This was followed up with a lab study, which found interpretations inversed: secure web pages were associated with *very cold* temperatures, while insecure web pages to *very warm*. Participants commented on an association of heat to danger and cold with calm. Napoli *et al.* [114] adopted this to inform on the security of TLS certificates. They built a prototype using a heating pad on the laptop, presenting temperature on the wrist, which needs additional evaluation to determine its usability.

Lately, thermal feedback was used to help users assess their own body's reaction, by projecting internally sensed temperature to a thermal waist belt [41] or by notifying users about stress [115].

Umair *et al.* [116, 117] used thermal feedback in connection with self-crafted worn smart artefacts to support the wearer's awareness of their own state of arousal. They discussed the materials used and described several stages of the crafting process. Participants had the chance to play around with different heating devices and some of their comments were shared. Participants commented negatively on their lack of controlling the return to neutral temperatures in the heating pads. Peltier devices were also used, but only one-directional with providing current to heat up and then letting them cool passively by removing the battery.

Trojan *et al.* [118] investigated, if the rubber hand illusion could also be induced by visualthermal stimulation. The rubber hand illusion traditionally is induced by touch: the real hand of participant, obstructed from view, and a rubber hand placed next to the real hand are repeatedly touched at the same time. This leads to induced feelings of ownership of the artificial hand, as if the hand were part of the participant's body. The same effect was observed when coloured spots were presented on the rubber hand and a thermal cue on the real hand, especially when the colour *matched* the temperature: warm/red, cold/blue. This effect could potentially be used to increase immersion in virtual reality.

The research presented in this section discussed a number of different, specific use cases for which thermal feedback was investigated. It can be seen that the interpretation of temperature is very use case dependent. The act of actively feeling the temperature change could also change a perception participants had when interpreting them theoretically, as shown in the case of the security of web pages. These findings showed that expected interpretations have to be experimentally proven, an approach which has been used for the research of this thesis. Furthermore, the idea of presenting danger in this previous research inspired the use case of notifications with different levels of urgency with thermal feedback within the thesis.

Summary of Affective Computing

Thermal feedback can evoke emotional responses. The direction of temperature change was in general associated with valence: warm temperature changes were interpreted with positive, cold temperatures with negative emotions. However, in specific use cases, these mappings were reversed. When the cue was associated with security, warm temperatures were connected to the feeling of danger, while cold represented security. In other areas, such as interpersonal communication, warm cues represented familiarity and social presence. The interpretation of thermal cues is context dependent and new use cases have to be tested to ensure that the intended meaning is communicated. However, when the association was made, the effect was quite consistent within participant groups.

2.2.3 Virtual Reality

One of the areas in which thermal feedback has been used the most, is Virtual Reality (VR). The main goal of thermal feedback in connection with VR is to increase immersion, the feeling of presence in the virtual world, by simulating touch sensations or parts of the thermal environment. It has been shown that even the visual representation of thermal cues in the virtual environment without any actual temperature difference in the real world can have an influence on grasp movements [119]. These aspects are less important in an in-car environment, were thermal feedback would be used to convey information, rather than aim to enhance any kind of immersion. Therefore, only a comparatively short overview will be given of thermal feedback in VR. As one of the areas with increased research in thermal feedback, examples from this area might be briefly discussed in other sections.

Research in this area often investigates effective delivery systems for temperature feedback, which would not interfere with the activities usually performed within VR. To this end, most commonly gloves or hand-worn devices were outfitted with thermal, and often other tactile, devices [33, 70, 120, 38, 34, 39, 121], parts of clothing fitted with thermal capabilities [37] or VR equipment enhanced [122, 123]. More rarely, the physical environment gets adapted to simulate the virtual environment [124].

Use cases vary, but most commonly thermal feedback was used to represent the virtual thermal environment [35, 122, 36, 125, 37], or object temperature for identification [126, 70, 38, 39, 34, 127]. More exotic use cases involve creating sensations in VR that are not possible in real life, such as the feeling of passing through an object [128].

This short summary of thermal feedback within VR was added because it is one of the most used applications for thermal feedback within research and should be mentioned at least briefly for a complete discussion of thermal feedback. The goal of the use cases within VR differ significantly from in-car feedback for drivers, but might be interesting when investigating the use of temperature within the car for passengers, especially in view of automated vehicles.

2.2.4 Stationary and Mobile Application

This section will introduce some thermal displays with a setup fixed to a single location, either because of their bulky shape or because their function connects to a specific location. Feedback in the car combines aspects of stationary and mobile applications. While the setup

inside the cabin is comparable to a stationary setup, the movement of the car adds requirements of mobile environments, such as the need to be aware of the surroundings. Therefore, both stationary and mobile applications will be discussed.

Kushiyama *et al.* [129] used a large array of Peltier devices to present or enhance pictures with temperature. They did not primarily aim to add a layer of affect, as in research discussed previously, but used this setup in collaboration with artists to allow them to add the layer of touch to their art. In further steps [130, 131], they used their setup with thermally sensitive sheets, changing their colour depending on temperature. These sheets would visualise the temperature of the Peltier devices under it, thus producing images which can be both seen and felt.

Thermal feedback on game controllers were investigated [132, 133]. Comparing visual, vibrotactile and thermal feedback for finding elements in a game, Löchtefeld *et al.* [133] found that while accuracy and time were worse for thermal feedback, it was still effective as an active game element. Thermal feedback was rated as less temporally demanding than both visual and vibrotactile, which could be useful for specific feedback within a game.

Temperature was also used to navigate a two dimensional maze game [134], where the feedback was presented on the arm rather than the controller. The feedback would stay warm, while the correct path through the maze was taken and turned colder, when the path was left. Temperature was here presented constantly, eliminating the need to return to a neutral temperature in between cues, only changing when the user left the shortest path through the maze. This feedback helped participants navigate the maze effectively.

Some investigations into design for mobile applications and influence of environmental factors have been discussed in previous sections. This part will focus on research discussing specific use cases or explorations of different locations on the body. Most experiments involve the design of a prototype for use at well defined locations. In contrast, Maeda & Kurahashi [135] designed a wearable prototype of several thermal devices, which could be worn at different locations all over the body and Niijima *et al.* [136] tested a setup with three small devices that held and moved to different locations over the body.

Wettach *et al.* [40] used a handheld thermal prototype for pedestrian navigation through a city. The testing of the device seemed to only have been done with one participant, and they seemed to have fairly intensive training with the device beforehand, as they tested identifying three different temperature steps in one sitting and then long-term over ten days with random stimuli presentations. Training did increase identification of the cues and the participant managed to successfully navigate an unknown city with the device.

Adding thermal capabilities to an intelligent voice-agent device, Kim *et al.* [137] investigated the benefits of thermal feedback to the delivery and enhancement of auditory content and the influence on the perception of the agent. Participants reported that they perceived the voice agent as friendlier when the interaction included thermal feedback. The thermally enhanced

content was noted to be more memorable, when warm temperature was presented, and felt more immersive and engaging.

High identification accuracy was found for feedback on four small cooling elements on a ring-like prototype, especially for single-spot feedback and specific patterns [138]. On the wrist with six different devices and configurations, thermal feedback outperformed vibrotac-tile feedback in terms of accuracy during walking and in distracted situations [139].

Thermal feedback on the ear was used to define open space [140]: specific areas in a large open space were associated with temperatures. This can be used to split open space without the use of physical barriers. Observation showed the warmer spots were preferred. Nasser *et al.* proposed thermal feedback on an ear-worn device to provide feedback for hearing and visually impaired [141].

The vehicle cabin can allow for fixed installation of thermal devices. Findings of this previous research suggest that thermal feedback was less temporally demanding than other modalities and could successfully be used for navigation purposes. Both findings inspired the use of thermal feedback for in-car navigation.

2.2.5 Thermal-Vibrotactile Feedback

Haptic displays sometimes combine various modalities for a richer experience and a broader design space [142], which would be of great value within the vehicle. The combination of thermal and vibrotactile feedback has been explored in some detail, as the fast onset and detection of vibration as opposed to slow, but more emotionally interpreted thermal cues combine two very different experiences of touch. In VR, thermal-vibrotactile feedback could lead to even more immersive experiences. For example, when simulating holding a cup which gets filled up with cold water, thermal feedback could present the temperature of the water, while vibration could simulate the impact of the liquid within the cup [143].

The two modalities presented together can, however, influence their perception. Some of the affective influences were already discussed in Section 2.2.2. The perception of vibro-tactile feedback differed for high frequency voltage cues: perception thresholds were higher for cold than for neutral and warm temperatures [144]. Skin temperature also had an effect on the identification accuracy of vibrotactile patterns. Warm skin aided the identification of vibration patterns with varied amplitude, while it hindered the identification of patterns with varied pulse duration [145].

In reverse, the number of vibration pulses can influence the perception of thermal patterns [30]. Perception of cold cues was found to be more heavily affected by concurrent vibration than warm cues.

These findings inspired and informed the use of bimodal tactile feedback for notifications presented in this thesis. Three experiments investigated temperature changes in combination

with vibration, each of which used different onset times for the modalities.

2.2.6 Temperature in Driving

In a vehicular environment, temperature was mostly discussed in terms of thermal comfort assessment and prediction models [146, 147, 148, 149]. Another focus lies within counteracting driving fatigue with thermal changes [150, 151]. It has been shown that 4min of cooling can mitigate driver fatigue. In these instances, the whole cabin and driver were subjected to temperature changes. While it was suggested that more localised cues at the hands might be effective and should be tested to counteract driver fatigue [152], using thermal cues for information purposes in a vehicular environment has not been tested. This thesis aims to fill this gap.

2.2.7 Summary

Thermal feedback possesses some unique characteristics, which suggest its use for many very different applications. In VR, its contribution to identify materials is interesting, as is the enhancement of the feeling of immersion when simulating temperatures either of the surroundings or interacting objects.

Affective associations play an important role both in- and outside of VR: warm temperatures were generally associated with positive, active and personal experiences, cold with negative, inactive, and distant ones. But warm can also increase the feeling of danger or insecurity, and the experience of feeling it can lead to different interpretations than were expected when contemplating them theoretically. The mapping of thermal associations can be delicate and needs to be tested for specific use cases. But when they have been found, they generally were interpreted consistently across participants. For the driving environment, interpretation of direction or danger are of special interest. In terms of navigation, thermal feedback was used to direct users by presenting warm temperatures while they stayed on the correct course and cool when they left it. This interaction cannot be adopted for driving, as turning around might not always be possible or potentially dangerous. Turn-by-turn navigation would be more effective and different approaches on feedback will have to be tested for that.

Investigations of the design space of thermal feedback showed that 30°C to 32°C as neutral temperature in between cues was suitable. Rate of change should be faster than 0.3°C/s, and 1°C/s and 3°C/s have been identified as effective. Thermal changes of 3°C and 6°C could be distinguished well in both directions, smaller changes were harder to detect. The thenar eminence was affirmed as the best location for thermal feedback on the hands, but cues on the fingers could be detected and distinguished as well, especially when presented on several fingers. Detection of warm and cold stimuli can differ significantly, especially

when presented simultaneously with vibrotactile feedback. These findings highly influenced the design of thermal cues within this thesis.

These are important basics for designing thermal cues for the driving environment. However, the environment in which these cues will have to be identified, are very different in nature. While environmental factors are less of an issue within cars, as the thermal environment can be measured easily and any feedback potentially adapted, identification and perception of thermal cues will be moved from a primary task, as it has been in most of the studies discussed in this section, to a secondary task with a highly demanding primary task. The increase in workload and actions done simultaneously during driving could have a distinct influence on perception and identification of thermal feedback. The following sections will focus on factors of driving and haptic feedback during driving.

2.3 Driving Background

This section delivers some background information on important issues of designing feedback for the driving environment. Driving is a highly demanding task, in which mistakes can lead to grave consequences or even death. There are many demands posed onto the driver, including the need of high situation awareness [153] to ensure the safe operation of the vehicle within its environment. When designing cues for use during manual driving, it is important to distract the driver as little as possible from their main task.

This section will first discuss driver distraction and its important role in evaluation of in-car applications. This will be followed by a brief overview of levels of autonomy and the change in demands on driver attention and driving.

2.3.1 Driver Distraction

Many different definitions of driver distraction can be found. Driver distraction is generally described as a subset of inattention [153, 154], where the attention is diverted from the driving task by another activity, as defined by Lee *et al.* [155, 156]:

Driver distraction is a diversion of attention away from activities critical for safe driving toward a competing activity.

The *activities critical for safe driving* are less well defined. Typically, driving related activities are described as primary, driving unrelated activities, such as eating or operating mobile

devices, as secondary tasks. The primary driving task includes driving manoeuvres, longitudinal (velocity) and latitudinal (lane keeping) control of the car and staying aware and reacting to traffic regulations and participants. Generally, everything outside this scope is counted at least as secondary task. The categorisation of tasks such as navigation, however, are not always clear. Navigation cues are often integrated into the in-vehicle infotainment systems (IVIS), which is in general regarded as providing secondary tasks [157, 158]. However, navigation is time and location specific and not always optional and could be considered driving related and, therefore, is sometimes included into the primary task [159, 158]. Secondary task distraction has been associated with the highest percentage of crashes in an extensive naturalistic driving study [160, 2].

The categorisation by the National Highway Traffic Safety Administration (NHTSA) is often applied to the discussion in how attention can be diverted [154]:

Visual distraction: Tasks that require the driver to look away from the roadway to visually obtain information;

Manual distraction: Tasks that require the driver to take a hand off the steering wheel and manipulate a device;

Cognitive distraction: Tasks that are defined as the mental workload associated with a task that involves thinking about something other than the driving task.

Some definitions include auditory distraction in this listing, such as Pettitt *et al.* [161], who comprised a more precise and detailed definition of driver distraction, including the different possible types of distraction:

Delay by the driver in the recognition of information necessary to safely maintain the lateral and longitudinal control of the vehicle (the driving task) (Impact)
Due to some event, activity, object or person, within or outside the vehicle (Agent)

- That compels or tends to induce the driver's shifting attention away from fundamental driving tasks (Mechanism)

- By compromising the driver's auditory, biomechanical, cognitive or visual faculties, or combinations thereof (**Type**).

The presentation of thermal feedback could possibly present cognitive distraction from the main driving task, posed by the thermal devices within the car, and could have an impact on the driving behaviour.

The measurement of distraction is often based on the influence of the distraction on the driving behaviour (performance based data) and/or the driver's behaviour (physiological

data) [162]. Hurts *et al.* [157] and Riener *et al.* [163] summarised methods mostly used to examine a task in the context of the primary driving task. Some methods are about measuring task-related data, such as task completion time, errors and response time.

Others observe physiological data, such as number of gazes needed to complete the task and the gaze time as well as overall eyes-off-the-road time. Visual distractions, which averts the gaze from the forward road, pose the highest risk of resulting in crashes [2, 164], especially when glances are longer than 2s [2].

In addition, the impact on the longitudinal and lateral control can be evaluated. Longitudinal control is typically measured by evaluating speed (mean speed, speed variance and/or speed compliance) or headway (and/or headway variance). Lateral control is usually measured through investigating the lateral position (or lane deviation), using either the standard deviation of the lateral position, lane exceedance or the root mean square error of the lane position [165]. Eyes-off-the-road time can significantly increase lane deviation [166].

Another suggested method investigates the miss rate and mean response time of the peripheral detection test (PDT), where items are presented to the driver in their visual periphery. As workload increases, peripheral awareness decreases, as the visual attention narrows [167, 168] and higher miss rates and longer detection times can be used as reliable indicators of driving distraction.

Measuring mental workload is another method to capture driver distraction [162, 169]. High mental workload can have an influence on the driving performance, as it can increase lane deviation [168]. The NASA task load index (TLX) questionnaire [170] was designed to capture self reported subjective workload. It captures six aspects of workload: mental demand, physical demand, temporal demand, performance, effort and frustration. An additional scale for *annoyance* has occasionally been added in the past [171].

The impact of a secondary task on the workload can be mitigated to some degree by using different modalities than those engaged in the primary task. This was suggested in the Multiple Resources Theory [3], which proposes that a more even spread of information over different sensory resources could help decrease mental demand. Wickens' theory presents three different dimensions that influence the performance in concurrent tasks: the *stages of processing* (Perception and Cognition / Responding), the *codes of processing* (Spatial / Verbal), and the *modalities* (Visual / Auditory). The theory suggests that *the extent that two tasks use different levels along each of the three dimensions, time-sharing will be better* [3], which in turn would lead to a better driving performance and less mental workload. A fourth dimension *visual channels* (Focal / Ambient) was added as an expansion of the original model and while the model primarily discussed the visual and auditory channels for the dimension *modalities*, the expansion including other sensory channels have been discussed [172] and appear to be widely accepted.

Driving engages primarily the visual and kinaesthetic sense [157]. Kinaesthesia describes

the perception of position and movement of limbs, and often includes force perception [173]. Secondary tasks should, therefore, be designed to address other senses.

Typically, the distraction of a secondary task or newly designed feedback is measured during simulated driving, as it allows to precisely control many aspects of the driving environment, leading to similar and repeatable scenarios and the possibility of presenting dangerous situations without the risk of injury [174].

The influence of thermal feedback would mostly be of cognitive nature. Changes in driving behaviour, captured by example with lane deviation, and perceived workload ratings were chosen to determine the level of distraction posed by thermal feedback.

2.3.2 Autonomous Vehicles

Autonomous vehicles are expected to outnumber manual vehicles in the near future. The change has already begun, with many cars already being outfitted with features that can take control over driving aspects, such as lane keeping assistants, which take over lateral control of the car. The Society of Automobile Engineers $(SAE)^1$ has devised a categorisation of autonomous cars (see Figure 2.4²), which has become a reference point for discussion and research. The different levels of automation will be shortly discussed here.

Level 0 is driving without any kind of automation involved, depicting classical manual driving. The car can give warnings and momentary assistance, for example lane departure warning or automatic emergency braking.

In Level 1, *Driver Assistance*, the driver transfers the control of one specific task to the car, for example velocity control or lateral control, but the driver can interfere at every point and has the obligation to do so in dangerous situations. Examples for this would be lane centring or adaptive cruise control.

The next level sees two functions transferred to the system at the same, while the driver is still being expected to be available at any time and is supervising the system. This is called *Partial Automation* and would be achieved, for example, when both the lane centring and

¹https://www.sae.org/news/2019/01/sae-updates-j3016-automated-driving-graphic (accessed 09/2020) ²https://www.nhtsa.gov/technology-innovation/automated-vehicles-safety (accessed 21/09/2020)



Figure 2.4: SAE Levels of Autonomy.

adaptive cruise control were activated at the same time.

The Levels 0 to 2 describe driving situations in which the driver is in charge of the car, being expected to take full responsibility of the driving performance.

Level 3, *Conditional Automation*, is described as the state in which all driving tasks are fulfilled by the car, but the driver has to be available for taking over control when needed. An example feature for this is a traffic jam chauffeur, which can take over lateral and longitudinal control during a traffic jam.

Level 4 addresses *High Automation* situations in which the car is able to drive fully autonomous under certain conditions, such as specific areas with local driverless taxis.

The last level, *Full Automation*, describes a state in which the car takes full control of the driving task and the driver is not expected to be available at all.

In Levels 3 to 5 the car is in control of the driving, but in the lower Levels 3 and 4 the driver can still be involved. Situations in which control might have to be transferred between car and driver can, therefore, occur in vehicles with conditional and high automation. The transfer of control can be distinguished between *handover*, when the driver hands the control to the car, and *takeover*, when the driver takes back control from the car. These situations can be driver-initiated or car-initiated (system-initiated) and scheduled or non-scheduled [175, 176]. Research is highly focused on takeover scenarios, especially takeover in emergency situations. These takeover requests would only occur when the car has either detected a dangerous situation it cannot resolve on its own or the car has diminished situation awareness due to loss of sensor information because of damage or extreme environmental or weather conditions. In either case, it is paramount to inform the driver rapidly and effectively on potentially upcoming danger, as well as traffic situation and vehicle state. Section 2.4.3 will discuss research on using haptic feedback to aid the transfer of control.

The location of and potential use cases for haptic feedback for vehicles with autonomous features depends on the level of autonomy. Haptic feedback on the steering wheel would only be effective for Levels 0 to 4, with limited use cases for Levels 3 and 4. Feedback on other locations, such as the seat or seat belt, could be used in any level of autonomy. As the thesis focuses on presenting thermal feedback directly on the skin, interaction will be investigated only for vehicles within the first 4 autonomy levels.

2.3.3 Summary

Driving is a highly visual primary task, which imposes many demands on the driver. Secondary tasks can compete with the resources engaged in safely manoeuvring the car and are one of the major contributors to crashes. Visual distraction with high eyes-off-the-road time are especially risky. Engaging other senses with a secondary task can mitigate this effect not only by leaving the visual channels open for the primary task, but also by decreasing mental workload. Haptic feedback can be used as one of those alternatives. Eyes-off-theroad time is a widely used measure to evaluate the impact of a secondary task on driving, but this measure would be not very insightful when that task does not involve any visual elements. Mental workload would be a more appropriate measure to explore the influence of a haptic modality such as thermal feedback on driving. Perceived workload can be easily measured with questionnaires, such as the NASA TLX. This can be only applied when the questionnaire can be presented after being exposed to a single type of feedback. Another commonly used measure is lane deviation, investigating the influence of a secondary task on the driving performance. There are several ways to measure this factor, one of them being the Root Mean Square Error (RMSE) of the lane deviation. In addition, task performance of the secondary task is used to evaluate the effectiveness of the task and the posed distraction.

When investigating feedback for autonomous cars, the moments during and shortly after control is transferred from driver to car (handover) or car to driver (takeover) are of special interest. These control transfers can occur in cars with automation levels 3 and 4 and initiated by either driver or system. The next section will discuss research on haptic feedback used in the car. Before summarising practical applications during transfer of control, more common application areas will be discussed: haptic feedback for navigation and notification purposes in the car.

2.4 Haptic Feedback for In-Car Applications

The section discusses haptic feedback within the vehicle environment. Haptic describes in general both kinaesthetic and tactile stimuli [177]. As discussed before, the kinaesthetic sense tracks body movements and is highly engaged during the driving task. In the driving environment, torque feedback is the most often used kinaesthetic feedback. Torque is typically embedded into the steering wheel, where forced movements or jerks of the steering wheel are used to convey warnings or aid lateral control [178, 179, 180, 181, 182, 183, 184, 185, 186, 187]. Some force feedback can also be found on pedals [188]. Tactile cues involve stimulation of the skin. This cutaneous stimulation can be achieved through mechanical, chemical, electrical or thermal cues [177]. The presentation of thermal feedback will be more closely related to the presentation of other tactile cues in the car and the discussed research in this section will, therefore, focus on tactile feedback.

There are several classes of information in cars that could potentially be conveyed with haptic feedback [189]: spatial information, warning signals, communication (of driving related information), coded information (mostly abstract presentation of car state information) and general (such as guiding during use of the dashboard). These different kinds of cues can be presented on a number of locations in the vehicle environment: seat, seat belt, steering wheel and pedals. Touch-less tactile sensations such as air streams [190] or ultrasound [191] can be presented on any part of the body, even though the sensation might be hindered by clothing. Vibration is one of the most widely investigated tactile feedback types. Outside of the car, exploration of vibrotactile feedback led to guidelines [192, 193, 194, 195, 196, 197] and design explorations [198, 199, 200, 201, 202]. Within the in-car environment, possible locations have been tested for abstract identification and design (seat [203, 204, 205, 206], seat (or waist) belt [207, 31, 208, 209], steering wheel [210, 211, 212], pedals [213, 214, 215, 216]) and compared for different use cases [217, 218, 219]. Literature on haptic in-car feedback is vast and a detailed discussion would be out of scope of the thesis. Therefore, three specific and often applied use cases will be discussed. Section 2.4.1 *Navigation* will discuss directional information for lane changes and turn-by-turn navigation. Section 2.4.2 *Notifications* will summarise warnings and notifications to the driver and Section 2.4.3 *Transfer of Control* will focus on haptic interaction for autonomous vehicles.

2.4.1 Navigation

Haptic pedestrian navigation systems usually work with vibrating wearable [220, 221, 222, 223] or mobile devices [224, 225, 226, 227, 228]. Vehicle types other than cars, such as motorcycle [229], helicopter or boat [230, 231] often presented vibrotactile navigation cues on the waist or thigh belt [232]. In the car, waist bands, mostly to simulate or substitute seat belts [233, 8, 9], seats (with cues under the thigh [6, 7]) and steering wheels [10, 234, 11, 12, 235] have been used to convey vibrotactile navigation information. Alternative tactile modalities, such as shear or stretch feedback, were more commonly presented on the steering wheel [13, 236, 14], but could also be presented on the seat [158].

In the car, spatial information for haptic navigation was generally presented by feedback position: the side of the driver's body on which the cue was presented indicated the direction of the intended lane change [10, 13, 11, 7] or turn [189, 6, 233, 8, 12, 235, 158]. Sometimes, the explicit information *go straight* was added to the tactile dictionary, typically with a dynamic activation of several actuators in a forwards direction [189].

When vibrotactile cues were presented on the steering wheel, a balance of vibration intensity had to be struck: the vibration had to be intense enough to be felt on the hand, but needed to be subtle enough to keep the whole steering wheel from vibrating to ensure one sided cues could be identified [10].

Some tactile displays presented the direction information with dynamic patterns. Hwang *et al.* [11] embedded 32 actuators into the steering wheel and tested vibration patterns, evoking tactile illusions. Their findings showed that vibration on a single actuator in progression along the steering wheel towards the turning direction, invoked the sensory saltation illusion (as discussed in Section 2.1.3) and had the highest recognition rate and fast response times.

With 20 vibrotactile actuators, Kim *et al.* [235] used clockwise activation on the right steering wheel side for right, and counterclockwise activation on the left side for left turn instructions, again counting on sensory saltation to enhance the feeling of directional motion. In pilot tests, they compared tactile feedback to audio and visual feedback and all combinations thereof and found that the multimodal conditions had better preference ratings and performance and less cognitive load than single modality feedback. A follow-up study with multimodal feedback compared navigation of elder and younger drivers. They found that elder drivers were less impacted by visual presentation of feedback than younger, but had higher workload in every condition. Younger drivers preferred the visual-auditory presentation, already known from conventional navigation systems, but feedback including haptic cues performed better and faster. For elder drivers, the combination of audio and haptic feedback worked best.

Studies showed that vibrotactile cues reduced navigation errors in situations with auditory noise or distraction [12]. Additionally, in cognitively demanding navigation tasks, reaction times were shorter for non-visual feedback and haptic was rated as less physically demanding than audio and visual. While the error rate was highest for haptic feedback, ratings still showed participants preferred vibration [7]. In situations with high workload, tactile navigation reduced workload compared with visual navigation. The multimodal combination led to fastest reaction time [6].

Some studies investigated how the factor distance could be encoded into vibration. The factors rhythm [189, 6], intensity and duration of vibration pulses (or combinations of them [233, 237, 8]) have successfully been used to convey distance information. In addition, more complex turning scenarios, such as roundabouts have been considered and successfully navigated [12, 8].

Few tactile non-vibrotactile feedback types have been tested for directional cues in the car. Medeiros-Ward *et al.* [13] presented shear feedback on the steering wheel: actuators under the index finger moved the skin in the direction of the intended lane change. This feedback proved especially effective in situations with audio distraction.

Ploch *et al.* [236, 14] presented skin stretch feedback on the steering wheel: a rotating ring on the rim of the steering wheel was embedded to move clockwise or counterclockwise, stretching the skin along the wheel with its movement. Navigation and second task accuracy were higher for stretch feedback than audio cues.

Rotary motors and vibration actuators on a seat were used for navigation by Farooq *et al.* [158]. The vibration was used to catch the driver's attention, while the direction of motor rotation indicated the turning direction. The degree of rotation encoded information on how sharp the upcoming turns were and could even prompt U-turns. Secondary task performance was faster in the haptic and audio condition than visual, but no other differences were found.

Summarising their extensive work on tactile navigation, in and outside of the car, van Erp

& Werkhoven [238] discussed the benefits of vibrotactile feedback (on the torso) and concluded that it can be effectively used for local guidance navigation tasks, independent of the amount of visual load. Furthermore, it can reduce the effect of additional cognitive tasks and be reliably used in the presence of external stressors (such as spatial disorientation and night operations). These findings mirror results of the in-car navigation studies discussed here: tactile feedback for navigation had high recognition rates and in general reduced workload and recognition time, while being less influenced by demanding driving situations and environmental factors. Multimodal feedback including vibration was often preferred and had better performance.

Thermal feedback could be used well for feedback such as navigation, as routes are generally calculated beforehand and the sluggish response of thermal feedback could be easily counteracted by presenting it early enough. The first explorations within this thesis will, therefore, focus on directional cues during driving.

2.4.2 Notifications

This section will discuss notifications presented haptically to the driver. Some of these had informational character, mostly aiming towards increasing situational awareness [239] or improving driving behaviour [215, 240] independent of hazardous situations, but the tactile modality was most often investigated for time critical warnings, such as lane departure [4, 181, 15, 16] or collision warnings [17, 31, 23, 24, 241, 242, 25, 26, 18, 208, 209, 19, 243, 244]. Most tactile stimuli for warnings were vibrotactile, but a system for pneumatic steering wheel alerts was proposed by Enriquez *et al.* [245]: inflatable pads were embedded into the steering wheel, allowing the wheel to pulsate with differing frequencies, leading to faster reaction times.

Shakeri *et al.* [210] provided cutaneous push feedback on the steering wheel: small pins could protrude from the steering wheel to poke the palm of the driver. They tested patterns with differing number of pins and used them to give informational feedback on mid-air gesture input [246, 247].

Situation awareness of rear obstacles could be increased by presenting them haptically with servomotors in the back of the seat [239]. This led to a shape-change of the seat, presenting obstacles in a continuous way.

Economical driving was encouraged by presenting vibration cues on the accelerator pedal, triggered when the driver exceeded the 50% throttle threshold [215]. This measure resulted in reduced acceleration, both mean and maximum, and excess throttle, in addition to reducing perceived workload. Vibration on the seat or seat belt was proposed to reduce CO_2 emissions [240]. Vibrotactile feedback, on the seat belt especially, led to more fuel efficient driving, even without participants being aware of the purpose of the vibration.

Feedback design of differing levels of urgency investigated audio, visual and tactile warnings [248, 249]. For tactile warnings, the perceived urgency increased with shorter pulse interval. In addition, tactile warnings were rated as more urgent than audio and visual cues, without the high increase in annoyance that could be observed for audio cues.

The influence of speech and tactile feedback on the steering wheel and accelerator pedal for speeding warnings was tested in a simulator study [250]. Warnings of both modalities reduced the frequency of speed violations, but severe violations were only reduced by speech feedback. If speed reduction would lead to safer driving, both modalities performed the same, but vibrotactile feedback was better for situations that needed fast responses. While speech feedback showed an increase in workload, this was not observed for tactile feedback.

When designing warnings, vibrotactile feedback can be used to generally alert the driver, or give additional direction information on the problem. In navigation, the location of tactile cues usually indicated the direction towards which the driver needed to turn. For warnings, however, there are two possible meanings to the position of the cue: it could indicate the direction the driver should steer towards to mitigate the danger, or it could indicate where the danger can be found. Beruscha et al. [234] investigated participant's reaction to vibrotactile cues on one side of the steering wheel, once without and then afterwards with instructions. Instructions for two use cases (with and without driving) were presented: when vibration was introduced as lane departure warning, the cue indicated the side on which the car was steering off the lane (steering in opposite direction of cue needed) and in the second case the car was following a line on the road and the cue indicated in which direction to steer to stay on the line. The results of steering without instructions showed no pattern, the direction seemed randomly chosen. During driving, almost half the participants steered toward the vibration, approximately one fourth away and the rest did not steer at all after scanning for danger and not finding anything. The instructed events had less steering errors for the line driving, where the vibration indicated the direction of steering. This suggests that vibration by itself can be interpreted either way, the context decides on how the driver reacts. Suzuki & Jansson [4] had different findings: in their study unexplained vibration of the steering wheel was generally interpreted as lane departure. However, the vibration in their case was not directed, but presented on the whole steering wheel. Lane departure warnings are already implemented in cars and prove useful: an evaluation of police reported crashes found that lane departure warnings prevented crashes of all severities [5].

Research has favoured looking into steering wheel torque over using vibration to warn the driver about an imminent lane departure. Brown *et al.* [16] used steering wheel vibration and directional seat vibration, as well as steering torque, audio and visual cues and no alert. All kinds of alerts reduced the severity of lane departures, but seat vibration did not influence the extent of the departure.

Designing vibrotactile lane departure warnings for truckers to replace often turned off audio

warnings, Dass *et al.* [15] presented vibration on the seat as well, both in a simulator test and real world driving. They found that the feedback was as effective as auditory signals and clearly preferred.

Tactile collision warnings have been shown to be unaffected by noise [19] and less affected by conversations [241] with faster reaction times than visual or audio cues [24], and also in combination with visual [243] or audio [25]. Furthermore, the brake reaction time for unior multimodal cues with vibration was unaffected by visual orientation [243].

Directed vibration, mostly presented on a waist band, with actuators on the front and back, have led to faster and more correct responses when warning for front or back collisions [17, 31]. Dynamic vibrotactile patterns [208, 209], or multimodal audio-tactile presentation [23], especially when presented from the same direction [242], decreased brake reaction times further.

Multimodal feedback with different levels of urgency, combining visual, audio and vibrotactile cues, have been evaluated by Politis *et al.* [26, 18]. The more urgent cues led to faster responses, as did the multimodal cues compared to the unimodal feedback. Visual feedback slowed response time. Thermal cues, with their inherent association with danger, could enhance the feeling of urgency and in a multimodal setting, where a fast response might be evoked by other modalities and the additional emotional aspect could provide a richer experience.

Many different kinds of notifications during driving have been presented haptically, mostly through vibration and most effectively in multimodal settings. Warnings have been of special interest and could be enhanced with thermal feedback, combining the aspect of multimodality with the advantage of affective associations. The work on feedback with different levels of urgency inspired one of the experiments of this thesis. In addition, the novel cutaneous push feedback, only tested for notification feedback, was chosen as comparison for thermal feedback for navigation, leading to new findings for both modalities. The range of different locations for tactile feedback was adapted for placement of vibrotactile devices for the last experiment of the thesis.

2.4.3 Transfer of Control

This section discusses tactile feedback for the process of control transfer in semi-autonomous cars. Semi-autonomous cars (SAE Autonomy Level 3 and 4) are cars with autonomous features that can take control of the full driving process for specific tasks or under certain conditions, but the control might still have to be transferred to a driver. In these cars the control of the vehicle can be handed over from the driver to the car (handover) or from the car to the driver (takeover). This can lead to situations on the road in which drivers are deeply engaged in a secondary task, while in autonomous mode, and might then need to take over control

of the vehicle without a long adjustment period. If these takeover requests (TORs) were prompted because of a system failure or the car encountering a situation it was not equipped to deal with, an immediate and effective reaction from the driver would be essential. As this is a critical and dangerous situation, research has focused on using tactile feedback to interrupt potentially very engaging tasks [20, 251, 252, 253, 22, 254], give information needed after the takeover and increase situation awareness during takeover [255, 256, 257, 258, 259, 21, 260]. All of these measures ultimately aim to reduce reaction time and increase the quality of driving manoeuvres after takeover and tactile feedback was almost exclusively presented with vibration, generally on the seat [255, 256, 259, 21, 260, 253, 22, 254] or wrist [20, 252], with typically non-directional cues for interruptions and directional vibration (on one side) to present information on upcoming obstacles.

An alternative to vibrotactile feedback to present directional information during the takeover process was presented by Sadeghian Borojeni *et al.* [257] and compared to vibrotactile feedback on the steering wheel. They created a shape-changing steering wheel with plates on both sides, which would be placed under the driver's hands. The plates were attached on a single point in the middle and could rotate the upper part to the left or right, with the lower part moving into the opposite direction. While steering wheel feedback did not decrease reaction times during takeover, vibrotactile (but not shape-changing) feedback led to lower workload.

The investigation of effective interruptions found that vibration (in form of speech tactons) alone were rated as more annoying than in combination with audio or visual feedback. Multimodal cues, designed for different urgencies, were more effective with increasing urgency [20]. When comparing the influence of audio and tactile warning, tactile only led to slower response times, while multimodal combinations with visual and/or audio cues had similar response times [252, 253]. Visual feedback alone had significantly slower response times [251, 252, 253]

Multimodal cues including directional vibration led to faster reactions and increased situational awareness [256, 259, 21]. Basic, static vibration patterns performed better than dynamic patterns, were sets of actuators were activated in succession [255, 260].

Few research has looked into *how* the control was transferred. Transfer mechanisms are generally only mentioned in passing within the study design, detailed discussions are rarely found. Most studies use a single button press [261, 262, 20, ?, 263, 251, 22] or lever pull or switch [264, 257, 265] to transfer control when the transfer is being actively initiated. Often the use of the steering wheel or brake was enough to transfer control back to the driver [261, 262, 256, 259, 260, 21, 252, 253]. These techniques could easily be done unintentionally, possibly leading to seconds of uncertainty in which the driver is unsure or unaware of who is in charge of driving. Simultaneous double button press [266] could overcome this problem, as it takes conscious effort to press two buttons with two hands at the

exact same time.

Tactile feedback informing on the process itself could further aid to ensure that the driver is aware of the progress and completion of control transfer. Kerschbaum *et al.* [267] proposed a transforming steering wheel, which would very prominently announce the driving state and enhance comfort during the autonomous drive. They tested the concept and participants found no problem with the steering wheel during takeover and reacted faster than in the control condition with a conventional steering wheel. Thermal feedback could also help reassure the driver during the transfer of control, where the temperature change could symbolically present the process of control change.

The slower recognition of thermal feedback would suggest this modality for progress information rather than urgent takeover requests. In combination with vibration, especially on different locations within the car, the bimodal tactile feedback could inform the driver non-visually, lessening the workload. This was explored in the last experiment of the thesis.

2.4.4 Summary

Vibrotactile feedback in the car was generally used to capture attention and effectively achieved this task, as reduced reaction times in most use cases confirm. Spatial placement can guide attention if the context of use is clear, and faster pulse rate of vibration patterns can increase the feeling of urgency of the cue. Multimodal stimuli, adding visual or audio to vibration, often achieved better results than unimodal tactile feedback, though the unimodal variety still outperformed visual only feedback.

Especially navigation information can be effectively presented by other tactile feedback types, such as shear feedback. Thermal feedback could be used in this context as well, as navigation in driving is usually routed and feedback could be timed accordingly, rendering attention grabbing vibration unnecessary.

2.5 Literature Review Conclusions

Thermal displays have recently seen a rise in attention. While one of the most popular uses for temperature is the simulation of virtual thermal environments or objects, thermal cue design has also been explored as feedback for stationary and mobile devices and a number of reliably recognised factors have been identified, best among them direction of temperature change. Thermal feedback has been associated with emotion and different concepts such as danger, familiarity and navigation goals could be linked to warm temperature changes. But still, thermal feedback has not been used in the driving context, even though other tactile modalities, vibrotactile being the most popular, have not only been investigated widely in research, but have already found their way into modern cars for commercial use. The car presents a closed and well known environment, which could be easily mapped and thermal cues adapted to the thermal environment for optimal use. Vibration is a very attention grabbing modality, but the thermal sense is more sluggish and could be used to present information that does not need immediate attention. Navigational cues and notifications could be encoded into temperature changes and presented on the steering wheel or seat.

As driving is a highly demanding task, any use of new modalities has to be properly investigated and some questions answered. Does the modality have a negative influence on driving behaviour? Is it too cognitively distracting? How well can it be recognised while driving? Does the modality fit the use case? This thesis aims to answer some of these questions.

The research questions posed in this thesis combine aspects of the different sections discussed in this literature review. The complexity and characteristics of thermal sensing (discussed in Section 2.1), in accordance with findings from Section 2.2, will be combined with the specifics of driving related tasks (Section 2.4) and requirements posed by the high demand environment (Section 2.3). Research Question 1 will establish the basic suitability of thermal feedback during driving, by determining if simple binary cues (designed in accordance with [29, 268]) for direction can be used effectively during driving and what influence the presentation of thermal cues has on driving behaviour and workload in a simple driving task, the importance of which was discussed in Section 2.3.1.

Research Question 1: *How accurately can thermal feedback give binary direction cues in an automotive setting (with one thermal device)?*

Adding complexity to both the driving task and the thermal cues, in accordance with findings from Section 2.2.1, Research Question 2 will determine, how suited thermal feedback is for basic in-car applications such as turn-by-turn navigation, when presented with the spatial information of multiple thermal devices on different locations, inspired by tactile feedback research discussed in Section 2.4.1. The comparison with another tactile modality, cutaneous push, was inspired by [246, 210, 247].

Research Question 2: *How accurately can thermal feedback give direction cues in an automotive setting with the spatial information from multiple thermal devices?*

In the next step, thermal feedback will be compared and used in combination with vibrotactile feedback, inspired by Section 2.2.5. The complexity of the cues will be increased, investigating cue designs with several factors and associations with temperatures, as discussed in Section 2.2.2, to convey more complex notifications (Section 2.4.2). First, notifications of different levels of urgency will be investigated, inspired by and adapted from [26, 18], where the different modalities encode the same information. In the next step, notifications with different levels of information will be compared, influenced by the use case of [44] and strengthened by mappings found in [112, 111]. Finally, the influence of bimodal tactile feedback in a multimodal environment was tested to inform on the transfer of control process, filling the gap identified in 2.4.3, taking inspiration for the location from [215] and mapping from [113].

Research Question 3: *How effectively can thermal feedback convey notifications in a car in a*

- a. unimodal setting?
- b. bimodal tactile setting?

This incremental introduction of thermal feedback into the driving environment will provide a solid basis for further investigation of thermal cues within the car. The next chapters will discuss the user experiments conducted to explore the novel application of thermal feedback in the vehicle.

Chapter 3

Thermal Binary Directional Cues

3.1 Introduction

Thermal feedback, as shown in the previous chapter, has been used in previous research to give directional cues. However, this was not in an automotive environment, where driving is a primary task demanding a high level of attention and identification of thermal directional cues becomes a secondary task. Therefore, experiments were designed and conducted to gain first insights into how thermal feedback influences driving and what kind of problems drivers face when identifying directional cues. Direct translation of thermal navigation as used in former research by, for example, Wettach *et al.* [40] or Tewell *et al.* [134], is not possible for the automotive sector, as their approach was reactive rather than anticipatory: when participants left the correct path the temperature would change. In cars, however, directional cues have to be given explicitly with ample time to make preparations for a safe turn or lane change to avoid dangerous incidents or crashes.

The first two studies presented in this chapter were conducted using a single device, comparable to former research, to investigate how binary instructions given by the direction of temperature change, i.e. warm or cold change, can be utilised in an automotive environment. This factor showed high recognition rates in previous research for mobile environments and was used to test the suitability of thermal feedback during driving. These experiments aim to answer Research Question 1:

How accurately can thermal feedback give binary direction cues in an automotive setting (with one thermal device)?

This chapter describes these first two experiments utilising a single device for binary feedback, starting with the description of the apparatus. A conclusion will discuss the findings in Section 3.6.

3.2 Apparatus

The hardware and setup used for these two first experiments were the same: participants were seated in an adjustable office chair facing an 23.6-inch HANNS-G HL249 screen. Attached to the table was a Logitech G920 Driving Force steering wheel, connected to a DELL XPS 15 9550 laptop using Windows 10. The thermal feedback was provided by a 2x2cm Peltier element, a thermoelectric device, which was mounted on a heat-sink, see Figure 3.1(a).



(a) Peltier element with heat-sink

(b) Experimental Setup

Figure 3.1: Experiment 1 and 2: Hardware and experimental setup.

The control board setup was built by SAMH Engineering and details can be found in Appendix A.1.1. Communication with the control board was facilitated via Python 2.7. The device was placed on the tabletop on the right side of the participants, enabling them to touch the Peltier with the index finger of their right hand while driving one-handed with their left, see Figure 3.1(b). Throughout the experiment, participants wore Sennheiser HD 25-1 II Basic Edition headphones playing car noises during driving and audio instructions, if applicable. The audio instructions were created with the CereProc Cloud API¹. The driving scenario was implemented in the Java-based OpenDS 3.5 free version² simulator, the communication with the Python-driven board was achieved via sockets. The driving environment depicted an empty five lane motorway, see Figure 3.2. The simulated car maintained a constant speed of 100km/h, participants were not provided with accelerator and brake pedals.

¹https://www.cereproc.com/products/cloud, accessed 06/04/2020

²https://opends.dfki.de/, accessed 06/04/2020



Figure 3.2: Empty five lane motorway, implemented in OpenDS.

3.3 Experiment 1: Thermal Lane Change

This first experiment investigated if the presentation of thermal feedback during driving influenced driving performance and how well thermal directional instructions could be followed. The state-of-the-art non-visual feedback for navigation currently used in cars is speech. Therefore, the performance of thermal feedback was compared to this as a control condition.

3.3.1 Methodology

The study was designed as a within-subjects experiment with feedback type as the independent variable, where feedback type was either thermal or audio. The main goals of the experiment were to explore if the presentation of thermal feedback impacted driving performance and perceived workload. Furthermore, recognition rate, preferences regarding the two feedback types and time differences between them were investigated.

The experiment was split into two parts with both feedback types being presented in each 10 times. In the first part, participants kept in the middle lane and orally reported the identified cue to the experimenter, in the second part participants fulfilled lane changes according to the feedback they perceived. The feedback types themselves were not mixed, participants had distinctive driving blocks within each experimental part with only one feedback type, or condition, presented. The order of thermal and audio driving blocks in each part of the study was decided by iterating through all possible permutations, counterbalancing the conditions.

It was hypothesised that thermal feedback, as a novel tactile feedback, would be rated as more distracting than audio feedback and show more lane deviation and higher perceived workload. The thermal cues gradually increased, whereas speech feedback is abrupt in nature, so thermal feedback was expected to be rated as less disruptive and more pleasant. As a consequence, however, the time needed to complete a lane change would differ between the two conditions, with the gradual thermal feedback taking longer, as detection time would be increased. In addition, as previous literature has shown that there might be differences in perception between genders [52], results were compared between participating genders. Participants also were posed a question on thermal preferences regarding navigation, which was added as preparation for later study designs. Therefore, the hypotheses for this study were:

Hypothesis 1: Thermal feedback will lead to higher lane deviation and perceived workload than auditory cues;

Hypothesis 2: The time taken to complete a lane change will be longer in the thermal condition;

Hypothesis 3: Thermal feedback will be rated as less disruptive and more pleasant than the audio feedback;

Hypothesis 4: More participants will choose to turn towards the direction of the warm side of a thermal steering wheel.

The dependent variables evaluated for this experiment were: lane deviation, recognition rate, time to complete lane change, perceived workload and subjective ratings. Detailed descriptions of the dependent variables can be found in Section 3.3.4 *Measures*.

3.3.2 Driving Task and Feedback

The experiment was divided into two parts: in the first, participants should stay in the middle lane and report orally any feedback they received. This would ensure the best capturing of lane deviation to evaluate the distraction posed by the presentation of the feedback itself. In the second part, participants were asked to change lanes according to the feedback they identified. Participants were asked to drive one-handed with their left, placing their right index finger on the thermal device throughout driving, independent of condition.

10 invisible trigger elements were placed on the simulated road in regular intervals. After triggering one of these elements, the lane change instruction was presented within a random time interval between 0s and 4s. The direction of lane change was decided randomly, only restricted when the car was in one of the outer lanes and needed to change towards a specific direction to stay on the motorway.

The thermal feedback in both parts was identical and based on previous research [29, 268]: the Peltier devices were set to the neutral base temperature of 30°C at the beginning of the driving block. This fixed temperature, presented before and in between each cue, would



Figure 3.3: Experiment 1: Thermal feedback.

ensure that the skin in contact with the device adapted to the device temperature and led to controlled temperature changes on this location, independent of previous skin temperature. When a lane change to the right was requested, the thermal device would warm to 36°C with a rate of change of 3°C/s, resulting in 2s of temperature change (Figure 3.3 red line). This temperature was kept constant for 8s and was then returned to neutral at the same rate of change. The thermal feedback for leftward lane changes was identical in structure, but the temperature would decrease and cool to 24°C, see Figure 3.3 blue line. The complete stimulus, including the return to neutral, was introduced to participants as a single cue before their first thermal driving block.

In the audio condition, the voice feedback, uttered by a synthesised male voice, differed in the two parts. In the second part, where actual lane changes were required, the instructions were *right* and *left*. However, in comparison to the more abstract and novel thermal feedback the audio instructions were considered to potentially elicit a stronger response and unintended lane changes and, therefore, increase lane deviation in the first part. The speech instructions of the first part were *warm* and *cold*, respectively, to counteract this and to adapt it to the thermal feedback.

3.3.3 Participants and Procedure

The experiment was completed by fourteen participants, for details consult Table 3.1. All participants were right-handed, had at least corrected vision and reported no sensory impairments in their hands. In addition, they all held valid driving licenses: nine participants

	Number of	Age	Driving	Experience	Experience	Experience
	Participants	(Years)	Experience	Simulator	Audio	Thermal
			(Years)			
All	14	<i>R</i> : 20-34	<i>R</i> : 1-17	<i>R</i> : 1-5	<i>R</i> : 1-5	<i>R</i> : 1-3
		M = 26.36;	M = 7.71	M = 2.00	M = 2.93	M = 1.18
		SD = 4.68	SD = 4.92	SD = 1.29	SD = 1.63	SD = 0.54
Female	8	<i>R</i> : 20-32	<i>R</i> : 1-17	<i>R</i> : 1-4	<i>R</i> : 1-5	<i>R</i> : 1
		M = 24.75	M = 6.88	M = 1.75	M = 2.75	M = 1.00
		SD = 4.77	SD = 6.03	SD = 1.16	SD = 0.75	SD = 0.00
Male	6	<i>R</i> : 23-34	<i>R</i> : 5-14	<i>R</i> : 1-5	<i>R</i> : 1-5	<i>R</i> : 1-3
		M = 28.50	M = 8.83	M = 2.33	M = 3.17	M = 1.42
		<i>SD</i> = 3.94	SD = 3.06	SD = 1.47	<i>SD</i> = 1.57	SD = 0.80

Table 3.1: Experiment 1: Detailed participant data, showing range (R), mean (M) and standard deviation (SD).

obtained them in countries with right-side driving, five in left-side driving countries.

Demographic data including gender, age, years of driving and were collected in addition to their experience with the technologies used (simulators, audio feedback and thermal feedback) prior to the experiment on a 5-point Likert scale (1 equalled none, 5 much). Participants reported their gender as either *female* or *male*, no participant chose the option *other*.

Recruitment of the participants was done via advertisement on university student pages and posters in the library, the School of Computing Science and the School of Psychology at the University of Glasgow. Participants were greeted in a usability testing room in the School Computing Science and presented with an information sheet. They signed a consent form and filled in the first questionnaire. Afterwards, participants were given time to get used to the driving simulator. They could drive on the empty five-lane motorway without any feedback present until they felt comfortable. Depending on the counterbalanced schedule, they then were introduced to their first feedback type, followed by the driving task for that condition. During driving, the feedback was presented 10 times, with the direction of lane change being random. After a short break the participants repeated this procedure for the second condition. At the introduction of the thermal feedback, it was made sure that participants could feel the feedback and identify the direction of the temperature change and also that they were familiar with the return to neutral.

In this first part, participants reported the identified feedback orally to the experimenter. After a short break, participants started the second part of the experiment, in which they were asked to turn one lane towards the direction indicated by the feedback, when prompted. Again, the counterbalanced schedule decided on the first condition.

In the break between the conditions of this second part, participants filled in the NASA TLX questionnaire and the additional scale for subjective ratings, capturing their impressions

of the just experienced feedback type. The same procedure was repeated for the second feedback type. At the end of the experiment, participants were presented with an additional questionnaire, capturing overall ratings and comments.

The documents used in the experiment, including information sheet, consent form and all questionnaires, can be found in Appendix A.2. The experiment design was approved by the Ethics Committee of the College of Science and Engineering of the University of Glasgow. The study was completed within one hour and each participant was paid £6.

3.3.4 Measures

The dependent variables were evaluated for all participants and then compared between the genders. One of the variables chosen to investigate the driver distraction was lane deviation: if drivers were distracted by the feedback identification, their driving performance would be less stable and lane deviation would increase. The driving logs generated by the driving simulator in the first part of the experiment provided the data for this. The deviation was calculated by comparing the Root Mean Square Error (RMSE) of the logged x-positions, with the value 0 describing the middle of the lane. A baseline lane deviation of 5s driving was taken before the onset of the stimulus and was then compared to time frames corresponding with the different levels of thermal changes: 2s of thermal change to 36°C or 24°C, 8s of constant temperature and 2s of returning to the neutral temperature of 30°C. This would allow a more detailed comparison of driving performance at different stages of the feedback.

Perceived workload was also captured as an indicator for distraction. The NASA TLX questionnaire [170] was utilised to collect perceived workload data. Participants rated the presented feedback on a 10-point Likert scale for the factors *Mental Demand*, *Physical Demand*, *Time Pressure*, *Effort*, *Performance*, *Frustration* and *Annoyance*. In addition, they were presented with a 5-point Likert scale, capturing subjective ratings on how *Pleasant*, *Comfortable*, *Disruptive* and *Complicated* the feedback felt to participants.

Furthermore, logging data from the second part of the experiment were utilised to calculate the time needed to complete a lane change. The time was taken from the onset of the stimulus triggering the lane change until the car had fully entered the lane. Recognition rates, i.e. correct lane changes, were also captured via logging data in the second part of the experiment. The oral reports of the first part were used to capture recognition there.

Participants were also asked to respond to the following question:

Imagine you were presented with thermal feedback on the steering wheel for navigation purposes, where the devices of one side of the wheel will be warmed, while the other side will be cooled, how would you interact:

If the right side of the steering wheel was warmed, while the left side was cooled, I would

turn to the: left/right

This question about participants' preference for the mapping of turning direction to direction of temperature change was posed to help design future experiments and confirm the mapping used in previous research [40, 134]. It was expected that participants would prefer to turn towards the warm side, as this was the mapping not only used in previous research, but also in children's games such as *Hot or Cold*. Participants also rated the different feedback types in the final questionnaire, asking *Please rate how much you liked the different instruction type* with an attached 5-point Likert scale, and asked to name reasons for their rating. In addition, they were asked to share ideas for applications using thermal feedback and locations where they thought it should be presented and were given ample space to comment on their experience. All questionnaire data (including NASA TLX) will be discussed in Section 3.3.5 *Results* under *Subjective Data*.

3.3.5 Results

All experiments in the thesis were newly evaluated using R³. This step was taken to ensure consistency and transparency throughout the thesis, as different types of software (SPSS⁴, JASP⁵ and R) were used during the course of the PhD, some of which were proprietary and/or the models used for evaluation were not reproducible. Results may therefore vary from results presented in the published papers. Substantial differences will be pointed out.

The comparisons were done with t-tests (t(df)), when the data were normally distributed, and Wilcoxon (V) tests otherwise. Shapiro Wilks tests were utilised throughout the thesis to determine normality. Paired versions of the tests were done between conditions, for all participants and for both gender groups individually, while unpaired tests were utilised for comparisons between the genders. For unpaired t-tests, R used Welch t-tests, for which the data was automatically adjusted for unequal variance between the two groups (adapted degree of freedom). Likert scale and NASA TLX data were compared with Wilcoxon tests. Participant comments throughout the thesis will be presented with typos removed and capitalisation adapted where needed.

Lane Deviation

The lane deviation of the four time frames *before* the stimulus, *during* the temperature change, during the *constant* temperature and at the *return* to neutral were compared with t-tests (t(df)) and Wilcoxon (V), for non-normally distributed data, between the feedback types

³https://www.r-project.org/

⁴https://www.ibm.com/products/spss-statistics

⁵https://jasp-stats.org/

	Before	During	Constant	Return
Condition All	<i>V</i> = 29	t(13) = 1.11	t(13) = 1.39	t(13) = 1.02
	p = 0.15	p = 0.29	p = 0.19	p = 0.33
Condition Female	t(7) = 0.66	t(7) = 1.07	t(7) = 1.82	t(7) = 0.20
	p = 0.53	p = 0.32	p = 0.11	p = 0.85
Condition Male	t(5) = 1.58	t(5) = 0.59	t(5) = 1.03	V = 4
	<i>p</i> = 0.18	p = 0.58	p = 0.35	p = 0.22
Gender Thermal	t(8.12) = 0.42	t(9.19) = 0.19	t(5.91) = 0.64	t(7.72) = 0.73
	p = 0.69	p = 0.85	p = 0.55	p = 0.49
Gender Audio	t(9.73) = 0.31	t(10.07) = 0.47	t(6.84) = 0.01	V = 21
	p = 0.76	p = 0.65	p = 0.99	p = 0.76

Table 3.2: Experiment 1: Statistics for Comparison of Lane Deviation, captured during the first part of the experiment.

for all (*Condition All*) and each individual gender (*Condition Female / Condition Male*) and for each condition between the genders (*Gender Thermal / Gender Audio*). No significant differences were found, see Table 3.2.

Time to Complete Lane Change

The overall mean time to complete a lane change was 4.37s. In the thermal condition, the lane changes took significantly longer (M = 5.28s, SD = 0.75) than audio (M = 3.46s, SD = 0.42) (t(13) = 12.82, p < 0.001). The lane change was 1.82s faster in the audio condition. Comparisons with t-tests of feedback type within genders showed significant results (female: t(7) = 8.34, p < 0.001; male: t(5) = 11, p < 0.001), where women changed 1.68s faster in the audio condition and men 2.02s, see Figure 3.4. The t-test comparison between the genders showed also a significant difference for both conditions (thermal: t(11.02) = 5.9, p < 0.001; audio: t(7.00) = 21.67, p < 0.001). In the thermal condition women changed 0.21s faster than men, while men changed 0.1s faster than women in the audio condition.

Recognition

Figure 3.5 shows all recognition rates. Recognition for audio feedback in both parts of the experiment was 100%: all speech stimuli were interpreted correctly. Thermal feedback had a recognition rate of 97% in the first part of the experiment. 4 out of 140 stimuli were erroneous: 3 stimuli were missed and one wrongly identified. Two of these (the wrong identification and one missed stimuli) occurred within the female, two within the male participants. This leads to a recognition rate of 98% for the female and 97% for the male participants. The difference in recognition between conditions was tested with Wilcoxon and not statistically significant (V = 6, p = 0.17) for the first part of the experiment, neither was the comparison



Figure 3.4: Experiment 1: Time to Complete Lane Change; error bars always show the standard error.

within (female: V = 1, p = 1; male: V = 3, p = 0.35) and between the genders (V = 20, p = 0.53). In addition to these identifications, 25 additional temperature changes were reported: 18 by women and 7 by men. These occurred as a result of misinterpreting the return to neutral as a new stimulus, a false positive. The difference in gender for this was not statistically significant (V = 29, p = 0.54).

While all speech stimuli were recognised correctly again with lane changing in the second part of the experiment, 16 thermal stimuli were missed (12 by women and 4 by men), resulting in a recognition rate of 88% overall, 85% for women and 93% for men. This difference was tested with Wilcoxon and statistically significant (V = 36, p = 0.01) for conditions for participants, but the comparisons within (female: V = 15, p = 0.06; male: V = 6, p = 0.17) and between genders (V = 29.5, p = 0.49) were not significant. The number of false positives went down to 8: 6 occurred when temperatures were returning to neutral from cold stimuli and 2 from warm. All these false positives were identified by female participants.

Subjective Data

The perceived workload was collected with NASA TLX questionnaires, see Figure 3.6, and evaluated with Wilcoxon tests. Overall workload showed significant differences (V = 11, p = 0.007). Detailed results of the comparisons can be seen in Table 3.3. Mental Demand (median(audio) = 3.00, median(thermal) = 7.00) showed statistically significant results between the two conditions thermal and audio for all participants, as did Performance (median(audio) = 8.00, median(thermal) = 5.00) and Frustration (median(audio) = 1.25, median(thermal) = 3.75). None of these categories was significantly different within the gender groups and between genders. The only statistically different results in the comparison between genders was found for Physical Demand for thermal feedback: the male participants rated thermal feedback significantly more physically demanding than female participants (median(female))



Figure 3.5: Experiment 1: Recognition Rate; significant results are in bold font.

	Mental	Physical	Time	Effort	Perfor-	Frustra-	Annoy-
	Demand	Demand	Effort		mance	tion	ance
Condition	<i>V</i> = 4.5	<i>V</i> = 33.5	<i>V</i> = 34	<i>V</i> = 20.5	<i>V</i> = 8.5	<i>V</i> = 7.5	<i>V</i> = 9.5
All	<i>p</i> = 0.007*	<i>p</i> = 0.69	<i>p</i> = 0.94	<i>p</i> = 0.16	<i>p</i> = 0.02*	<i>p</i> = 0.03*	p = 0.14
Condition	<i>V</i> = 3	<i>V</i> = 25.5	<i>V</i> = 6.5	<i>V</i> = 11	V = 6	V = 4	<i>V</i> = 2
Female	p = 0.08	<i>p</i> = 0.32	<i>p</i> = 0.46	<i>p</i> = 0.67	p = 0.20	<i>p</i> = 0.11	<i>p</i> = 0.17
Condition	V = 0	V = 0	<i>V</i> = 10.5	<i>V</i> = 1	V = 0	<i>V</i> = 1	<i>V</i> = 3
Male	p = 0.06	p = 0.10	p = 0.50	<i>p</i> = 0.10	<i>p</i> = 0.06	p = 0.20	<i>p</i> = 0.58
Gender	<i>V</i> = 18	<i>V</i> = 40	<i>V</i> = 14.5	V =27.5	<i>V</i> = 30.5	<i>V</i> = 24.5	<i>V</i> = 18
Thermal	p = 0.47	<i>p</i> = 0.04*	<i>p</i> = 0.24	p = 0.70	p = 0.44	<i>p</i> = 1.00	<i>p</i> = 0.47
Gender	<i>V</i> = 24	V = 21	<i>V</i> = 27.5	<i>V</i> = 31	V = 27	<i>V</i> = 19.5	V = 26.5
Audio	p = 1.00	p = 0.75	<i>p</i> = 0.69	p = 0.40	p = 0.7	p = 0.60	<i>p</i> = 0.79

= 2.25, median(male) = 4.75).

Table 3.3: Experiment 1: Statistics for NASA TLX rating of workload; significant results are bold and marked with an asterisk.

The additional subjective ratings were collected on a 5-point Likert scale and the results can be seen in Figure 3.4. Results of the statistical analysis with Wilcoxon tests can be found in Table 3.4. Only the factor Complicated was found to be statistically significant for all participants as well as within both gender groups (all: median(audio) = 1, median(thermal) = 3.00; female: median(audio) = 1.25, median(thermal) = 2.25; male: median(audio) = 1.00, median(thermal) = 3.50). In the comparison between genders, Disruptive was found to be significantly different, where male participants rated audio to be more disruptive than female participants (median(female) = 1.00, median(male) = 2.25).



Figure 3.6: Experiment 1: Perceived workload; significant results are in bold font.

The final rating at the end of the experiment, see Figure 3.8, showed no significant difference between conditions for all participants (V = 73, p = 0.54) and within gender (female: V = 22.5, p = 0.17; male: V = 17, p = 0.20), as well as between genders (V = 20, p = 0.64).

When asked for reasons of their rating, participants mentioned that thermal required more concentration. P03, for example, stated they had to concentrate a lot more for thermal difficult to judge if it had gone hot/cold whereas audio was much clearer and PO2 wrote that they found the thermal instructions more difficult to follow, requiring much more concentration. Sometimes it was hard to distinguish if the temperature was going back to neutral or was getting warm. On the other hand, audio instructions were clear to understand. Problems with distinguishing between the return to neutral and new stimuli were also reported by P08: Thermal instructions are unclear - not sure it an actual instruction, or just returning to 30 degree and P09 agreed: Thermal was tricky sometimes because I wasn't sure when the temperature had gone back to neutral. Participants also commented on uncertainty in the thermal condition that they did not experience in the audio condition: At times I wasn't so sure about the temperature. Basically, didn't trust my sense of touch (P06) and the audio made me nervous, but I clearly new were I was supposed to go. Because the thermal stimuli was similar to me I was not sure when I was supposed to turn although I was less nervous (P05). The more disruptive nature of the audio feedback described here by P09 was affirmed by other participants, who wrote audio can be a bit annoying and demands some concentration while thermal cannot interfere with other senses or activity (P01). The speech feedback was familiar to participants, as P10 mentioned that it was easier to follow the audio instructions, possibly because I am used to driving with GPS. P14 wrote that audio was like


Figure 3.7: Experiment 1: Subjective Additional Ratings; significant results are in bold font.



Figure 3.8: Experiment 1: Final rating for both feedback types.

listening to a sat nav and *P12* commented that they *found the thermal type distracts from my focusing on driving. Maybe because I was not familiar with it.*

When answering the question about preferred turning direction, most participants answered *Right*, favouring the mapping of warm temperature changes to direction of turn. Only one participant chose *Left*.

Participants had several suggestions for the use of thermal feedback in the car. *P02* suggested using thermal interaction to give feedback on distances, for example when parking, which was also mentioned by *P07*. *P03* suggested that warming the steering wheel when speeding might be useful. *P14* had several ideas: *How far away you were from hazards - plan routes, avoid traffic, outside weather conditions*. Hazard warnings were also mentioned by *P11*,

	Complicated	Pleasant	Disruptive	Comfortable
Condition All	V = 0	<i>V</i> = 52	<i>V</i> = 17	V = 30.5
	p = 0.002*	<i>p</i> = 0.09	p = 0.16	p = 0.37
Condition Female	V = 0	V = 23.5	V = 2	V = 6
	p = 0.04*	p = 0.12	p = 0.09	p = 0.78
Condition Male	V = 0	<i>V</i> = 7	V = 6.5	<i>V</i> = 9
	p = 0.04*	<i>p</i> = 0.58	p = 0.90	p = 0.20
Gender Thermal	V = 36.5	V = 27.5	<i>V</i> = 31.5	V =21
	p = 0.11	<i>p</i> = 0.67	<i>p</i> = 0.36	p = 0.74
Gender Audio	<i>V</i> = 26	V = 27.5	<i>V</i> = 5.8	<i>V</i> = 13.5
	p = 0.83	<i>p</i> = 0.67	<i>p</i> = 0.04*	p = 0.19

Table 3.4: Experiment 1: Statistics for Subjective Additional Ratings; significant results are bold and marked with an asterisk.

P10 and *P05*, whereas *P09* would also see use in providing information regarding weather conditions. *P11* would like to be reminded to turn on the lights, while *P13* would prefer thermal feedback on the gear stick to indicate they should change gears.

Overall, 5 participants could imagine thermal feedback being presented on the gear stick. However, the steering wheel was the most popular location (9 participants). In addition, 4 participants named the seat, and one participant each the seat belt, gear pedals and door handle.

3.3.6 Experiment 1 Discussion

This first experiment compared thermal and speech feedback for lane change instructions to gain some first insights into the suitability of thermal feedback during driving. The sample size in the experiment (and all others discussed in this thesis) was comparably small, a higher and more diverse number of participants could make the results more robust and might show different results. However, the experiments in this thesis present interesting first insights into the basics of thermal feedback within the car environment. More discussion on limitations of the studies can be found in Section 6.5 *Limitations and Future Work*.

Temperature changes for in-car interaction, opposite to speech feedback, was a novel concept and needed additional mapping. Therefore, it was hypothesised that thermal feedback would show an increased level of distraction from the primary task of driving, which would influence the driving performance and show differences in lane deviation as well as perceived workload. However, the lane deviation showed no statistically significant differences between the conditions for any of the observed time frames. No definite conclusion could therefore be drawn. However, participants commented on having to concentrate more to follow the thermal instructions, which could have a greater impact on driving in more complex driving scenarios. Perceived workload showed differences in mental demand, performance and frustration. Thermal feedback was rated worse than speech in all these categories, indicating higher workload. *Hypothesis 1* could, therefore, be partly corroborated.

Lane changes took an average of 1.82s longer in the thermal condition than in the audio condition, as claimed in *Hypothesis 2*. The difference for the female participants was statistically significantly smaller than for the male participants, with 1.68s and 2.02s, respectively. The gender differences within the conditions were also different: men were faster than women in the audio and women faster than men in the thermal condition.

The recognition rates did not significantly differ between the genders. However, the faster lane change times of the female participants could indicate a faster detection of temperature changes. Furthermore, male participants rated thermal feedback as more physically demanding, which adds to the impression that women could be more sensitive to thermal feedback than men. Interestingly, only the male participants rated audio feedback as more disruptive than thermal feedback, seemingly showing more sensitivity towards audio feedback than female participants.

Hypothesis 3 expected thermal feedback to be rated as less disruptive and more pleasant than speech. However, there were no significant differences found for pleasantness ratings. Some participants commented on how the audio feedback made them nervous or could be annoying. However, only the comparisons between genders found differences for disruptiveness: men rated audio feedback as more disruptive then women, but not more than thermal feedback.

Participants of both genders agreed on thermal feedback being more complicated than audio feedback. One of the reasons for this was the additional mapping needed to translate temperature change to lane change direction. Familiarity with speech navigation systems was named as another one. However, the most prominent reason seems to be uncertainty connected to thermal feedback, but not speech navigation. The return to the base temperature needed between thermal cues was confusing to participants and resulted in additional 25 reported temperature changes in the first and 8 additional lane changes in the second part of the study. The cues had the same rate of change for the temperature changes towards the goal temperature and when returning to the base temperature, which made the return feel the same as a new stimulus in the opposite direction of change. A less symmetrical design could help minimise the number of false positives. In addition to this, some thermal cues were also missed in both parts of the study, while audio feedback had perfect recognition rate, indicating the need to further improve the cue design.

At the end of the experiment, a question regarding the preferred mapping of direction of temperature change to turning direction was posed. *Hypothesis 4* claimed that more participants would choose to turn towards the warm side, and all but one participant confirmed this mapping.

The results of this first study clearly show that thermal feedback has potential for in-car use, with recognition rates of 97% in the first and 88% in the second part. However, the return to the neutral temperature was identified as a source of confusion and misinterpretation. A second experiment was therefore devised, aiming to explore different aspects of thermal cue design in order to minimise the number of false positives and maximise the recognition rate.

3.4 Experiment 2: Different Stimuli Designs

In this experiment, several design factors for thermal feedback were explored to investigate their influence on recognition rate and number of false positives. In addition to direction of temperature change, three other factors where tested: length of temperature presentation, extent of temperature change and rate of temperature change at the return to the neutral temperature.

3.4.1 Methodology

The study was designed as a within-subjects experiment with the four independent variables direction (*DIR*), extent (*EXT*), length (*LEN*) and rate (*ROC*) of temperature change.

The rate of temperature change was expected to have an influence on the number of false positive recognitions: the slower the temperature would return to neutral, the less conspicuous the change would be and the less it would be interpreted as a new stimulus. In addition, it was hypothesised that a shorter presentation of the temperature would lead to less false positives, as the skin had less time to adapt to the presented temperature and would therefore be less sensitive to the change back. The recognition rate was expected to be mostly influenced by the extent of temperature change: the bigger the change in temperature, the better it should be recognised. The hypotheses for this experiment, therefore, were defined as:

Hypothesis 1: The slowest return to the neutral temperature will have the smaller number of false positive lane changes at the return to the neutral temperature;

Hypothesis 2: A shorter presentation time will have the smallest number of false positive lane changes at the return to the neutral temperature;

Hypothesis 3: The higher extent of temperature change will have the better recognition rate.

Dependent variables were: recognition rate, number of false positives and time to complete lane change. As in the first experiment, the results were analysed for the two genders male and female. More information on the dependent variables can be found in Section 3.4.4 *Measures*.

3.4.2 Driving Task and Feedback

The driving scenario was similar to the first experiment. Participants again drove on a fivelane motorway and asked to change lanes depending on the temperature change they felt: warm thermal cues directed towards the next rightward lane, cold temperature changes towards the next leftward lane.

The factors for the thermal feedback were:

- *DIR*: direction of temperature change, either warm or cold
- *LEN*: length of temperature presentation, either $\underline{0}$ s, $\underline{3}$ s, or $\underline{6}$ s
- *EXT*: extent of temperature change, either $\underline{3}^{\circ}C$ or $\underline{6}^{\circ}C$
- ROC: rate of temperature change, either 1°C/s (slow) or simulated 0.5°C/s (angled).

Figure 3.9 visualises all combinations of these factors. Because of limitations of the hardware, which only enabled rate of changes of either 1°C/s or 3°C/s, any rate slower than this had to be simulated. The angled rate of change of 0.5°C/s was achieved by changing 1°C with a rate of change of 1°C/s and then keeping this temperature for 1s. This procedure was repeated until the neutral temperature was reached, see dashed lines in Figure 3.9.



Figure 3.9: Experiment 2: Thermal feedback.

A labelling system was introduced with describing DIR-LEN-EXT-ROC. A cold temperature change of 3° C, lasting 3s and returning to neutral with an angled rate of change would be labelled *c*-3-3-*a*. Every combination was presented to participants three times, resulting in 72 cues in total. The order of the stimuli was randomised.

	Number of	Age	Driving	Experience	Experience
	Participants	(Years)	Experience	Simulator	Thermal
			(Years)		
All	16	<i>R</i> : 19-35	<i>R</i> : 0-17	<i>R</i> : 1-5	<i>R</i> : 1-4
		M = 25.88;	M = 5.28	M = 3.25	M = 2.00
		<i>SD</i> = 5.06	SD = 5.20	<i>SD</i> = 1.29	<i>SD</i> = 1.32
Female	8	<i>R</i> : 19-35	<i>R</i> : 1-17	<i>R</i> : 1-5	<i>R</i> : 1-4
		M = 24.75	M = 6.13	M = 3.00	M = 2.00
		<i>SD</i> = 5.63	SD = 6.17	SD = 1.31	SD = 1.31
Male	8	<i>R</i> : 21-35	<i>R</i> : 0-14	<i>R</i> : 1-5	<i>R</i> : 1-4
		M = 27.00	M = 4.44	M = 3.55	M = 2.00
		SD = 4.50	SD = 4.27	SD = 1.31	SD = 1.41

Table 3.5: Experiment 2: Detailed participant data, showing range (R), mean (M) and standard deviation (SD).

3.4.3 Participants and Procedure

Sixteen newly recruited participants, eight female and eight male, completed the experiment. Their age range, driving experience and prior experience with driving simulators and thermal feedback (on 5- point Likert scale with 1 being no experience) are presented in Table 3.5. All participants held valid driving licenses and at least corrected vision. They reported no sensory impairments and were right-handed. Eleven participants obtained their driving licenses in countries with right-side driving.

Recruitment was done through mailing lists and university web pages. The study was conducted in the School of Computing Science at the University of Glasgow. Participants were greeted and presented with the information sheet. After thoroughly reading through it, they signed the consent form. Participants were given some time to get used to the driving simulator, driving on the empty motorway until they felt comfortable. This took approximately between 1min and 3min. Afterwards, the different stimuli were introduced in detail. The driving task itself was divided into 8 blocks with 9 randomly chosen stimuli each, lasting about 5min and divided by short breaks. At the end of the experiment participants filled in a questionnaire collecting demographic data. The questionnaire, as well as the information sheet and consent form can be found in Appendix A.3. The experiment took approximately an hour and participants were paid £6.

3.4.4 Measures

The dependent variables investigated in this study were extracted from the logging files of the driving simulator and calculated similarly to the first study. Recognition rates and the number of false positives were the main interest. In addition, the time to complete lane change was calculated, as affirmation of the value found in the first experiment. The results were again evaluated for all participants and the two represented genders.

3.4.5 Results

Thermal perception of warm and cold temperatures varies, so the data were evaluated separately for warm and cold stimuli with repeated measures analysis of variance (ANOVA), testing for variables LEN, EXT and ROC. For comparison of gender differences, Gender was added as between subject factor. *Post hoc* tests were conducted with Tukey adjustments, automatically provided by R, to adjust for multiple comparisons. Outliers varying more than two standard deviations from the mean were removed before evaluation. There was only one outlier for Time to Complete Lane Change.

Recognition

Recognition rates for all cold and warm stimuli can be seen in Figures 3.10 and 3.11, respectively. Only one stimulus (DIR-LEN-EXT-ROC), *w-6-6-a*, had a recognition of 100% overall. Six more stimuli were also always recognised correctly by female participants (see Figures 3.10 and 3.11), but not by male participants. The worst recognition rate was found for *w-0-3-s* with 58%. All other stimuli had recognition rates over 70%.

Statistical evaluation with repeated measures ANOVA for all participants (see Table 3.6) showed significant results for cold stimuli only for the variable EXT, while recognition of warm stimuli was influenced by both LEN and EXT and also showed significant differences for the interaction between LEN and EXT. Significant gender differences were not found (see Table 3.7). Higher temperature changes led to a better recognition for both warm (t(15) = 4.81, p = 0.0002) and cold (t(15) = 2.89, p = 0.01) stimuli. *Post hoc* tests with Tukey adjustments for LEN of warm stimuli showed significant differences for 0 - 3 (t(30) = 4.32, p = 0.0005) and 0 - 6: t(30) = 3.63, p = 0.003), but not 3 - 6 (t(30) = 0.69, p = 0.77). Stimuli with LEN of 0 had lower recognition rates.

Post hoc test statistics with Tukey adjustments for the interaction between LEN and EXT for warm stimuli can be found in Table 3.8. The combination 0,3 was significantly worse recognised than all other combinations.



Figure 3.10: Experiment 2: Recognition Rate for all Cold Stimuli.

	LEN	EXT	ROC	LEN:	LEN:	EXT:	LEN:
				EXT	ROC	ROC	EXT:
							ROC
Cold	<i>F</i> (2,30)	<i>F</i> (1,15)	F(1,15)	<i>F</i> (2,30)	F(2,30)	F(1,15)	<i>F</i> (2,30)
All	= 1.34	= 8.37	= 1.47	= 1.33	= 1.34	= 0.85	= 1.22
	p = 0.28	<i>p</i> = 0.01*	<i>p</i> = 0.24	p = 0.28	<i>p</i> = 0.28	<i>p</i> = 0.37	<i>p</i> = 0.31
Warm	<i>F</i> (2,30)	<i>F</i> (1,15)	F(1,15)	<i>F</i> (2,30)	F(2,30)	<i>F</i> (1,15)	F(2,30)
All	= 10.75	= 23.13	= 2.36	= 7.33	= 1.12	= 0.60	= 1.54
	<i>p</i> < 0.001*	<i>p</i> < 0.001*	<i>p</i> = 0.15	<i>p</i> = 0.003*	p = 0.34	<i>p</i> = 0.45	<i>p</i> = 0.23

Table 3.6: Experiment 2: Statistics for Recognition Rates for all participants; significant results are bold and marked with an asterisk.



Figure 3.11: Experiment 2: Recognition Rate for all Warm Stimuli.

	Gender:	Gender:	Gender:	Gender:	Gender:	Gender:	Gender:
	LEN	EXT	ROC	LEN:	LEN:	EXT:	LEN:
				EXT	ROC	ROC	EXT:
							ROC
Cold	<i>F</i> (2,28)	<i>F</i> (1,14)	<i>F</i> (1,14)	<i>F</i> (2,28)	<i>F</i> (2,28)	<i>F</i> (<i>1</i> , <i>1</i> 4)	F(2,28)
	= 0.15	= 0.05	= 0.06	= 1.71	= 0.27	= 0.09	= 0.94
	<i>p</i> = 0.86	<i>p</i> = 0.83	p = 0.82	p = 0.20	<i>p</i> = 0.77	<i>p</i> = 0.77	p = 0.40
Warm	<i>F</i> (2,28)	F(1, 14)	F(1, 14)	<i>F</i> (2,28)	<i>F</i> (2,28)	F(1, 14)	F(2,28)
	= 2.89	= 0.04	= 0.09	= 2.33	= 0.44	= 0.00	= 0.51
	<i>p</i> = 0.07	<i>p</i> = 0.84	<i>p</i> = 0.77	<i>p</i> = 0.16	<i>p</i> = 0.65	<i>p</i> = 1.00	<i>p</i> = 0.61

Table 3.7: Experiment 2: Statistics for Recognition Rates with Gender as between subject factor; significant results are bold and marked with an asterisk.

LEN,EXT	t	p
0,3 - 3,3	t(60.0) = 5.60	< 0.0001*
0,3 - 6,3	t(60.0) = 4.63	0.0003*
0,3 - 0,6	t(41.4) = 6.04	< 0.0001*
0,3 - 3,6	t(48.1) = 6.49	< 0.0001*
0,3 - 6,6	t(48.1) = 6.49	< 0.0001*
3,3 - 6,3	t(60.0) = 0.97	0.92
3,3 - 0,6	t(48.1) = 1.08	0.89
3,3 - 3,6	t(41.4) = 1.51	0.66
3,3 - 6,6	t(48.1) = 1.51	0.66
6,3 - 0,6	t(48.1) = 1.95	0.39
6,3 - 3,6	t(48.1) = 2.38	0.18
6,3 - 6,6	t(41.4) = 2.37	0.19
0,6 - 3,6	t(60.0) = 0.49	1.00
0,6 - 6,6	t(60.0) = 0.49	1.00
3,6 - 6,6	t(60.0) = 0.00	1.00

Table 3.8: Experiment 2: Statistics for Recognition Rate *post hoc* tests for the Interaction between LEN and EXT for warm stimuli of all participants; significant results are bold and marked with an asterisk.

Number of False Positive Lane Changes

Figures 3.12 and 3.13 show rates for all false positives for cold and warm temperature changes, respectively. The highest numbers of false positives can be found for stimuli *c*-6-6-a and w-6-6-s with 17%. The lowest rate was 2%, which was presented for *c*-0-6-s and w-0-6-s. Statistical evaluations with repeated measures ANOVA for all participants (see Table 3.9) and gender influences (see Table 3.10) revealed no significant effects.



Figure 3.12: Experiment 2: False Positives for all Cold Stimuli.



Figure 3.13: Experiment 2: False Positives for all Warm Stimuli.

	LEN	EXT	ROC	LEN:	LEN:	EXT:	LEN:
				EXT	ROC	ROC	EXT:
							ROC
Cold	F(2,30)	<i>F</i> (1,15)	F(1,15)	F(2,30)	F(2,30)	F(1, 15)	F(2,30)
All	= 1.92	= 0.35	= 0.39	= 0.64	= 0.10	= 0.39	= 3.05
	<i>p</i> = 0.16	<i>p</i> = 0.57	<i>p</i> = 0.54	<i>p</i> = 0.53	<i>p</i> = 0.90	<i>p</i> = 0.54	<i>p</i> = 0.06
Warm	F(2,30)	<i>F</i> (1,15)	<i>F</i> (1,15)	F(2,30)	F(2,30)	F(1,15)	F(2,30)
All	= 2.42	= 1.20	= 0.00	= 0.74	= 3.11	= 0.77	= 0.66
	<i>p</i> = 0.11	<i>p</i> = 0.29	<i>p</i> = 1.00	<i>p</i> = 0.49	<i>p</i> = 0.06	<i>p</i> = 0.39	<i>p</i> = 0.52

Table 3.9: Experiment 2: Statistics for False Positives for all participants; significant results are bold and marked with an asterisk.

	Gender:	Gender:	Gender:	Gender:	Gender:	Gender:	Gender:
	LEN	EXT	ROC	LEN:	LEN:	EXT:	LEN:
				EXT	ROC	ROC	EXT:
							ROC
Cold	<i>F</i> (2,28)	<i>F</i> (1,14)	<i>F</i> (1,14)	<i>F</i> (2,28)	<i>F</i> (2,28)	<i>F</i> (<i>1</i> , <i>1</i> 4)	F(2,28
	= 2.79	= 3.35	= 0.04	= 0.81	= 1.36	= 0.64	= 1.25
	<i>p</i> = 0.08	<i>p</i> = 0.09	<i>p</i> = 0.85	<i>p</i> = 0.46	<i>p</i> = 0.27	p = 0.44	p = 0.30
Warm	<i>F</i> (2,28)	F(1, 14)	<i>F</i> (1,14)	<i>F</i> (2,28)	<i>F</i> (2,28)	F(1, 14)	F(2,28
	= 3.06	= 0.16	= 0.26	= 0.36	= 0.44	= 0.47	= 0.98
	p = 0.06	p = 0.69	p = 0.62	p = 0.70	p = 0.65	p = 0.50	p = 0.39

Table 3.10: Experiment 2: Statistics for False Positives with Gender as between subject factor; significant results are bold and marked with an asterisk.

Time to Complete Lane Change

Results of one male participant had to be removed, as the data differed more than two standard deviations from the mean. The Time to Complete Lane Change for all cold and warm stimuli can be seen in Figure 3.14 and 3.15, respectively. Stimulus c-6-3-a led, on average, to the fastest lane change (4.20s), while both w-3-6-s and w-6-6-s took the longest (4.97s).



Figure 3.14: Experiment 2: Time to Complete Lane Change for all Cold Stimuli.



Figure 3.15: Experiment 2: Time to Complete Lane Change for all Warm Stimuli.

When evaluating the times for all participants with repeated measures ANOVA (see Table 3.11), no significant influence of the variables was found for cold stimuli, whereas LEN and EXT influenced the time for warm stimuli. *Post hoc* tests with Tukey adjustments for

LEN showed significant differences for 0 - 6 (t(28) = 2.75, p = 0.03), where the longer stimulus presentation led to longer lane change times, but not 0 - 3 (t(28) = 2.43, p = 0.05) or 3 - 6 (t(28) = 0.32, p = 0.95). *Post hoc* tests with Tukey adjustments for EXT showed that the larger extent of temperature change led to longer lane change times (t(14) = 2.41, p = 0.03). The evaluation of gender differences for Time to Complete Lane Change showed no significant results, see Table 3.12.

	LEN	EXT	ROC	LEN:	LEN:	EXT:	LEN:
				EXT	ROC	ROC	EXT:
							ROC
Cold	<i>F</i> (2,28)	<i>F</i> (1,14)	<i>F</i> (1,14)	<i>F</i> (2,28)	F(2,28)	<i>F</i> (1,14)	<i>F</i> (2,28)
All	= 1.27	= 0.16	= 0.72	= 0.33	= 0.12	= 0.94	= 0.32
	<i>p</i> = 0.30	<i>p</i> = 0.69	<i>p</i> = 0.41	p = 0.72	<i>p</i> = 0.89	p = 0.35	p = 0.73
Warm	F(2,28)	<i>F</i> (1,14)	<i>F</i> (1,14)	<i>F</i> (2,28)	<i>F</i> (2,28)	<i>F</i> (1,14)	<i>F</i> (2,28)
All	= 4.53	= 5.83	= 4.52	= 2.82	= 2.14	= 0.60	= 2.35
	p = 0.02*	<i>p</i> = 0.03*	<i>p</i> = 0.05	<i>p</i> = 0.08	p = 0.14	p = 0.45	<i>p</i> = 0.11

Table 3.11: Experiment 2: Statistics for Time to Complete Lane Change for all participants; significant results are bold and marked with an asterisk.

	Gender:	Gender:	Gender:	Gender:	Gender:	Gender:	Gender:
	LEN	EXT	ROC	LEN:	LEN:	EXT:	LEN:
				EXT	ROC	ROC	EXT:
							ROC
Cold	<i>F</i> (2,26)	<i>F</i> (1,13)	<i>F</i> (1,13)	<i>F</i> (2,26)	F(2,26)	<i>F</i> (1,13)	F(2,26)
	= 1.99	= 0.02	= 0.58	= 1.80	= 0.31	= 0.87	= 0.07
	<i>p</i> = 0.16	<i>p</i> = 0.88	<i>p</i> = 0.46	<i>p</i> = 0.19	p = 0.74	<i>p</i> = 0.37	<i>p</i> = 0.94
Warm	<i>F</i> (2,26)	<i>F</i> (1,13)	F(1,13)	F(2,26)	F(2,26)	<i>F</i> (1,13)	F(2,26)
	= 0.24	= 1.38	= 1.00	= 0.09	= 0.17	= 2.19	= 0.10
	<i>p</i> = 0.79	p = 0.26	p = 0.34	<i>p</i> = 0.91	p = 0.84	<i>p</i> = 0.16	p = 0.90

Table 3.12: Experiment 2: Statistics for Time to Complete Lane Change with Gender as between subject factor; significant results are bold and marked with an asterisk.

3.4.6 Experiment 2 Discussion

This second experiment explored different stimuli designs and primarily investigated their influence on recognition rate and false positives at the return to neutral. No statistically significant difference was found to indicate an influence of any of the evaluated factors on the number of false positives. Therefore, *Hypotheses 1* and 2 could not be corroborated. There was also no influence of gender found on the number of false positives. The highest rate of false positives over all participants for cold and warm stimuli was discovered for c-6-6-a and

w-6-6-s, respectively, with 17% each. The variation in number of false positives, both for the directions of change and gender gave no indication of what design factors influenced this value.

There were no gender differences found for recognition rates. Recognition rate for both cold and warm stimuli was influenced by the extent of temperature change: higher temperature changes increased the recognition for both directions of change. No other factor influenced the recognition for cold stimuli, whereas both extent and lengths of temperature change were important for recognition of warm stimuli. The higher temperature extent of 6°C led to a higher recognition, which corroborated *Hypothesis 3* only partly, as this effect was seen exclusively for warm temperature changes. Further exploration of the influence of LEN showed that recognition was reduced for warm changes of only 3°C, if the temperature was returned to neutral immediately after reaching the goal temperature. This, again, was only found for warm temperature changes, showing that a mirroring of stimuli in both directions of change might not lead to expected results.

No solution was found to both increase recognition and at the same time decrease the number of false positives: w-6-6-a, as the only feedback with a recognition rate of 100%, presented with 15% one of the highest numbers of false positives at the return to neutral. Designing thermal feedback for interaction with binary information of one device, therefore, has to be adapted to maximise the prioritised outcome.

Differences in task time, here represented by the time to complete lane change, might also influence the design process. While the lane change time was not influenced by any of the factors for cold stimuli, length and extent of temperature change had a significant effect for warm temperature changes. These findings again point out the discrepancy between cold and warm stimuli and the need to adapt the design depending on the preferred outcome.

3.5 Chapter 3 Discussion

Two experiments explored the usefulness of thermal feedback to convey binary directional information on a single device during driving. As this was the first investigation of thermal feedback during driving, the influence of the feedback on driving measures such as lane deviation, as indication of distraction, and recognition as a secondary task were one focus of the first experiment, in addition to time to complete lane change and subjective feedback, capturing the user ratings and workload. Lane deviation was found to not differ statistically from baseline driving with audio feedback. Recognition rates were high, with 97% and 88%, however, a surprising number of false positive identifications were made. Participants had problems differentiating between the return to neutral and a new stimulus. The time to complete lane change was increased by 1.82s for thermal feedback, when compared to the base-

line of speech. This increased time requirement showed an unsuitability of thermal feedback for time-sensitive feedback. Significant differences between the genders were found here, which could indicate a slightly enhanced sensitivity of female participants towards thermal feedback. This conclusion was backed up by the higher number of false positives reported by women, but more research is needed for clarification. Workload showed no statistically significant difference. However, subjective ratings indicated that most participants found thermal feedback more complicated, which can be explained by both the novelty of the feedback and the additional mapping needed to translate direction of temperature change to direction of lane change. In addition, participants reported on a level of uncertainty connected to thermal feedback, which they did not experience for speech.

A second experiment was designed to explore a set of design aspects and evaluate their influence on recognition rate and false positives at the return to neutral. The parameters direction of change, length of cue presentation, extent of temperature change and rate of change at the return to neutral were investigated. Recognition rates varied, with one stimulus noticeably worse than all other, with a recognition rate of 58%. All other stimuli had recognition rates between 70% and 100%, the latter achieved by only one stimulus. The recognition rate was influenced by different parameters for warm and cold stimuli: while only extent influenced cold stimuli, length and extent of temperature change additionally influenced warm stimuli, as did their interaction. Design of thermal stimuli, therefore, cannot be simply mirrored for both directions of temperature change. No differences for gender were found, neither for recognition rate nor for false positives. None of the parameters had a statistically significant influence on false positives, which varied for stimuli between 2% and 17%. The stimulus with the highest recognition rate showed one of the highest numbers of false positives, illustrating that the design of thermal feedback will have to focus on optimising one of the aspects.

As there were no significant gender differences found for recognition or number of false positives in the second experiment, this line of investigation will not be pursued in the following experiments.

3.6 Conclusions and Research Question 1

The previously discussed two studies aimed to answer Research Question 1:

How accurately can thermal feedback give binary direction cues in an automotive setting (with one thermal device)?

The results show that the design of binary thermal directional cues has to be carefully considered. Some of the tested stimuli showed exceptionally high recognition rates, however, the problem of false positives highly impacted the accuracy of those cues. Additional lane changes occurred when the temperature changed back to neutral. Unfortunately, no solution could accommodate best results for both measures. For directional cues, this poses a considerable problem, as any false positives would be treated the same as the original cue, negating the desired effect by prompting a lane change into the original lane, or in the case of turn-by-turn navigation, lead to driver confusion and to potentially dangerous situations. An additional level of information will have to be added to ensure more robust directional cues. One possible solution could be adding spatial information to directional cues. This will require the use of at least two different thermal devices at different locations. The location of the thermal feedback could give the directional information while the design of the feedback could give information such as distance. Turn-by-turn navigation would benefit from such a design. Routes are usually planned in advance, providing a suitable use case for the less time-sensitive type of feedback. Chapter 4 explores such feedback for navigation with multiple thermal devices.

Chapter 4

Thermal Directional Cues with Multiple Devices on the Steering Wheel

4.1 Introduction

The previous chapter investigated the accuracy of binary directional cues provided by one thermal device. The binary feedback was found to have difficulties in conveying accurate directional cues, as the return to neutral was sometimes identified as an additional directional prompt. Spatial information, provided by more than one device, could help overcome this problem. The following two experiments have been designed to explore this aspect and aim to answer Research Question 2:

How accurately can thermal feedback give direction cues in an automotive setting with the spatial information from multiple thermal devices?

The directional cues have been changed from indicating simple lane changes to turn-byturn navigation in a city environment, to provide a more complex and more realistic driving scenario. The thermal devices have been mounted on the steering wheel, to ensure constant contact with the hands and fixed spatial locations. Both experiments exploring this feedback will be described in detail in the next sections, followed by a conclusion in Section 4.5.

4.2 Experiment 3: Thermal Navigation

This user study compared thermal feedback on the steering wheel to audio feedback, as speech instructions, for turn-by-turn navigation. Speech was used as baseline, as it is the most common non-visual feedback type used for in-car navigation.

4.2.1 Methodology

The experiment was designed as a within-subjects study with one independent variable feedback type, being either thermal or audio. As navigation routes are often planned in advance, the use case was deemed to be suitable for the slowly increasing nature of thermal feedback and it was expected to perform as well as speech. While there were differences found for workload and subjective ratings in previous experiments, those might have been due to the additional mapping of direction of temperature change to direction of lane change. The addition of spatial information was hypothesised to alleviate these problems, as it eliminated the mapping. Driving performance in the previous experiments was not influenced by the feedback condition, observations in this study were not expected to be different. The mapping of destination direction to warm temperature changes was informed by the findings of Experiment 1, in addition to mappings used by both Wettach *et al.* [40] and Tewell *et al.* [134]. One of the goals of this experiment was to confirm this mapping. The hypotheses for this experiment, therefore, were:

Hypothesis 1: Thermal and auditory navigation cues will be equally effective in a navigation task;

Hypothesis 2: The workload and subjective rating for audio and thermal feedback will not differ significantly;

Hypothesis 3: The deviation from the ideal path will not differ significantly for the feedback types;

Hypothesis 4: The use of warm (opposed to cool) temperature changes to indicate the destination will be preferred by participants.

These questions were answered by investigating the dependent variables: stimulus recognition, deviation from the ideal path, perceived workload and subjective ratings. More detailed descriptions of the variables can be found in Section 4.2.5 *Measures*.

4.2.2 Apparatus

For this experiment, four smaller 1x1cm Peltier devices were used, see Figure 4.1(a). The devices were attached to the steering wheel using Velcro cable binders, see Figure 4.1(b): two on each side of the steering wheel (Figure 4.2(a)). They were positioned in a way that one device was facing the driver, while the position of the other was mirrored on the back of the steering wheel. In this way, the devices on one side could be touched by the thumb and the index finger of the same hand, leaving the rest of the hand free to hold the steering wheel, ensuring a comfortable grip, as seen in Figure 4.2(b). The Peltier control board was controlled via USB and built by SAMH Engineering, for details please consult



(a) 1x1cm Peltier device (gold) attached to heat-sink (black)



(b) Peltier device mounted on steering wheel with Velcro cable binders

Figure 4.1: Experiment 3: Hardware.



(a) Steering Wheel



(b) Experimental Setup

Figure 4.2: Experiment 3: Steering Wheel and experimental setup.

Appendix A.1.2. The rest of the technical setup was identical to Experiment 1 and 2, apart from the driving scenario: instead of a five-lane motorway the driving took place in a city environment, see Figure 4.3, also implemented in OpenDS 3.5 free version.

Driving Task and Feedback 4.2.3

In this experiment, participants followed instructions for turn-by-turn navigation in a simulated city environment. The driving instructions were kept as simple as possible: follow the road straight ahead in the right-most lane, until the feedback indicates the turning point; then turn into the next upcoming road on the side pointed to by the feedback.

The speed of the car was fixed to 30km/h, no pedals were provided. While the lack of braking capabilities for turning scenarios was slightly unnatural, the control of driving variation was



(a) City environment, suburbs



(b) City environment, downtown



deemed more important in order to secure a valid comparison of the feedback types. 30km/h was found to be slow enough to turn, yet still represented a valid speed limit in European cities.

The simulator environment used was set for right-side driving and was kept unchanged, as the number of participants used to right-side driving in previous experiments exceeded the number of participants used to left-side driving. The car was reset to a different location after each turn, always back in the right-most lane, so even participants who were used to left-side driving were regularly reminded to keep on the right. This resetting approach was, however, mainly chosen to overcome the difficulties of implementing rerouting algorithms for the limited map space of the simulation environment and to ensure comparable turning points for both feedback types. After each reset, the participants had ample time to settle back into driving, before any new feedback was presented. In order to simulate standard navigation systems, the turning instructions were given in two steps: first by an initial ahead notification 200m before the turn, and then again when the turning point was reached. To capture the recognition of the ahead warnings, participants were asked to pull the gear paddle on the steering wheel on the side of the upcoming turn. Wrong turns were possible at any point, resetting the car for the next turn.

The turning points themselves were selected to represent the four major types of junctions and crossing: T-junctions, crossings (see Figure 4.3), right-only and left-only junctions. One turn of each type was chosen for the training sessions to introduce the feedback, while three of each were presented in random order for the main driving task. The training set was the same for both feedback types, while two different sets of twelve turns were used for the main driving tasks, counterbalanced over the feedback types and order of use. The drive towards each turning point usually passed several junctions and crossings, so the route could not be guessed without the feedback presented.

The speech instructions given to participants were spoken by a synthesised female voice and consisted of *Turn right/left in 200 metres* and *Turn right/left* when the turning point was reached.

The thermal feedback, see Figure 4.4, started from the neutral temperature of 30° C for all devices. 200m before the turn, the devices on both sides of the steering wheel changed their temperature for an extent of 6°C with a rate of change of 3°C/s. On the turning side, the devices warmed for 6°C (Figure 4.4 red line), while they cooled 6°C on the opposite side (Figure 4.4 blue line). This temperature was presented for 3s and then turned off, see black line in Figure 4.4.

When the turning point was reached, the feedback started the same, but the temperature was presented until the turn was completed and then returned to the neutral temperature. In this design, the distance to the turning point was encoded by length of temperature presentation,



Figure 4.4: Experiment 3: Thermal feedback.

the short burst an ahead notification of the upcoming turn. The simultaneous presentation of warm and cold stimuli was expected to aid the recognition of the direction of temperature change, as the temperature difference between both hands would be increased and should be easier to recognise.

4.2.4 Participants and Procedure

Thirteen participants completed the experiment. They all held a valid driving license, all but one were right-handed. Four participants obtained their driving licenses in countries with right-side driving. None reported sensory impairments in the hands and all had at least corrected vision. For more details and on prior experience with the technologies (on 5- point Likert scale with 1 being no experience), consult Table 4.1.

Number of	Age	Driving	Experience	Experience	Experience
Participants	(Years)	Experience	Simulator	Audio	Thermal
		(Years)			
13	<i>R</i> : 19-38	<i>R</i> : 1-11	<i>R</i> : 1-5	<i>R</i> : 1-4	<i>R</i> : 1-5
(Female: 6	M = 25.38;	M = 4.88	M = 2.23	M = 2.92	M = 1.85
Male: 7)	SD = 5.24	SD = 3.11	SD = 1.48	SD = 0.86	SD = 1.28

Table 4.1: Experiment 3: Participant data, showing range (R), mean (M) and standard deviation (SD).

Participants were students and staff recruited through email lists, university online platforms and the participant pool provided by the School of Psychology of the University of Glasgow. The participants were welcomed to the School of Computing Science at the university, where the experiments were conducted. They were presented with an information sheet, signed a consent form and filled in a questionnaire, capturing some demographic data such as gender, age and years of driving experience. In addition, they rated their experience with the technologies and were asked to choose their preferred mapping of direction of temperature change to destination direction.

Participants were given up to 5min to drive through the city environment freely, to get familiar with the driving setup. This was followed by the training phase of the first condition, consisting of four turns, succeeded by the main driving task with 12 turns, arranged in random order. After filling in the NASA TLX questionnaire and some additional ratings, the procedure was repeated for the second feedback type. At the end of the experiment participants filled in an additional questionnaire, capturing ratings and inquiring what temperature mapping participants would prefer after having finished the experiment with our design.

The documents used during this experiment can be found in Appendix A.4. The experiment design was approved the Ethics Committee of the College of Science and Engineering of the University of Glasgow. Participants were paid £6 for completing the one hour experiment.

4.2.5 Measures

The effectiveness of the feedback types was evaluated by comparing the recognition rates of both the 200m warnings and the actual turns.

NASA TLX questionnaires were used to capture the perceived workload, while ratings on a 5-point Likert scale asking how pleasant, comfortable, disruptive and complicated the feedback felt was used for additional subjective feedback. At the end of the experiment, participants were asked to rate how much they liked the feedback types on a 10-point Likert scale.

The influence on the driving behaviour was calculated by observing the deviation from an ideal path. The ideal path followed the road along the centre of the right-most lane, until the turn was made. The road leading towards the turn was usually not perfectly straight and had crossings and junctions (see Figure 4.3(b)). This fact, in addition to the complex city map, made an easy calculation of lane deviation, as used in the previous experiments, impossible. This alternative was used to enable a comparison between the two feedback types. The deviation was calculated using the Root Mean Square Error (RMSE) of the ideal path and the participant's driving path. 8s of driving both before and after the stimulus start were compared for the 200 metres ahead warnings. The turns themselves were not compared, as the deviation would be too heavily influenced by wrong turns or differing turn strategy in the simulator environment without braking (such as moving the vehicle slightly towards the middle of the road before turning).

To confirm the mapping of direction of temperature to turn direction, participants were asked about their preference at the beginning of the experiment. This was not asked directly, participants were rather presented with the following scenario:

In the following experiment, you will be presented with thermal feedback for navigation purposes on the steering wheel, where the device of one side of the wheel will be warmed, while the other side will be cooled.

Please decide: If the right side of the steering wheel was warmed, while the left side was cooled, I would turn to the: Left / Right

In the questionnaire at the end of the experiment, they were asked their opinion again, more direct this time:

Having finished the experiment, which direction would you turn towards, if presented with thermal feedback for navigation purposes? Warm / Cold. Participants were asked twice to see if the use of the feedback during the experiment would alter their preference. In addition, open questions asked for possible locations and use cases in an in-car environment.

4.2.6 Results

The data were compared using paired t-tests (t(df)), if they were normally distributed, and paired Wilcoxon tests (V) otherwise. Results of the perceived workload parameters *Physical Demand* and *Performance* differed from previously published work due to the reanalysis of the data.

Recognition

The ahead warnings were correctly recognised 94% in the thermal and 100% in the audio condition. The recognition rate for the turns was 91% and 100% for thermal and audio, respectively. Nine ahead warnings and thirteen turns were interpreted incorrectly in the thermal condition. This difference between feedback types was found to be significant with Wilcoxon tests for both ahead warnings (V = 0, p = 0.03) and turns (V = 0, p = 0.01).

Subjective Data

The comparison of workload with Wilcoxon tests showed significant results for overall workload (V = 1, p = 0.002). The more detailed analysis for the different workload parameters can be seen in Table 4.2, their values are presented in Figure 4.5. All parameters apart from Physical Demand and Performance showed significant differences, with thermal navigation rated more negatively than speech.



Figure 4.5: Experiment 3: Perceived workload; significant results are in bold font.

Mental	Physical	Time	Effort	Perfor-	Frustra-	Annoy-
Demand	Demand	Effort		mance	tion	ance
V = 0	<i>V</i> = 14	<i>V</i> = 10	<i>V</i> = 3.5	<i>V</i> = 64	<i>V</i> = 5	<i>V</i> = 6
<i>p</i> = 0.001*	<i>p</i> = 0.05	<i>p</i> = 0.02*	p = 0.005*	<i>p</i> = 0.05	<i>p</i> = 0.005*	<i>p</i> = 0.01*

Table 4.2: Experiment 3: Statistics for NASA TLX rating of workload; significant results are bold and marked with an asterisk.



Figure 4.6: Experiment 3: Subjective Additional Ratings; significant results are in bold font.

Pleasant	Comfortable	Disruptive	Complicated
<i>V</i> = 58	V = 42	<i>V</i> = 4.5	<i>V</i> = 7.5
<i>p</i> = 0.02*	<i>p</i> = 0.02*	<i>p</i> = 0.06	<i>p</i> = 0.01*

Table 4.3: Experiment 3: Statistics for Subjective Additional Ratings; significant results are bold and marked with an asterisk.

The values for the additional subjective ratings can be found in Figure 4.3, the statistical data of the Wilcoxon tests in Table 4.3. The parameters Pleasant, Comfortable and Complicated showed significant differences, favouring speech feedback, but not Disruptive.

The comparative rating at the end of the experiment showed significantly better rating for audio feedback (V = 76, p = 0.004): median(audio) = 7.5, median(thermal) = 3).

Participant comments indicated that the audio feedback was considered to be *simple (P09, P11, P12, P13, P07), clear (P03, P10, P12, P13), easy (P08, P06, P04)* and *familiar (P02, P07, P11)*, but also *disruptive (P01)*. Thermal feedback was described as *difficult (P02, P03, P07, P13), requiring more concentration (P06, P08, P11, P12), varying in sensation (P02, P11)* and *confusing (P05)*. *P04* commented: *I felt like sometimes my brain just registered intensity of temperature rather than intense heat or intense cold* and *P05* mentioned that it was *sometimes difficult to tell hot from cold. Often felt hot all the time. P07* reported similar problems: *I felt i was difficult at times to tell the difference between the hot and cold feedback and I really had to think about it. I may well have turned the wrong way once or twice.* While many participants conveyed a sense of overall uncertainty concerning thermal feedback, it was also described as *pleasant and soothing (P10)* and not *as disruptive (P01)*.

Thermal Before -	Thermal After -	Thermal Before -	Audio Before -
Audio Before	Audio After	Thermal After	Audio After
V = 80;	<i>V</i> = 74	V = 0	V = 1
p = 0.01*	p = 0.048*	<i>p</i> < 0.001*	<i>p</i> < 0.001*

Table 4.4: Experiment 3: Statistics for Deviation from Ideal Path; significant results are bold and marked with an asterisk.

back as a factor of their rating (P02, P03).

P02 named a few use cases for thermal feedback in the car: more emergency things. An ambulance approaching from behind, a collision ahead, a stop sign or red light ahead perhaps. Or maybe if you are breaking speed limit. The idea of collision warning was shared by P09. Others had suggestions related to temperature: Engine overheating warning (P03), air-conditioning (P08) and current temperature setting. [...] Road ice or grip/traction conditions so I know to drive more carefully (P11). Distance to destination (P11) and low fuel (P06) were named in addition. P03 suggested to present thermal feedback on the door handle if you leave electronics on when leaving. Seat (P07), pedal (P09), side/shoulder (P10) and aircon dial (P11) were named as possible locations for thermal feedback.

Deviation from Ideal Path

The deviation from the ideal path was compared between the feedback types, both before and after the stimulus onset, as well as within the conditions. All comparisons with Wilcoxon tests showed significant results, see Table 4.4. The deviation in the thermal condition was significantly higher than in the audio conditions and within both condition the deviation after the presentation of the feedback was higher than the deviation before, see Figure 4.7.

Direction Mapping

When asked about the mapping at the beginning of the experiment, eight participants stated they would turn towards the warm side, five towards the cold. After the experiment, eleven stated that they would now turn towards the warm, while two participants stated, they would turn towards the cold. Interestingly, one of those two changed their answer after the experiment from warm to cold.

4.2.7 Experiment 3 Discussion

This experiment explored thermal feedback for turn-by-turn navigation and compared it to speech feedback. Recognition rates for thermal feedback were high, 94% and 91%, but they



Figure 4.7: Experiment 3: Deviation from Ideal Path; significant results are in **bold** font.

were still statistically significantly worse than speech with a recognition rate of 100%. *Hypothesis 1* claiming they would show equal efficiency, therefore, could not be corroborated. The novelty of the feedback and inherent difference of perception for audio and haptic feedback, made a direct comparison of the two feedback types complex. The comparison of thermal with another haptic condition might give a better understanding of the efficacy and usefulness of the feedback.

Furthermore, both perceived workload and three of four additional subjective ratings showed statistically significant differences between the feedback types, so *Hypothesis 2* could not be corroborated either. Participants reported general uncertainty and increased need for concentration for thermal feedback. Both sentiments would be increased by the novelty of the feedback. They also commented that the simultaneous temperature change on both sides of the steering wheel confused rather than enforced the identification. Temperature change of only one side of the steering wheel might aid in decreasing the confusion and increase recognition and ratings. Furthermore, participants commented on differences of hand positioning needed for the feedback types and named this as contributing factors for the rating. The comparison with a feedback type with similar physical constraints could balance this factor.

The deviation from the ideal path was different for the two feedback types, not corroborating *Hypothesis 3*. Thermal feedback had a higher deviation than audio, mirroring the comments and workload ratings and indicating more distraction posed by the novel thermal feedback, compared to the familiar speech. Both feedback types showed higher deviation after being presented with the feedback. This might be slightly influenced by the paddle pull, with which participants indicated their recognition. While more participants in this experiment

were used to left-side driving, none commented on encountering problems with driving on the right side.

Hypothesis 4, claiming the mapping of warm temperature change to destination would be preferred by most, was proven correct: eight out of thirteen participants stated they would prefer to turn toward the warm side at the beginning of the experiment and even more confirmed the mapping after the experiment, with all but two agreeing with the mapping.

The findings of this experiment suggested that a change in thermal design is needed to improve the navigation efficiency. A follow up study therefore explored an improved design, comparing it to another tactile feedback type, which would allow for the design of more similar feedback and presentation conditions.

4.3 Experiment 4: Haptic Navigation

Thermal feedback for navigation with an altered design was compared to another haptic feedback type in this experiment: cutaneous push [210]. This is feedback elicited through retractable solenoid pins embedded in a steering wheel.

4.3.1 Methodology

The within-subjects study had one independent variable: feedback type, consisting of thermal and cutaneous push feedback. These two tactile modalities could be presented at the same location on the steering wheel and would both be equally unfamiliar to the driver, as cutaneous push feedback had not been tested for navigation purposes in the past. The comfort rating of both was, therefore, expected not to be different. In addition, no difference in performance was expected, as both feedback types utilised the same sensory channel, while addressing different receptors in the skin. The change in thermal feedback design was expected to improve the performance compared to the previous experiment. The hypotheses for this study were:

Hypothesis 1: Thermal and cutaneous push navigation cues will be equally effective in a simulated navigation task;

Hypothesis 2: *The subjective rating of comfort of cutaneous push feedback and thermal will not differ;*

Hypothesis 3: The simplified design of the second thermal navigation feedback will perform better than the design in Experiment 3.

Recognition rates will be evaluated to determine the effectiveness of the feedback. Perceived workload and additional subjective ratings will be explored, as will the deviation from the

ideal path. Some data will in addition be compared with results of Experiment 3, to determine if the design changes of the thermal feedback increased performance. More details of the dependent variables will be discussed in Section 4.3.5 *Measures*.

4.3.2 Apparatus

This experiment utilised 2x2cm Peltier devices, as were used in Experiments 1 and 2, see Figure 4.10(a), controlled via USB boards, as in Experiment 3. One device was attached to the steering wheel on each side with sports bands and could be grasped with the hand, ensuring the Peltier would touch the palm of the hand, see Figure 4.8(a).



(a) Thermal Steering Wheel

(b) Cutaneous Push Steering Wheel

Figure 4.8: Experiment 4: Steering Wheel.

The steering wheel consisted of a pre-drilled metal steering wheel¹, mounted onto a Logitech G27 racing wheel base. One Push Action Tubular Solenoid² was embedded into each side, see Figure 4.8(b). The solenoids were retracted into the steering wheel and could protrude to gently tap the drivers palm, see Figure 4.9. Small plastic balls taken off sewing needles topped the solenoids to ensure comfort. The cutaneous push feedback was controlled with an Arduino Uno board connected through USB.

Participants were seated in an RSeat RS1 gaming racing chair³, with the steering wheel attached securely in front of them. Logitech racing pedals were placed on the ground under

¹http://www.longacreracing.com/products.aspx?prodid=7620, accessed 06/04/2020

²https://uk.rs-online.com/web/p/dc-d-frame-solenoid/2501280/, accessed 06/04/2020

³http://www.rseat.net/rs1-racing-cockpit/rs1-m4a-black/, accessed 30/09/2020



Figure 4.9: Experiment 4: Solenoid embedded into the Steering Wheel for Cutaneous Push Feedback.



(a) Peltier Device

(b) Experimental Setup

Figure 4.10: Experiment 4: Peltier device and setup.

the steering wheel. The driving environment was the same as in Experiment 3 and projected onto the wall of the usability lab using a BENQ DLP projector attached to a Windows desktop computer. Sennheiser HD 25 Basic Edition headphones played driving noises throughout the experiment. The complete setup can be seen in Figure 4.10(b).

4.3.3 Driving Task and Feedback

The driving task in this study was the same as in Experiment 3. The same training set was used, as were the two sets of turns, again counterbalanced. The thermal feedback was changed, however, and only showed warm feedback on the turning side. 200m before the turn, the device warmed 6° C from the neutral temperature of 30° C with a rate of change

of 3°C/s. This temperature was presented for 6s and then returned with a rate of change of 1°C/s, see Figure 4.11. The feedback at the turning point was of the same design, but the temperature of 36°C was presented until the turn was completed.



Figure 4.11: Experiment 4: Thermal feedback.

The cutaneous push feedback for the 200m ahead warning was adapted from the thermal feedback: the pin protruded from the steering wheel on the turning side and stayed in this position for 6s, before retracting into the steering wheel again. When the turning point was reached, the pin protruded and retracted rhythmically with a frequency of 1s until the turn was completed.

Participants were asked to use the foot pedals to indicate the turning direction after the first warning. The brake pedal indicated a turn towards the left side, and the accelerator pedal a turn towards the right side. The clutch pedal was not used, as the pressure setting differed from the other two, and the separated location made the use of the other two pedals easier and more comfortable. This approach was adapted as the steering wheel was bigger and the gear paddles could not be reached easily, especially in the thermal condition, and to eliminate any influence of the paddle pull on the deviation of the ideal path.

4.3.4 Participants and Procedure

Seventeen participants were newly recruited for the experiment. All participants held a valid driving license, six of them were obtained in right-side driving countries. Two participants were left-handed, all others right-handed. No participants reported sensory impairments in their hands and all had at least corrected vision. Their experience prior to the study (on 5-point Likert scale with 1 being no experience) and more detailed demographic data can be found in Table 4.5.

Number of	Age	Driving	Experience	Experience	Experience
Participants	(Years)	Experience	Simulator	Cutaneous	Thermal
		(Years)		Push	
17	<i>R</i> : 20-65	<i>R</i> : 1-45	<i>R</i> : 1-4	<i>R</i> : 1-4	<i>R</i> : 1-4
(Female: 10	M = 29.00	M = 9.12	M = 2.35	<i>M</i> = 1.65	<i>M</i> = 1.53
Male: 7)	SD = 10.35	SD = 10.49	SD = 1.17	SD = 1.06	SD = 0.94

Table 4.5: Experiment 4: Participant data, showing range (R), mean (M) and standard deviation (SD).

Participants were recruited through email lists and the participant pool of the School of Psychology and consisted of students and staff. The participants were welcomed in a university room of the School of Computing Science and presented with the information sheet and the consent form they signed. They were seated and presented with the first feedback type, followed by the training set. During the training, participants did not wear headphones, instead they were actively guided by the experimenter. The main driving task of the first feedback task followed, during which participants were presented with driving noises via headphones. After filling in a questionnaire, the procedure was repeated with the second feedback type. At the end of the experiment, participants filled in an additional questionnaire. Affirmation for the association of warm to destination was sought again. Participants received £6 at the end of this one hour experiment. Please see Appendix A.5 for more details of the documents used. The experiment was approved by the Ethics Committee of the College of Science and Engineering of the University of Glasgow.

4.3.5 Measures

The recognition rates were collected for both the first warnings 200m ahead of the turn, indicated by the pedal presses, and the turns themselves.

Perceived workload and subjective rating were collected through questionnaires after each driving task, consisting of the NASA TLX questionnaire and additional subjective feedback asking how pleasant, comfortable, complicated and disruptive the feedback felt. Another questionnaire was presented at the end of the experiment, capturing demographic data, ratings and ideas for further use of thermal feedback.

As in Experiment 3, the deviation from the ideal path was calculated as the Root Mean Square Error between the ideal path and the position of the participant's car of the 8s before and after the first warning 200m before the turn.

The comparison between the thermal designs of the first and second navigation experiment was evaluated for the recognition and the subjective ratings. As two different steering wheels had to be used for the two experiments, it could not be excluded that the deviation from the

Mental	Physical	Time	Effort	Perfor-	Frustra-	Annoy-
Demand	Demand	Effort		mance	tion	ance
V = 0	<i>V</i> = 15	<i>V</i> = 25	<i>V</i> = 12	<i>V</i> = 103.5	<i>V</i> = 13.5	<i>V</i> = 12
<i>p</i> < 0.001*	<i>p</i> = 0.04*	<i>p</i> = 0.03*	p = 0.002*	<i>p</i> = 0.01*	<i>p</i> = 0.009*	p = 0.007*

Table 4.6: Experiment 4: Statistics for NASA TLX rating of workload; significant results are bold and marked with an asterisk.

ideal path might be influenced by differences in steering wheel sensitivity and hardware behaviour.

4.3.6 Results

The data were evaluated using paired t-tests (t(df)), if the data were normally distributed, and paired Wilcoxon (V) otherwise and for subjective data. The comparison between the designs was evaluated with the unpaired versions of the tests.

Recognition

The ahead warnings had a perfect recognition of 100% for cutaneous push feedback and 97% for thermal feedback. Four thermal ahead warnings were missed. The recognition rate for the turns was 98% for cutaneous push and 87% for thermal feedback. Eight turns in the cutaneous push condition were made incorrectly. Most of these incorrect turns were made when participants prematurely turned into roads, which were already almost passed when the feedback started. These sudden turns were not made in the thermal condition. In the thermal condition, four 200m warnings were missed and 25 turns made incorrectly. Some of those turns occurred because participants turned when the 200m warnings were given. The difference between the feedback types with Wilcoxon tests was statistically significant for turns (V = 4, p = 0.01), but not ahead warnings (V = 0, p = 0.35).

Subjective Data

The overall perceived workload showed statistically significant differences with Wilcoxon tests between the conditions (V = 9, p = 0.002). More detailed evaluation showed that all parameters were significantly different, see statistical results in Table 4.6 and the parameter values in Figure 4.12. Cutaneous push feedback was rated more positively than thermal feedback for every parameter.

Subjective ratings for pleasantness, comfort, disruptiveness and complexity (see Figure 4.13) showed significant differences with Wilcoxon tests between the conditions for pleasantness



Figure 4.12: Experiment 4: Perceived workload; significant results are in bold font.

and complexity, see Table 4.7. Cutaneous push feedback was rated as more pleasant and less complicated than thermal feedback.

Pleasant	Comfortable	Disruptive	Complicated
V = 33.5	<i>V</i> = 51	<i>V</i> = 42	V = 0
<i>p</i> = 0.03*	p = 0.10	<i>p</i> = 0.84	<i>p</i> = 0.003*

Table 4.7: Experiment 4: Statistics for Subjective Additional Ratings; significant results are bold and marked with an asterisk.

At the end of the experiment participants rated thermal feedback significantly worse than cutaneous push feedback (median(push) = 9, median(thermal) = 2.5; Statistics with Wilcoxon tests: V = 1, p < 0.001)

Participants commented on the thermal feedback that they had to actively pay attention to notice the temperature change (P14), a view that was shared by P02 and P03. Several participants noted that it often took them a while to be completely certain the temperature was actually going up (P07). P07 commented further that they were often uneasy at first if signal was given, so reaction time felt longer and required more focus on the navigation, a feeling that was shared by P17 and P02. P01 wrote: I found that after getting a thermal signal, that hand would feel odd, almost as if it was getting more thermal signals in the next task, even if it wasn't. I kept feeling the urge to press a pedal until the actual thermal feedback activated and was obviously the real feedback due to its intensity. This uncertainty connected to thermal feedback, which was also voiced in the previous experiments, was mirrored here in comments from several other participants as well. P12 wrote: I sometimes



Figure 4.13: Experiment 4: Subjective Additional Ratings; significant results are in bold font.

wasn't sure whether the steering wheel was getting hot or whether it was just my hand and P04 mentioned that they felt too much mental hesitation or uncertainty regarding the warming up sensation. P01 commented that The more I used it, the less clear it was what temperature I was feeling. Also sometimes the sensation snuck up on me. The feeling of the warming itself was annotated with almost opposite remarks. P05 noted that the heat felt nice, but it was hard to detect / feel it for sure and P11 that it felt nice, but [...] sometimes [...] difficult to detect, while P09 mentioned that thermal feedback warms the palms, very annoying. P08 commented that they found it less distracting than cutaneous feedback.

The cutaneous push feedback on the other hand was mostly described as *easy* (*P01*, *P07*, *P10*, *P12*, *P14*, *P17*) and *clear* (*P01*, *P07*, *P11*). *P03* commented that there was *no need to pay attention to it - its sounds and feel similar to normal indicator, it feels more natural than the thermal one*. The lack of additional attention needed was mentioned also by *P07* and *P16*. The latter went on to write that they *didn't feel the pressure* they felt with thermal feedback. *P14* described the push feedback as if *it was almost like the car indicator was physically clicking*. While both *P01* and *P02* remarked on the non-disruptive nature of cutaneous push feedback, *P08* described it as *a bit disorientating*, their *instinct* [...] *to push the 'button' back into the wheel when given the 200m indication. P16* noted that *it could actually be beneficial as a disruptive feedback because if you are falling asleep it could save your life to feel the poke, instead of warm thermal feeling. P09* commented on a drawback of both feedback types: *The steering wheel* [...] *is really annoying and made me uncomfortable while driving as I had to keep both hands on the steering wheel at all times.*
Most participants suggested the use of thermal feedback on the seat (*P05*, *P09*, *P14*), *P13* suggested pedals. *P01* and *P08* agreed that the steering wheel already is the most appropriate location. The use case most often named was a warning when speeding (*P01*, *P02*, *P11*, *P13*). Several types of other warnings were suggested as well: handbrake-on warning (*P08*), front proximity warning (*P05*, *P09*), blind-spot warning (*P10*) or a general *emergency feedback* [...] *if any of the control lamps light up* (*P07*). Some of those were inspired by metaphors, such as the blind-spot warning, *as the person would give off thermal radiation in a literal sense, so would be appropriate* (*P10*) or *driving too fast. It would be useful, and it seems like a good metaphor (engine getting too hot or something...)* (*P11*). A different use was suggested by *P14*: *It could be used to draw attention to non-essential locations during the journey. E.g., alert to something like a service station. Or if the satnav knows I'm going to the shops or the bank, then the thermal could indicate that there's an alternative one nearby. So it's not really important if I recognise or not, but it's nice to become aware of that stuff if I do happen to feel it.*

Deviation from Ideal Path

The deviation from the ideal path was different within the two condition, with the deviation being smaller after the presentation of the feedback. The comparison with Wilcoxon tests between the conditions did not differ significantly, see Table 4.8 for statistical data and Figure 4.14 for values.

Thermal Before -	Thermal After -	Thermal Before -	Push Before -
Push Before	Push After	Thermal After	Push After
V = 75	V = 85	<i>V</i> = 135	<i>V</i> = 148
p = 0.96	p = 0.71	p = 0.003*	<i>p</i> < 0.001*

Table 4.8: Experiment 4: Statistics for Deviation from Ideal Path; significant results are bold and marked with an asterisk.

Comparison of Thermal Feedback Design

Comparing the two thermal designs of Experiment 3 and 4 showed no statistically significant differences with Wilcoxon tests, see Table 4.9 for the perceived workload statistics and Table 4.10 for additional ratings and recognition statistics.



Figure 4.14: Experiment 4: Deviation from Ideal Path; significant results are in bold font.

Mental	Physical	Time	Effort	Perfor-	Frustra-	Annoy-
Demand	Demand	Effort		mance	tion	ance
V = 76.5	<i>V</i> = 133.5	<i>V</i> = 111	<i>V</i> = 101	<i>V</i> = 67	<i>V</i> = 112	<i>V</i> = 132
<i>p</i> = 0.16	<i>p</i> = 0.34	<i>p</i> = 1.00	<i>p</i> = 0.70	<i>p</i> = 0.07	<i>p</i> = 0.97	<i>p</i> = 0.39

Table 4.9: Experiment 3 and 4: Statistics for the Comparison of NASA TLX rating of work-load; significant results are bold and marked with an asterisk.

Pleasant	Comfortable	Disruptive	Complicated	Rating	Warnings	Turns
<i>V</i> = 94.5	<i>V</i> = 78.5	<i>V</i> = 153	<i>V</i> = 123.5	<i>V</i> = 119	<i>V</i> = 75.5	<i>V</i> = 111
<i>p</i> = 0.49	p = 0.17	<i>p</i> = 0.06	<i>p</i> = 0.59	<i>p</i> = 0.74	<i>p</i> = 0.06	<i>p</i> = 1.00

Table 4.10: Experiment 3 and 4: Statistics for Comparison of Subjective Additional Ratings; significant results are bold and marked with an asterisk.

4.3.7 Experiment 4 Discussion

In this simulator experiment thermal feedback was compared to cutaneous push feedback in the context of turn-by-turn navigation in a city environment. The 200m ahead warnings were equally well understood and their recognition reported in both feedback types. The different recognition rates of 97% in the thermal and 100% in the push condition were not statistically significantly different. The main turning events, however, showed a significant difference between the recognition rate of 87% in the thermal and 98% in the cutaneous push condition, negating the predictions made in *Hypothesis 1*. While participants reported that they felt their reaction time was longer for thermal feedback, the wrong turns made in the cutaneous push condition were due to the opposite effect: some participants turned too early, manoeuvring the car into roads that were already mostly passed at the onset of the feedback. This led to situations which would be highly dangerous in real life scenarios.

These sudden reactions were not observed in the thermal condition, suggesting that the feedback was experienced as calmer and less abrupt. This was reflected by only one participant in their comments, however. Most participants reported that cutaneous push feedback was easier and left them feeling more relaxed, as they did not have to invest as much attention to the feedback. This sentiment was mirrored in the subjective ratings, both for perceived workload, with cutaneous push feedback rating significantly better for all parameters, and the additional ratings, where both pleasantness and complexity were rated significantly better. Ratings for disruptiveness and comfort showed no significant difference, so no conclusions can be drawn on *Hypothesis 2*. The presentation of both tactile feedback types at the same location on the steering wheel might have levelled this parameter. However, one participant noted on how the need to keep both hands on the steering wheel throughout made them feel annoyed and uncomfortable.

The change in the thermal design for Experiment 4 showed no statistical difference when compared to Experiment 3, not corroborating *Hypothesis 3*. Neither recognition rates, nor subjective ratings showed significant improvement by the change in design and the presentation of thermal feedback on only one hand. The change between the thermal design from Experiment 3 to Experiment 4 was accompanied by a change of hardware as well: four 1x1cm Peltier devices were swapped with two 2x2cm devices. This further changed the location of the feedback presentation: while thermal feedback was presented on the fingertips of index finger and thumb in Experiment 3, this was changed to the thenar region of the palm for Experiment 4, a slightly more perceptive part of the hand for tactile feedback. While this change influenced the direct comparability of the thermal designs, the changes were made to ensure a better mirroring of thermal and cutaneous push feedback types.

4.4 Chapter 4 Discussion

Thermal feedback was explored with multiple devices on the steering wheel for turn-by-turn navigation in two user experiments. While recognition rates were high with 87% to 97%, navigation with speech or cutaneous push feedback still outperformed thermal feedback for the actual turns, but not the 200m warnings for the second experiment. The presentation of warm and cold stimuli simultaneously in Experiment 3 was described as confusing, but the change to presentation of only warm stimuli on one hand did not seem to improve the recognition significantly. Furthermore, thermal feedback was continuously described as more demanding as both audio and cutaneous push feedback, mostly due to uncertainty connected to thermal feedback that was not reported for either audio or cutaneous push feedback. This could partly be a side effect of the novelty of thermal feedback. While cutaneous push also represents a novel feedback type, the idea of being pushed or tapped as feedback is more ubiquitous then being actively warmed up or cooled down to present information other than the temperature of the item. However, comments indicated that they sometimes were not sure, if their hands naturally warmed on the steering wheel, an effect that would not influence either of the other feedback types used here. While warming seemed to be the more natural choice to direct movement, reinforced by results presented in this thesis, cooling might be a better choice to present at locations which would be held for a longer time. Warming might result naturally from holding something, the hands warming up the surface of the item being held, but a cooling sensation might be unexpected enough to overcome the uncertainty. In addition, long term use of thermal feedback would have to be tested, to explore how the detection of the feedback is influenced by longer exposure. One of the participants mentioned that uncertainty increased with longer use, rather than decreased as would be imagined with increasing familiarity.

An interesting difference between the tactile feedback types of Experiment 4 could be observed: while thermal feedback led to missed turns due to its unobtrusive nature, cutaneous push feedback led to participants reacting in a sudden and unexpected way: they turned into roads that had already been partly passed by the time the feedback started. This kind of immediate reaction might be unsuitable for navigational purposes, with sudden reactions leading to dangerous situations. Other use cases could easily be imagined in a driving environment, where the calmer nature of thermal feedback could be more suitable than the attention grabbing cutaneous push feedback, such as for non-urgent notifications.

4.5 Conclusions and Research Question 2

Two experiments with thermal devices attached to the steering wheel aimed to answer Research Question 2:

How accurately can thermal feedback give direction cues in an automotive setting with the spatial information from multiple thermal devices?

Both experiments used spatial information presented by the locations of the thermal devices on the steering wheel for turn-by-turn navigation cues. While thermal feedback was presented on both hands simultaneously in the first experiment, with directional cues given by the direction of temperature change presented, the second experiment only presented thermal feedback on one side, the presentation location itself informing the driver of the turning direction. Recognition rates were high, over 87%, showing that thermal feedback on multiple devices can be utilised to present directional cues. The presentation of warm and cold stimuli simultaneously was described as confusing. In line with feedback from the user experiments discussed in Chapter 3, participants reported an increased level of uncertainty when compared to audio and cutaneous push. Furthermore, they mentioned an increased need for paying attention, as is mirrored in the perceived workload ratings. Even with high recognition rates, the use of thermal feedback by itself might lead to unsatisfied and fatigued drivers if these ratings hold true even after familiarisation with the novel feedback. Longitudinal studies will have to explore this aspect.

The exploration of thermal feedback in combination with other tactile feedback, such as vibration, could lead to a decrease in perceived workload and take advantage of the familiarity of the second feedback type. As one of the use cases consistently suggested by participants, notifications present a suitable test bed for the investigation of this bimodal tactile aspect. Chapter 5 will therefore discuss experiments exploring bimodal thermal and vibrotactile no-tifications in a driving environment.

Chapter 5

Unimodal and Multimodal Notifications with Thermal Feedback

5.1 Introduction

The previous chapters investigated the use of unimodal thermal feedback for directional cues. This chapter will move away from this use case and explore more complex thermal feedback for notifications, both unimodal and in combination with vibration, as bimodal tactile feedback. Three experiments will aim to answer Research Question 3:

How effectively can thermal feedback convey notifications in a car in aa. unimodal setting?b. bimodal tactile setting?

In all three experiments, the feedback was presented on the steering wheel, enabling the driver in the simulated environment to safely keep both hands on the steering wheel. The focus of the notifications differs in all three experiments, to allow for a broader investigation and explore several useful combinations and use cases. In the first, the influence of thermal feedback on the feeling of urgency was explored, while the second investigated how easily several layers of information could be conveyed. The third experiment aimed to enrich multimodal feedback for transfer of control with bimodal tactile feedback. Section 5.6 will follow the discussion of all three experiments with a conclusion.

In this experiment, the effectiveness of thermal and bimodal tactile feedback for notifications was tested with the affective association of urgency embedded into the feedback. Thermal and vibrotactile feedback was presented unimodally and as bimodal combination, encoding different levels of urgency. The same urgency levels were encoded in all modalities, leading to redundant information presentation in the bimodal condition.

5.2.1 Methodology

The experiment was designed as a 5x3 within-subjects user study, with Modality (Bimodal Warm, Bimodal Cold, Thermal Warm, Thermal Cold, Vibration) and Urgency (High, Medium, Low) as the independent variables.

The influence of the thermal modality on the feeling of urgency and the identification rate and time during driving were investigated to evaluate how effective notifications of different urgency levels can be conveyed. The combination of the two feedback types, thermal and vibrotactile, was expected to increase the feeling of urgency. Adding temperature changes, which have been shown to increase dominance and arousal, to the already attention-grabbing vibrotactile cues would have an influence on the perceived urgency. Due to the slowly increasing thermal cues and the temperature range used, thermal feedback was expected to be rated as less urgent than vibration. Recognition rates for the combination of the two modalities should increase, as the same information was presented in both. As shown in previous experiments, the recognition time for thermal feedback was slow, so it was expected to be slower than the other modalities here as well. However, as there was no differences in lane deviation between modalities in the previous experiment with two tactile modalities, the same was expected here. The hypotheses for this experiment, therefore, were:

Hypothesis 1: The bimodal stimuli will be rated as more urgent than the individual thermal and vibrotactile ones;

Hypothesis 2: Thermal feedback will be rated as less urgent than vibrotactile feedback;

Hypothesis 3: The recognition rate of bimodal notifications will be higher than vibrotactile; Hypothesis 4: The recognition time will be longer for thermal feedback.

In order to investigate those hypotheses, subjective urgency ratings, recognition rate, recognition time, lane deviation and subjective data were evaluated as dependent variables. More details can be found in Section 5.2.5 *Measures*.

5.2.2 Apparatus

For this experiment, two 1x1cm USB Peltier devices (Figure 5.1(a)), were attached to a Logitech G920 Driving Force Racing Wheel with elastic bands, one on each side. They were



(a) Hardware: Peltier device and vibrotactile actuator

(b) Steering Wheel

Figure 5.1: Experiment 5: Hardware and Steering Wheel.

placed so that the driver holding the wheel could place the thenar regions of their palms on the Peltiers, see Figure 5.1(b). The heat-sinks were isolated with cardboard, so participants would only feel the temperature of the Peltier device. In addition, one Haptuator Mark II^1 vibrotactile actuator (Figure 5.1(a) black bar) was taped onto the heat-sinks.

The subjective rating framework was implemented in Python 2.7, the interface can be seen in Figure 5.2. The slider was moved across the scale from 0 to 100 by moving the steering



Figure 5.2: Experiment 5: Interface for Subjective Rating in Part 1.

wheel: the slider would start in the centre of the scale, steering to the left would move it to the left, steering to right to the right of the scale. The amount of steering influenced the speed of the slider movement. The range of the axis movement ranged from -1 to 1, with 0 representing the neutral starting position. An axis movement of 0.004 to the right, or -0.004 to the left, would move the slider 1 point in the desired direction. The speed increased to 3 points on the scale for an axis movement of bigger than 0.01. Pulling either of the gear

¹http://tactilelabs.com/products/haptics/haptuator-mark-ii-v2/, accessed 31/05/2020



Figure 5.3: Experiment 5: Experimental setup.

paddles submitted the rating and triggered the presentation of the next stimulus.

The simulator environment was again implemented in OpenDS 3.5 free version and depicted a five-lane motorway with bridges (Figure 5.4). The RSeat RS1 gaming racing chair was re-used in this experiment, as was the BENQ DLP projector and the Sennheiser HD 25 Basic Edition headphones, playing driving noises during driving and white noise, generated with Audacity², during the first part. The equipment was run on a Dell XPS Windows laptop with Windows 10. The complete setup can be seen in Figure 5.3.

5.2.3 Driving Task and Feedback

The experiment was split into two parts. In Part 1, participants were asked to subjectively rate the perceived urgency of the feedback on a scale from 0 to 100. They were presented with a scale (see Figure 5.2) and could move the pointer by turning the steering wheel to the left or right. Their rating was submitted by pulling either gear paddle. The feedback was not introduced prior to the rating; participants were just told to expect the five different modalities. After each rating submission, the next feedback started randomly within 3s, and participants were asked to rate 0 if they had not felt anything after 5s. This was introduced to allow for missed stimuli and to ensure the experiment would go on. Each stimulus was presented three times in randomised order. Participants could take a break by not submitting their rating until they felt ready to go on. They were listening to white noise throughout this part of the experiment.

In Part 2 of the experiment, participants were asked to identify the correct stimuli during driving with a constant speed of 100km/h and report them by pressing corresponding but-

²https://www.audacityteam.org/, accessed 08/06/2020



Figure 5.4: Experiment 5: Simulator Environment, arrow on gantry indicating the destination lane of the desired lane change.

tons on the steering wheel. To do this, all stimuli were first presented to them twice, making sure that they could feel each cue. They were asked to report the level of urgency: the high urgency by pressing the highest button on the steering wheel (Y, see Figure 5.1(b)), medium urgency by pressing the middle button (B) and low urgency by pressing the lowest button (A). To keep participants engaged with the driving, they were prompted to change lanes. Arrows would appear on gantries placed over the five-lane motorway, see Figure 5.4, indicating the lane the car should change into. The participants were encouraged to change into these lanes before arriving at the signs. The destination lanes were selected randomly. Between the lane changes, one stimulus was presented, randomly within a time frame of 3s. The lane changes and feedback cues did not overlap. Before starting the main driving task, participants fulfilled a short training drive, in which three lane changes occurred, presenting three representative stimuli: UrgentBiendoalCold, MediumThermalWarm, LowVibration. During the main driving task, each feedback cue was presented three times in randomised order, leading to 45 cues in total. The driving part was broken up into driving blocks with nine cues each. Breaks could be taken in between the driving blocks.

The vibrotactile feedback was matched to the cues tested by Politis et al. [26], see Figure 5.5.

The sound files used for the vibration were generated with 250Hz using Audacity. The patterns encoded levels or urgency with differing number and lengths of vibrations within approximately 1.5s. The high urgency pattern had eight short vibrations of 0.1s with 0.1s intervals, see Figure 5.5(a), medium urgency was represented by five 0.17s vibration with 0.17s intervals, resulting in an overall cue time of 1.53s, see Figure 5.5(b). The low urgency only had two vibrations of 0.5s with an interval of also 0.5s, see Figure 5.5(c).

The thermal feedback was designed to match this time frame. The highest round number temperature change achievable with the hardware used within 1.5s was 4°C. This was used for high urgency rather than 4.5°C, to ensure that the temperature would be constant for some

5.2. Experiment 5: Uni- and Bimodal Tactile Notifications with Different Levels of Urgency



Figure 5.5: Experiment 5: Vibration patterns.

time, as this was shown to be more easily recognised in Experiment 2. The medium and low urgency temperature changes were 2°C and 1°C, respectively. As the high urgency warning would always be the most important one, it was decided that the temperature change between high and medium should be more detectable. The low urgency, represented by 1°C temperature change, would be very subtle and potentially hard to detect, it was therefore important to make sure that participants had the possibility to report missed stimuli in the first part of the experiment. It was of interest to see how well these subtle cues could be identified. The temperature changed with 3°C/s from the neutral temperature of 30°C and the temperature was then kept constant until the vibration was finished after 1.5s. At which point the temperature returned to neutral with 1°C/s. The thermal feedback can be seen in Figure 5.6: full line for high urgency, dashed line for medium urgency and dotted line for low urgency.

5.2.4 Participants and Procedure

Eighteen participants completed the experiment, no sensory impairments in their hands were reported and they had at least corrected vision. All participants obtained valid driving licences, fourteen in left-side driving countries. All but two participants were right-handed. Their data can be found in Table 5.1, with prior experience captured on a 5-point Likert scale (1 equalled no experience).

5.2. Experiment 5: Uni- and Bimodal Tactile Notifications with Different Levels of Urgency 10



Figure 5.6: Experiment 5: Thermal feedback.

Number of	Age	Driving	Experience	Experience	Experience
Participants	(Years)	Experience	Simulator	Vibration	Thermal
		(Years)			
18	<i>R</i> : 19-48	<i>R</i> : 0-30	<i>R</i> : 1-5	<i>R</i> : 1-5	<i>R</i> : 1-4
(Female: 10	M = 27.56	M = 8.89	M = 2.28	M = 1.94	M = 1.58
Male: 8)	SD = 7.59	<i>SD</i> = 7.47	SD = 1.07	SD = 1.06	SD = 0.86

Table 5.1: Experiment 5: Participant data, showing range (R), mean (M) and standard deviation (SD).

The participants were recruited through email lists and university advertisement services, being mostly university staff and students. On arrival at the university lab at the School of Computing Science, the participants were given the information sheet and then asked to sign the consent form. They were then seated in the racing chair and started the first part of the experiment, the subjective rating. After this, all stimuli were introduced to the participants and they had time to explore the driving part in the training session. This was followed by the driving task, consisting of five driving blocks interrupted by short breaks. At the end of the study participants filled in a questionnaire. The experiment lasted approximately one hour and £6 were offered as compensation.

All documents, including the questionnaire, can be found in Appendix A.6. The experiment design was approved by the Ethics Committee of the College of Science and Engineering of the University of Glasgow.

5.2.5 Measures

In this experiment participants were presented with stimuli outside of the driving task and without prior introduction. They were rating the subjective urgency on a scale of 0 to 100, with instructions to rate 0 when they could not feel anything within 5s.

Recognition rates for the correct urgency recognition during the driving task were counted and evaluated. Participants reported the identified urgency by pressing corresponding buttons on the steering wheel.

Recognition time reports the time frame from the start of the cue until the button on the steering wheel was pressed.

As in previous experiments, the lane deviation was calculated using the Root Mean Square Error for the x-position of the car compared to the middle of the occupied lane. The deviation was calculated for three time frames: *Before* the cue, starting at the bridge that defined the end of the lane change task until the beginning of the stimulus, *During* the 1.5s of the stimulus presentation and *After* the stimulus representation, starting when the cue finished and ending right before the next lane change. The time frame labelled *After* included the time in which the temperature of the thermal device returned to neutral.

Subjective data summarised answers captured in the questionnaire at the end of the experiment. Participants were asked to rank the modalities and give reasons for their ranking in free text and to choose which modality felt most urgent and most comfortable. For these questions, the modalities have been congregated and presented as *thermal*, *vibrotactile*, *bimodal (thermal and vibrotactile together)* and *all equally*. The ranking assigned numbers to the modalities, according to participants' preference: 1 being their most favourite and 3 their least favourite. The sums of these rankings therefore assign the lowest number to the most preferred modalities, while the highest number declares the least favourite.

In addition, participants could choose which direction of temperature change felt more urgent and more comfortable, presented with the options *warm*, *cold* and *both equally*. At the very end participants had the chance to leave comments in free text.

5.2.6 Results

The influence of modality and urgency were evaluated with repeated measure ANOVA, with Tukey corrections for *post hoc* tests, automatically applied in R to adjust for multiple comparisons. Outliers varying more than two standard deviations from the mean were excluded from the evaluation. Lane Deviation was evaluated with multivariate analysis of variance (MANOVA).

Subjective Urgency Ratings

In the first part of the experiment participants rated the subjective urgency of the cues without prior introduction. They were asked to rate 0, if they did not perceive anything within 5s. However, one participant commented after the task that they rated some cues with 0, even when they felt it, but rather found them not urgent at all.

Therefore, the 0 ratings could not be treated as missed stimuli, as intended, without additional investigation. The evaluation was made twice, once with the ratings excluded, once with them included. As the differences in the statistical results were minimal, the results of the tests for the subjective rating with the 0 ratings included were moved to Appendix A.6.4 and only the differences will be outlined here. Figure 5.7 presents subjective rating data both with and without 0 ratings. 9 Outliers were excluded for the Subjective Urgency Ratings, as they differed more than two standard deviations from the mean (1 LowBimodal-Warm, 1 LowBimodalCold, 1 LowThermalWarm, 1 LowThermalCold, 1 LowVibration, 1 MediumBimodalWarm, 1 MediumThermalWarm, 1 MediumThermalCold, 1 UrgentThermalCold). The missed stimuli themselves will be evaluated as well.



Figure 5.7: Experiment 5: Subjective Rating with and without *0* Ratings included; significant results are in **bold** font.

The subjective rating without 0 ratings with repeated measures ANOVA showed significant differences for Urgency (F(2,27) = 21.48, p < 0.001), Modality (F(4,60) = 112.17, p < 0.001) and the interaction of the two factors (F(8,116) = 3.34, p = 0.002). Post hoc tests

with Tukey corrections for Urgency showed significant differences between low - urgent (t(34) = 5.40, p = < 0.0001) and medium - urgent (t(34) = 3.22, p = 0.009), but not low - medium (t(34) = 2.06, p = 0.12). When including the 0 ratings (see Table A.1 in Appendix A.6.4), differences could be found between low - medium and low - urgent, but not medium - urgent. *Post hoc* tests with Tukey corrections for Modality showed significant results between thermal and the other modalities, see Table 5.2. A visual representation of results for *post hoc* tests with Tukey corrections of the interaction of the two factors as a heat map can be seen in Table 5.3 (all results in Appendix A.6.5). All thermal cues (light gray background in labels of the heat map) were significantly different from bimodal cues and vibration. Within the thermal conditions, LowThermalCold (LTC) was different from both urgentThermalWarm (UTW). The evaluation with 0 ratings included had similar results, see Appendix A.6.4: only LowThermalCold was not found to be different from UrgentThermalCold.

Modality - Modality	t(60)	p
Bimodal Cold - Bimodal Warm	1.34	0.67
Bimodal Cold - Thermal Cold	12.62	< 0.0001*
Bimodal Cold - Thermal Warm	10.86	< 0.0001*
Bimodal Cold - Vibration	0.65	0.97
Bimodal Warm - Thermal Cold	13.82	< 0.0001*
Bimodal Warm - Thermal Warm	12.03	< 0.0001*
Bimodal Warm - Vibration	0.69	0.96
Thermal Cold - Thermal Warm	1.67	0.46
Thermal Cold - Vibration	13.23	< 0.0001*
Thermal Warm - Vibration	11.63	< 0.0001*

Table 5.2: Experiment 5: Statistics of *post hoc* test results for Modality for Subjective Ratings without *0* ratings; significant results are bold and marked with an asterisk.

	L B C	M B C	U B C	L B W	M B W	U B W	L T C	M T C	U T C	L T W	M T W	U T W	L V	M V	U V
L															
M															
B-C															
U															
B-C															
L B-W															
M															
B-W															
U															
B-W															
L T-C															
М															
T-C															
U T-C															
L															
T-W															
M T-W															
U	_														
1-W															
L V															
M V															
U															
V															

5.2. Experiment 5: Uni- and Bimodal Tactile Notifications with Different Levels of Urgency

Table 5.3: Experiment 5: Heat Map of post hoc tests for the Interaction between Urgency and Modality for Subjective Ratings without 0 ratings included, all results can be found in Appendix A.6.5; Low, Medium and Urgent notifications for Bimodal, Thermal, Cold and <u>W</u>arm, and <u>V</u>ibration are coloured yellow (p < 0.05), orange (p < 0.001) and red (p < 0.001) 0.0001).

Missed Stimuli (*0* Ratings)

Figure 5.8 shows the number of missed stimuli or 0 ratings during the Subjective Rating at the beginning of the experiment.



Figure 5.8: Experiment 5: Number of Missed Stimuli (0 Ratings) during the Subjective Rating.

The evaluation with repeated measures ANOVA showed significant differences for Urgency (F(2,34) = 13.21, p < 0.001), Modality (F(4,68) = 20.66, p < 0.001) and the interaction of Urgency and Modality (F(8, 136) = 6.01, p < 0.001). Post hoc tests with Tukey corrections for Urgency showed significant differences between low - urgent (t(34) = 4.97, p = 0.0001) and medium - urgent (t(34) = 3.63, p = 0.003), but not low - medium (t(34) = 1.38, p = 0.39). Post hoc tests for Modality showed significant results between the thermal conditions and the other modalities. Results of the post hoc tests with Tukey corrections for the interaction are represented in the heat map in Table 5.5. All results for missed stimuli can be found in Appendix A.6.6. Differences were found between all thermal medium and low cues and the bimodal and vibrotactile cues. Urgent thermal cues were an exception, where only thermal cold cues were different from the bimodal cues, except for the low bimodal warm cues. Within the thermal cues, differences were found only for the urgent thermal cues: Urgent-ThermalCold (UTC) was different from LowThermalCold (LTC), and UrgentThermalWarm (UTW) from LowThermalCold (LTC) and Low and MediumThermalWarm (MTW).

Modality - Modality	t(68)	р
Bimodal Cold - Bimodal Warm	0.11	1.00
Bimodal Cold - Thermal Cold	5.74	< 0.0001*
Bimodal Cold - Thermal Warm	6.07	< 0.0001*
Bimodal Cold - Vibration	0.00	1.00
Bimodal Warm - Thermal Cold	5.63	< 0.0001*
Bimodal Warm - Thermal Warm	5.96	< 0.0001*
Bimodal Warm - Vibration	0.11	1.00
Thermal Cold - Thermal Warm	0.33	1.00
Thermal Cold - Vibration	5.74	< 0.0001*
Thermal Warm - Vibration	6.07	< 0.0001*

Table 5.4: Experiment 5: Statistics of *post hoc* test results for Modality for 0 Ratings; significant results are bold and marked with an asterisk.

	L B C	M B C	U B C	L B W	M B W	U B W	L T C	M T C	U T C	L T W	M T W	U T W	L V	M V	U V
L B-C															
M B-C															
U B-C															
L B-W															
M B-W															
U B-W															
L T-C															
M T-C															
U T-C															
L T-W															
M T-W															
U T-W															
L V															
M V															
U V															

Table 5.5: Experiment 5: Heat Map of *post hoc* tests for the Interaction between Urgency and Modality for Missed Stimuli (0 Ratings) during the Subjective Rating, all results can be found in Appendix A.6.6; <u>Low</u>, <u>M</u>edium and <u>U</u>rgent notifications for <u>B</u>imodal, <u>T</u>hermal, <u>C</u>old and <u>W</u>arm, and <u>V</u>ibration are coloured yellow (p < 0.05), orange (p < 0.001) and red (p < 0.0001).

Recognition Rate

The recognition rate of the second part of the experiment can be seen in Figure 5.9. The highest recognition rate was achieved for UrgentBimodalWarm with 93%, the lowest for UrgentThermalCold with 20%. The statistical evaluation with repeated measures ANOVA



Figure 5.9: Experiment 5: Recognition Rate.

showed significant differences for Modality (F(4,68) = 21.41, p < 0.0001) and the interaction between Urgency and Modality (F(8,136) = 3.91, p = 0.0004), but not Urgency (F(2,34) =2.48, p = 0.10).

Post hoc tests with Tukey correction for the factor Modality showed significant differences between both thermal conditions and the other cues, see Table 5.6.

Results for the *post hoc* tests with Tukey corrections of the interaction can be found as a heat map in Table 5.7, all results can be seen in Appendix A.6.7.

No differences between vibrotactile and bimodal cues was found, most significant results were in connection with thermal cues, some between bimodal cues. The urgent bimodal cold cue was significantly different from all thermal stimuli, while MediumBimodalCold (MBC) was different from all thermal warm stimuli and only the urgent thermal cold one and the low bimodal cold stimulus was different from all warm thermal stimuli and the medium and urgent bimodal cold cues. The low bimodal warm stimulus was only found to be different from both urgent thermal stimuli and the urgent bimodal warm cue. The medium bimodal warm stimulus was different from the urgent thermal stimuli and the medium warm thermal cue, while UrgentBimodalWarm (UBW) was different from all thermal stimuli. Low thermal warm and medium thermal cold were found to only be different from UrgentVibation

Modality - Modality	t(68)	p
Bimodal Cold - Bimodal Warm	0.35	1.00
Bimodal Cold - Thermal Cold	5.82	< 0.0001*
Bimodal Cold - Thermal Warm	6.88	< 0.0001*
Bimodal Cold - Vibration	0.964	0.87
Bimodal Warm - Thermal Cold	5.47	< 0.0001*
Bimodal Warm - Thermal Warm	6.51	< 0.0001*
Bimodal Warm - Vibration	0.61	0.97
Thermal Cold - Thermal Warm	1.04	0.84
Thermal Cold - Vibration	4.86	0.0001*
Thermal Warm - Vibration	5.90	< 0.0001*

5.2. Experiment 5: Uni- and Bimodal Tactile Notifications with Different Levels of Urgency

Table 5.6: Experiment 5: Statistics of *post hoc* test results for Modality for Recognition Rate; significant results are bold and marked with an asterisk.

(UV), while UrgentThermalCold (UTC) was different from vibrotactile stimuli of all urgencies. Medium thermal warm stimuli were different from both medium and urgent vibration, while UrgentThermalWarm (UTW) was found to be different from vibrotactile stimuli of all urgencies. Thermal cues, especially of lower urgency, were recognised correctly less often than the other modalities.

Figure 5.10 shows the rated urgencies for each cue during the second part. In the urgent thermal modalities, the stimuli were rated more often as medium than urgent, for cold temperatures even more often low than urgent. In the medium condition, the rating for thermal almost as often medium as low, only the low thermal condition seemed to be easier to recognise correctly. The number of wrongly rated vibrotactile stimuli was higher than anticipated, especially for medium cues which were rated as urgent.

	L B C	M B C	U B C	L B W	M B W	U B W	L T C	M T C	U T C	L T W	M T W	U T W	L V	M V	U V
L															
M															
B-C															
U															
B-C															
L B-W															
M B-W															
U B-W															
L T-C															
M T-C															
U T-C															
L T-W															
M T-W															
U T-W															
L V															
M V															
U V															

Table 5.7: Experiment 5: Heat Map of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Rate, all results can be found in Appendix A.6.7; <u>Low</u>, <u>M</u>edium and <u>Urgent notifications for <u>B</u>imodal, <u>T</u>hermal, <u>C</u>old and <u>W</u>arm, and <u>V</u>ibration are coloured yellow (p < 0.05), orange (p < 0.001) and red (p < 0.0001).</u>



5.2. Experiment 5: Uni- and Bimodal Tactile Notifications with Different Levels of Urgency

Figure 5.10: Experiment 5: Recognition: All Rating in Numbers.

Recognition Time

An overview of the recognition times can be found in Figure 5.11.

The evaluation with repeated measures ANOVA showed significant results for Modality $(F(2,28) = 7.71 \ p = 0.0002)$ and the interaction of the factors $(F(4,28) = 6.19, \ p = 0.001)$, but not Urgency (F(2,28) = 2.93, p = 0.07). Post hoc tests with Tukey corrections showed significant differences between almost all modalities, see Table 5.8. Only the differences between Bimodal Warm and Cold and between Thermal Warm and Cold were not found to

Modality - Modality	t(63)	p
Bimodal Cold - Bimodal Warm	0.15	1.00
Bimodal Cold - Thermal Cold	3.26	0.01*
Bimodal Cold - Thermal Warm	5.02	< 0.0001*
Bimodal Cold - Vibration	3.14	0.02*
Bimodal Warm - Thermal Cold	3.40	0.01*
Bimodal Warm - Thermal Warm	5.16	< 0.0001*
Bimodal Warm - Vibration	2.99	0.03*
Thermal Cold - Thermal Warm	1.77	0.40
Thermal Cold - Vibration	6.25	< 0.0001*
Thermal Warm - Vibration	7.96	< 0.0001*

Table 5.8: Experiment 5: Statistics of post hoc test results for Modality for Recognition Time; significant results are bold and marked with an asterisk.

111



5.2. Experiment 5: Uni- and Bimodal Tactile Notifications with Different Levels of Urgency 112

Figure 5.11: Experiment 5: Recognition Time.

be different. The thermal cues (warm: M = 6.01s, SD = 2.28; cold: M = 5.27s, SD = 32.01) took longer to be recognised than the other modalities and the bimodal modalities (warm M = 4.47s, SD = 0.81; cold: M = 4.50s, SD = 1.17) took in turn longer than vibration (M = 3.67, SD = 0.59). Significant results of *post hoc* tests with Tukey corrections for the interaction can be seen in Table 5.9 (all results in Appendix A.6.8). Low and medium bimodal cues were only different from the urgent vibrotactile stimulus. UrgentBimodalCold and Urgent-BimodalWarm were different from all thermal cues apart from UrgentThermalCold. Low and medium thermal stimuli and UrgentThermalWarm were different from all vibrotactile cues, while UrgentThermalCold was only different from UrgentVibration.

Urgency, Modality - Urgency, Modality	t	p
Low,Bimodal Cold - Urgent,Vibration	t(196) = 4.35	0.002*
Medium, Bimodal Cold - Urgent, Vibration	t(195) = 3.96	0.009*
Urgent, Bimodal Cold - Low, Thermal Cold	t(194) = 4.51	0.001*
Urgent, Bimodal Cold - Medium, Thermal Cold	t(193) = 4.03	0.007*
Urgent, Bimodal Cold - Low, Thermal Warm	t(192) = 5.21	<0.0001*
Urgent,Bimodal Cold - Medium,Thermal Warm	t(191) = 4.90	0.0002*
Urgent, Bimodal Cold - Urgent, Thermal Warm	t(181) = 4.66	0.0006*
Low,Bimodal Warm - Urgent,Vibration	t(196) = 3.65	0.03*
Medium, Bimodal Warm - Urgent, Vibration	t(195) = 3.81	0.02*
Urgent,Bimodal Warm - Low,Thermal Cold	t(194) = 3.97	0.009*
Urgent,Bimodal Warm - Medium,Thermal Cold	t(193) = 3.49	0.04*
Urgent,Bimodal Warm - Low,Thermal Warm	t(192) = 4.67	0.0005*
Urgent,Bimodal Warm - Medium,Thermal Warm	t(191) = 4.27	0.002*
Urgent,Bimodal Warm - Urgent,Thermal Warm	t(181) = 4.13	0.005*
Low, Thermal Cold - Low, Vibration	t(183) = 4.12	0.005*
Low, Thermal Cold - Medium, Vibration	t(194) = 4.32	0.002*
Low, Thermal Cold - Urgent, Vibration	t(194) = 6.11	<0.0001*
Medium, Thermal Cold - Low, Vibration	t(191) = 3.80	0.02*
Medium, Thermal Cold - Medium, Vibration	t(181) = 3.75	0.02*
Medium, Thermal Cold - Urgent, Vibration	t(193) = 5.64	<0.0001*
Urgent, Thermal Cold - Urgent, Vibration	t(181) = 4.29	0.003*
Low, Thermal Warm - Low, Vibration	t(182) = 4.81	0.0003*
Low, Thermal Warm - Medium, Vibration	t(191) = 5.03	0.0001*
Low, Thermal Warm - Urgent, Vibration	t(192) = 6.78	<0.0001*
Medium, Thermal Warm - Low, Vibration	t(191) = 4.66	0.0006*
Medium, Thermal Warm - Medium, Vibration	t(182) = 4.56	0.0009*
Medium, Thermal Warm - Urgent, Vibration	t(191) = 6.49	<0.0001*
Urgent, Thermal Warm - Low, Vibration	t(191) = 4.52	0.001*
Urgent, Thermal Warm - Medium, Vibration	t(190) = 4.58	0.0008*
Urgent, Thermal Warm - Urgent, Vibration	t(181) = 6.20	<0.0001*

Table 5.9: Experiment 5: Significant results of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Time, all results can be found in Appendix A.6.8.

Lane Deviation

The lane deviation can be seen in Figure 5.12. Evaluation with MANOVA showed no significant differences for any factor (Urgency: F(2,508) = 1.80, p = 0.10; Modality: F(4,765)= 0.35, p = 0.98; Urgency-Modality: F(8,765) = 1.05, p = 0.39).



Figure 5.12: Experiment 5: Lane Deviation.

Subjective Data

At the end of the experiment participants ranked the modalities (congregated as thermal, bimodal and vibrotactile) according to their preference. Vibration achieved the lowest number in the ranking with 27, declaring it to be the most preferred modalities. It was followed closely by bimodal with 29 and thermal with 52.

Some of the reasons were named by P10: Vibration was very clear. With thermal, I didn't know if it (the warmth) was the heat from my hands, the warm steering wheel or the thermal warning. P17 wrote that the thermal feedback confused my senses. The same sentiment was expressed by P14: the vibrotactile feedback gave me more clear info about the urgency, while the other two ones tended to confuse me. P06 wrote: I found it hard to distinguish between the intensity of the thermal rating, and also because I felt the steering wheel remained hot, it didn't have much of affect for me. The vibrations were easiest as you could count how many, a sentiment that was shared by P11: Much easier to detect and differentiate between each

vibrotactile sensation, deeming bimodal unnecessary. However, some participants preferred the combination of vibrotactile and thermal cues. P07 commented: With thermal feedback after a while you are not really feeling the temperature change especially if it is subtle. With the vibrotactile feedback I still need to think whether its urgent or not whereas the bimodal feedback gives you two feelings which I prefer (temperature change and the vibration). And P18 mentioned that some of the temperature changes at lower intensities were hard to detect, vibrations were always detectable. However, when combined with vibration, the temperature changes made the overall warning feel more urgent. P01 wrote that the more feedback the better - both at once makes things seem very urgent, vibrating next and lastly thermal as sometimes hard to feel. P05 also commented on the uncertainty of thermal feedback, as was mentioned in previous experiments: I couldn't always feel the thermal feedback. Combination of thermal and vibration feedback did not improve it either, since I was relying primarily on the vibration feedback. I was sure I am not going to miss [it], while with thermal feedback I was always uncertain. As for preferences of warm or cold stimuli, some participants mentioned that the cold stimuli were uncomfortable (P13) or even a little bit painful (P09), while P04 wrote that cold changes were better felt I think than warm.

After the ranking participants rated which modality felt most urgent: 11 named bimodal, 6 vibrotactile and 1 thermal. However, when asked for the most comfortable modality, most participants preferred vibrotactile (10), followed by bimodal with 6. Two participants found all modalities equally comfortable. None rated thermal as most comfortable.

When asked about which direction of temperature change felt more urgent and more comfortable, more participants named warm as being more urgent (12, to 5 naming cold and 1 both equally), while they were tied for being most comfortable: 8 participants each named warm and cold, 2 found both equally comfortable.

In overall comments at the end of the experiment *P18* remarked that *it was sometimes very difficult to detect the smallest temperature changes - following a temp change of lowest intensity, I was sometimes aware that a temp change had occurred but wasn't sure whether it had gotten hotter or colder*, again pointing out a certain perceived uncertainty connected with thermal stimuli. A sentiment shared by *P10: just having a heated steering wheel wasn't very comfortable, and it seemed to make temperature differences less perceptible. I was never sure if I was feeling it until it got really warm, but then again - maybe it wasn't the warning after all. The vibration was a more visceral stimulus that made me more alert immediately. P04* pointed out the different perception of the directions of change, which was also mentioned in previous experiments: *cold more noticeable. Unsure whether warm change or from gripping steering wheel. P05* mentioned *thermal feedback was perceived binary and it was hard to distinguish the levels. [...] In some situations however I was paying more attention to the presence of the thermal feedback, when the vibration came. Every time the vibration came I was asking myself: is thermal feedback also here?*, suggesting that thermal and vibrotactile feedback in combination might be used in a different way.

5.2.7 **Experiment 5 Discussion**

This experiment was conducted to evaluate thermal and tactile bimodal feedback for three different levels of urgency. In the first part of the experiment, participants rated the perceived urgency of the stimuli on a scale from 0 to 100 without prior introduction to them. They were instructed to rate 0, when they could not perceive any cue within 5s, but comments from a participant suggested that 0 was also used to rate thermal feedback which did not feel urgent at all. So these 0 ratings could not be used as a measure for missed stimuli without additional exploration.

The difference found by including 0 ratings was observed between thermal conditions: the low thermal cold cue was found to be different from the urgent thermal cold stimulus without the 0 ratings, but not with them included. Other differences found in both evaluations were between the thermal and the bimodal and vibrotactile modalities. Thermal cues were rated consistently as less urgent than any other modality corroborating *Hypothesis 2*. But they were also missed, or rated 0, most often. In fact, only one non-thermal cue was missed: a low bimodal warm cue. These differences were significant. The number of misses were higher for the more subtle thermal cues, as was expected. But bimodal cues were not rated as more urgent than vibrotactile cues, so Hypothesis 1 could not be corroborated. However, rankings and comments at the end of the experiment did suggest that most participants consider bimodal feedback to represent more urgent feedback. They also rated warm temperature changes to feel more urgent than cold ones, even though this was not mirrored in the subjective rating at the beginning.

The recognition during driving of the second part of the experiment showed an influence of the interaction between the two factors Urgency and Modality, so no easy observations can be made for the recognition rate. However, there were only differences found between thermal and bimodal cues and thermal and vibrotactile cues, even though not always of all urgencies. But there were no significant differences between bimodal and vibrotactile cues, so *Hypothesis 3* could not be corroborated. These interactions could be a result of the small difference between the low and medium thermal cues. The numbers of wrongly rated cues indicated that especially the medium thermal cues were often mistaken for low urgency ones. In addition, urgent thermal cues were rated medium and even low often, with the urgent cold cue being rated urgent less often than both other urgency levels. Participants also reported on problems differentiating the thermal cues. In addition, participants misidentified an unanticipated high number of vibrotactile cues during driving. Medium and low vibrotactile patterns were mixed up almost one third of the time, with medium often interpreted as urgent and low as medium. The chosen cue design in general seemed to not have produced easily dis-

tinguishable levels of information. Decreasing the number of levels from three to two and lengthening the time for cue presentation of the thermal feedback could aid in creating more distinct levels of information.

As in previous experiments, it can also be observed that the perception of warm and cold stimuli was not always simply mirrored. As the preferences of the two directions of change differ, so does the perception of them. As thermal preferences can differ between individuals, the feedback might have to be personalisable for practical use. In an in-car environment, this could be easily accomplished by an additional setting for each driver, which could be set once and only adapted when needed.

Evaluation of recognition time showed that thermal cues took longer to be recognised than both bimodal and vibrotactile cues, corroborating *Hypothesis 4*. It was also seen that vibrotactile feedback was recognised faster than bimodal cues. The advantage of increased reaction time of vibration found in previous research seemed to have been compromised by the addition of thermal feedback. This addition in recognition time renders bimodal feedback, and in extension thermal feedback, unsuitable for urgent notifications in a driving environment, as an increase of reaction time might have fatal consequences. However, these modalities could be very useful for non-urgent notifications. One participant mentioned that their attention towards thermal cues was heightened after feeling vibration. Combining thermal and vibrotactile cues to convey different kinds of information might benefit from the certainty participants feel towards vibration and help them recognise additional thermal feedback without the uncertainty reported with it. Recognition rates for some thermal cues have been high and could reliably present information different from the one given through vibration. The next experiment will explore this idea and present two different types of information, both unimodally with thermal alone and bimodally in combination with vibration.

5.3 Experiment 6: Uni- and Bimodal Tactile Notifications with Different Types of Information

This experiment investigated thermal and bimodal tactile cues to present information of two different types, with the bimodal cues consisting of thermal and vibrotactile feedback. The cue design is comparable to the study conducted by Wilson *et al.* [44].

5.3.1 Methodology

The study was designed as a 2x2x2 within-subjects experiment, with Modality (thermal/bimodal), Importance (high/low) and Nature (personal/work) of a message as the independent variables.

As bimodal feedback has been identified more easily in the previous experiments, it is expected to fare the same in this experiment. However, as the thermal cues were chosen from the highest performing cues from Experiment 2, the difference might not be very much pronounced. Preference ratings at the end of the experiment were, however, still expected to favour bimodal feedback, as participants have felt more assured in their senses when vibration was involved. The hypotheses for this study, therefore, were:

Hypothesis 1: Recognition of bimodal stimuli will be higher; Hypothesis 2: Preference ratings will favour the bimodal condition.

The dependent variables used to discuss these hypotheses were recognition rate and subjective ratings. More details on these can be found in Section 5.3.5 *Measures*.

5.3.2 Apparatus

The experimental setup was similar to Experiment 5, the only difference being that large 2x2cm Peltier devices were attached to the steering wheel with sports bands, see Figure 5.13.



Figure 5.13: Experiment 6: Steering Wheel.

5.3.3 Driving Task and Feedback

The driving task adopted for this experiment was the same as in Experiment 5. Participants navigated the simulated car with a constant speed of 100km/h on a five-lane motorway and

changed lanes when prompted by appearing arrows on gantries in front of them. The destination lanes were chosen randomly and participants were asked to change into the lanes before they reached the gantry. The feedback was more complex than in Experiment 5 and the used Peltier devices bigger, which made reaching the buttons on the steering wheel without moving the hands off the thermal devices unfeasible. The participants, therefore, reported the identified Importance and Nature of the feedback orally back to the experimenter, who logged them by pressing corresponding keys on the keyboard. All cues were presented 8 times, resulting in 40 cues overall. These were presented within 45 lane changes (as in Experiment 5), resulting in 5 *empty* lane changes without any feedback presented. This was adopted to ensure that participants would not feel pressured to report a cue at each lane change. The order of the presented stimuli and empty lane changes was randomised.

The feedback combined thermal stimuli from Experiment 2 and vibrotactile stimuli from Experiment 5. The thermal patterns were chosen from the cues with the highest recognition rates (over 90%) and the lowest rates for false positives (under 10%).

The factor Importance in the bimodal condition was presented by the vibrotactile pattern used in Experiment 5: the high urgency pattern for high importance (see Figure 5.14(a)) and the low urgency pattern for low importance (see Figure 5.14(b)).



Figure 5.14: Experiment 6: High and Low Importance Vibration Patterns.

The vibration of the bimodal cues was combined with the thermal patterns c/w-3-6-s from Experiment 2, see Figure 5.15, dashed lines. The temperature was changed 6°C with a rate of 3°C/s, kept constant for 3s and then returned to the neutral temperature of 30°C with a rate of 1°C/s.

The direction of temperature change encoded the factor Nature: a cold temperature change was used for work messages, while a warm temperature change was used for personal messages. This mapping was chosen, because previous research showed that warm temperature increased perceived friendship and emotional stability [111] and even reminded participants

5.3. Experiment 6: Uni- and Bimodal Tactile Notifications with Different Types of Information



Figure 5.15: Experiment 6: Thermal feedback.

of a *loving home* [112] and it was assumed that mapping warm to home would feel more natural. The thermal feedback in the bimodal condition started once the vibration was finished. While the direction of change also depicted the Nature in the thermal condition, two different thermal patterns were used to encode the Importance of a message. Low importance messages were presented by very short thermal changes, c/w-0-6-s, where the temperature was changed 6°C with 3°C/s and returned to neutral with 1°C/s directly after reaching the goal temperature, see Figure 5.15, dotted lines. High importance messages were presented with the thermal pattern c/w-3-6-a, where the temperature was changed 6°C with 3°C/s, held at that temperature for 3s and then returned to the neutral temperature at the *angled* rate. At this rate the temperature was changed for 1°C at a rate of 1°C/s, held constant for 1s and then changed again, keeping this pattern until the temperature reached the neutral temperature of 30°C, see Figure 5.15 solid line.

5.3.4 Participants and Procedure

Eighteen newly recruited participants completed the experiment. None had sensory impairments in their hands and only corrected vision. They all had valid driving licenses, twelve obtained theirs in left-side driving countries. Table 5.10 shows more detailed participant data.

The study was advertised via email lists and using university services and therefore attracted mostly students and university employees. The study was conducted in labs within the School of Computing Science and after welcoming the participants, they were presented with an information sheet and a consent form, which they signed. While seated in the gaming racing chair, participants were introduced to the feedback, showing them all stimuli. This

Number of Participants	Age (Years)	Driving Experience (Years)	Experience Simulator	Experience Vibration	Experience Thermal
18 (Male: 8	<i>R</i> : 18-36	<i>R</i> : 0.5-18	<i>R</i> : 1-3	<i>R</i> : 1-5	<i>R</i> : 1-3
Female: 9 Non-Binary: 1)	M = 25.72 SD = 4.93	M = 6.33 SD = 4.80	M = 2.00 SD = 0.77	M = 2.77 SD = 1.64	M = 1.56 SD = 0.78

5.3. Experiment 6: Uni- and Bimodal Tactile Notifications with Different Types of Information 121

Table 5.10: Experiment 6: Participant data, showing range (R), mean (M) and standard deviation (SD).

was followed by a training session before the main driving task started, which was broken up in five blocks of driving with short breaks in between. At the end of the experiment, participants filled in the questionnaire and were compensated with a £10 Amazon voucher for taking part in the one hour study.

The documents for this experiment can be found in Appendix A.7. The experimental design was approved by the College Ethics Committee.

5.3.5 Measures

The most important dependent variable in this experiment was recognition rate. The recognition of the two factors Nature and Importance will be investigated independently and together.

The recognised cue was reported orally to the experimenter and then recorded by them, so the recognition time could not be evaluated because of the occurring delay.

Lane Deviation was calculated in the same way as in Experiment 5.

Subjective data and ratings were collected at the end of the experiment, including a ranking of how well the stimuli could be identified, as well as how much they liked the modalities and how easy they could differentiate the thermal and vibrotactile patterns, each on a 5-point Likert scale.

5.3.6 Results

The influence of the three independent variables was evaluated with repeated measures ANOVA. *Post hoc* tests were done with Tukey corrections, automatically applied in R to adjust for multiple comparisons. Lane Deviation was calculated with MANOVA. The subjective data were evaluated with Friedman's ANOVA, with Wilcoxon pairwise comparisons as *post hoc* tests, or Wilcoxon for comparison of only two factors.



5.3. Experiment 6: Uni- and Bimodal Tactile Notifications with Different Types of Information

Figure 5.16: Experiment 6: Recognition Rate.

Recognition Rate

The recognition rates for all stimuli can be seen in Figure 5.16. The cue with the lowest recognition was ThermalLowPersonal, the short warm thermal stimulus, with a recognition rate of 60% (single factors: Nature 64%, Importance 63%). The highest recognition was achieved for ThermalHighWork, the long cold thermal cue, with a recognition rate of 92% (single factors: Nature 94%, Importance 93%). All other thermal cues were recognised correctly between 88% and 89%, the bimodal cues between 86% and 87%. Recognition of single factors over thermal stimuli (excluding ThermalLowPersonal) was between 91% and 96% for Nature and 88% and 93% for Importance. There was no significant difference found with repeated measures ANOVA for factor Modality, but all other factors and interactions, see Table 5.11.

High important notifications (M = 288.33, SD = 23.31) were better recognised than low im-

Modality	Importance	Nature	Modality:	Modality:	Importance:	Modality:
			Importance	Nature	Nature	Importance:
						Nature
F(1,17)	<i>F</i> (1,17)	<i>F</i> (1,17)	<i>F</i> (1,17)	<i>F</i> (1,17)	<i>F</i> (1,17)	<i>F</i> (1,17)
= 0.44	= 17.94	= 6.60	= 18.3	= 7.61	= 15.72	= 8.67
<i>p</i> = 0.51	<i>p</i> < 0.001*	<i>p</i> = 0.02*	<i>p</i> < 0.001*	<i>p</i> = 0.01*	<i>p</i> = 0.001*	<i>p</i> = 0.009*

Table 5.11: Experiment 6: Statistics for Recognition Rate; significant results are bold and marked with an asterisk.

122

portant notifications (M = 256.11, SD = 25.95) and work notifications (M = 288.33, SD = 21.35) better than personal ones (M = 256.11, SD = 27.91).

Modality:Importance	t	р
BimodalHigh - ThermalHigh	<i>t</i> (20.5)=0.67	0.36
BimodalHigh - BimodalLow	<i>t</i> (<i>33.9</i>)=0.20	1.00
BimodalHigh - ThermalLow	<i>t</i> (<i>20.1</i>)=1.87	0.27
ThermalHigh - BimodalLow	<i>t</i> (<i>20.1</i>)=0.60	0.93
ThermalHigh - ThermalLow	<i>t</i> (<i>33.9</i>)=6.02	< 0.001*
BimodalLow - ThermalLow	<i>t</i> (20.5)=1.95	0.24

Table 5.12: Experiment 6: Statistics of *post hoc* test results of Recognition Rate for the interaction between Modality and Importance; significant results are bold and marked with an asterisk.

The *post hoc* tests with Tukey corrections for the interaction for Modality and Importance can be seen in Table 5.12. The only significant result was found between thermal of high and low importance, where ThermalHigh (M = 135.00, SD = 17.45) had a better recognition than ThemalLow (M = 103.89, SD = 21.82).

Modality:Nature	t	p
BimodalPersonal - ThermalPersonal	<i>t</i> (23.2)=1.71	0.34
BimodalPersonal - BimodalWork	<i>t</i> (<i>33.3</i>)=0.13	1.00
BimodalPersonal - ThermalWork	t(25.0)=0.56	0.94
ThermalPersonal - BimodalWork	t(25.0)=1.75	0.32
ThermalPersonal - ThermalWork	<i>t</i> (<i>33.3</i>)=3.75	0.004*
BimodalWork - ThermalWork	<i>t</i> (<i>23.2</i>)=0.49	0.96

Table 5.13: Experiment 6: Statistics of *post hoc* test results of Recognition Rate for the interaction between Modality and Nature; significant results are bold and marked with an asterisk.

The *post hoc* tests with Tukey corrections for the interaction between Modality and Nature can be seen in Table 5.13 and only showed significant differences between ThermalPersonal (M = 104.44, SD = 24.65) and ThermalWork (M = 136.11, SD = 14.63), where work were recognised better than personal notifications. The *post hoc* tests with Tukey corrections for the interaction between Importance and Nature can be seen in Table 5.14. There were significant differences between high and low personal notifications, where HighPersonal (M = 131.67, SD = 25.28) was recognised better than LowPersonal (M = 103.33, SD = 30.54), between LowPersonal and HighWork (M = 135.56, SD = 21.33), with the former was recognised worse than the latter and between low personal and work notifications, where LowWork (M = 131.11, SD = 21.37) was recognised better than LowPersonal. The *post hoc* tests with Tukey corrections for the interaction between all three factors shows significant differences between ThermalLowPersonal and all other stimuli, but no other cues (Table 5.15).

5.3.	Experiment 6:	Uni- and Bimodal '	Tactile Notifications	with Different Ty	pes of
Info	rmation				124

Importance:Nature	t	p
HighPersonal - LowPersonal	<i>t</i> (<i>32.0</i>)=5.78	< 0.001*
HighPersonal -HighWork	t(24.1)=0.64	0.92
HighPersonal - LowWork	t(28.0) = 0.00	1.00
LowPersonal - HighWork	t(28.0)=4.39	0.001*
LowPersonal - LowWork	<i>t</i> (<i>24.1</i>)=4.01	0.003*
HighWork - LowWork	t(32.0)=0.92	0.79

Table 5.14: Experiment 6: Statistics of *post hoc* test results of Recognition Rate for the interaction between Importance and Nature; significant results are bold and marked with an asterisk.

Modality:Importance:Nature	t	p
BimodalHighPersonal - ThermalHighPersonal	<i>t</i> (<i>32.0</i>)=0.45	1.00
BimodalHighPersonal - BimodalLowPersonal	<i>t</i> (<i>63.8</i>)=0.29	1.00
BimodalHighPersonal - ThermalLowPersonal	<i>t</i> (29.0)=3.53	0.03*
BimodalHighPersonal - BimodalHighWork	<i>t</i> (55.4)=0.23	1.00
BimodalHighPersonal - ThermalHighWork	<i>t</i> (<i>31.4</i>)=0.90	0.98
BimodalHighPersonal - BimodalLowWork	<i>t</i> (58.3)=0.22	1.00
BimodalHighPersonal - ThermalLowWork	<i>t</i> (<i>33.6</i>)=0.29	1.00
ThermalHighPersonal - BimodalLowPersonal	<i>t</i> (29.0)=0.31	1.00
ThermalHighPersonal - ThermalLowPersonal	<i>t</i> (<i>63.8</i>)=7.58	< 0.001*
ThermalHighPersonal - BimodalHighWork	<i>t</i> (<i>31.4</i>)=0.30	1.00
ThermalHighPersonal - ThermalHighWork	<i>t</i> (55.4)=0.68	1.00
ThermalHighPersonal - BimodalLowWork	<i>t</i> (<i>33.6</i>)=0.29	1.00
ThermalHighPersonal - ThermalLowWork	<i>t</i> (58.3)=0.22	1.00
BimodalLowPersonal - ThermalLowPersonal	t(32.0) = 3.59	0.02*
BimodalLowPersonal - BimodalHighWork	t(58.3) = 0.00	1.00
BimodalLowPersonal - ThermalHighWork	<i>t</i> (<i>33.6</i>)=0.73	1.00
BimodalLowPersonal - BimodalLowWork	t(55.4) = 0.00	1.00
BimodalLowPersonal - ThermalLowWork	<i>t</i> (<i>31.4</i>)=0.15	1.00
ThermalLowPersonal - BimodalHighWork	<i>t</i> (<i>33.6</i>)=3.53	0.02*
ThermalLowPersonal - ThermalHighWork	<i>t</i> (58.3)=6.46	< 0.001*
ThermalLowPersonal - BimodalLowWork	<i>t</i> (<i>31.4</i>)=3.59	0.02*
ThermalLowPersonal - ThermalLowWork	<i>t</i> (<i>55.4</i>)=5.66	< 0.001*
BimodalHighWork - ThermalHighWork	<i>t</i> (<i>32.0</i>)=0.75	1.00
BimodalHighWork - BimodalLowWork	<i>t</i> (<i>63.8</i>)=0.00	1.00
BimodalHighWork - ThermalLowWork	<i>t</i> (<i>29.0</i>)=0.15	1.00
ThermalHighWork - BimodalLowWork	<i>t</i> (29.0)=0.77	1.00
ThermalHighWork - ThermalLowWork	<i>t</i> (<i>63.8</i>)=1.17	0.94
BimodalLowWork - ThermalLowWork	<i>t</i> (<i>32.0</i>)=0.15	1.00

Table 5.15: Experiment 6: Statistics of *post hoc* test results of Recognition Rate for the interaction between Modality, Importance and Nature; significant results are bold and marked with an asterisk.
Lane Deviation

The evaluation of the lane deviation with MANOVA (see Figure 5.17) showed no significant differences for any factor or time frame, see Table 5.16.



Figure 5.17: Experiment 6: Lane Deviation.

Modality	Importance	Nature	Modality:	Modality:	Importance:	Importance:
			Importance	Nature	Nature	Modality:
						Nature
<i>F</i> (<i>3</i> , <i>134</i>)	<i>F</i> (<i>3</i> , <i>134</i>)	F(3,134)	<i>F</i> (<i>3</i> , <i>134</i>)			
= 0.87	= 1.57	= 2.14	= 0.86	= 1.34	= 0.05	= 1.07
<i>p</i> = 0.46	p = 0.20	<i>p</i> = 0.10	p = 0.46	<i>p</i> = 0.26	p = 0.98	<i>p</i> = 0.36

Table 5.16: Experiment 6: Statistics for Lane Deviation; significant results are bold and marked with an asterisk.

Subjective Data

All subjective ratings can be seen in Figure 5.18. Participants' rating of how much they liked the two modalities was compared with Wilcoxon and showed significant differences (V = 0, p < 0.001). Bimodal feedback (*median* = 5) was liked more than thermal feedback (*median* = 3).



5.3. Experiment 6: Uni- and Bimodal Tactile Notifications with Different Types of Information

Figure 5.18: Experiment 6: All Subjective Ratings.

The ratings of how easily the direction of temperature change (Nature), the lengths of temperature change (T-Importance) and vibration patterns (V-Importance) were evaluated with Friedman's ANOVA, showing significant differences ($\chi^2(2) = 27.74$, p < 0.001). Post hoc tests with Tukey corrections showed significant differences for Nature - T-Importance (V = 120, p = 0.001) and T-Importance - V-Importance (V = 0, p < 0.001), but not Nature - V-Importance (V = 8, p = 0.16). T-Importance (*median* = 3) was differentiated with more difficulty than both Nature (*median* = 4) and V-Importance (*median* = 5).

The ranking of the stimuli at the end of the experiment can be seen in Figure 5.19, with numbers showing how often each stimulus was sorted into at a specific rank, with *Rank 1* being the highest rank and, therefore, the stimulus deemed best to identify by participants. The bimodal stimuli were more often ranked into the top positions, with the high importance vibration pattern preferred, while thermal feedback was more often ranked towards the bottom. Also, notifications for personal messages were ranked higher than work messages, showing a preference for warm stimuli.

P07 summarised this in their comment Longer vibrations easier to feel and heat easier to feel than cold, as did P08: Found the bimodal stimuli much clearer - less likely to think I had received a message when I hadn't. This highlights the uncertainty around thermal feedback that was already addressed in previous experiments, picked up as well by P06: Although I could differentiate between all of the bimodal ones, sometimes I wondered whether the thermal ones were just my hands being over-sensitive / getting used to the temperature. P11 commented thermal feedback only can be hard to distinguish, especially the low importance ones as it can be confused for normal hand temperature fluctuation. Bimodal made this eas-



Figure 5.19: Experiment 6: Ranking of Stimuli.

5.3. Experiment 6: Uni- and Bimodal Tactile Notifications with Different Types of Information

ier as you can expect a change in temperature. P05 wrote I found it much easier to identify the bimodal prompts. I struggled to differentiate between the lengths of time of the thermal prompts

The combination of vibration and temperature changes was commented on favourably overall: The vibration was more clear than thermal and also feeling it let me know I should now expect thermal which took away the worry of missing the thermal (P01); the bimodal gave a clear indication that there was a notification, and gave a cue to keep an eye for the thermal changes to see what type of notification it was (P02); Bimodal was easier to identify because it raises your attention with the vibrations and then informs you whether the message was private or work-related. Thermal only could not be identified at times (P04).

Participants noted on the time it took to identify the stimuli. While *P12* wrote that *the vibrations were always obvious and you could tell their meaning instantaneously, unlike the thermal ones where you had to wait till they had ended to be sure and also sometimes it could be hard to detect the low importance thermal ones, <i>P16* shared that they felt that it was easier to quickly identify the form of stimulus from just the thermal ones, since the timing and heat sensation were provided together. Although it was sometimes difficult to quickly determine the length of the thermal stimuli, which is why having the vibration made it easier. *P11* mentioned that the temp change in bimodal should start at the same time as the vibrations so that it doesn't take so long to get info about the notification.

Another observation made in the previous experiments was affirmed in the comments as well: preferences for warm and cold stimuli differ throughout the participants. *P02* noted that they *had a bit more trouble identifying the cold change than the hot one*, requiring more focus. *P08* commented that they *found the hot sensation clearer than the cold*. *P01*, on the other hand, noted that they *found cold easier to feel than hot* and *P16* concurs that *the cold thermal feeling was easier to identify than the warm one*. *P03* mentioned that they *could identify the work and home stimuli equally as well*.

There were a few comments on the vibration patterns. *P11* noted: *The vibration pattern for low importance seems more urgent than high importance to me. The high importance one is just like a normal phone notification. P17* wrote about the low importance vibration pattern that *the style of vibration can be confusing, like it still feels important (like an alarm).* They suggested that the pattern should be changed.

5.3.7 Experiment 6 Discussion

In this experiment thermal and vibrotactile cues were combined to present two types of information on the steering wheel. The cues were chosen from previous experiments and combined to present either unimodal thermal or bimodal tactile notifications about the Nature and Importance of a message. The evaluation showed that all factors apart from Modality

5.3. Experiment 6: Uni- and Bimodal Tactile Notifications with Different Types of Information 129

had an influence on the recognition, including interactions between the different factors. A closer look showed that ThermalLowPersonal stimuli, the short warm thermal cues, were recognised significantly worse than any other stimulus. This finding was mirrored in the interactions with the aggregated factors. The recognition rate of this cue was with 60% the lowest by far, with almost all other cues having recognition rates between 86% and 89%, with ThermalHighWork, the long cold thermal cue, being the only one above this with 92%. Bimodal cues had slightly smaller recognition rates than thermal cues. Factor Nature had very high recognition, higher than Importance for both thermal and vibration patterns. Direction of temperature was especially well recognised after a vibration in the bimodal feedback. In Experiment 2, ThermalLowPersonal, there labelled w-0-6-s, had a recognition rate of 94%. This substantial drop in recognition was surprising. The cues in Experiment 2 were presented on one Peltier device of the same size, with a finger tip touching the device. In this experiment, the temperature was presented on both hands and on the thenar region, which is usually more susceptible to tactile feedback. If anything, this setup should have led to an increase in recognition. The driving task in this experiment was slightly more challenging than in Experiment 2, as the lane change prompts in this study were presented visually and not haptically. This could have led to an increase in cognitive load of the driving task and influenced the recognition of the presented cues. In addition, the presentation on the hands gripping the wheel might have increased the feeling of warmness emanating naturally from the hands instead of the devices.

Hypothesis 1 claimed that the recognition for bimodal stimuli would be higher, which could not be corroborated. Participants ranked the stimuli according to their ease of identification and consistently ranked bimodal cues higher than thermal cues, only one participant ranked ThermalHighWork as the cue that could be identified the best, yet the recognition rate would suggest that this stimulus was most easily distinguished. In the ranking, this cue, however, was most often rated on Rank 6, after all bimodal cues and ThermalHighPersonal. The stimulus with the lowest recognition rate, ThermalLowPersonal, was also not found most often in the last spot, as would be expected, but on Rank 7, before ThermalLowWork. Some participants commented, as in previous experiments, how they were not sure sometimes if the warming of the steering wheel was a cue or due to the hands getting warm, yet the short warm stimulus was still ranked higher than the cold one. Preferences of direction of temperature change differed between participants.

Hypothesis 2 could be corroborated: participants very clearly rated that they preferred the bimodal to the unimodal thermal condition. Comments suggested that participants appreciated the certainty presented by the vibration and could use it to concentrate on the following temperature change. Some participants mentioned that the vibration patterns did not convey the intended level of importance to them. This issue could be overcome in practical applications by allowing to personalise vibration patterns. There was, however, no confusion mentioned

5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars

on the mapping of direction of temperature change for Nature of message: warm for personal and cold for work messages. This could mean that this association was familiar enough to be easily remembered, while the vibration patterns were more generic. Utilising inherent associations of temperature changes could increase recognition of cues immensely.

Experiment 5 and 6 investigated unimodal thermal and bimodal thermal and vibrotactile feedback for notifications. In the next experiment, bimodal haptic feedback will be investigated in combination with the non-tactile modalities audio and visual, and in a more applied context.

5.4 Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars

In this experiment bimodal thermal and vibrotactile feedback was added to a pre-tested set of multimodal cues for transfer of control. The tactile feedback was used to inform on the status of control transfer, rather than inform on obstacles or danger on the road. The autonomous feature was a level 3 system, Traffic Jam Pilot, much like the traffic jam chauffeur mentioned in 2.3.2. It took over full control of the car in a traffic jam situation on the motorway, when the speed was slower than 40mph. Drivers were required to be available to take over control when needed. In this experiment only handover and driver-initiated takeover were investigated.

5.4.1 Methodology

The study was designed as 2x2 within-subjects experiment, with Haptic (on/off) and Visual (on/off) as the independent variables. Not all combinations were tested, the version without haptic and visual feedback was skipped. Haptic feedback was bimodal with thermal and vibrotactile feedback, while visual feedback described a visual progress bar behind the steering wheel on the instrument cluster, which was either visible or not. The observed conditions were:

- Visual progress bar, without haptics (NoHapVis)
- Visual progress bar, with haptics (HapVis)
- No visual progress bar, with haptics (HapNoVis)

The transfer of control was triggered by the simultaneous pulling of the gear paddles on the steering wheel. Pilot tests showed that participants took a long time to take their hands of

5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars

the steering wheel and their foot off the accelerator pedal after the handover of control to the car. The use of haptic feedback was expected to shorten this time as vibrotactile cues would remind the driver that their manual control was no longer needed and reaffirm the successful control transfer, potentially leading to safer driving during the transfer and increased trust in and preference of the system. Thermal feedback informed on the progress of the control transfer, much like the visual progress bar. It was of interest to see if the tactile presentation would render the visual feedback unnecessary, which could lead to less eyes-off-the-road time and as a consequence to safer driving.

An additional condition **Buttons** was added, to compare if the use of the gear paddles would have benefits over triggering the control transfer by simultaneously pressing two buttons on the steering wheel, which was used in pilot tests. The feedback for this condition was similar to the condition NoHapVis and was only compared directly to this condition. In pilot tests utilising the button press to transfer the control, participants were taking their eyes off the road for long glances in order to find the correct buttons on the steering wheel. It was hypothesised that the use of the gear paddles would reduce this. The hypotheses for this experiment, therefore, were:

Hypothesis 1: Haptic feedback will lead to faster disengagement with the steering wheel and accelerator pedal after handover;

Hypothesis 2: *Haptic feedback will positively influence driving behaviour and increase trust in the system;*

Hypothesis 3: Participants will prefer the conditions with additional haptic feedback;

Hypothesis 4: The lack of visual feedback will lead to less eyes-off-the-road time;

Hypothesis 5: Transferring control with buttons will lead to longer eyes-off-road time and more long glances away from the road.

The dependent variables used to investigate these questions were: time to disengage with the steering wheel and accelerator pedal, lane deviation and number of micromotions of the steering wheel, eyes-off-the-road time and number of long glances away from the road, subjective ratings after each condition and a ranking at the end of the experiment. More detailed information on the dependent variables can be found in Section 5.4.5 *Measures*.

5.4.2 Apparatus

The hardware used for the tactile feedback were the 2x2cm Peltier devices and the Haptuators Mark II vibrotactile actuators used in previous experiments. One vibrotactile actuator was taped onto the heat-sink of each Peltiers. The top side of the heat-sinks were insulated with cork, so fingers would only feel the temperature of the thermal device. The device was



(a) Hardware

(b) Setup

Figure 5.20: Experiment 7: Hardware and experimental setup.

taped onto a copper sheet, which was in turn attached to the gear paddle with Velcro tape, see Figure 5.20(a). An additional vibrotactile actuator was placed under the accelerator pedal.

The study was conducted in a high-fidelity driving simulator setup, see Figure 5.20(b), at the Jaguar Land Rover Driving Lab at the University of Warwick. The high-fidelity simulator environment was made available by Jaguar Land Rover for the experiment as part of an internship, and full details of the system cannot be provided.

The simulator environment was implemented using STI Sim³ software, with the driving scenario being presented on a large screen in front of the driving cabin. The driving environment can be seen in Figure 5.21. Additional small screens were integrated into the setup as instrument cluster, centre console and side windows. Three webcams were filming the driver, each focusing on a different region of interest. One camera was pointed at the face of the driver to capture gaze data, one filmed the feet and their pedal use and one pointed at the hands to capture their interaction with the steering wheel. Video logs captured the input of those cameras as well as the output of the instrument cluster, centre console and side windows.

³https://stisimdrive.com/, accessed 01/06/2020



Figure 5.21: Experiment 7: Driving simulator environment as captured in the video log. The side mirror content was presented to drivers on screens at the side of the driving cabin.

5.4.3 Driving Task and Feedback

The driving task in this experiment was the same for all conditions. The drive started with the car being placed on the left shoulder of a three lane motorway. Participants were asked to move the car to, and stay within, the middle lane, driving with a constant speed of around 40mph. After approximately 1.5min the traffic intensified, forcing the driver to slow down. When the speed of the car moved under 40mph, the autonomous feature became available. The feature was activated by pulling both gear paddles (or two buttons on the steering wheel in condition Buttons) for 2s. When the control was transferred to the car, 2min video snippets of movies (Up^4 , $Skyfall^5$, The Hunger Games⁶ and Frozen⁷) were automatically played on the centre console without prompting. The order of the videos was counterbalanced over the conditions. When the videos stopped, the experimenter asked the participants to take back control from the car the same way they handed it over. They then drove manually for approximately 2min, after which they were asked to move the car back to the left shoulder of the motorway and stop the car.

Visual and audio feedback was provided in all conditions. When the autonomous feature

⁴Up (2009), https://www.imdb.com/title/tt1049413/

⁵Skyfall (2012), https://www.imdb.com/title/tt1074638/

⁶The Hunger Games (2012), https://www.imdb.com/title/tt1392170/

⁷Frozen (2013), https://www.imdb.com/title/tt2294629/

5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars



Figure 5.22: Experiment 7: Thermal feedback: Handover (left) and Takeover (right).

became available, a (proprietary) visual icon appeared on the instrument cluster, accompanied by a non-vocal sound notification (ding). When the driver started the control transfer by pulling the gear paddles (or pressing the buttons), the progress was presented by showing a visual progress bar in the instrument cluster behind the steering wheel, in the Visual-On conditions (HapVis and NoHapVis/Buttons). The haptic feedback presented in the Haptic-On conditions (HapVis and HapNoVis) consisted of thermal feedback and vibrotactile feedback. The thermal devices were set to 32°C during manual driving and changed during the handover to 26°C with 3°C/s, see Figure 5.22 left. When the control was transferred, the vibrotactile actuators on the gear paddles and the accelerator pedal vibrated for 0.5s. The process during takeover was similar, only the temperature changed from 26°C back to 32°C, see Figure 5.22 right. The warmer temperature was chosen for manual driving, as previous research has linked warm temperatures to activity and presence [113], which corresponds with manual rather than automated driving. The completion of the transfer process was accompanied by speech feedback in all conditions, uttering Traffic Jam Pilot activated after handover and Traffic Jam Pilot deactivated after takeover. In addition, the instrument cluster changed its appearance from presenting the classical round speedometer design to a (proprietary) presentation of the car in its surroundings⁸.

The order of the conditions was counterbalanced over the participant pool and the order of movie snippets was counterbalanced over conditions.

⁸Comparable to the Tesla design, https://teslatap.com/wp-content/uploads/2015/10/ TeslaTap_instrument_v7_p85d_720.jpg, accessed 14/08/2020

5.4.4 Participants and Procedure

Fourteen participants, all employees of Jaguar Land Rover with valid UK driving licenses, completed the study. Their job descriptions ranged from market researchers, to patent attorneys to training developers and human factors specialists. All but one were right-handed, the last one named both right and left as dominant hands, see Table 5.17 for more detailed demographic data.

Number of	Age	Driving	Experience	Experience	Experience	Experience
Participants	(Years)	Experience	Autonomous	Simulator	Vibration	Thermal
		(Years)				
14	<i>R</i> : 24-58	<i>R</i> : 3-41	<i>R</i> : 1-5	<i>R</i> : 1-5	<i>R</i> : 1-5	<i>R</i> : 1-2
(Female: 3	<i>M</i> = 38.21	<i>M</i> = 18.43	M = 3.14	M = 2.50	M = 3.36	<i>M</i> = 1.43
Male 11)	SD = 12.98	SD = 12.33	SD = 1.46	SD = 1.51	SD = 1.36	SD = 0.51

Table 5.17: Experiment 7: Participant data, showing range (R), mean (M) and standard deviation (SD).

The participants were recruited via internal organisation mails. The study was conducted at Jaguar Land Rover labs located at the University of Warwick in Coventry. Participants were welcomed and presented with an information sheet, before they were asked to sign a consent form. They were then guided to the high fidelity driving simulator and seated. Participants were given a chance to get used to the simulator with a 2min drive without any feedback. This was followed by an introduction of the first feedback and the corresponding driving part, after which participants were asked to fill in a questionnaire. This was repeated for the remaining feedback types. At the end of the experiment, participants filled in an additional questionnaire, capturing demographic data and subjective ratings. The experiment took approximately one hour, participants were not compensated for their time.

For all documents, please refer to Appendix A.8. This experiment was approved both by the College Ethics Committee of the University of Glasgow as well as the internal Jaguar Land Rover Ethics procedures.

5.4.5 Measures

The evaluation of the video data provided some of the dependent variables. The videos were manually labelled and the time data extracted to discern how long participants took to let go off the steering wheel (HandsOff) and the accelerator pedal (FootOff) and to return the foot after takeover (FootOn). Additionally, the gaze behaviour was extracted from the videos, labelling the time frames participants averted their gaze from the forward road from the moment the autonomous feature became available until the control had been fully handed

5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars

over to the car (CompleteHandover) and from the start of the takeover process until the driver was driving fully manual (CompleteTakeover). There were several areas of interest that the gaze would turn to: down to the steering wheel, to the instrument cluster behind the steering wheel just under the front window, upwards towards the rear mirror or left/right to the two mirrors on the side. Overall eyes-off-the-road (Eyes-Off-Road) time was calculated as the sum of these. Any glances longer than 2s during the handover process were counted as well, as these long glances have been shown to be major contributor to crashes and should be avoided. Long glances and eyes-off-the-road time during the takeover process were ignored, as the car was still driving autonomously and glances away from the road, therefore, posed no danger.

The lane deviation was extracted from the driving log provided by the driving simulator and was calculated, as before, using the RSME to summarise the driving behaviour again for the time the autonomous feature became available until the control was fully transferred. Driving behaviour was compared both between conditions, evaluating the influence of the independent variables, and within conditions, comparing the baseline driving with the driving during the handover process. The baseline of the same length as the handover process was captured just before the feature was available, to allow for a close comparison. While the data for the user initiated takeover process was captured as well, it could not be compared to the baseline, as the road conditions had changed noticeably during the autonomous drive. A baseline right before the takeover would have only captured the cars' driving behaviour and a baseline right after takeover might have unduly been influenced by the driver settling back into driving and adjusting the cars trajectory to the personally preferred position within the lane. Driving behaviour, therefore, was only investigated for the handover process.

The driving simulator log captured changes in driving rather than capturing a screenshot of the data at specific time intervals. This provided additional information on the driving behaviour: the number of micromotions used to achieve the presented level of lane deviation. Every new move of the steering wheel prompted a new entry into the log, therefore, many such movements could indicate fidgeting which in turn indicates anxiety, uncertainty or distraction from the driving. A low number of micromotions were interpreted as positive.

After the driving task of each condition, participants filled in questionnaires capturing perceived workload, with the NASA TLX questionnaire, as well as the four additional ratings from previous experiments (pleasantness, comfort, complexity and disruptiveness) on a 10point Likert scale. Additionally, an open-text comment section was available.

Another questionnaire was presented at the end of the experiment, capturing a ranking of all four conditions, as well as a rating of how well the thermal feedback could be felt with and without the visual progress bar (on a 10-point Likert scale). Additionally, participants indicated whether the condition influenced their trust in the system and were then asked to

rate their trust for each condition on a 10-point Likert scale. A free text comment section was also provided.

5.4.6 Results

The evaluation of the data was done with repeated measures ANOVA for gaze data, driving behaviour between conditions and subjective ratings. *Post hoc* tests were done with Tukey corrections, automatically calculated in R to adjust for multiple comparisons. Outliers of more than two standard deviations have been excluded for these statistical evaluations. Condition *Buttons* will be included in the figures, but will only be evaluated in section 5.4.6. The comparisons of driving within conditions and between Buttons and NoHapVis were done with t-tests (t(df)), when the data were normally distributed, or Wilcoxon tests (V) otherwise.

Time Observations

The timings extracted for the video logs can be seen in Figure 5.23. For the gaze evaluation there were outliers for Eyes-Off-Road (1 HapNoVis, 1 NoHapVis), CompleteHandover (1 NoHapVis), HandsOff (1 HapNoVis), FootOff (1 NoHapVis), CompleteTakeover (2 HapVis) and FootOn (1 NoHapVis, 1 HapNoVis), were the values differed more than two standard deviations from the mean. The results of the statistical evaluation with repeated measures ANOVA, see Table 5.18, showed significant differences for FootOff for Visual, where participants took longer for Visual-Off (M = 1.13, SD = 0.50) than Visual-On (M = 1.01, SD = 0.70). Additionally, FootOn showed significant differences for Haptic, where Haptic-Off was faster (M = 0.79, SD = 0.39) than Haptic-On (M = 1.22, SD = 0.71). The HandsOff outlier removed for condition HapNoVis was very much higher (141.21s) than the mean for this condition (11.21s), which can be seen in Figure 5.23.

	Eyes-Off-	Complete	Hands Off	Foot Off	Complete	Foot On
	Road	Handover			Takeover	
Haptic	<i>F</i> (1,11)	<i>F</i> (1,12)	<i>F</i> (1,12)	<i>F</i> (1,12)	<i>F</i> (1,12)	<i>F</i> (1,11)
	= 0.90	= 1.24	= 0.57	= 4,49	= 0.40	= 6.66
	<i>p</i> = 0.36	<i>p</i> = 0.29	p = 0.47	<i>p</i> = 0.06	p = 0.54	<i>p</i> = 0.03*
Visual	<i>F</i> (1,12)	F(1,13)	F(1,12)	F(1,13)	<i>F</i> (1,11)	<i>F</i> (1,12)
	= 2.06	= 0.00	= 0.11	= 6.51	= 1.53	= 2.79
	<i>p</i> = 0.18	<i>p</i> = 0.96	<i>p</i> = 0.75	<i>p</i> = 0.02*	<i>p</i> = 0.24	p = 0.12

Table 5.18: Experiment 7: Statistics for Time Observations; significant results are bold and marked with an asterisk.



5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars

Figure 5.23: Experiment 7: Time Observations (including Buttons).

3 long glances during handover occurred (1 HapNoVis, 2 NoHapVis), usually due to interruptions in the process (when, for example, the participants stopped pulling the paddles before the handover was complete).

Driving Behaviour

Driving had outliers in number of micromotions (2 HapNoVis), where the values were excluded for evaluation as they differed more than two standard deviations from the mean. The evaluation with repeated measures ANOVA of the lane deviation during handover between conditions showed significant differences for Haptic (F(1,13) = 5.88, p = 0.03), where Haptic-On had less deviation (M = 21.58, SD = 0.74) than Haptic-Off (M = 21.99, SD = 0.62). No differences with repeated measures ANOVA were found for Visual (F(1,13) = 1.02, p = 0.33). The number of micromotions evaluated with repeated measures ANOVA showed no significant differences (Haptic: F(1,12) = 0.15, p = 0.70; Visual: F(1,11) = 0.09, p = 0.77).

Driving within the conditions, evaluated with t-tests (t(df)) or Wilcoxon tests (V), for not normally distributed data, showed no significant results for the number of micromotions of HapVis (V = 29, p = 0.12), but lane deviation (t(13) = 2.58, p = 0.02), where the handover had less deviation (M = 21.48, SD = 0.72) than the baseline (M = 21.94, SD = 0.87). Neither NoHapVis (lane deviation: t(13) = 1.77, p = 0.02; number of micromotions: V = 12, p = 0.23) nor HapNoVis (lane deviation: t(13) = 1.54, p = 0.15; number of micromotions: V = 12



5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars

Figure 5.24: Experiment 7: Perceived workload; significant results are in bold font.

20.5, p = 0.30) showed significant differences for driving.

Subjective Data

The evaluation of overall perceived workload showed no statistical difference with repeated measures ANOVA for Haptic (F(1,13) = 0.56, p = 0.47) or Visual (F(1,13) = 1.96, p = 0.19). The ratings of the single categories can be seen in Figure 5.24. The evaluation of the additional ratings with repeated measures ANOVA showed no significant differences for complicated (Haptic: F(1,13) = 0.09, p = 0.77; Visual: F(1,13) = 0.00, p = 1.00), pleasant (Haptic: F(1,13) = 0.41, p = 0.53; Visual: F(1,13) = 1.59, p = 0.23) and comfortable (Haptic: F(1,13) = 0.89, p = 0.36; Visual: F(1,13) = 0.80, p = 0.39). For disruptiveness there was no difference found for Haptic (F(1,13) = 2.07, p = 0.17), but for Visual (F(1,13) = 8.97, p = 0.01), where Visual-Off was rated as more disruptive (*median* = 3.5) than Visual-On (*median* = 2). When asked if the condition influenced the trust in the system, 6 participants chose *no*, 5 *yes* and 3 *maybe*. The evaluation with repeated measures ANOVA of the trust ratings showed no significant differences (Haptic: F(1,13) = 0.15, p = 0.71; Visual: F(1,13) = 1.89, p = 0.19). Additional subjective and trust ratings can be seen in Figure 5.25.

The rating of how well participants felt the thermal feedback with and without the visual bar with Wilcoxon test showed no significant differences (V = 10, p = 0.57, both *median* = 2). The ranking of the conditions, see Figure 5.26, included Buttons, which was ranked least favourite (Rank 4) most often. HapNoVis was the least preferred condition using the gear paddles. NoHapVis was rated most often as most favourite, closely followed by HapVis.

10 9 Likert Scale (1-10) 8 7 6 54 $\frac{3}{2}$ 1 0 Comfortable Complicated Disruptive Trust Pleasant HapVis Buttons

5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars

Figure 5.25: Experiment 7: Additional Subjective and Trust Ratings; significant results are in bold font.

HapVis was most often ranked into the top two positions.

The addition of haptic feedback was received diversely. *P01* disliked the vibration and commented that *the haptic feedback felt like the system was faulty, that [...] it was buzzing, and when the steering wheel moved, it just felt [like] the system was broken. The buttons and the voice over was much cleaner* and *P13* noted that *the haptic feedback [on] the steering wheel does not help me trust the system more and it's just annoying and irritating (unnecessary).* However, *P08* wrote that *phone and various controllers use vibrations as a form of feedback, so this is the form which felt most natural. P05* mentioned that *all options felt very similar and none were difficult or complicated. I didn't notice the thermal feedback.* Additionally, participants commented that *the system itself is generally what would need to build/hold my trust... Not necessarily the handover method. Once this is set and understood - then I would trust it (P09)* and *the longer the experiments went on the more comfortable I became with the systems. Towards the end I was aware that I was taking my eyes off the road and traffic conditions for long periods of time compared to the first test condition (P07).*

The lack of visual progress bar was noted as *fine as I was using the vibration rather then progress bar to know when the system was activated (P08). P09* commented: *On reflection, maybe nicer to have the progress bar, but not essential at all. The other confirmations (vibrations and audio) felt like enough. But on the other hand, progress bar is more akin to regular steering wheel functions that I'm used to. P06* noted that *the visual cue made me feel more confident at the system in that I can see the progressing of the operation via my input. P12 felt that without the visual timer bar it took longer, although just a mind trick, the timer*

5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars



Figure 5.26: Experiment 7: Ranking of Conditions.

bar I feel, helps, and P13 was looking for feedback on cluster and it wasn't there - I trust more the visual feedback with the progress bar than waiting for other form of feedback. P08 wrote that thermal had little impact as this was insignificant over such a short period, but vibration was good as this allowed me to focus on the road rather then watching [the] progress bar and P03 found having the vibration was more involving. P12 noted that they had slightly more confidence in this due to the vibration feedback, [but] didn't really notice the temperature change. Other participants shared this view on thermal feedback: Heat sensor... Not really needed? (P09); The warm to cool and back controls seem pointless? (P10); didn't really feel hot/cold that much, doubt its a very good indicate of an affirmative action (P01) and couldn't personally feel the change in temperature on the handles (P11).

Comments on Buttons described the condition ambiguously. On the one hand Buttons were more fiddly (P07) and not great - too many buttons in close proximity, I had to think which one to choose and there was a high chance I'd get it wrong (P13). P10 noted that while I felt buttons would be easier to use than the paddles in fact I found the buttons harder to locate the correct spot. But on the other hand, they were most natural (P05) and slightly more effort than simply tapping a paddle, but... I kind of preferred it! Paddle seems to me something I

could tap by accident, this felt more definitive as an action and almost impossible to do by accident. Would have preferred vibration confirmation though (P09). Still, most participants prefer using the paddles (P04),

Comparison of Buttons and Gear Paddles

The comparison of time observations between using buttons and gear paddles to initiate the control transfer with t-tests (t(df)) or Wilcoxon tests (V), for not normally distributed data, showed a significant difference for eyes-off-the-road time, see Table 5.19. A more detailed look showed that the time to look down was significantly different between Buttons and NoHapVis, where the time to look down was longer in Buttons (M = 1.03, SD = 0.59) than NoHapVis (M = 0.13, SD = 0.24), resulting in longer overall eyes-off-the-road-time (Buttons: M = 3.16, SD = 1.60; NoHapVis: M = 2.16, SD = 1.24).

Eyes-Off-	Down	Complete	Hands Off	Foot Off	Complete	Foot On
Road		Handover			Takeover	
<i>V</i> = 87	<i>V</i> = 78	V = 71	<i>V</i> = 52	<i>V</i> = 43	<i>V</i> = 22.5	<i>V</i> = 53
<i>p</i> = 0.03*	<i>p</i> = 0.003*	<i>p</i> = 0.26	<i>p</i> = 1.00	<i>p</i> = 0.58	<i>p</i> = 0.65	<i>p</i> = 1.00

Table 5.19: Experiment 7: Statistics for the Comparison of Time Observations between Buttons and Gear Paddles; significant results are bold and marked with an asterisk.

The comparison of the driving evaluated with t-tests (t(df)) or Wilcoxon tests (V), for not normally distributed data, showed no significant differences for lane deviation (t(13) = 0.12, p = 0.90) and number of micromotions (V = 33, p = 0.23) and there were no significant differences found for subjective data between Buttons and NoHapVis, see Table 5.20.

Overall Workload	Complicated	Pleasant	Disruptive	Comfortable	Trust
<i>V</i> = 71.5	<i>V</i> = 24	<i>V</i> = 33	V = 45.5	<i>V</i> = 29	<i>V</i> = 27
p = 0.07	p = 0.44	<i>p</i> = 1.00	p = 0.27	p = 0.44	<i>p</i> = 1.00

Table 5.20: Experiment 7: Statistics for the Comparison of Subjective Data between Buttons and Gear Paddles; significant results are bold and marked with an asterisk.

5.4.7 Experiment 7 Discussion

This experiment applied thermal and vibrotactile feedback to inform on the process of control transfer for autonomous features in a vehicle. It was expected that the localised haptic feedback would help drivers to let go of the steering wheel and accelerator pedal faster after completed handover. But these reactions were not influenced by the presence of haptic feedback, so *Hypothesis 1* could not be corroborated. Furthermore, it took participants longer

5.4. Experiment 7: Multimodal Feedback for Transfer of Control in Semi-Autonomous Cars

to return their foot to the accelerator pedal after takeover in the haptic condition. Haptic feedback did not improve reaction times. However, changes in the visual feedback had an effect on the timings: drivers took longer to take their foot off the accelerator pedal without the visual progress bar present. The thermal feedback did not seem to be enough to replace the information presented by the progress bar. Some participants mentioned in the open text comments that they could not feel the thermal feedback or considered it unnecessary and when rating how well participants could feel the temperature changes, the rating was on the lower end of the scale. Even though the temperature change was at the fastest rate and highest extent we observed so far, it seems to still have been too subtle within the short time frame of 2s for some participants.

Hypothesis 2 claimed that haptic feedback would increase trust in the system and positively influence driving behaviour. Participants commented that the trust in the system was not influenced by the condition, but rather their familiarity with the system itself. When asked directly, some remarked that the control transfer feedback might have influenced their trust, but the evaluation of the trust rating between conditions did not corroborate this. The driving behaviour was influenced by the presence of haptic feedback: less lane deviation was found with haptic feedback than without. With both haptic and visual feedback present, the lane deviation was improved during handover compared to baseline driving. The number of micromotions was not found to be different, even when lane deviation was improved. This would suggest that participants steered more steadily with both haptic and visual feedback present, increasing driving behaviour. *Hypothesis 2* could, therefore, be partly corroborated.

Participants reacted ambiguously towards haptic feedback. Some participants preferred it, other deemed it too much or unnecessary, especially the thermal feedback. Vibration was more often considered a reassuring addition, thermal feedback was almost exclusively remarked on negatively. The ranking at the end of the experiment almost tied the two conditions HapVis and NoHapVis for the first two spots, suggesting that the visual feedback was the more important factor, not haptic. *Hypothesis 3* could, therefore, not be corroborated. This aspect might, however, be influenced by the participant pool: as JLR employees, the participants were potentially biased towards new technologies, as they would not experience them naively as a novel interaction, but connected to consequences or experiences attached to their specific roles within the corporation. This limitation might hold true for several observations of the experiment, but especially for preference ratings. The different conditions should be tested with more naive users in the future, to gain more versatile and diverse feedback.

Hypothesis 4, expecting that the lack of visual feedback would lead to less eyes-off-theroad time, was not corroborated. Neither haptic nor visual feedback influenced the overall eyes-off-the-road time. Long glances away from the road occurred both with and without visual feedback and were brought on by difficulties during the control transfer. Comments and rankings indicate that participants felt more confident with visual feedback present, even when they did feel comfortable with relying on vibration. Furthermore, the lack of visual feedback was rated as significantly more disruptive, highlighting again that thermal feedback alone seemingly did not convey enough information to prepare participants for the upcoming vibration.

The comparison between buttons and gear paddles as transfer mechanisms showed differences in overall eyes-off-the-road time by 1s, caused by a significantly longer amount of time spent looking down on the steering wheel when using buttons. There were, however, no long glances away from the road in condition Buttons, while there were two in condition NoHapVis. This corroborates *Hypothesis 5* only partly.

5.5 Chapter 5 Discussion

Three experiments explored thermal feedback in combination with vibrotactile feedback on the steering wheel for notifications. The first two experiments additionally looked into unimodal thermal cues, exploring their use for notifications of different levels of urgencies and for two types of information. Thermal stimuli were consistently rated as less urgent than vibrotactile or bimodal haptic stimuli, suggesting their use for non-urgent notifications. But they also were missed more often. The stimuli design of Experiment 5 was aligned to vibrotactile feedback used in previous research, in which they combined vibration with audio and visual feedback for urgent notifications. The cues, therefore, were short and, especially for low and medium urgency notifications, very subtle, leading to many missed stimuli. While participants rated in the questionnaire at the end that bimodal cues felt more urgent, this sentiment was not mirrored in the rating of the perceived urgency of the cues at the beginning of the experiment. Furthermore, the recognition of bimodal stimuli took longer than vibrotactile stimuli alone, which makes the cues unsuitable for urgent notifications during driving, as fast reaction would generally be needed to avoid danger. In addition, the bimodal representation did not increase recognition rate compared to vibrotactile. The expected benefits of the bimodal condition, therefore, could not be observed. Participants reported a feeling of uncertainty connected to thermal feedback: they had sometimes problems distinguishing especially subtle warm temperature changes from the natural warming of their hands on the steering wheel. Additionally, they reported problems with the differentiation of the length of temperature presentation, which encoded the different levels or urgency. Three levels, in combination with the subtle changes of medium and low urgency cues, led to a high number of missed and falsely identified thermal stimuli.

The second experiment used longer thermal cues, chosen from the most successful combinations from Experiment 2. Three sets of stimuli were chosen, all of which showed high recognition rates and low number of false positives, for both directions of temperature changes. Two different types of information were presented: the Nature of a message, represented by the direction of temperature change (warm/personal, cold/work) in both thermal and bimodal tactile cues, and the Importance of a message, represented by either the combination of length of temperature presentation and the style of return to neutral in the thermal, or vibration pattern in the bimodal haptic condition. The study led to high recognition rates for all stimuli, apart from the thermal low importance personal cue, the short warm temperature change. Again, participants reported that they had problems distinguishing the warm, especially the short warm, cues from the natural warming of the hand. The other three thermal cues were easier recognised, with rates between 88% and 92% even slightly higher than the bimodal cues with 86% to 87%. Even though participants ranked the bimodal stimuli as easiest to identify, the thermal cold high importance cue had the highest recognition rate (92%). This could be due to the fact that some participants mixed up the two vibration patterns. The direction of temperature change, the Nature of the message, seemed to be easy to recognise by participants, mirrored in their rating at the end of the experiment. The mapping of warm to personal and cold to work seemed to benefit from natural associations with temperature that could not be as easily replicated with vibration. Furthermore, the uncertainty of thermal feedback was overcome by starting the cue with vibration in the bimodal condition, alerting participants to an upcoming thermal cue. This combination was reported to work well and showed very high recognition rates, especially for direction of temperature change. With increased vibration patterns, maybe even designed by participants themselves, this could be extremely effective. As thermal preferences differ between individuals, adaptive thermal feedback could increase recognition of thermal cues as well.

In the last experiment, thermal and vibrotactile cues were added to a multimodal design for the transfer of control of an autonomous feature. The thermal feedback represented the progress of the control transfer, while vibrotactile feedback announced the completion of the transfer. The transfer of control was achieved by simultaneously pulling both gear paddles, on which the thermal and vibrotactile feedback was presented, with additional vibration on the accelerator pedal. The successful transfer was completed after 2s, leaving a rather short time period for the thermal cue. While vibration was welcomed by most participants, the thermal cue was not considered necessary or helpful. In one condition, the visual progress bar was completely replaced by the temperature change, which led to an increased feeling of disruptiveness. However, the addition of bimodal tactile feedback to the multimodal cue improved the driving behaviour during the handover of control to the car.

5.6 Conclusions and Research Question 3

In this chapter, three experiments were presented to answer Research Question 3:

How effectively can thermal feedback convey notifications in a car in a a. unimodal setting?

b. bimodal tactile setting?

The effectiveness of thermal unimodal notifications depends highly on the stimuli design and the intended use case. Short, especially warm, temperature changes are often missed or mistaken for the natural warming of the hands of the steering wheel. Even cues that have been successful in previous experiments, where they were presented on the fingertips of one hand, have reached low recognition rates when presented on the thenar of both hands while grasping the steering wheel. The driving task in those two experiments differed as well, which could have had an influence on the workload and in turn could have influenced the recognition of the cues. This highlights that not all factors impacting the recognition of thermal cues have been explored and fully understood yet. While the presentation of three levels of information led to many recognition errors, especially for thermal only cues, two levels were well recognised. Overall, thermal feedback has been consistently rated as less urgent than vibration and bimodal tactile cues. In combination with the longer recognition time compared to other modalities, this recommends thermal as feedback for non-urgent information. High and long temperature changes led to high recognition rates, in some cases even higher than bimodal tactile feedback.

The addition of vibrotactile feedback to thermal cues can be highly beneficial to the recognition of thermal changes: participants commented that they were alerted by the vibration and then focused on the temperature. The uncertainty reported with thermal feedback could be overcome by prompting the cue with vibration. Not only does this eliminate the uncertainty, it also minimises the effect of false positives, as the initial temperature change is highlighted by the vibration and the return to neutral can be more easily distinguished. This led to high recognition rates and preference ratings. In addition, thermal feedback can have associations that can increase the recognition, as shown by the mapping of warm to personal and cold to work, which seemed to be more memorable than vibration patterns. Both factors, however, might benefit from adjustable and personalisable cues, to accommodate preferences and personal associations. Haptic feedback can improve driving behaviour, as shown in the last experiment. However, more detailed studies would have to determine if thermal feedback influenced this effect or if it was mostly due to the vibrotactile cue. The next chapter will summarise the findings and conclude this thesis.

Chapter 6

Conclusions

This thesis investigated the feasibility of thermal feedback use during the demanding task of driving. The thesis statement in *Introduction* read as follows:

Haptic feedback within cars is used to reduce visual distraction during driving, one of the main contributors to crashes. This research explores aspects and applications of thermal feedback for in-car use to convey information to the driver non-visually. Results show that thermal cues can be used to convey information to the driver accurately and non-urgently and its effectiveness can be enhanced through multimodal combination with vibrotactile feedback and offers a rich, new modality to broaden the design space of tactile in-car feedback.

This statement was supported by research discussed in the following chapters. The exploration of thermal feedback during driving started in Chapter 3 with a single device to convey direction, presenting binary information in the form of direction of temperature change (warm/cold) to see how well this minimal information could be perceived and what influence it would have on the driving performance. Chapter 4 introduced the information provided by spatial placement of multiple devices on the steering wheel and explored its influence on directional information. Chapter 5 described three studies on thermal and vibrotactile bimodal cues and their effectiveness for different kinds of notifications. These user experiments were conducted to answer three Research Questions and their results will be summarised and discussed in this chapter. In addition, contributions and limitations of the thesis will be discussed and future work outlined.

6.1 Research Question 1

How accurately can thermal feedback give binary direction cues in an automotive setting (with one thermal device)?

To answer this question, two simulator studies were conducted, as discussed in Chapter 3. One thermal device was presented to participants on the desk in front of them, while they drove one-handed. Experiment 1 investigated fixed warm and cold cues, where warm prompted a lane change towards the right and cold towards the left. Thermal feedback was compared to speech and showed higher workload for factors mental and physical demand, performance and frustration. Thermal cues were recognised with high accuracy during driving, with of up to 97% correct recognition. However, the return of the temperature to neutral after the cues was often misinterpreted as a new cue, a false positive.

Experiment 2 investigated several stimuli designs, testing the influence of several factors on improving recognition and reducing the number of false positives. No solution was found that ensured high recognition rate and low rate of false positives. The design process for binary thermal feedback has to prioritise one aspect. In addition, recognition time for warm stimuli was influenced by how much the temperature was changed and how long it was presented. Recognition time of cold stimuli was not influenced by any design factor.

High recognition rates would suggest that the modality can be effectively used with direction of temperature change providing binary information. However, false positives can have a reducing influence on the accuracy of the feedback. This might not pose a problem for feedback of informational character, but can effectively negate directional cues. In answer to Research Question 1: binary thermal feedback is not suitable to provide accurate directional cues in a driving context.

6.2 Research Question 2

How accurately can thermal feedback give direction cues in an automotive setting with the spatial information from multiple thermal devices?

Experiments 3 and 4 in Chapter 4 used multiple devices on the steering wheel for turn-bytun navigation to answer this question. The navigation in both experiments included ahead warnings 200m before the turn and another cue right at the turn. The first thermal navigation design, used in Experiment 3, was compared to audio feedback. It indicated the direction of upcoming turns by warming the corresponding side of the steering wheel, while cooling the opposite side. The association of warm towards destination was used in previous research and confirmed by participants in the first two studies. Thermal feedback had high recognition rates of over 90%, but was rated higher on workload and lower in additional subjective ratings. Participants commented on the simultaneous presentation of warm and cold stimuli as confusing.

Therefore, Experiment 4 only presented warm stimuli on the turning side of the steering wheel and compared the navigation to cutaneous push feedback, an equally novel cue which

also engages the tactile sense. Recognition for thermal feedback was more divergent, as recognition for ahead warnings reached 98%, while only 87% of turns were correct. In addition, workload for thermal feedback was higher in every rated aspect and participants preferred cutaneous push feedback. In both experiments, participants commented on an overall sense of uncertainty around thermal feedback, not being sure if the warming was natural warming of the hand grasping the wheel or a stimulus. This led to an increased need to pay attention to the cues.

The spatial information presented through the location of multiple thermal devices could be used effectively to give navigation information in both studies. Therefore, to answer to Research Question 2: spatial thermal directional cues, provided by multiple thermal devices, can be accurately used during driving. Further studies should, however, investigate the influence of long term use on familiarisation and different workload conditions to ensure that accuracy levels can be sustained.

6.3 Research Question 3

How effectively can thermal feedback convey notifications in a car in a a. unimodal setting? b. bimodal tactile setting?

This questions was explored with three simulator studies with thermal and vibrotactile feedback for different kinds of notifications, discussed in Chapter 5. Experiment 5 investigated the effectiveness of thermal feedback to increase the feeling of urgency. The thermal cues were designed to fit vibration patterns of three differing urgency levels and were presented at the same time, resulting in some subtle cues. Perceived urgency was rated for uni- and bimodal cues and showed that thermal cues were consistently rated as less urgent than both vibrotactile and bimodal tactile cues, but also missed more often. The addition of thermal feedback did not increase the feeling of urgency compared to vibration in this subjective rating, even though participants commented in the questionnaire that bimodal cues felt more urgent. Recognition was in general better for cues including vibration. Bimodal feedback led to slower reaction times than vibration alone, which makes bimodal feedback unsuitable for urgent notifications that require fast reactions to avoid dangerous situations.

Experiment 6, therefore, explored the use of bimodal tactile and unimodal thermal feedback to convey non-urgent notifications: message information with two levels of information. Direction of temperature change encoded the Nature of a message (warm-personal / cold-work), vibration patterns in the bimodal, or length of cue in the thermal condition, encoded Importance. The ThermalLowPersonal cue (short warm cue) had the single worst recognition (60%) and ThermalHighWork the best (92%), while all other cues had recognition rates

between 86% and 89%. The factor Nature achieved especially high recognition in the bimodal tactile modality.

Experiment 7 added bimodal tactile feedback to a set of multimodal cues for transfer of control. The feedback was introduced to inform drivers on the progress of the control transfer, to affirm when drivers could disengage the steering wheel and pedals. Thermal feedback was used to inform on the progress of the transfer, vibration on its completion. The addition of bimodal feedback positively influenced driving behaviour, but thermal cues could not successfully replace a visual progress bar.

These results show that thermal feedback can be used to convey notifications unimodally and bimodally with vibration, dependent on cue design. Bimodal feedback was generally preferred and had best results when presenting vibration before the temperature change for two levels of information. To answer Research Question 3: thermal feedback can effectively convey notifications in the driving context, both in a uni- and bimodal setting with vibration.

6.4 Contributions and Recommendations

This thesis contributes novel insights into the use of thermal feedback during driving. The main contributions are: (1) first examination of design factors for binary thermal feedback during driving; (2) exploration and effective use of spatial thermal feedback for in-car navigation purposes; (3) investigation of perceived urgency for uni- and bimodal tactile notifications on the steering wheel; (4) evaluation of perceptability of uni- and bimodal tactile cues during driving.

Design recommendations have been extracted from the findings of the thesis and are summarised in Table 6.1. These recommendations are based on observations with specific design factors and ranges used in this thesis and need to be interpreted with this in mind.

Design Recommendation	Chapter
 (1) Thermal cues took 1.82s longer to be recognised than audio cues. Thermal feedback should be used for non-time-critical cues. (2) Binary cues of high extent (6°C) and length (3s and 6s) of warm temperature changes led to high recognition, but also to a high number of false positive recognition at the return to neutral and long recognition times. Cue design has to prioritise between high recognition and low number of false positives. 	3
 (3) Direction information with thermal spatial information achieved high recognition rates of 87% to 97%. Thermal spatial information is suitable for navigation. (4) Participants reported confusion with warm and cold temperatures presented an opposing hands at the same time. Thermal feedback should not be presented on both hands with opposing direction of temperature change for navigation. 	4
 (5) Thermal cues, in the range tested, were rated as less urgent than vibro-tactile cues. They should be used for non-urgent feedback. (6) Warm temperature changes of short duration were often missed when presented during vibration. Short thermal cues, especially warm temperature changes, should not be presented during vibration. (7) Bimodal thermal and vibrotactile cues take longer to recognise than vibration alone. Urgent vibrotactile warnings should not be combined with thermal feedback. (8) Vibration patterns followed by temperature can be used effectively and reduced the feeling of uncertainty sometimes experienced with thermal unimodal cues. Bimodal tactile cues should present the vibration before the temperature changes on the steering wheel had poor recognition. They should not be presented on the steering wheel during driving. (10) Long cold thermal changes (6°C presented for 3s with angled return to neutral) were recognised with the highest accuracy (even when compared to bimodal cues). These cues should be used for important thermal only feedback. (11) Short temperature change during transfer of control could not compensate the lack of a visual progress bar. Short temperature changes should not be used to replace visual information of processes. 	5

Table 6.1: Design recommendations overview collected from the experimental findings of this thesis.

6.5 Limitations and Future Work

This section will discuss limitations of the thesis and propose ideas for future work.

6.5.1 Simulator Studies

Thermal feedback was only investigated in simulator studies, a necessity to ensure that the presentation of this feedback, novel within the car, would not lead to hazardous situations and endanger participants and experimenters. However, the effects of real world driving on the perception and usability of thermal feedback is unknown. No significant differences in lane deviation have been found for any studies but the last, which was the only one including visual feedback, therefore, no definite conclusion can be drawn on the influence of thermal cues on driving behaviour. However, thermal feedback increased workload, which has been shown to have negative effects on driving [162, 169]. More studies are needed to evaluate the impact on thermal feedback on driving, before the feedback can be safely tested in real world driving.

Most studies presented in this thesis were conducted in a low fidelity simulator, which could have impacted immersion into the driving task. High fidelity simulators could provide a more realistic driving experience, which could ultimately have an influence on the perception of the thermal modality.

6.5.2 Simple Driving Tasks

Most driving tasks in this thesis were limited to lane change scenarios and turn-by-turn navigation within empty streets. More complex tasks should be evaluated in future experiments. The last experiment used an environment with heavy traffic, but thermal changes were presented only for a few seconds. Perception studies with light and heavy traffic on different kinds of roads and potentially with pedestrians should be conducted in the future. All these factors would potentially increase the workload of participants during normal driving and the impact of additional workload added by thermal feedback could be evaluated in detail.

6.5.3 Selection of Use Cases

Use cases for the experiments were chosen from a vast number of possible scenarios within the car. Directional cues can be used in many different scenarios and are not limited on prompting lane changes or navigation. The notifications tested in this thesis were designed for very specific use, with many more possibilities awaiting. Thermal feedback could be used for presentation of the car state, indicating low fuel or tyre pressure. Or it could indicate low outside temperature, warning the driver of possible ice on the road. Participants suggested many different use cases, which could all benefit from thermal presentation and warrant investigation in future studies.

6.5.4 Pre-Set Thermal Cues

The thermal feedback used within this thesis was chosen in accordance with previous work or to allow for cues within a set time frame. Choosing a neutral base temperature led to consistent changes for all participants. But thermal perception can vary between individuals, as can thermal comfort. Temperature changes which some declare barely noticeable are described as very hot by others. For practical and commercial use, this variability will have to be taken into account and adaptable, personalisable cues need to be possible. Future research needs to investigate what cue factors should be adaptable and how to still ensure common associations and interpretations of these differing stimuli.

6.5.5 Different Thermal Devices

Peltier devices of two different sizes were used for the experiments. While the type of device and the capabilities were the same, the area of thermal stimulus presentation has an impact on the perception due to spatial summation [69, 67]. Experiment 3 and 4 use different sized Peltiers, but twice the number of devices for the smaller ones. It is not clear if this balances perception or not. Experiment 5 only used one small device and had some very short cues, while Experiment 6 used the large devices with longer cues. The impact of the size of the different Peltier devices should be tested in detail. Furthermore, a more fitting prototype, such as a thermal steering wheel, instead of Peltier devices manually taped to the wheel, could influence perception and especially comfort enormously.

6.5.6 Influence of Hand Placement

The first two experiments evaluated thermal stimuli on a device placed in front of the participants, later experiments had devices attached to the steering wheel. Results of Experiment 6 showed that cues that tested well under the first condition had significantly lower recognition rate when presented on the steering wheel. Participants commented that they had trouble distinguishing between thermal cues and the natural warming of their hand caused by grasping the steering wheel. Future research should look into this effect and determine how sensing the actual temperature of the hand and adapting cues according to this as the new base temperature influences perception and comfort.

6.5.7 Limited Location

Thermal feedback was only presented on the hands, either in front of the driver or on the steering wheel. If autonomous cars become more prevalent, hands might be rarely placed on the steering wheel or other parts of the car. However, drivers will still be seated and using seat belts. These locations could be used for thermal feedback, the same way they are used for vibrotactile feedback. Perception and comfort studies would have to confirm the feasibility of such feedback. The influence of clothing and temperature build up on the seat through long-time sitting would also have to be investigated in more detail.

6.5.8 Short Duration Studies

Experiments usually took around an hour and thermal feedback was presented for a fraction of the time. It is unknown if and how thermal perception for differing cues on a small area of the skin would change over a longer period of time. Perception could increase due to better familiarity with the cue. However, longer exposure could lead to unexpected reactions of the skin's thermal sensing. Skin adapts to temperatures over time, but how does it react to many differing cues presented on the same location? Future work should evaluate the effect of long exposure to differing thermal cues.

6.5.9 Participant Demographic and Number

Participants were small in number and biased towards higher education and similar in age and gender, as the focus of the experiments was to establish a basic understanding of the effect of thermal feedback in the car rather than ensuring diverse participant groups. Recruitment was primarily within institution (university or company), no specific focus was put into recruiting more diverse groups from different backgrounds and diverse demographics. Investigating perception over a more diverse group and in higher number could yield important insights. Age has been shown to influence thermal perception and cultural background (such as land of origin and its climate) could have an influence on perception and workload.

6.6 Conclusions

Driver visual distraction can be mitigated with tactile feedback. This thesis offers the first insights into the effectiveness of novel thermal feedback for in-car applications. Several factors of thermal perception were investigated during the highly demanding task of driving and thermal feedback was applied to some of the use cases most common in the vehicle.

Findings showed that some thermal cue designs can be identified with high accuracy, both uni- and bimodally, and should be used for non-urgent feedback due to longer recognition times. Direction of temperature change proved to be the best recognised factor of thermal cues on the steering wheel, further enhanced when presented after vibration. These findings provide the basis for a richer and more diverse design space for tactile in-car feedback and can help to improve driver safety.

Appendix A

Appendix

- A.1 Hardware Manuals
- A.1.1 Thermal Bluetooth Board Manual

UGLA Bluetooth 2 Channel Heat Pump Driver User Manual Rev 0.00



Figure 1 - Device with enclosure cover removed

Bluetooth 2-ch Heat Pump Driver



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

OPERATIONAL OVERVIEW:	3
DEVICE CONNECTIONS AND SETUP:	4
PROGRAMMING / COMMUNICATION REFERENCE:	5
The command CMD_SET_CHx_TEMP	7
The command CMD_READ_CURRENT_TEMP	8
The command CMD_READ_SETPOINT_TEMP	9
The command CMD_SET_LEDs	10
The command CMD_SET_ROC_LIMIT	11
DATA STREAMING:	12
FAULT EVENTS:	13
FIRMWARE UPGRADE:	14
DOCUMENT HISTORY	15

Bluetooth 2-ch Heat Pump Driver



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

Operational Overview:

This device allows up to 2 Peltier heat pump devices to be driven in either direction (hot or cold). Each channel has a temperature sensor which is used to close a PID feedback loop. Each channel can have its temperature set point adjusted on the fly by sending simple commands over a serial interface (via USB). The maximum range over which the temperature can be set is from -20degC to +45degC. The temperature for each channel can also be read back using similar commands. The drive electronics can source up to about 4W per channel – the recommended minimum load resistance of the Peltier devices is 3.5ohms and they should be rated for 5V or more. The PID loop in the firmware has been tuned and tested when using the CP20251 heat pump from CUI INC.

As a safety mechanism there is a dedicated circuit separate from the PID control loop that monitors the temperature sensor of both channels. If

- 1) the temperature rises above about 48degreesC on any channel OR
- 2) any of the temperature sensors are disconnected OR
- 3) there is a short circuit across any of the temperature sensors

this circuit will shut down all power to the Peltier heat pumps and flag an exception (the LED will flash red).

Additionally, in the event that the thermistor becomes detached from the surface of a heat pump of which it is supposed to be sensing, then the heat pump will be shut down 12 seconds after the detachment (by looking for expected changes in temperature).

Note: It is important that the thermistor remains properly bonded to the exposed surface of the heat pump. Super-glue is sufficient for this purpose once the ceramic surface of the heat pump has been roughened with sand paper. If the thermistor becomes detached from the heat pump surface there is a danger that the temperature could reach 60 degrees C for a second or two before the detached thermistor protection mechanism kicks in.

The circuit and firmware have been designed and written to support the Panasonic ERT-J1VG103FA thermistor. No other thermistors are supported by this hardware.

Normally the thermistor will be placed close to an edge of the exposed surface of the heat pump. For correct operation of a temperature display to a person's finger it is important that the finger is not placed on top of the thermistor as this will distort the temperature reading from the surface.

The steady state error between the set point and actual channel temperature is typically less than 1degC. Step changes of large magnitude (20 degrees) result in overshoot of about 1 degC before settling to the set point temperature. The rate of change of temperature of the surface of the heat pump is typically 3 degrees C / second.

There is also an internal temperature sensor that measures the ambient temperature inside the enclosure. It is located just inside the roof of the top enclosure half.

Bluetooth 2-ch Heat Pump Driver



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

Device Connections and Setup:

This device is designed to connect to a host PC using Bluetooth serial port profile (SPP). It obtains power from a 4*AA NiMh battery pack that is specially wired to a mini USB jack which should be connected to the mini USB socket on the device. If the device is connected to a USB socket of a PC using a standard USB cable then there will not be enough current available to drive the heat pumps.

Important: In the battery pack it is essential only to use either NiCd or NiMh cells. Alkaline or Lithium AA cells may have too high an internal resistance and may not be capable of providing enough power to drive the heat pumps.

The Peltier heat pumps and their temperature sensing thermistors are supplied on separate heat sinks. They are connected to the device with a pair of 4-pole 2.5mm jack terminated cables. They should be connected to the 2.5mm sockets at the ends of the device.

Important: To avoid over heating and a burning hazard always ensure that the opposite side of the heat pumps is connected to a heatsink. Ideally that heatsink should have a rating of at least 10 degC / Watt.

Note: Both heat pumps (or at least their temperature sensors) must be connected to the device or else it will shut down to a fault state.


1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

Programming / Communication Reference:

The board has an onboard microcontroller that performs commands on demand from a host PC. The host talks to the board using a serial COM port over a Bluetooth SPP serial port profile wireless connection.

The COM port settings are BAUD = 460800, 1 start bit, 8 data bits, 1 stop bit, no parity and hardware flow control.

The host configures the device by sending command packets, and the device will return acknowledge packets with the result if the command executed successfully and not-acknowledge packets if the command was not executed. (reasons that a command may not execute is that a previous command has not completed execution, or the command argument or type are invalid)

The board will also send back event packets in the event that a fault condition was detected (either an over temperature fault or a supply brown out fault)

All command packets to the board have the following format -

\$CMD,tttt,cc

where tttt is the command argument (a 16 bit signed integer) and cc is the command type (an 8 bit unsigned integer) and the format is in Hex.

A detailed description for each command are listed at the end of this document with several examples cited for clarity.

The acknowledge packet returned by the board to the hosts always takes this format

\$ACK,tttt,cc

where tttt is the acknowledge value (a 16 bit signed integer) and cc is the command type that owns this acknowledge (an 8 bit unsigned integer) and the format is in Hex.

Finally, a not-acknowledge packet takes this format -

\$NAK, tttt, cc

where tttt is 0000 and cc is the command type that caused this not-acknowledge (an 8 bit unsigned integer) and the format is in Hex.

When the host first opens the COM port to the board, it will receive the following text (note that the firmware revision numbers may be different to that printed here) -

Bluetooth 2-ch Heat Pump Driver	1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com
---------------------------------------	--

UGLA Bluetooth 2 Channel Heat Pump Driver v0.00 Copyright 2011 SAMH Engineering Services Firmware Revision 00.00

where the firmware revision will vary but always have the format AA.bb where A and b are decimal digits.



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

The command CMD_SET_CHx_TEMP

This command is used to set the temperature of channel x where x is either 1 or 2 depending on which channel you want to set the temperature for. The command argument field is signed 16 bit integer in units of tenths of a degree Celsius. The maximum range over which the temperature can be set is from -20degC to +45degC.

The command type has the value of the channel number to read.

Example:

To set the temperature of channel 1 to 45 degreesC send this command -

\$CMD,01C2,01

To set the temperature of channel 2 to 0 degreesC send this command -

\$CMD,0000,02

To set the temperature of channel 2 to -20 degreesC send this command -

\$CMD, FF38,02

To disable a channel, set the value field to 0xFFFF. For example to disable channels 1 and 2 send -

\$CMD, FFFF, 01
\$CMD, FFFF, 02



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

The command CMD_READ_CURRENT_TEMP

This command is used to read the temperature of channel x where x is either 1 or 2 depending on which channel you want to set the temperature for.

The command type has the value 0x05, and the command argument is the channel number to read.

The temperature is returned in the value field of an \$ACK packet. It is in hex format, signed 16 bit, and units of tenths of a degree Celsius.

Example:

To read the temperature of channel 1 send this command -

\$CMD,0001,05

If the temperature is 20 degrees C, then the returned packet is -

\$ACK,00CB,05

Example: To read the temperature of channel 2 send this command –

\$CMD,0002,05

If the temperature is 20 degrees C, then the returned packet is -

\$ACK,00CB,05



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

The command CMD_READ_SETPOINT_TEMP

This command is used to read the temperature set-point of either channel.

The command type has the value 0x06, and the command argument is the channel number to read.

The temperature is returned in the value field of an \$ACK packet. It is in hex format, signed 16 bit, and units of tenths of a degree Celsius.

Example:

To read the temperature set-point of channel 1 send this command -

\$CMD,0001,06

If the temperature set-point is 20 degrees C then the returned packet is -

\$ACK,00CB,06

The default set-point temperatures after power up is the value 0xFFFF which implies that the channel is disabled.

Bluetooth 2-ch Heat Pump Driver	Services	1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com
---------------------------------------	----------	--

The command CMD_SET_LEDs

Use this command to set the state of the RGB LED on the board. Once the LED state is set, it remains so until it is changed again regardless of any other commands that are performed.

The command type has the value 0x00, and the command argument is-

Revert to firmware control = 0x0000RED = 0x0001 (i.e. bit 1) GREEN = 0x0002 (i.e. bit 2) BLUE = 0x0004 (i.e. bit 3)

So the LED colours can be combined by adding (ORing) the values together.

If successful, an ACK packet will be returned with 0 as its first argument and the CMD_SET_LEDs enumeration code as the second argument.

Example: To set the LED to ORANGE send this command -

\$CMD,0003,00



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

The command CMD_SET_ROC_LIMIT

Use this command to limit the rate of change of temperature to the heat pumps. Normally the rate of change can be up to 3 degrees/second with the specified Peltiers, but in certain circumstances this rate of change may need to be reduced. If the limit is enabled, the maximum rate will typically be reduced by a factor of three to about 1 degree per second, although variations will manifest depending on the Peltiers used and the starting and ending temperature during the change.

The command type has the value 0x08, and the command argument is-

DISABLE_RATE_LIMIT = 0x0000, ENABLE_RATE_LIMIT = 0x0001

If successful, an ACK packet will be returned with 8 as its first argument and the CMD_SET_ROC_LIMIT enumeration code as the second argument.

Example: To set the rate limit for all 2 channels to 1 degree per second send this command -

\$CMD,0001,08



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

Data Streaming:

For debugging purposes it is possible to enable streaming of the current temperature and set point temperatures for all channels. The temperatures are output in units of degrees Celsius. To enable this feature send the following command –

\$CMD,0001,07

To disable it send this command -

\$CMD,0000,07

This feature is enabled by default.

The following is typical of the output data, and it is output a couple of times per second(the third temperature is the ambient temperature) -

\$SET POINT ,+20,+20 \$TEMPERATURE,+21,+19,+23



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

Fault Events:

There are five circumstances that can cause a fault event -

- If any one of the active channels has an over temperature condition
 If the temperature sense thermistor for any active channel is electrically
 - disconnected
- If the temperature sense thermistor for any active channel is electrically short circuited
- 4) If the thermistor for any active channel becomes detached from the heat pump and if that heat pump is driven in the hot direction for more than about 12 seconds
- 5) If there is a brown out on the supply voltage (when supplied from a power supply) or if the battery is low (when supplied with the battery pack)

In this circumstance the following packet will be output -

\$EVT,000X,00

Where X is the fault reason number (as per the list above).

The board will also shut down to a low power state and indicate a flashing pattern on the red LED. This pattern is a long flash followed by a series of short flashes where the number of short flashes is the same as the number of the fault event as listed above.

In order to recover from a fault condition the power to the board must be cycled by removing then re-inserting the power cable.

Bluetooth 2-ch Heat Pump Driver Heat Pump

Firmware Upgrade:



To perform a firmware upgrade, unzip the .exe and .hex firmware files to a folder on a windows PC. Make a note of the COM port number that the device is normally connected to the PC. Turn on the power switch while holding the navigation switch in. Run the .exe bootloader program. Follow the instructions displayed. Wait until the bootloader program has finished execution, then re-connect and use the board as normal.

Bluetooth 2-ch
Heat Pump
Driver



1 Leopardstown Drive, Blackrock, Co. Dublin, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com W: www.samh-engineering.com

Document History

Revision #	Date	Changes
0.00	27/08/2011	Initial Draft

A.1.2 Thermal USB Board Manual

QUUTEC Quad Universal USB ThermoElectric Controller -User Manual Version 0.1 (Draft)



OPERATIONAL OVERVIEW:	3
DEVICE CONNECTIONS AND SETUP:	5
PROGRAMMING / COMMUNICATION REFERENCE:	6
Device Reset Command	8
Set Temperature Command	8
Read Temperature Command	9
Read Temperature Setpoint Command	
Modify Rate of Change Limit Command	
DATA STREAMING:	12
FAULT EVENTS:	13
FIRMWARE UPGRADE:	14
DOCUMENT HISTORY	15

4-ch Heat Pump Driver Board	Engineering Services	34 S Irela Ph: E: s
-----------------------------------	----------------------	------------------------------

34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Note: This hardware is for research purposes only



34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Operational Overview:

The QUUTEC device allows up to 4 Peltier heat pump devices to be driven in either direction (hot or cold). Each channel has a temperature sensor which is used to close a PID feedback loop. Each channel can have its temperature set point adjusted on the fly by sending simple commands over a serial interface (via USB). The maximum range over which the temperature can be set is from -20degC to +45degC. The temperature for each channel can also be read back using similar commands. The drive electronics can source up to about 5W of power per channel – the recommended minimum load resistance of the Peltier devices is 3.5 ohms and they should be rated for 5V or more. The PID loop in the firmware has been tuned and tested when using the CP20251 heat pump from CUI INC.

As a safety mechanism there is dedicated electronics that are separate to the microcontroller to monitor the temperature of each channel. If the temperature rises above about 60 degrees Celsius on any channel this circuit will shut down all power to the Peltier heat pumps. This ensures that in the event of the microcontroller 'crashing' a situation that could cause burns will not arise.

Additionally the microcontroller implements safety mechanisms that will shut off power to the heat pumps if their temperature rises above 50 degC under normal operation. It will also shut off the heat pumps if the thermistor that senses the temperature of an active channel becomes electrically disconnected.

Furthermore, in the event that the thermistor becomes detached from the surface of a heat pump of which it is supposed to be sensing, then the heat pump will be shut down 12 seconds after the detachment.

Note: It is important that the thermistor remains properly bonded to the exposed surface of the heat pump. Super-glue is sufficient for this purpose once the ceramic surface of the heat pump has been roughened with sand paper. If the thermistor becomes detached from the heat pump surface there is a danger that the temperature could reach 70 degrees C for a second or two before the detached thermistor protection mechanism kicks in.

The circuit and firmware have been designed and written to support a thermistor with the following characteristics:

- Resistance in Ohms @ 25°C 10k
- Resistance Tolerance ±1%
- B Value Tolerance ±1%
- B25/50 3375K +/- 5K
- B25/85 3435K +/- 5K

For example, the following thermistors can be used:

- Panasonic ERT-J0EG103FA
- Murata NXFT15XH103FA1B025





34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Thermistors with characteristics other than those listed above are not supported by this hardware.

Normally the thermistor will be placed close to an edge of the exposed surface of the heat pump. For correct operation of a temperature display to a person's finger it is important that the finger is not placed on top of the thermistor as this will distort the temperature reading from the surface.

The steady state error between the set point and actual channel temperature is typically less than 1degC. Step changes of large magnitude (20 degrees) can result in overshoot of about 2 degC before settling to the set point temperature. The rate of change of temperature of the surface of the heat pump is typically 3 degrees C / second, however this can be limited to about 1 degree C / second as described in the Modify Rate of Change Limit Command Section below.



Device Connections and Setup:

The QUUTEC device is designed to connect to a host PC using a USB cable (mini-B plug) connected to the "Host USB" socket:



The device obtains power for the Peltiers from 5V power supplies with a micro-USB plug and with 1.6A or greater output capability (recommended minimum is 2A). They connect to the sockets shown here:



The Peltier heat pumps and their temperature sensing thermistors are supplied individually, attached to heat sinks. They are connected to the device via the 4 micro-USB sockets on the front of the PCB:





34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Programming / Communication Reference:

The device includes a microcontroller that responds to commands originating from a connected (host) computer. This host communicates with the device via a serial COM port over a USB physical interface.

The USB interface is based on the FT232R IC and drivers can be downloaded here - <u>http://www.ftdichip.com/Drivers/VCP.htm.</u>

The COM port settings are BAUD = 460800, 1 start bit, 8 data bits, 1 stop bit, no parity and hardware flow control.

When the COM port is opened, the MODE LED on the device should illuminate green, and when the COM port is closed, the MODE LED should extinguish.



The host configures the device by sending command packets, and the device will return acknowledge packets with the result if the command executed successfully and not-acknowledge packets if the command was not executed.

Note: reasons that a command may not execute is that a previous command has not completed execution, or the command argument or type are invalid.

The device will also send back event packets in the event that a fault condition was detected (either an over temperature fault or a supply brown out fault).

All command packets to the device have the following format -

\$CMD,tttt,cc

where tttt is the command argument (a 16 bit signed integer) and cc is the command type (an 8 bit unsigned integer) and the format is in Hex.

A detailed description for each command are listed at the end of this document with several examples cited for clarity.

The acknowledge packet returned by the device to the hosts always takes this format -

\$ACK,tttt,cc

where <code>tttt</code> is the acknowledge value (a 16 bit signed integer) and <code>cc</code> is the command type that owns this acknowledge (an 8 bit unsigned integer) and the format is in Hex.



34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Finally, a not-acknowledge packet takes this format -

\$NAK, tttt, cc

where tttt is 0000 and cc is the command type that caused this not-acknowledge (an 8 bit unsigned integer) and the format is in Hexadecimal.

When the host first opens the COM port to the device, it will receive the following text (note that the firmware revision numbers may be different to that printed here) –

QUUTEC Quad Universal USB Thermoelectric Controller v2.00 Copyright 2017 SAMH Engineering Services Firmware Revision 00.00

where the firmware revision will vary but always have the format AA.bb where A and b are decimal digits.





34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Device Reset Command

In the event of an error condition, or to immediately disable all heat pumps, the reset command can be invoked by sending a '!' character (ASCII code 33).

Once received, the device will

- Reset the previous error condition, if any
- Output the startup splash (see above)
- Revert all heat pump drivers to the OFF state

Set Temperature Command

This command is used to set the temperature of channel X where X is either 1,2 3 or 4 depending on which channel the temperature setpoint is to be configured.

The command argument field is signed 16 bit integer in units of tenths of a degree Celsius. The maximum range over which the temperature can be set is from -20degC to +45degC.

The command type has the value of the channel number to read.

Example:

To set the temperature of channel 1 to 45 degreesC send this command -

\$CMD,01C2,01

To set the temperature of channel 4 to 0 degreesC send this command -

\$CMD,0000,04

To set the temperature of channel 2 to -20 degreesC send this command -

\$CMD, FF38,02

To disable a channel, set the value field to 0xFFFF. For example to disable channels 3 and 4 send –

\$CMD, FFFF, 03
\$CMD, FFFF, 04

Note: In practice, it may not be possible for the temperature delta between ambient and the surface of the Peltier to exceed a magnitude of 20 deg C. For example, if the ambient temperature is 25 deg C, it may only be possible for the Peltier to drive the temperature to 5 deg C, due to the power constraints of the Peltier modules and the heatsinking ability of the heatsink to which the Peltier is attached.



34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Read Temperature Command

This command is used to read the temperature of channel x where x is either 1,2 3 or 4 depending on which channel you want to set the temperature for.

The command type has the value 0x05, and the command argument is the channel number to read.

The temperature is returned in the value field of an \$ACK packet. It is in hex format, signed 16 bit, and units of tenths of a degree Celsius.

Example: To read the temperature of channel 1 send this command –

\$CMD,0001,05

If the temperature is 20 degrees C, then the returned packet is -

\$ACK,00CB,05

Example: To read the temperature of channel 4 send this command –

\$CMD,0004,05

If the temperature is 20 degrees C, then the returned packet is -

\$ACK,00CB,05



Read Temperature Setpoint Command

This command is used to read the temperature set-point of any one of the 4 channels.

The command type has the value 0x06, and the command argument is the channel number to read.

The temperature is returned in the value field of an \$ACK packet. It is in hex format, signed 16 bit, and units of tenths of a degree Celsius.

Example:

To read the temperature set-point of channel 1 send this command -

\$CMD,0001,06

If the temperature set-point is 20 degrees C then the returned packet is -

\$ACK,00CB,06

The default set-point temperatures after power up is the value 0xFFFF which implies that the channel is disabled.



34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Modify Rate of Change Limit Command

Use this command to enable or disable the limiting of the rate of change of temperature to the heat pumps. Normally the rate of change can be up to 3 degrees/second with the specified Peltiers, but in certain circumstances this rate of change may need to be reduced. If the limit is enabled, the maximum rate will typically be reduced by a factor of three to about 1 degree per second, although variations will manifest depending on the Peltiers used and the starting and ending temperature during the change.

The command type has the value 0x08, and the command argument is-

DISABLE_RATE_LIMIT = 0x0000, ENABLE_RATE_LIMIT = 0x0001

If successful, an ACK packet will be returned with 8 as its first argument and the CMD_SET_ROC_LIMIT enumeration code as the second argument.

Example: To set the rate limit for all 4 channels to 1 degree per second send this command –

\$CMD,0001,08



34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Data Streaming:

For debugging purposes it is possible to enable streaming of the current temperature and set point temperatures for all channels. The temperatures are output in units of degrees Celsius.

To disable this feature send the following command -

\$CMD,0000,07

To re-enable it send this command -

\$CMD,0001,07

This feature is **enabled** by default.

The following is typical of the output data, and it is output a couple of times per second - $% \left({{{\mathbf{x}}_{i}}} \right) = {{\mathbf{x}}_{i}} \right)$

\$SET POINT ,+20,+20,###,####
\$TEMPERATURE,+21,+19,+21,+20

Note: If the temperature setpoint has not been configured for a channel, the set point shall be listed as ###. This indicates that this particular channel is disabled, i.e. no heating or cooling is active for that channel.



34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Fault Events:

There are four circumstances that can cause a fault event -

- 1) If any one of the active channels has an over temperature condition
- 2) If the temperature sense thermistor for any active channel is electrically disconnected
- If the thermistor for any active channel becomes detached from the heat pump and if that heat pump is driven in the hot direction for more than about 12 seconds
- If there is a brown out on the supply voltage (when supplied from a power supply) or if the battery is low (when supplied with the battery pack)

In this circumstance the following packet will be output -

\$EVT,000X,00

Where X is the fault reason number.

- 1 Over Temperature Fault
- 2 Disconnected Thermistor Fault
- 3 Detached Thermistor Fault
- 4 Brown-out Fault

The device will also shut down to a low power state and indicate a flashing pattern on the FAULT LED.



This pattern is a long flash followed by a series of short flashes where the number of short flashes is the same as the number of the fault event as listed above.

In order to recover from a fault condition either

- Sending a reset command (i.e. by sending the '!' character, see Reset Command section above).
- the power to the device must be cycled by removing then re-inserting the host USB cable



34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Firmware Upgrade:

To perform a firmware upgrade:

1. unzip the .exe and .hex firmware files to a folder on a windows PC.

QUUTEC Firmware Upgrade Tool.exe	28/02/2017 11:24	Application	13 KB
QUUTEC.hex	28/02/2017 11:28	UltraEdit Docume	70 KB

- 2. Make a note of the COM port number that the device is normally connected to the PC.
- 3. Disconnect the power and USB cables from the device.
- 4. Run the QUUTEC Firmware Upgrade Tool.exe program.
- 5. Follow the instructions displayed.
- 6. Wait until the bootloader program has finished execution, then re-connect and use the device as normal.



4-ch Heat	
Pump Driver	
Board	



34 Stepaside Park, Dublin 18, Ireland Ph: +353 86 8240409 E: stephenahughes@gmail.com

Document History

Revision #	Date	Changes
0.1	27/02/2017	Initial Draft

A.2 Experiment 1

A.2.1 Experiment 1: Information Sheet

TempLane – Investigating Thermal Feedback for Lane Change Scenarios



INFORMATION SHEET

BACKGROUND

The goal of this user study is to improve safety in driving situations by providing navigational information through thermal feedback on the steering wheel.

TASK

The study is conducted in a driving simulator and will consist of four blocks. You will be asked to follow instructions given either through thermal feedback, both warm and cold, or audio instructions. You will perform the tasks sitting in a padded, adjustable chair. There will be a steering wheel securely attached to the desk in front of you. You will be holding the wheel throughout the experiment with one hand while driving. The other hand will be placed on the table next to the steering wheel and you will be asked to place your finger on the Peltier device (used for the thermal feedback). The simulator requires you to keep a virtual vehicle within the limits of the road and keep it inside a lane. You will get instructions to change lanes, either through audio commands ("right" or "left") or through thermal feedback (warm for right and cold for left). Please change to the next lane on the indicated side as fast and accurately as possible and try to stay inside the lane, whenever you are not actively changing into another. Please always only change one lane.

You will have time to get used to the simulator and the thermal feedback and the audio navigation in the first two blocks of the experiment, while keeping in the middle lane and reporting back the stimuli. In the third and fourth block you will be asked to follow instructions given by either audio or the thermal commands. In the beginning and after each block you will be asked to fill in questionnaires and will you have time to take a break.

INSTRUCTIONS

- Please place one hand on the steering wheel and the other on the table while driving.
- Please try to follow the instructions given by the system to the best of your ability and try to keep inside the lane. If not indicated otherwise, please follow the road straight ahead.

MISCELLANEOUS

- You will be wearing headphones throughout the experiment.
- The experiment is divided into blocks. You can rest between the blocks for as long as you need.
- You can withdraw from the study at any time without penalty. The data recorded until then will be destroyed.

IMPORTANT

The systems you interact with will not ask or store personal information and all data recorded are anonymised. The study will take approximately 60 minutes to complete and £6 will be paid at the end. If you are a student, no course credits will be awarded for completing this study. To take part in this study, the following criteria must be met:

- I am at least 18 years old
- I hold a full driving licence
- I have no visual impairments (not including wearing glasses or contact lenses)
- I have no neurological disorders (e.g. epilepsy)
- I do not suffer from motion sickness
- I do not have any sensory impairments

You will be given resting sessions between experimental conditions but please do not be afraid to ask for a break if required at any time during the study. Furthermore, if you feel unwell at any point, please let the experimenter know immediately so that medical attention can be given as soon as possible If you have any questions about the study or would like more information about our research in general, contact details are provided below.

CONTACT DETAILS

Name:	Patrizia Di Campli San Vito (PhD student/Experimenter)
Email:	p.di-campli-san-vito.1@research.gla.ac.uk
Office:	Room F131, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Tel:	+44 (0)141-330-8430
Supervisor:	Professor Stephen Brewster
Email:	stephen.brewster@glasgow.ac.uk
Office:	Room S131, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Tel:	+44 (0)141-330-4966

This study has been approved by the Ethics Committee (Reference Number - D1483958803605).

This research is part funded by the EPSRC and Jaguar Land Rover.

Version No. 3 (June 2017)

A.2.2 Experiment 1: Consent Form

TempLane – Investigating Thermal Feedback for Lane Change Scenarios

CONSENT FORM

I confirm that I have read and understood the Study Information Sheet provided to me for the above study and have had the opportunity to ask questions.
The study has been explained to me, and I understand the explanation given and what my participation will involve.
I understand that during the course of the study the experimenter will give additional instructions concerning the experiment and that the purpose of some of the instructions will only be explained at the end of the study.
I understand that my participation is voluntary and that I am free to withdraw at any time, without giving a reason.
I understand that the research data may be accessed by researchers working at the GIST, but that at all times my personal data will be kept confidential in accordance with data protection guidelines.

I have initialled the above boxes myself and I agree to take part in the study.			
FULL NAME:			
SIGNATURE:			
DATE:			
EMAIL:			

Experimenter:	Ms Patrizia Di Campli San Vito (p.di-campli-san-vito.1@research.gla.ac.uk)
DATE, SIGNATURE:	
Supervisor details:	Professor Stephen Brewster (stephen.brewster.glasgow.gla.ac.uk, 0141-330-4966)

This study has been approved by the Ethics Committee (Reference Number -).

This research is part funded by the EPSRC and Jaguar Land Rover.



Version No. 2 (March 2017)

A.2.3 Experiment 1: Questionnaires

Participant Number:	TempLane_Beginning					
Age:						
Gender: male female other						
Occupation / field of study:						
Years of driving experience:						
Country, where driving licence was obtained:						
Please rate your experience with the following:						
(none) 1 2 3 4 5 (much)						
Experience with a driving simulator						
Experience with audio navigation						
Experience with thermal feedback						
Please indicate which hand is dominant:						
left right both						
Any comments?						

1



Participant Number:						Instruction Type:	
Please rate the instruction type you just finished using the following attributes:							
(not at all)1 2	2 3		4	5 (much)		
Pleasant	¦	¦	¦	¦			
Comfortable	¦	¦	¦	¦			
Disruptive	¦	¦	¦	¦			
Complicated	¦	¦	¦	¦			

Any comments?
Participant Number: _____

TempLane _End

Please rate how much you liked the different instruction type:

(not at all) 1 2 3 4 5 (much) Audio |---|---|---|---|---|

Please name reasons for your rating:

Imagine you were presented with thermal feedback on the steering wheel for navigation purposes, where the devices of one side of the wheel will be warmed, while the other side will be cooled, how would you interact:

If the right side of the steering wheel was warmed, while the left side was cooled, I would turn to the:

Left	Right	

Is there any other kind of information in a car that you would like to receive through thermal feedback? Why?

At what places in the car would you like to receive thermal feedback?

Any other comments?

A.3 Experiment 2

A.3.1 Experiment 2: Information Sheet

StimLen – Investigating Thermal Feedback for Lane Change Scenarios



INFORMATION SHEET

BACKGROUND

The goal of this user study is to improve safety in driving situations by providing navigational information through thermal feedback on the steering wheel. To achieve this, we compare different types of stimuli representation.

TASK

The study is conducted in a driving simulator and will consist of eight blocks. You will be asked to follow instructions given through thermal feedback, both warm and cold. You will perform the tasks sitting in a padded chair. There will be a steering wheel securely attached to the desk in front of you. You will be holding the wheel throughout the experiment with one hand while driving. The other hand will be placed on the table next to the steering wheel and you will be asked to place your finger on the Peltier device (used for the thermal feedback). The simulator requires you to keep a virtual vehicle within the limits of the road and keep it inside a lane. You will get instructions to change lanes through thermal feedback (warm for right and cold for left). Please change to the next lane on the indicated side as fast and accurately as possible and try to stay inside the lane, whenever you are not actively changing into another. Please always only change one lane.

You will have time to get used to the simulator and the thermal feedback before the start of the first block of the experiment. Between the blocks you will have time to take a break. After the eight blocks you will be asked to fill in a questionnaire.

INSTRUCTIONS

- Please place one hand on the steering wheel and the other on the table while driving.
- Please try to follow the instructions given by the system to the best of your ability and try to keep inside the lane. If not indicated otherwise, please follow the road straight ahead.

MISCELLANEOUS

- You will be wearing headphones throughout the experiment.
- The experiment is divided into blocks. You can rest between the blocks for as long as you need.
- You can withdraw from the study at any time without penalty. The data recorded until then will be destroyed.

IMPORTANT

The systems you interact with will not ask or store personal information and all data recorded are anonymised. The study will take approximately 60 minutes to complete and £6 will be paid at the end. If you are a student, no course credits will be awarded for completing this study. To take part in this study, the following criteria must be met:

- I am at least 18 years old
- I hold a full driving licence
- I have no visual impairments (not including wearing glasses or contact lenses)
- I have no neurological disorders (e.g. epilepsy)
- I do not suffer from motion sickness
- I do not have any sensory impairments

You will be given resting sessions between experimental blocks but please do not be afraid to ask for a break if required at any time during the study. Furthermore, if you feel unwell at any point, please let the experimenter know immediately so that medical attention can be given as soon as possible. If you have any questions about the study or would like more information about our research in general, contact details are provided below.

CONTACT DETAILS

Name:	Patrizia Di Campli San Vito (PhD student/Experimenter)
Email:	p.di-campli-san-vito.1@research.gla.ac.uk
Office:	Room F141, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Supervisor:	Professor Stephen Brewster
Email:	stephen.brewster@glasgow.ac.uk
Office:	Room S131, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Tel:	+44 (0)141-330-4966
This study has b	een approved by the Ethics Committee (Reference Number - 300160073).

This research is part funded by the EPSRC and Jaguar Land Rover.

Version No. 1 (October 2017)

A.3.2 Experiment 2: Consent Form

StimLen – Investigating Thermal Feedback for Lane Change Scenarios

CONSENT FORM

I confirm that I have read and understood the Study Information Sheet provided to me for the above study and have had the opportunity to ask questions.
The study has been explained to me, and I understand the explanation given and what my participation will involve.
I understand that during the course of the study the experimenter will give additional instructions concerning the experiment and that the purpose of some of the instructions will only be explained at the end of the study.
I understand that my participation is voluntary and that I am free to withdraw at any time, without giving a reason.
I understand that the research data may be accessed by researchers working at the GIST, but that at all times my personal data will be kept confidential in accordance with data protection guidelines.

I ha	ve initialled the above boxes myself and I agree to take part in the	study.
FULL NAME:		
SIGNATURE:		
DATE:		
EMAIL:		

Experimenter:	Ms Patrizia Di Campli San Vito (p.di-campli-san-vito.1@research.gla.ac.uk)
DATE, SIGNATURE:	
Supervisor details:	Professor Stephen Brewster (stephen.brewster.glasgow.gla.ac.uk, 0141-330-4966)

This study has been approved by the Ethics Committee (Reference Number -300160073).

This research is part funded by the EPSRC and Jaguar Land Rover.



Version No. 1 (October 2017)

A.3.3 Experiment 2: Questionnaires

QuestionnaireStim	ıLen				
At the end					
* 1. Participant Numbe	er				
* 2. Age					
* 3. Gender					
male female	other				
* 4 0		P I I X			
* 4. Occupation (and f	ield of study, if ap	olicable)			
* 5. Years of driving e	xperience				
* 6. Country, where th	e driving license w	vas obtained			
* 7. Please indicate w	hich hand is domir	nant			
left right I	both				
* 8. Please rate your e	experience with the	e following			
	1 (none)	2	3	4	5 (very much)
Experience with a driving simulator	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Experience with thermal feedback	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
9. Any comments?					

A.4 Experiment 3

A.4.1 Experiment 3: Information Sheet

TempWheel – Investigating Thermal Feedback on the Steering Wheel



INFORMATION SHEET

BACKGROUND

The goal of this user study is to improve safety in driving situations by providing navigational information through thermal feedback on the steering wheel.

TASK

The study is conducted in a driving simulator and will consist of three blocks. You will be asked to follow instructions given by a navigation system, either through thermal feedback, both warm and cold, or audio instructions. You will perform the tasks sitting in a padded, adjustable chair. There will be a steering wheel securely attached to the desk in front of you. You will be holding the wheel throughout the experiment while driving. The simulator requires you to keep a virtual vehicle within the limits of the road, driving on the right hand side of the road. You will follow navigational instructions by turning on specified points.

You will have time to get used to the simulator and the thermal feedback and the audio navigation in the first block of the experiment. In the second and third block you will be asked to follow instructions given by either an audio or the thermal navigation system. In the beginning and after each block you will be asked to fill in questionnaires and will you have time to take a break.

The navigation instructions will be given in two stages: first you will be given a short indicator 200 metres before the turning point and then another directly before you have to turn. After turning, the virtual car will move to a different location, where you should follow the road straight ahead until the next turning event. The simulation will stop automatically after 12 turns.

INSTRUCTIONS

- Please place both hands at the indicated spaces on the steering wheel while driving.
- Please drive in the rightmost lane and ignore any road signs, including painted arrows or stop signs painted on the road itself.
- Please try to follow the instructions given by the navigation system to the best of your ability and try to keep inside the lane. If not indicated otherwise, please follow the road straight ahead.

MISCELLANEOUS

- You will be wearing headphones throughout the experiment.
- The experiment is divided into blocks. You can rest between the blocks for as long as you need.
- You can withdraw from the study at any time without penalty. The data recorded until then will be destroyed.

IMPORTANT

The systems you interact with will not ask or store personal information and all data recorded are anonymised. The study will take approximately 60 minutes to complete and £6 will be paid at the end. If you are a student, no course credits will be awarded for completing this study. To take part in this study, the following criteria must be met:

- I am at least 18 years old
- I hold a full driving licence
- I have no visual impairments (not including wearing glasses or contact lenses)
- I have no neurological disorders (e.g. epilepsy)
- I do not suffer from motion sickness
- I do not have any sensory impairments

You will be given resting sessions between experimental conditions but please do not be afraid to ask for a break if required at any time during the study. Furthermore, if you feel unwell at any point, please let the experimenter know immediately so that medical attention can be given as soon as possible If you have any questions about the study or would like more information about our research in general, contact details are provided below.

CONTACT DETAILS

Name:	Patrizia Di Campli San Vito (PhD student/Experimenter)
Email:	p.di-campli-san-vito.1@research.gla.ac.uk
Office:	Room F131, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Tel:	+44 (0)141-330-8430
Supervisor:	Professor Stephen Brewster
Email:	stephen.brewster@glasgow.ac.uk
Office:	Room \$131 Sir Alwyn Williams Building (Computing Science) Glasgow G12 887
	Scotland

This study has been approved by the Ethics Committee (Reference Number - D1483958803605).

This research is part funded by the EPSRC and Jaguar Land Rover.

Version No. 3 (June 2017)

A.4.2 Experiment 3: Consent Form

TempWheel – Investigating Thermal Feedback on the Steering Wheel

CONSENT FORM

I confirm that I have read and understood the Study Information Sheet provided to me for the above study and have had the opportunity to ask questions.
The study has been explained to me, and I understand the explanation given and what my participation will involve.
I understand that during the course of the study the experimenter will give additional instructions concerning the experiment and that the purpose of some of the instructions will only be explained at the end of the study.
I understand that my participation is voluntary and that I am free to withdraw at any time, without giving a reason.
I understand that the research data may be accessed by researchers working at the GIST, but that at all times my personal data will be kept confidential in accordance with data protection guidelines.

I ha	ve initialled the above boxes myself and I agree to take part in the	study.
FULL NAME:		
SIGNATURE:		
DATE:		
EMAIL:		

Experimenter:	Ms Patrizia Di Campli San Vito (p.di-campli-san-vito.1@research.gla.ac.uk)
DATE, SIGNATURE:	
Supervisor details:	Professor Stephen Brewster (stephen.brewster.glasgow.gla.ac.uk, 0141-330-4966)

This study has been approved by the Ethics Committee (Reference Number - D1483958803605).

This research is part funded by the EPSRC and Jaguar Land Rover.



Version No. 2 (March 2017)

A.4.3 Experiment 3: Questionnaires

Participant Number:	TempWheel_Beginning	
Age:		
Gender: male	female other	
Occupation / field of study:		
Years of driving experience:		
Country, where driving licence wa	s obtained:	
Please rate your experience with th	e following:	
(none) 1	2 3 4 5 (much)	
Experience with a driving simulator		
Experience with audio navigation	- ¦ ¦	
Experience with thermal feedback	- ¦ ¦	
In the following experiment, you will be presented with thermal feedback for navigation purposes on the steering wheel, where the device of one side of the wheel will be warmed, while the other side will be cooled. Please decide: If the right side of the steering wheel was warmed, while the left side was cooled, I would turn to the:		
Left	Right	
Any comments?		

1



Participant Number:	Navigation Type:
Please rate the naviga	ation you just finished using the following attributes:
(not at all)) 1 2 3 4 5 (much)
Pleasant	
Comfortable	
Disruptive	
Complicated	! ! !

Any comments?

Participant Number: _____

TempWheel_End

Please rate how much you liked the different navigation types:

(not at all) 1 2 3 4 5 (much)

Audio |---|---|---|---|---|

Thermal |----|----|----|----|----|----|

Please name reasons for your rating:

Having finished the experiment, which direction would you turn towards, if presented with thermal feedback for navigation purposes?

Warm	
------	--

Cold

Is there any other kind of information in a car that you would like to receive through thermal feedback? Why?

Is there any other place in the car where you would like to receive thermal feedback?

Any other comments?

A.5 Experiment 4

A.5.1 Experiment 4: Information Sheet

HapNav – Investigating Thermal and Cutaneous Push Feedback on the Steering Wheel



INFORMATION SHEET

BACKGROUND

The goal of this user study is to improve safety in driving situations by providing navigational information through thermal feedback on the steering wheel.

TASK

The study is conducted in a driving simulator and will consist of two blocks. You will be asked to follow instructions given by a navigation system, either through thermal feedback (warm temperature change) or cutaneous push (a pin poking your palm gently). You will perform the tasks sitting in a gaming racing chair. There will be a steering wheel securely attached in front of you and foot pedals on the ground. You will be holding the wheel throughout the experiment while driving. The simulator requires you to keep a virtual vehicle within the limits of the road, driving on the right-hand side of the road. You will follow navigational instructions by turning on specified points.

You will have time to get used to the simulator, the thermal feedback and the push feedback. In the first block you will be asked to follow instructions given by either push feedback or thermal, in the second block it will be the other condition. Each block consists of an introduction to the stimuli, a short training period and the task in the assigned condition. In the beginning and after each block you will be asked to fill in questionnaires and will you have time to take a break.

The navigation instructions will be given in two stages: first you will be given a short indicator 200 metres before the turning point and then another directly before you the turn. When you identify the indicator 200 metres before the turn, please press the right or middle foot pedal on the side of the turning direction (for left, please press the middle pedal). At the turning point, please turn towards the indicated direction. After turning, the virtual car will move (reset) to a different location, where you should follow the road straight ahead until the next turning event. The simulation will stop automatically after 12 turns.

INSTRUCTIONS

- Please place both hands at the indicated spaces on the steering wheel while driving.
- Please drive in the rightmost lane and ignore any road signs, including painted arrows or stop signs painted on the road itself.
- Please try to follow the instructions given by the navigation system to the best of your ability and try to keep inside the lane. If not indicated otherwise, please follow the road straight ahead.
- If a resetting error occurs, please wait for the experimenter to reset the car manually.

MISCELLANEOUS

- You will be wearing headphones throughout the experiment.
- The experiment is divided into blocks. You can rest between the blocks for as long as you need.

Version No. 2 (July 2018)

- You can withdraw from the study at any time without penalty. The data recorded until then will be destroyed.

IMPORTANT

The systems you interact with will not ask or store personal information and all data recorded are anonymised. The study will take approximately 60 minutes to complete and £6 will be paid at the end. If you are a student, no course credits will be awarded for completing this study. To take part in this study, the following criteria must be met:

- I am at least 18 years old
- I hold a full driving licence
- I have no visual impairments (not including wearing glasses or contact lenses)
- I have no neurological disorders (e.g. epilepsy)
- I do not suffer from motion sickness
- I do not have any sensory impairments

You will be given resting sessions between experimental conditions but please do not be afraid to ask for a break if required at any time during the study. Furthermore, if you feel unwell at any point, please let the experimenter know immediately so that medical attention can be given as soon as possible If you have any questions about the study or would like more information about our research in general, contact details are provided below.

CONTACT DETAILS

Name:	Patrizia Di Campli San Vito (PhD student/Experimenter)
Email:	p.di-campli-san-vito.1@research.gla.ac.uk
Office:	Room F141, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Supervisor:	Professor Stephen Brewster
Email:	stephen.brewster@glasgow.ac.uk
Office:	Room S131, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Tel:	+44 (0)141-330-4966

This study has been approved by the Ethics Committee (Reference Number - 300160073).

This research is part funded by the EPSRC and Jaguar Land Rover.

Version No. 2 (July 2018)

A.5.2 Experiment 4: Consent Form

HapNav – Investigating Thermal and Cutaneous Push Feedback on the Steering Wheel

CONSENT FORM

	firm that I have read and understood the Study Information Sheet provid	ed to me for the
above	e study and have had the opportunity to ask questions.	
The s partic	tudy has been explained to me, and I understand the explanation given a cipation will involve.	and what my
I unde instru only b	erstand that during the course of the study the experimenter will give ad actions concerning the experiment and that the purpose of some of the i be explained at the end of the study.	ditional nstructions will
l unde witho	erstand that my participation is voluntary and that I am free to withdraw out giving a reason.	at any time,
I unde that a guide	erstand that the research data may be accessed by researchers working at all times my personal data will be kept confidential in accordance with lines.	at the GIST, but data protection
	I have initialled the above boxes myself and I agree to take part in the	study.
		·
FULL NAME	:	
SIGNATURE		
DATE:		
I want to be	added to the participant's pool: YES NO	
EMAIL:		
L		
Experiment	er: Ms Patrizia Di Campli San Vito (p.di-campli-san-vito.1@research.gla.ac.uk)	

Version No. 1 (May 2018)

GLASGOW INTERACTIVE SYSTEMS GROUP

(stephen.brewster.glasgow.gla.ac.uk, 0141-330-4966)

This study has been approved by the Ethics Committee (Reference Number - 300160073).

Professor Stephen Brewster

University School of of Of Glasgow

This research is part funded by the EPSRC and Jaguar Land Rover.

DATE, SIGNATURE:

Supervisor details:

A.5.3 Experiment 4: Questionnaires

HapNav

BeforeQuestionnaire

Fill in before giving to participant!

1

* 1. Participant Number

HapNav	
Workload and Preference 1	
* 2. Navigation Type	
Cutaneous Push	
C Thermal	
* 3. Mental demand	
How much mental, visual and auditory activity was required? (e.g. thinking, deciding, calculating, listeni	ing, scanning, searching)
Low	High
0	
* 4. Physical demand	
How much physical activity was required? (e.g. pressing, controlling)	
Low	High
0	
* 5. Time pressure How much time pressure did you feel because of the rate at which things occurred or the time limit impo leisurely, rapid, frantic)	osed on the task? (e.g. slow,
Low	High
0	
* 6. Effort expended How hard did you work (mentally and physically) to accomplish your level of performance?	
	High
	- iigii
* 7. Performance level achieved	
How successful do you think you were in doing the task set by the experimenter? How satisfied were yo	ou with your performance?
Poor	Good
\bigcirc	

Low					High
\bigcirc					
). Annoyance exp low annoying did you	erienced find the system used in the	experiment?			
Low					High
\bigcirc					
10. Please rate the	e navigation type you j	ust finished for t	he following attrib	utes	
Pleasant	1 (not at all)	2	3	4	5 (very mucl
Comfortable	0	0	0	0	0
Disruptive	\bigcirc	0	0	0	0
Complicated	0	\bigcirc	0	0	

ad and Preference 2	
Navigation Type	
Cutaneous Push	
Thermal	
Mental demand	
much mental, visual and auditory activity was required? (e.g. thinking, deciding, calculating, listening	ı, scanning, searching)
Low	High
)	
Physical demand	
much physical activity was required? (e.g. pressing, controlling)	
Low	High
Low	Hign
LOW Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic)	Hign ed on the task? (e.g. slo
LOW Fime pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) LOW	Hign ed on the task? (e.g. slo High
Low Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) Low	Hign ed on the task? (e.g. slo High
Low Fime pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) Low	High High
Low Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) Low Effort expended	Hign ed on the task? (e.g. slo High
LOW Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) LOW Effort expended hard did you work (mentally and physically) to accomplish your level of performance?	High
LOW Fime pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) LOW Effort expended hard did you work (mentally and physically) to accomplish your level of performance? LOW	Hign ed on the task? (e.g. slo High
LOW Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) LOW Effort expended hard did you work (mentally and physically) to accomplish your level of performance? LOW	High
Low Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) Low Effort expended hard did you work (mentally and physically) to accomplish your level of performance? Low	High
LOW Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) LOW Effort expended hard did you work (mentally and physically) to accomplish your level of performance? LOW Performance level achieved	High
Low Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) Low Effort expended hard did you work (mentally and physically) to accomplish your level of performance? Low Performance level achieved successful do you think you were in doing the task set by the experimenter? How satisfied were you	High High With your performance?
Low Time pressure much time pressure did you feel because of the rate at which things occurred or the time limit impose rely, rapid, frantic) Low Effort expended hard did you work (mentally and physically) to accomplish your level of performance? Low Performance level achieved successful do you think you were in doing the task set by the experimenter? How satisfied were you Poor	High High With your performance? Good

Low					High
\bigcirc					
.9. Annoyance exp low annoving did vou	perienced find the system used in the	experiment?			
Low					High
0					
0. Please rate the	navigation type you j	ust finished for tl	he following attrib	utes	
Pleasant	1 (not at all)	2	3	4	5 (very much
Comfortable	\bigcirc	0	0	0	0
Disruptive		0	0	0	0
Complicated	\bigcirc	\bigcirc	0	\bigcirc	\bigcirc

uestionnaire End					
* 22. Age					
* 23. Gender					
	non-binary				
* 24. Occupation (and fiel	ld of study, if app	olicable)			
		,			
* 25. Vears of driving even	orioneo				
	enence				
* 26. Country, where the o	driving license w	as obtained			
* 26. Country, where the o	driving license w	vas obtained			
* 26. Country, where the o	driving license w	nant			
* 26. Country, where the o	driving license w	nant			
* 26. Country, where the o	driving license w ch hand is domir st experience wil	nant the following			
* 26. Country, where the o	driving license w ch hand is domir st experience wir 1 (none)	hant the following	3	4	5 (very much)
 * 26. Country, where the operation of the second second	driving license w ch hand is domin st experience wit 1 (none)	hant th the following 2	3	4	5 (very much)
 * 26. Country, where the operation of the second sec	driving license w ch hand is domin st experience wir 1 (none)	hant 2 2	3 ()	4	5 (very much)
 * 26. Country, where the operation of the second second	driving license w	ras obtained	3 () ()	4	5 (very much)
 * 26. Country, where the operation of the second state where the operation of the second state where the second state with a driving simulator Experience with thermal feedback * 29. Please rate how mu 	driving license w ch hand is domin st experience wit 1 (none)	ras obtained	3 O Navigation	4 () ()	5 (very much)
 * 26. Country, where the operation of the second second	driving license w ch hand is domin st experience wit 1 (none)	ras obtained	3 O Navigation	4 () () () () () () () () () () () () ()	5 (very much)
 * 26. Country, where the operation of the second state where the second state s	driving license w ch hand is domin st experience wit 1 (none)	ras obtained	3 O Inavigation	4 	5 (very much) Compared to the second

31. Please rate how much you like	d the thermal navigation	
not at all		very much
0		
32. Please name reasons for your	rating	
33. Which feedback type did you p	refer?	
Cutaneous Push		
Thermal		
34. Did associating warm with the α	lirection of turning feel appropriate	?
 34. Did associating warm with the organization of the second se	lirection of turning feel appropriate nation in a car that you would like t	? o receive through thermal feedback?
 34. Did associating warm with the organization of the provident of the organization of the provident of the organization of the organizat	lirection of turning feel appropriate nation in a car that you would like t	? o receive through thermal feedback?
34. Did associating warm with the o yes no 35. Is there any other kind of inform Why?	lirection of turning feel appropriate	? o receive through thermal feedback?
 34. Did associating warm with the original system of the original system of the original system. 35. Is there any other kind of inform Why? 36. Is there any other place in the original system. 	lirection of turning feel appropriate nation in a car that you would like t	? o receive through thermal feedback?
 34. Did associating warm with the one of the second seco	lirection of turning feel appropriate nation in a car that you would like t	? o receive through thermal feedback? e thermal feedback?
34. Did associating warm with the o yes no 35. Is there any other kind of inform Why? 36. Is there any other place in the o	lirection of turning feel appropriate nation in a car that you would like t	? o receive through thermal feedback? e thermal feedback?
 34. Did associating warm with the original system of the system	lirection of turning feel appropriate nation in a car that you would like t	? o receive through thermal feedback? e thermal feedback?
 34. Did associating warm with the original system of the product of the	lirection of turning feel appropriate nation in a car that you would like t ar where you would like to receive	? o receive through thermal feedback? e thermal feedback?
 34. Did associating warm with the original system of the system	lirection of turning feel appropriate nation in a car that you would like t ar where you would like to receive	? o receive through thermal feedback? e thermal feedback?
 34. Did associating warm with the original system of the system	lirection of turning feel appropriate	? o receive through thermal feedback? e thermal feedback?

A.6 Experiment 5

A.6.1 Experiment 5: Information Sheet

VibraTherm – Investigating Thermal and Vibrotactile Feedback on the Steering Wheel



INFORMATION SHEET

BACKGROUND

The goal of this user study is to improve safety in driving situations by providing warning cues through thermal and/or vibrotactile feedback on the steering wheel.

TASK

The study is conducted in a driving simulator and will consist of two major blocks. You will be asked to rate warnings, given either through thermal feedback, warm or cold, or vibration or a combination of thermal and vibrotactile feedback (bimodal). You will perform the tasks sitting in a gaming racing chair. There will be a steering wheel securely attached in front of you and you will be holding the wheel throughout the experiment. Please position your hands on the indicated location of the steering wheel and keep them on the devices during the rating and driving phases of the experiment. You will be wearing headphones throughout the experiment, either playing white noise or car sounds.

The first block of the experiment consists of subjectively rating presented stimuli. You will be presented with random stimuli. Please rate their urgency by moving the presented slider through turning of the steering wheel. When you have reached your perceived level of urgency, please select it by pressing the one of the two paddles on the steering wheel. There will be 45 stimuli in total. Please let the experimenter know, when you want to take a rest.

The second major block consist of a driving task. The simulator requires you to keep a virtual vehicle within the limits of the road. The car will be placed on a five-lane motorway. You will start in the middle lane and should swiftly and precisely change to another lane, when it is indicated by an arrow on one of the bridges. Please keep then within that lane, until another arrow appears. In between these lane changes you will be presented with urgency warnings, which you should rate swiftly by pressing one of three buttons on the steering wheel. The upper button (Y) should be pressed for warnings of high urgency, the middle button (B) for warnings of medium urgency and the lowest button (A) for warnings of low urgency. There will be a training phase before the main driving task begins. The driving task itself will be separated into five blocks of approximately 5 min each. You can break between these blocks, but please let the experimenter know, if you need to rest within a block.

INSTRUCTIONS

- Please place both hands at the indicated spaces on the steering wheel while driving.
- Please try to follow the instructions given by the arrows on the bridges to the best of your ability and try to keep inside the lane.

MISCELLANEOUS

- You will be wearing headphones throughout the experiment.

Version No. 1 (March 2018)

- The experiment is divided into blocks. You can rest between the blocks for as long as you need. If additional breaks are required, please let the experimenter know.
- You can withdraw from the study at any time without penalty. The data recorded until then will be destroyed.

IMPORTANT

The systems you interact with will not ask or store personal information and all data recorded are anonymised. The study will take approximately 60 minutes to complete and £6 will be paid at the end. If you are a student, no course credits will be awarded for completing this study. To take part in this study, the following criteria must be met:

- I am at least 18 years old
- I hold a full driving licence
- I have no visual impairments (not including wearing glasses or contact lenses)
- I have no neurological disorders (e.g. epilepsy)
- I do not suffer from motion sickness
- I do not have any sensory impairments

You will be given resting sessions between experimental conditions but please do not be afraid to ask for a break if required at any time during the study. Furthermore, if you feel unwell at any point, please let the experimenter know immediately so that medical attention can be given as soon as possible. Should you have any questions about the study or like more information about our research in general, contact details are provided below.

CONTACT DETAILS

Name:	Patrizia Di Campli San Vito (PhD student/Experimenter)
Email:	p.di-campli-san-vito.1@research.gla.ac.uk
Office:	Room F141, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Supervisor:	Professor Stephen Brewster
Email:	stephen.brewster@glasgow.ac.uk
Office:	Room S131, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Tel:	+44 (0)141-330-4966

This study has been approved by the Ethics Committee (Reference Number - 300160073).

This research is part funded by the EPSRC and Jaguar Land Rover.

Version No. 1 (March 2018)

A.6.2 Experiment 5: Consent Form

VibraTherm – Investigating Thermal and Vibrotactile Feedback on the Steering Wheel

CONSENT FORM

l con abov	firm that I have read and understood the Study Information Sheet provided to me for the re study and have had the opportunity to ask questions.
The s	study has been explained to me, and I understand the explanation given and what my cipation will involve.
l und instr only	lerstand that during the course of the study the experimenter will give additional uctions concerning the experiment and that the purpose of some of the instructions will be explained at the end of the study.
l und with	lerstand that my participation is voluntary and that I am free to withdraw at any time, out giving a reason.
l und that guide	lerstand that the research data may be accessed by researchers working at the GIST, but at all times my personal data will be kept confidential in accordance with data protection elines.
	I have initialled the above boxes myself and I agree to take part in the study.
FULL NAME	E:
SIGNATURE	E:
DATE:	
I want to b	e added to the participant's pool: YES NO
EMAIL:	

Experimenter:	Ms Patrizia Di Campli San Vito (p.di-campli-san-vito.1@research.gla.ac.uk)
DATE, SIGNATURE:	
Supervisor details:	Professor Stephen Brewster (stephen.brewster.glasgow.gla.ac.uk, 0141-330-4966)
This study has been ap	proved by the Ethics Committee (Reference Number - 300160073).

This research is part funded by the EPSRC and Jaguar Land Rover.



Version No. 1 (March 2018)
A.6.3 Experiment 5: Questionnaires

Vib	VibraTherm						
Qu	QuestionnaireVibraTherm						
*	* 1. Participant Number						
*	* 2. Age						
*	3. Gender						
	male female r	ion-binary					
*	4. Occupation (and field	of study, if app	licable)				
*	5. Years of driving expe	ience					
*	* 6. Country, where the driving license was obtained						
* *	 * 7. Please indicate which hand is dominant left right both * 8. Please rate how much experience you had with the following prior to the experiment: 						
	Everyone with a	1 (none)	2	3	4	5 (very much)	
	driving simulator	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
	Experience with vibrotactile feedback	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	
	Experience with thermal feedback	\bigcirc	\bigcirc	\bigcirc	\bigcirc	0	

	Thermal feedback
	Vibrotactile feedback
	Bimodal feedback
10. Ple	ase name reasons for your rating
11. Wh	ich feedback type felt most urgent ?
) the	mal 🔷 vibrotactile 📄 bimodal (thermal and vibrotactile together) 📄 all equally
12. Wh	ich feedback type felt most comfortable ?
the	mal 🔿 vibrotactile 🔿 bimodal (thermal and vibrotactile together) 🔵 all equally
13. Wh	ich direction of temperature felt more urgent ?
war	m 🔵 cold 🔵 both equally
14. Wh	ich direction of temperature felt more comfortable ?
war	m 🔵 cold 🔵 both equally
15. Any	/ comments?

A.6.4 Experiment 5: Statistics of Subjective Ratings with 0 Ratings Included

Experiment 5: Statistics of Independent Variables

Urgency	Modality	Urgency: Modality
F(2,27) = 23.35	F(4,61) = 135.13	F(8, 127) = 2.20
<i>p</i> < 0.0001*	<i>p</i> < 0.001*	p = 0.003*

Table A.1: Experiment 5: Statistics for the Subjective Ratings including 0 Ratings; significant results are bold and marked with an asterisk.

Experiment 5: Statistics of *post hoc* tests for Urgency for Subjective Ratings with *0* ratings included

Urgency - Urgency	t(27)	р
Low - Medium	3.33	0.007*
Low - Urgent	6.05	< 0.0001*
Medium - Urgent	2.38	0.06

Table A.2: Experiment 5: Statistics of *post hoc* test results for Urgency for Subjective Ratings with 0; significant results are bold and marked with an asterisk.

Experiment 5: Statistics of *post hoc* tests for Modality for Subjective Ratings with *0* ratings included

Modality - Modality	t(61)	р
Bimodal Cold - Bimodal Warm	1.30	0.69
Bimodal Cold - Thermal Cold	14.79	< 0.0001*
Bimodal Cold - Thermal Warm	13.70	< 0.0001*
Bimodal Cold - Vibration	0.66	0.96
Bimodal Warm - Thermal Cold	15.78	< 0.0001*
Bimodal Warm - Thermal Warm	14.69	< 0.0001*
Bimodal Warm - Vibration	0.65	0.97
Thermal Cold - Thermal Warm	0.80	0.93
Thermal Cold - Vibration	15.45	< 0.0001*
Thermal Warm - Vibration	14.35	< 0.0001*

Table A.3: Experiment 5: Statistics of *post hoc* test results for Modality for Subjective Ratings with 0; significant results are bold and marked with an asterisk.

Urgency, Modality - Urgency, Modality	t	p
Low,Bimodal Cold - Medium,Bimodal Cold	t(134) = 1.61	0.95
Low,Bimodal Cold - Urgent,Bimodal Cold	t(139) = 1.71	0.93
Low,Bimodal Cold - Low,Bimodal Warm	t(132) = 0.21	1.00
Low,Bimodal Cold - Medium,Bimodal Warm	t(140) = 2.51	0.44
Low,Bimodal Cold - Urgent,Bimodal Warm	t(139) = 2.79	0.26
Low,Bimodal Cold - Low,Thermal Cold	t(135) = 12.08	< 0.0001*
Low,Bimodal Cold - Medium,Thermal Cold	t(142) = 10.10	< 0.0001*
Low,Bimodal Cold - Urgent,Thermal Cold	t(141) = 9.24	< 0.0001*
Low,Bimodal Cold - Low,Thermal Warm	t(132) = 12.05	< 0.0001*
Low,Bimodal Cold - Medium,Thermal Warm	t(139) = 10.45	< 0.0001*
Low,Bimodal Cold - Urgent,Thermal Warm	t(138) = 6.78	< 0.0001*
Low,Bimodal Cold - Low,Vibration	t(134) = 0.46	1.00
Low,Bimodal Cold - Medium,Vibration	t(140) = 1.68	0.94
Low, Bimodal Cold - Urgent, Vibration	t(139) = 1.95	0.83
Medium, Bimodal Cold - Urgent, Bimodal Cold	t(130) = 0.07	1.00
Medium,Bimodal Cold - Low,Bimodal Warm	t(139) = 1.04	1.00
Medium,Bimodal Cold - Medium,Bimodal Warm	t(131) = 1.35	0.99
Medium,Bimodal Cold - Urgent,Bimodal Warm	t(138) = 1.53	0.97
Medium,Bimodal Cold - Low,Thermal Cold	t(140) = 1.53	< 0.0001*
Medium,Bimodal Cold - Medium,Thermal Cold	t(135) = 11.90	< 0.0001*
Medium, Bimodal Cold - Urgent, Thermal Cold	t(140) = 10.39	< 0.0001*
Medium,Bimodal Cold - Low,Thermal Warm	t(138) = 12.79	< 0.0001*
Medium,Bimodal Cold - Medium,Thermal Warm	t(130) = 12.28	< 0.0001*
Medium,Bimodal Cold - Urgent,Thermal Warm	t(138) = 7.93	< 0.0001*
Medium, Bimodal Cold - Low, Vibration	t(140) = 0.81	1.00
Medium, Bimodal Cold - Medium, Vibration	t(130) = 0.46	1.00
Medium, Bimodal Cold - Urgent, Vibration	t(134) = 0.68	1.00

Experiment 5: Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Subjective Ratings with *0* ratings included

Table A.4: Experiment 5: Part 1 Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Subjective Ratings with 0 ratings included; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t	р
Urgent,Bimodal Cold - Low,Bimodal Warm	t(134) = 1.10	1.00
Urgent,Bimodal Cold - Medium,Bimodal Warm	t(134) = 1.23	1.00
Urgent,Bimodal Cold - Urgent,Bimodal Warm	t(129) = 1.56	0.96
Urgent, Bimodal Cold - Low, Thermal Cold	t(140) = 12.99	< 0.0001*
Urgent, Bimodal Cold - Medium, Thermal Cold	t(141) = 11.38	< 0.0001*
Urgent, Bimodal Cold - Urgent, Thermal Cold	t(133) = 11.00	< 0.0001*
Urgent,Bimodal Cold - Low,Thermal Warm	t(138) = 12.96	< 0.0001*
Urgent, Bimodal Cold - Medium, Thermal Warm	t(139) = 11.70	< 0.0001*
Urgent, Bimodal Cold - Urgent, Thermal Warm	t(129) = 8.42	< 0.0001*
Urgent, Bimodal Cold - Low, Vibration	t(139) = 0.87	1.00
Urgent, Bimodal Cold - Medium, Vibration	t(139) = 0.38	1.00
Urgent, Bimodal Cold - Urgent, Vibration	t(130) = 0.66	1.00
Low,Bimodal Warm - Medium,Bimodal Warm	t(136) = 2.97	0.18
Low,Bimodal Warm - Urgent,Bimodal Warm	t(139) = 3.39	0.06
Low,Bimodal Warm - Low,Thermal Cold	t(133) = 12.11	< 0.0001*
Low,Bimodal Warm - Medium,Thermal Cold	t(140) = 10.19	< 0.0001*
Low,Bimodal Warm - Urgent,Thermal Cold	t(139) = 9.32	< 0.0001*
Low,Bimodal Warm - Low,Thermal Warm	t(130) = 12.08	< 0.0001*
Low,Bimodal Warm - Medium,Thermal Warm	t(137) = 10.53	< 0.0001*
Low,Bimodal Warm - Urgent,Thermal Warm	t(136) = 6.88	< 0.0001*
Low,Bimodal Warm - Low,Vibration	t(133) = 0.25	1.00
Low,Bimodal Warm - Medium,Vibration	t(139) = 1.47	0.98
Low,Bimodal Warm - Urgent,Vibration	t(139) = 1.73	0.92
Medium,Bimodal Warm - Urgent,Bimodal Warm	t(132) = 0.32	1.00
Medium, Bimodal Warm - Low, Thermal Cold	t(139) = 13.92	< 0.0001*
Medium,Bimodal Warm - Medium,Thermal Cold	t(134) = 12.99	< 0.0001*
Medium,Bimodal Warm - Urgent,Thermal Cold	t(139) = 11.51	< 0.0001*
Medium,Bimodal Warm - Low,Thermal Warm	t(137) = 13.87	< 0.0001*
Medium,Bimodal Warm - Medium,Thermal Warm	t(130) = 13.35	< 0.0001*
Medium,Bimodal Warm - Urgent,Thermal Warm	t(137) = 9.07	< 0.0001*
Medium, Bimodal Warm - Low, Vibration	t(140) = 2.07	0.75
Medium, Bimodal Warm - Medium, Vibration	t(131) = 0.90	1.00
Medium, Bimodal Warm - Urgent, Vibration	t(139) = 0.61	1.00
Urgent,Bimodal Warm - Low,Thermal Cold	t(139) = 14.29	< 0.0001*
Urgent, Bimodal Warm - Medium, Thermal Cold	t(139) = 12.71	< 0.0001*
Urgent,Bimodal Warm - Urgent,Thermal Cold	t(131) = 12.38	< 0.0001*
Urgent,Bimodal Warm - Low,Thermal Warm	t(136) = 14.24	< 0.0001*
Urgent,Bimodal Warm - Medium,Thermal Warm	t(137) = 13.01	< 0.0001*
Urgent,Bimodal Warm - Urgent,Thermal Warm	t(127) = 9.83	< 0.0001*
Urgent,Bimodal Warm - Low,Vibration	t(138) = 2.35	0.55
Urgent,Bimodal Warm - Medium,Vibration	t(139) = 1.10	1.00
Urgent,Bimodal Warm - Urgent,Vibration	t(130) = 0.90	1.00

Table A.5: Experiment 5: Part 2 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Subjective Ratings with θ ratings included; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t(229)	р
Low, Thermal Cold - Medium, Thermal Cold	t(137) = 1.79	0.90
Low, Thermal Cold - Urgent, Thermal Cold	t(140) = 3.12	0.13
Low, Thermal Cold - Low, Thermal Warm	t(133) = 0.12	1.00
Low, Thermal Cold - Medium, Thermal Warm	t(140) = 0.96	1.00
Low, Thermal Cold - Urgent, Thermal Warm	t(139) = 4.77	0.0004*
Low, Thermal Cold - Low, Vibration	t(134) = 12.54	< 0.0001*
Low, Thermal Cold - Medium, Vibration	t(140) = 13.25	< 0.0001*
Low, Thermal Cold - Urgent, Vibration	t(140) = 13.62	< 0.0001*
Medium, Thermal Cold - Urgent, Thermal Cold	t(135) = 1.23	1.00
Medium, Thermal Cold - Low, Thermal Warm	t(141) = 1.51	0.97
Medium, Thermal Cold - Medium, Thermal Warm	t(135) = 0.45	1.00
Medium, Thermal Cold - Urgent, Thermal Warm	t(139) = 3.31	0.08
Medium, Thermal Cold - Low, Vibration	t(143) = 10.54	< 0.0001*
Medium, Thermal Cold - Medium, Vibration	t(135) = 12.35	< 0.0001*
Medium, Thermal Cold - Urgent, Vibration	t(141) = 12.00	< 0.0001*
Urgent, Thermal Cold - Low, Thermal Warm	t(140) = 2.49	0.45
Urgent, Thermal Cold - Medium, Thermal Warm	t(139) = 1.39	0.99
Urgent, Thermal Cold - Urgent, Thermal Warm	t(131) = 2.48	0.46
Urgent, Thermal Cold - Low, Vibration	t(141) = 9.68	< 0.0001*
Urgent, Thermal Cold - Medium, Vibration	t(140) = 10.83	< 0.0001*
Urgent, Thermal Cold - Urgent, Vibration	t(133) = 11.66	< 0.0001*
Low, Thermal Warm - Medium, Thermal Warm	t(137) = 1.38	0.99
Low, Thermal Warm - Urgent, Thermal Warm	t(139) = 6.42	< 0.0001*
Low, Thermal Warm - Low, Vibration	t(132) = 12.51	< 0.0001*
Low, Thermal Warm - Medium, Vibration	t(138) = 13.21	< 0.0001*
Low, Thermal Warm - Urgent, Vibration	t(138) = 13.58	< 0.0001*
Medium, Thermal Warm - Urgent, Thermal Warm	t(132) = 4.89	0.0003*
Medium, Thermal Warm - Low, Vibration	t(139) = 10.89	< 0.0001*
Medium, Thermal Warm - Medium, Vibration	t(130) = 12.72	< 0.0001*
Medium, Thermal Warm - Urgent, Vibration	t(139) = 12.31	< 0.0001*
Urgent, Thermal Warm - Low, Vibration	t(138) = 7.21	< 0.0001*
Urgent, Thermal Warm - Medium, Vibration	t(138) = 8.35	< 0.0001*
Urgent, Thermal Warm - Urgent, Vibration	t(129) = 9.08	< 0.0001*
Low, Vibration - Medium, Vibration	t(134) = 1.59	0.96
Low, Vibration - Urgent, Vibration	t(139) = 1.94	0.83
Medium, Vibration - Urgent, Vibration	t(130) = 0.32	1.00

Table A.6: Experiment 5: Part 3 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Subjective Ratings with 0 ratings included; significant results are bold and marked with an asterisk.

A.6.5 Experiment 5: Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Subjective Ratings without 0 ratings; significant results and discussion can be found in section 5.2.6

Urgency,Modality - Urgency,Modality	t	p
Low,Bimodal Cold - Medium,Bimodal Cold	t(138) = 0.60	1.00
Low,Bimodal Cold - Urgent,Bimodal Cold	t(140) = 0.81	1.00
Low,Bimodal Cold - Low,Bimodal Warm	t(125) = 0.24	1.00
Low,Bimodal Cold - Medium,Bimodal Warm	t(136) = 1.79	0.90
Low,Bimodal Cold - Urgent,Bimodal Warm	t(133) = 2.16	0.69
Low, Bimodal Cold - Low, Thermal Cold	t(127) = 11.28	< 0.0001*
Low,Bimodal Cold - Medium,Thermal Cold	t(129) = 9.34	< 0.0001*
Low,Bimodal Cold - Urgent,Thermal Cold	t(124) = 8.62	< 0.0001*
Low,Bimodal Cold - Low,Thermal Warm	t(124) = 9.48	< 0.0001*
Low,Bimodal Cold - Medium,Thermal Warm	t(129) = 9.72	< 0.0001*
Low,Bimodal Cold - Urgent,Thermal Warm	t(125) = 5.90	< 0.0001*
Low,Bimodal Cold - Low,Vibration	t(127) = 0.52	1.00
Low, Bimodal Cold - Medium, Vibration	t(135) = 0.86	1.00
Low,Bimodal Cold - Urgent,Vibration	t(132) = 1.22	1.00
Medium,Bimodal Cold - Urgent,Bimodal Cold	t(139) = 0.20	1.00
Medium,Bimodal Cold - Low,Bimodal Warm	t(135) = 0.22	1.00
Medium,Bimodal Cold - Medium,Bimodal Warm	t(124) = 1.38	0.99
Medium,Bimodal Cold - Urgent,Bimodal Warm	t(131) = 1.71	0.93
Medium, Bimodal Cold - Low, Thermal Cold	t(129) = 11.65	< 0.0001*
Medium, Bimodal Cold - Medium, Thermal Cold	t(124) = 9.91	< 0.0001*
Medium,Bimodal Cold - Urgent,Thermal Cold	t(125) = 9.03	< 0.0001*
Medium,Bimodal Cold - Low,Thermal Warm	t(128) = 9.81	< 0.0001*
Medium,Bimodal Cold - Medium,Thermal Warm	t(124) = 10.28	< 0.0001*
Medium, Bimodal Cold - Urgent, Thermal Warm	t(123) = 6.36	< 0.0001*
Medium, Bimodal Cold - Low, Vibration	t(135) = 0.05	1.00
Medium, Bimodal Cold - Medium, Vibration	t(123) = 0.43	1.00
Medium, Bimodal Cold - Urgent, Vibration	t(133) = 0.75	1.00

Table A.7: Experiment 5: Part 1 Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Subjective Ratings without 0 ratings included; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t	p
Urgent,Bimodal Cold - Low,Bimodal Warm	t(132) = 0.36	1.00
Urgent,Bimodal Cold - Medium,Bimodal Warm	t(132) = 1.21	1.00
Urgent,Bimodal Cold - Urgent,Bimodal Warm	t(123) = 1.61	0.95
Urgent,Bimodal Cold - Low,Thermal Cold	t(128) = 11.86	< 0.0001*
Urgent,Bimodal Cold - Medium,Thermal Cold	t(128) = 9.98	< 0.0001*
Urgent, Bimodal Cold - Urgent, Thermal Cold	t(120) = 9.29	< 0.0001*
Urgent,Bimodal Cold - Low,Thermal Warm	t(127) = 10.00	< 0.0001*
Urgent,Bimodal Cold - Medium,Thermal Warm	t(128) = 10.35	< 0.0001*
Urgent, Bimodal Cold - Urgent, Thermal Warm	t(118) = 6.59	< 0.0001*
Urgent, Bimodal Cold - Low, Vibration	t(132) = 0.09	1.00
Urgent, Bimodal Cold - Medium, Vibration	t(132) = 0.27	1.00
Urgent, Bimodal Cold - Urgent, Vibration	t(123) = 0.63	1.00
Low,Bimodal Warm - Medium,Bimodal Warm	t(138) = 2.05	0.77
Low,Bimodal Warm - Urgent,Bimodal Warm	t(140) = 2.59	0.38
Low,Bimodal Warm - Low,Thermal Cold	t(127) = 11.48	< 0.0001*
Low,Bimodal Warm - Medium,Thermal Cold	t(128) = 9.56	< 0.0001*
Low,Bimodal Warm - Urgent,Thermal Cold	t(125) = 8.81	< 0.0001*
Low,Bimodal Warm - Low,Thermal Warm	t(123) = 9.66	< 0.0001*
Low,Bimodal Warm - Medium,Thermal Warm	t(128) = 9.90	< 0.0001*
Low,Bimodal Warm - Urgent,Thermal Warm	t(125) = 6.10	< 0.0001*
Low,Bimodal Warm - Low,Vibration	t(126) = 0.28	1.00
Low,Bimodal Warm - Medium,Vibration	t(134) = 0.62	1.00
Low,Bimodal Warm - Urgent,Vibration	t(132) = 0.97	1.00
Medium,Bimodal Warm - Urgent,Bimodal Warm	t(140) = 0.47	1.00
Medium, Bimodal Warm - Low, Thermal Cold	t(130) = 12.86	< 0.0001*
Medium,Bimodal Warm - Medium,Thermal Cold	t(125) = 11.11	< 0.0001*
Medium,Bimodal Warm - Urgent,Thermal Cold	t(125) = 10.29	< 0.0001*
Medium,Bimodal Warm - Low,Thermal Warm	t(128) = 11.02	< 0.0001*
Medium,Bimodal Warm - Medium,Thermal Warm	t(124) = 11.48	< 0.0001*
Medium,Bimodal Warm - Urgent,Thermal Warm	t(124) = 7.59	< 0.0001*
Medium, Bimodal Warm - Low, Vibration	t(136) = 1.28	0.99
Medium, Bimodal Warm - Medium, Vibration	t(124) = 0.97	1.00
Medium, Bimodal Warm - Urgent, Vibration	t(133) = 0.60	1.00
Urgent,Bimodal Warm - Low,Thermal Cold	t(128) = 13.28	< 0.0001*
Urgent, Bimodal Warm - Medium, Thermal Cold	t(127) = 11.43	< 0.0001*
Urgent,Bimodal Warm - Urgent,Thermal Cold	t(120) = 10.74	< 0.0001*
Urgent,Bimodal Warm - Low,Thermal Warm	t(127) = 11.41	< 0.0001*
Urgent,Bimodal Warm - Medium,Thermal Warm	t(127) = 11.76	< 0.0001*
Urgent,Bimodal Warm - Urgent,Thermal Warm	t(118) = 8.06	< 0.0001*
Urgent,Bimodal Warm - Low,Vibration	t(132) = 1.65	0.94
Urgent, Bimodal Warm - Medium, Vibration	t(132) = 1.30	0.99
Urgent,Bimodal Warm - Urgent,Vibration	t(124) = 0.98	1.00

Table A.8: Experiment 5: Part 2 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Subjective Ratings without 0 ratings included; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t	р
Low, Thermal Cold - Medium, Thermal Cold	t(143) = 2.38	0.54
Low, Thermal Cold - Urgent, Thermal Cold	t(143) = 3.67	0.03*
Low, Thermal Cold - Low, Thermal Warm	t(125) = 1.81	0.89
Low, Thermal Cold - Medium, Thermal Warm	t(124) = 1.47	0.98
Low, Thermal Cold - Urgent, Thermal Warm	t(122) = 5.19	0.0001*
Low, Thermal Cold - Low, Vibration	t(127) = 11.76	< 0.0001*
Low, Thermal Cold - Medium, Vibration	t(129) = 12.05	< 0.0001*
Low, Thermal Cold - Urgent, Vibration	t(128) = 12.43	< 0.0001*
Medium, Thermal Cold - Urgent, Thermal Cold	t(142) = 1.18	1.00
Medium, Thermal Cold - Low, Thermal Warm	t(122) = 0.04	1.00
Medium, Thermal Cold - Medium, Thermal Warm	t(124) = 0.31	1.00
Medium, Thermal Cold - Urgent, Thermal Warm	t(121) = 3.40	0.06
Medium, Thermal Cold - Low, Vibration	t(129) = 9.84	< 0.0001*
Medium, Thermal Cold - Medium, Vibration	t(124) = 10.30	< 0.0001*
Medium, Thermal Cold - Urgent, Vibration	t(128) = 10.54	< 0.0001*
Urgent, Thermal Cold - Low, Thermal Warm	t(121) = 0.79	1.00
Urgent, Thermal Cold - Medium, Thermal Warm	t(121) = 1.14	1.00
Urgent, Thermal Cold - Urgent, Thermal Warm	t(116) = 2.60	0.38
Urgent, Thermal Cold - Low, Vibration	t(125) = 9.09	< 0.0001*
Urgent, Thermal Cold - Medium, Vibration	t(125) = 9.43	< 0.0001*
Urgent, Thermal Cold - Urgent, Vibration	t(120) = 9.88	< 0.0001*
Low, Thermal Warm - Medium, Thermal Warm	t(142) = 0.47	1.00
Low, Thermal Warm - Urgent, Thermal Warm	t(142) = 4.88	0.0003*
Low, Thermal Warm - Low, Vibration	t(125) = 10.09	< 0.0001*
Low, Thermal Warm - Medium, Vibration	t(130) = 10.27	< 0.0001*
Low, Thermal Warm - Urgent, Vibration	t(129) = 10.65	< 0.0001*
Medium, Thermal Warm - Urgent, Thermal Warm	t(142) = 5.32	< 0.0001*
Medium, Thermal Warm - Low, Vibration	t(132) = 10.25	< 0.0001*
Medium, Thermal Warm - Medium, Vibration	t(125) = 10.78	< 0.0001*
Medium, Thermal Warm - Urgent, Vibration	t(129) = 11.03	< 0.0001*
Urgent, Thermal Warm - Low, Vibration	t(127) = 6.44	< 0.0001*
Urgent, Thermal Warm - Medium, Vibration	t(125) = 6.81	< 0.0001*
Urgent, Thermal Warm - Urgent, Vibration	t(120) = 7.24	< 0.0001*
Low, Vibration - Medium, Vibration	t(138) = 0.47	1.00
Low, Vibration - Urgent, Vibration	t(140) = 0.94	1.00
Medium, Vibration - Urgent, Vibration	t(139) = 0.47	1.00

Table A.9: Experiment 5: Part 3 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Subjective Ratings without 0 ratings included; significant results are bold and marked with an asterisk.

A.6.6 Experiment 5: Statistics of *post hoc* tests for the Interaction between Urgency and Modality for for Missed Stimuli (*0* Ratings) of the Subjective Rating; significant results and discussion can be found in section 5.2.6

Urgency,Modality - Urgency,Modality	t	p
Low,Bimodal Cold - Medium,Bimodal Cold	t(170) = 0.00	1.00
Low,Bimodal Cold - Urgent,Bimodal Cold	t(170) = 0.00	1.00
Low,Bimodal Cold - Low,Bimodal Warm	t(127) = 0.28	1.00
Low,Bimodal Cold - Medium,Bimodal Warm	t(128) = 0.00	1.00
Low,Bimodal Cold - Urgent,Bimodal Warm	t(128) = 0.00	1.00
Low,Bimodal Cold - Low,Thermal Cold	t(127) = 6.36	<0.0001*
Low,Bimodal Cold - Medium,Thermal Cold	t(128) = 4.44	0.002*
Low,Bimodal Cold - Urgent,Thermal Cold	t(128) = 3.61	0.03*
Low,Bimodal Cold - Low,Thermal Warm	t(127) = 6.36	<0.0001*
Low,Bimodal Cold - Medium,Thermal Warm	t(128) = 6.66	<0.0001*
Low,Bimodal Cold - Urgent,Thermal Warm	t(128) = 2.22	0.65
Low,Bimodal Cold - Low,Vibration	t(127) = 0.00	1.00
Low,Bimodal Cold - Medium,Vibration	t(128) = 0.00	1.00
Low,Bimodal Cold - Urgent,Vibration	t(128) = 0.00	1.00
Medium,Bimodal Cold - Urgent,Bimodal Cold	t(170) = 0.00	1.00
Medium,Bimodal Cold - Low,Bimodal Warm	t(128) = 0.28	1.00
Medium,Bimodal Cold - Medium,Bimodal Warm	t(127) = 0.00	1.00
Medium,Bimodal Cold - Urgent,Bimodal Warm	t(128) = 0.00	1.00
Medium, Bimodal Cold - Low, Thermal Cold	t(128) = 6.38	<0.0001*
Medium, Bimodal Cold - Medium, Thermal Cold	t(127) = 4.43	0.002*
Medium, Bimodal Cold - Urgent, Thermal Cold	t(128) = 3.61	0.03*
Medium,Bimodal Cold - Low,Thermal Warm	t(128) = 6.38	<0.0001*
Medium,Bimodal Cold - Medium,Thermal Warm	t(127) = 6.64	<0.0001*
Medium,Bimodal Cold - Urgent,Thermal Warm	t(128) = 2.22	0.65
Medium, Bimodal Cold - Low, Vibration	t(128) = 0.00	1.00
Medium, Bimodal Cold - Medium, Vibration	t(127) = 0.00	1.00
Medium, Bimodal Cold - Urgent, Vibration	t(128) = 0.00	1.00

Table A.10: Experiment 5: Part 1 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Missed Stimuli (*0* Ratings) of the Subjective Rating; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t	р
Urgent,Bimodal Cold - Low,Bimodal Warm	t(128) = 0.28	1.00
Urgent,Bimodal Cold - Medium,Bimodal Warm	t(128) = 0.00	1.00
Urgent, Bimodal Cold - Urgent, Bimodal Warm	t(127) = 0.00	1.00
Urgent, Bimodal Cold - Low, Thermal Cold	t(128) = 6.38	<0.0001*
Urgent, Bimodal Cold - Medium, Thermal Cold	t(128) = 4.44	0.002*
Urgent, Bimodal Cold - Urgent, Thermal Cold	t(127) = 3.60	0.03*
Urgent, Bimodal Cold - Low, Thermal Warm	t(128) = 6.38	<0.0001*
Urgent, Bimodal Cold - Medium, Thermal Warm	t(128) = 6.66	<0.0001*
Urgent, Bimodal Cold - Urgent, Thermal Warm	t(127) = 2.21	0.66
Urgent, Bimodal Cold - Low, Vibration	t(128) = 0.00	1.00
Urgent, Bimodal Cold - Medium, Vibration	t(128) = 0.00	1.00
Urgent, Bimodal Cold - Urgent, Vibration	t(127) = 0.00	1.00
Low,Bimodal Warm - Medium,Bimodal Warm	t(170) = 0.42	1.00
Low,Bimodal Warm - Urgent,Bimodal Warm	t(170) = 0.42	1.00
Low,Bimodal Warm - Low,Thermal Cold	t(128) = 6.10	<0.0001*
Low,Bimodal Warm - Medium,Thermal Cold	t(128) = 4.16	0.005*
Low,Bimodal Warm - Urgent,Thermal Cold	t(128) = 3.33	0.07
Low,Bimodal Warm - Low,Thermal Warm	t(127) = 6.10	<0.0001*
Low,Bimodal Warm - Medium,Thermal Warm	t(128) = 6.38	<0.0001*
Low,Bimodal Warm - Urgent,Thermal Warm	t(128) = 1.94	0.83
Low,Bimodal Warm - Low,Vibration	t(127) = 0.28	1.00
Low,Bimodal Warm - Medium,Vibration	t(128) = 0.28	1.00
Low,Bimodal Warm - Urgent,Vibration	t(128) = 0.28	1.00
Medium,Bimodal Warm - Urgent,Bimodal Warm	t(170) = 0.00	1.00
Medium,Bimodal Warm - Low,Thermal Cold	t(128) = 6.38	<0.0001*
Medium, Bimodal Warm - Medium, Thermal Cold	t(127) = 4.43	0.002*
Medium,Bimodal Warm - Urgent,Thermal Cold	t(128) = 3.61	0.03*
Medium,Bimodal Warm - Low,Thermal Warm	t(128) = 6.38	<0.0001*
Medium,Bimodal Warm - Medium,Thermal Warm	t(128) = 6.64	<0.0001*
Medium,Bimodal Warm - Urgent,Thermal Warm	t(128) = 2.22	0.65
Medium, Bimodal Warm - Low, Vibration	t(128) = 0.00	1.00
Medium, Bimodal Warm - Medium, Vibration	t(127) = 0.00	1.00
Medium, Bimodal Warm - Urgent, Vibration	t(128) = 0.00	1.00
Urgent,Bimodal Warm - Low,Thermal Cold	t(128) = 6.38	<0.0001*
Urgent, Bimodal Warm - Medium, Thermal Cold	t(128) = 4.44	0.002*
Urgent,Bimodal Warm - Urgent,Thermal Cold	t(127) = 3.60	0.03*
Urgent,Bimodal Warm - Low,Thermal Warm	t(128) = 6.38	<0.0001*
Urgent,Bimodal Warm - Medium,Thermal Warm	t(128) = 6.66	<0.0001*
Urgent,Bimodal Warm - Urgent,Thermal Warm	t(128) = 2.21	0.66
Urgent,Bimodal Warm - Low,Vibration	t(128) = 0.00	1.00
Urgent,Bimodal Warm - Medium,Vibration	t(128) = 0.00	1.00
Urgent,Bimodal Warm - Urgent,Vibration	t(127) = 0.00	1.00

Table A.11: Experiment 5: Part 2 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Missed Stimuli (*0* Ratings) of the Subjective Rating; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t	р
Low, Thermal Cold - Medium, Thermal Cold	t(170) = 2.90	0.21
Low, Thermal Cold - Urgent, Thermal Cold	t(170) = 4.15	0.005*
Low, Thermal Cold - Low, Thermal Warm	t(127) = 0.00	1.00
Low, Thermal Cold - Medium, Thermal Warm	t(128) = 0.28	1.00
Low, Thermal Cold - Urgent, Thermal Warm	t(128) = 4.16	0.005*
Low, Thermal Cold - Low, Vibration	t(127) = 6.36	<0.0001*
Low, Thermal Cold - Medium, Vibration	t(128) = 6.38	<0.0001*
Low, Thermal Cold - Urgent, Vibration	t(128) = 6.38	<0.0001*
Medium, Thermal Cold - Urgent, Thermal Cold	t(170) = 1.24	1.00
Medium, Thermal Cold - Low, Thermal Warm	t(128) = 1.94	0.83
Medium, Thermal Cold - Medium, Thermal Warm	t(127) = 2.21	0.66
Medium, Thermal Cold - Urgent, Thermal Warm	t(128) = 2.22	0.65
Medium, Thermal Cold - Low, Vibration	t(128) = 4.44	0.002*
Medium, Thermal Cold - Medium, Vibration	t(127) = 4.43	0.002*
Medium, Thermal Cold - Urgent, Vibration	t(128) = 4.44	0.002*
Urgent, Thermal Cold - Low, Thermal Warm	t(128) = 2.78	0.27
Urgent, Thermal Cold - Medium, Thermal Warm	t(128) = 3.05	0.15
Urgent, Thermal Cold - Urgent, Thermal Warm	t(127) = 1.38	0.99
Urgent, Thermal Cold - Low, Vibration	t(128) = 3.61	0.03*
Urgent, Thermal Cold - Medium, Vibration	t(128) = 3.61	0.03*
Urgent, Thermal Cold - Urgent, Vibration	t(127) = 3.60	0.03*
Low, Thermal Warm - Medium, Thermal Warm	t(170) = 0.42	1.00
Low, Thermal Warm - Urgent, Thermal Warm	t(170) = 6.22	<0.0001*
Low, Thermal Warm - Low, Vibration	t(127) = 6.36	<0.0001*
Low, Thermal Warm - Medium, Vibration	t(128) = 6.38	<0.0001*
Low, Thermal Warm - Urgent, Vibration	t(128) = 6.38	<0.0001*
Medium, Thermal Warm - Urgent, Thermal Warm	t(170) = 6.64	<0.0001*
Medium, Thermal Warm - Low, Vibration	t(128) = 6.66	<0.0001*
Medium, Thermal Warm - Medium, Vibration	t(127) = 6.64	<0.0001*
Medium, Thermal Warm - Urgent, Vibration	t(128) = 6.66	<0.0001*
Urgent, Thermal Warm - Low, Vibration	t(128) = 2.22	0.65
Urgent, Thermal Warm - Medium, Vibration	t(128) = 2.22	0.65
Urgent, Thermal Warm - Urgent, Vibration	t(127) = 2.21	0.66
Low, Vibration - Medium, Vibration	t(170) = 0.00	1.00
Low, Vibration - Urgent, Vibration	t(170) = 0.00	1.00
Medium, Vibration - Urgent, Vibration	t(170) = 0.00	1.00

Table A.12: Experiment 5: Part 3 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for for Missed Stimuli (*0* Ratings) of the Subjective Rating; significant results are bold and marked with an asterisk.

A.6.7 Experiment 5: Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Rate; significant results and discussion can be found in section 5.2.6

Urgency,Modality - Urgency,Modality	t	р
Low,Bimodal Cold - Medium,Bimodal Cold	t(166) = 0.42	1.00
Low,Bimodal Cold - Urgent,Bimodal Cold	t(166) = 1.66	0.94
Low,Bimodal Cold - Low,Bimodal Warm	t(179) = 1.12	1.00
Low,Bimodal Cold - Medium,Bimodal Warm	t(189) = 0.72	1.00
Low,Bimodal Cold - Urgent,Bimodal Warm	t(189) = 2.16	0.69
Low, Bimodal Cold - Low, Thermal Cold	t(179) = 2.60	0.38
Low,Bimodal Cold - Medium,Thermal Cold	t(189) = 3.60	0.03*
Low, Bimodal Cold - Urgent, Thermal Cold	t(189) = 5.87	0.0002*
Low,Bimodal Cold - Low,Thermal Warm	t(179) = 4.28	0.003*
Low,Bimodal Cold - Medium,Thermal Warm	t(189) = 4.32	0.002*
Low,Bimodal Cold - Urgent,Thermal Warm	t(189) = 4.69	0.0005*
Low, Bimodal Cold - Low, Vibration	t(179) = 1.12	1.00
Low,Bimodal Cold - Medium,Vibration	t(189) = 0.72	1.00
Low, Bimodal Cold - Urgent, Vibration	t(189) = 0.90	1.00
Medium, Bimodal Cold - Urgent, Bimodal Cold	t(166) = 2.08	0.75
Medium,Bimodal Cold - Low,Bimodal Warm	t(189) = 0.72	1.00
Medium,Bimodal Cold - Medium,Bimodal Warm	t(179) = 0.37	1.00
Medium,Bimodal Cold - Urgent,Bimodal Warm	t(189) = 2.52	0.43
Medium, Bimodal Cold - Low, Thermal Cold	t(189) = 2.16	0.69
Medium,Bimodal Cold - Medium,Thermal Cold	t(179) = 3.35	0.07
Medium, Bimodal Cold - Urgent, Thermal Cold	t(189) = 4.50	0.001*
Medium,Bimodal Cold - Low,Thermal Warm	t(189) = 3.78	0.02*
Medium,Bimodal Cold - Medium,Thermal Warm	t(179) = 4.09	0.006*
Medium,Bimodal Cold - Urgent,Thermal Warm	t(189) = 4.32	0.002*
Medium,Bimodal Cold - Low,Vibration	t(189) = 0.72	1.00
Medium, Bimodal Cold - Medium, Vibration	t(179) = 0.37	1.00
Medium, Bimodal Cold - Urgent, Vibration	t(189) = 1.26	1.00

Table A.13: Experiment 5: Part 1 Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Rate; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t(238)	p
Urgent,Bimodal Cold - Low,Bimodal Warm	t(189) = 2.52	0.43
Urgent,Bimodal Cold - Medium,Bimodal Warm	t(189) = 2.16	0.69
Urgent,Bimodal Cold - Urgent,Bimodal Warm	t(179) = 0.74	1.00
Urgent, Bimodal Cold - Low, Thermal Cold	t(189) = 3.96	0.009*
Urgent,Bimodal Cold - Medium,Thermal Cold	t(189) = 5.05	0.0001*
Urgent, Bimodal Cold - Urgent, Thermal Cold	t(179) = 6.51	<0.0001*
Urgent,Bimodal Cold - Low,Thermal Warm	t(189) = 5.59	<0.0001*
Urgent,Bimodal Cold - Medium,Thermal Warm	t(189) = 5.77	<0.0001*
Urgent, Bimodal Cold - Urgent, Thermal Warm	t(179) = 6.32	<0.0001*
Urgent, Bimodal Cold - Low, Vibration	t(189) = 2.52	0.43
Urgent, Bimodal Cold - Medium, Vibration	t(189) = 2.16	0.69
Urgent, Bimodal Cold - Urgent, Vibration	t(179) = 0.56	1.00
Low,Bimodal Warm - Medium,Bimodal Warm	t(166) = 0.42	1.00
Low,Bimodal Warm - Urgent,Bimodal Warm	t(166) = 3.74	0.02*
Low, Bimodal Warm - Low, Thermal Cold	t(179) = 1.49	0.98
Low,Bimodal Warm - Medium,Thermal Cold	t(189) = 2.52	0.43
Low,Bimodal Warm - Urgent,Thermal Cold	t(189) = 3.78	0.02*
Low,Bimodal Warm - Low,Thermal Warm	t(179) = 3.78	0.11
Low,Bimodal Warm - Medium,Thermal Warm	t(189) = 3.24	0.09
Low,Bimodal Warm - Urgent,Thermal Warm	t(189) = 3.60	0.03*
Low, Bimodal Warm - Low, Vibration	t(179) = 0.00	1.00
Low, Bimodal Warm - Medium, Vibration	t(189) = 0.36	1.00
Low, Bimodal Warm - Urgent, Vibration	t(189) = 1.98	0.81
Medium,Bimodal Warm - Urgent,Bimodal Warm	t(166) = 3.33	0.07
Medium, Bimodal Warm - Low, Thermal Cold	t(189) = 1.80	0.90
Medium,Bimodal Warm - Medium,Thermal Cold	t(179) = 2.98	0.17
Medium, Bimodal Warm - Urgent, Thermal Cold	t(189) = 4.14	0.005*
Medium,Bimodal Warm - Low,Thermal Warm	t(189) = 3.42	0.05
Medium,Bimodal Warm - Medium,Thermal Warm	t(179) = 3.72	0.02*
Medium,Bimodal Warm - Urgent,Thermal Warm	t(189) = 3.96	0.009*
Medium, Bimodal Warm - Low, Vibration	t(189) = 0.36	1.00
Medium, Bimodal Warm - Medium, Vibration	t(179) = 0.00	1.00
Medium, Bimodal Warm - Urgent, Vibration	t(189) = 1.62	0.95
Urgent, Bimodal Warm - Low, Thermal Cold	t(189) = 4.69	0.0005*
Urgent, Bimodal Warm - Medium, Thermal Cold	t(189) = 5.77	<0.0001*
Urgent, Bimodal Warm - Urgent, Thermal Cold	t(179) = 7.25	<0.0001*
Urgent,Bimodal Warm - Low,Thermal Warm	t(189) = 6.31	<0.0001*
Urgent,Bimodal Warm - Medium,Thermal Warm	t(189) = 6.49	<0.0001*
Urgent,Bimodal Warm - Urgent,Thermal Warm	t(179) = 7.07	<0.0001*
Urgent,Bimodal Warm - Low,Vibration	t(189) = 3.24	0.09
Urgent, Bimodal Warm - Medium, Vibration	t(189) = 2.88	0.21
Urgent,Bimodal Warm - Urgent,Vibration	t(179) = 1.30	0.99

Table A.14: Experiment 5: Part 2 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Rate; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t(238)	р
Low, Thermal Cold - Medium, Thermal Cold	t(166) = 1.25	1.00
Low, Thermal Cold - Urgent, Thermal Cold	t(166) = 2.70	0.31
Low, Thermal Cold - Low, Thermal Warm	t(179) = 1.67	0.94
Low, Thermal Cold - Medium, Thermal Warm	t(189) = 1.80	0.90
Low, Thermal Cold - Urgent, Thermal Warm	t(189) = 2.16	0.69
Low, Thermal Cold - Low, Vibration	t(179) = 1.49	0.98
Low, Thermal Cold - Medium, Vibration	t(189) = 1.80	0.90
Low, Thermal Cold - Urgent, Vibration	t(189) = 3.42	0.05
Medium, Thermal Cold - Urgent, Thermal Cold	t(166) = 1.46	0.98
Medium, Thermal Cold - Low, Thermal Warm	t(189) = 0.54	1.00
Medium, Thermal Cold - Medium, Thermal Warm	t(179) = 0.74	1.00
Medium, Thermal Cold - Urgent, Thermal Warm	t(189) = 1.08	1.00
Medium, Thermal Cold - Low, Vibration	t(189) = 2.52	0.43
Medium, Thermal Cold - Medium, Vibration	t(179) = 2.98	0.17
Medium, Thermal Cold - Urgent, Vibration	t(189) = 4.50	0.001*
Urgent, Thermal Cold - Low, Thermal Warm	t(189) = 0.72	1.00
Urgent, Thermal Cold - Medium, Thermal Warm	t(189) = 0.54	1.00
Urgent, Thermal Cold - Urgent, Thermal Warm	t(179) = 0.19	1.00
Urgent, Thermal Cold - Low, Vibration	t(189) = 3.78	0.02*
Urgent, Thermal Cold - Medium, Vibration	t(189) = 4.14	0.005*
Urgent, Thermal Cold - Urgent, Vibration	t(179) = 5.95	<0.0001*
Low, Thermal Warm - Medium, Thermal Warm	t(166) = 0.21	1.00
Low, Thermal Warm - Urgent, Thermal Warm	t(166) = 0.62	1.00
Low, Thermal Warm - Low, Vibration	t(179) = 3.16	0.11
Low, Thermal Warm - Medium, Vibration	t(189) = 3.42	0.05
Low, Thermal Warm - Urgent, Vibration	t(189) = 5.05	0.0001*
Medium, Thermal Warm - Urgent, Thermal Warm	t(166) = 0.42	1.00
Medium, Thermal Warm - Low, Vibration	t(189) = 3.24	0.09
Medium, Thermal Warm - Medium, Vibration	t(179) = 3.72	0.02*
Medium, Thermal Warm - Urgent, Vibration	t(189) = 5.23	<0.0001*
Urgent, Thermal Warm - Low, Vibration	t(189) = 3.70	0.03*
Urgent, Thermal Warm - Medium, Vibration	t(189) = 3.96	0.009*
Urgent, Thermal Warm - Urgent, Vibration	t(179) = 5.76	<0.0001*
Low, Vibration - Medium, Vibration	t(166) = 0.42	1.00
Low, Vibration - Urgent, Vibration	t(166) = 2.29	0.60
Medium, Vibration - Urgent, Vibration	t(166) = 1.87	0.86

Table A.15: Experiment 5: Part 3 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Rate; significant results are bold and marked with an asterisk.

A.6.8 Experiment 5: Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Time; significant results and discussion can be found in section 5.2.6

Urgency,Modality - Urgency,Modality	t	p
Low,Bimodal Cold - Medium,Bimodal Cold	t(152) = 0.44	1.00
Low,Bimodal Cold - Urgent,Bimodal Cold	t(153) = 2.89	0.21
Low,Bimodal Cold - Low,Bimodal Warm	t(182) = 0.69	1.00
Low,Bimodal Cold - Medium,Bimodal Warm	t(196) = 0.55	1.00
Low,Bimodal Cold - Urgent,Bimodal Warm	t(196) = 2.13	0.71
Low, Bimodal Cold - Low, Thermal Cold	t(183) = 1.85	0.87
Low,Bimodal Cold - Medium,Thermal Cold	t(191) = 1.43	0.98
Low, Bimodal Cold - Urgent, Thermal Cold	t(192) = 0.16	1.00
Low, Bimodal Cold - Low, Thermal Warm	t(182) = 2.56	0.40
Low,Bimodal Cold - Medium,Thermal Warm	t(191) = 2.33	0.57
Low,Bimodal Cold - Urgent,Thermal Warm	t(191) = 2.19	0.68
Low,Bimodal Cold - Low,Vibration	t(182) = 2.41	0.51
Low,Bimodal Cold - Medium,Vibration	t(196) = 2.49	0.46
Low, Bimodal Cold - Urgent, Vibration	t(196) = 4.35	0.002*
Medium, Bimodal Cold - Urgent, Bimodal Cold	t(153) = 2.47	0.47
Medium,Bimodal Cold - Low,Bimodal Warm	t(196) = 0.30	1.00
Medium,Bimodal Cold - Medium,Bimodal Warm	t(182) = 0.15	1.00
Medium, Bimodal Cold - Urgent, Bimodal Warm	t(195) = 1.74	0.92
Medium, Bimodal Cold - Low, Thermal Cold	t(194) = 2.31	0.59
Medium, Bimodal Cold - Medium, Thermal Cold	t(181) = 1.77	0.91
Medium, Bimodal Cold - Urgent, Thermal Cold	t(193) = 0.56	1.00
Medium, Bimodal Cold - Low, Thermal Warm	t(191) = 3.04	0.15
Medium, Bimodal Cold - Medium, Thermal Warm	t(182) = 2.63	0.36
Medium, Bimodal Cold - Urgent, Thermal Warm	t(190) = 2.58	0.39
Medium, Bimodal Cold - Low, Vibration	t(196) = 2.03	0.78
Medium, Bimodal Cold - Medium, Vibration	t(182) = 2.06	0.76
Medium, Bimodal Cold - Urgent, Vibration	t(195) = 3.96	0.009*

Table A.16: Experiment 5: Part 1 Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Time; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t	р
Urgent,Bimodal Cold - Low,Bimodal Warm	t(195) = 1.99	0.80
Urgent,Bimodal Cold - Medium,Bimodal Warm	t(195) = 2.15	0.70
Urgent,Bimodal Cold - Urgent,Bimodal Warm	t(182) = 0.56	1.00
Urgent, Bimodal Cold - Low, Thermal Cold	t(194) = 4.51	0.001*
Urgent, Bimodal Cold - Medium, Thermal Cold	t(193) = 4.03	0.007*
Urgent, Bimodal Cold - Urgent, Thermal Cold	t(181) = 2.72	0.30
Urgent, Bimodal Cold - Low, Thermal Warm	t(192) = 5.21	<0.0001*
Urgent,Bimodal Cold - Medium,Thermal Warm	t(191) = 4.90	0.0002*
Urgent, Bimodal Cold - Urgent, Thermal Warm	t(181) = 4.66	0.0006*
Urgent, Bimodal Cold - Low, Vibration	t(196) = 0.26	1.00
Urgent, Bimodal Cold - Medium, Vibration	t(195) = 0.21	1.00
Urgent, Bimodal Cold - Urgent, Vibration	t(182) = 1.64	0.95
Low,Bimodal Warm - Medium,Bimodal Warm	t(152) = 0.16	1.00
Low,Bimodal Warm - Urgent,Bimodal Warm	t(153) = 1.54	0.97
Low,Bimodal Warm - Low,Thermal Cold	t(183) = 2.50	0.44
Low,Bimodal Warm - Medium,Thermal Cold	t(191) = 2.11	0.73
Low,Bimodal Warm - Urgent,Thermal Cold	t(192) = 0.85	1.00
Low,Bimodal Warm - Low,Thermal Warm	t(182) = 3.21	0.10
Low,Bimodal Warm - Medium,Thermal Warm	t(191) = 3.00	0.16
Low,Bimodal Warm - Urgent,Thermal Warm	t(191) = 2.86	0.23
Low,Bimodal Warm - Low,Vibration	t(182) = 1.72	0.93
Low,Bimodal Warm - Medium,Vibration	t(196) = 1.79	0.90
Low,Bimodal Warm - Urgent,Vibration	t(196) = 3.65	0.03*
Medium,Bimodal Warm - Urgent,Bimodal Warm	t(153) = 1.71	0.93
Medium, Bimodal Warm - Low, Thermal Cold	t(194) = 2.45	0.48
Medium, Bimodal Warm - Medium, Thermal Cold	t(181) = 1.91	0.84
Medium, Bimodal Warm - Urgent, Thermal Cold	t(193) = 0.70	1.00
Medium,Bimodal Warm - Low,Thermal Warm	t(191) = 3.18	0.10
Medium,Bimodal Warm - Medium,Thermal Warm	t(182) = 2.77	0.27
Medium,Bimodal Warm - Urgent,Thermal Warm	t(190) = 2.72	0.30
Medium, Bimodal Warm - Low, Vibration	t(196) = 1.88	0.86
Medium, Bimodal Warm - Medium, Vibration	t(182) = 1.92	0.84
Medium, Bimodal Warm - Urgent, Vibration	t(195) = 3.81	0.02*
Urgent,Bimodal Warm - Low,Thermal Cold	t(194) = 3.97	0.009*
Urgent, Bimodal Warm - Medium, Thermal Cold	t(193) = 3.49	0.04*
Urgent, Bimodal Warm - Urgent, Thermal Cold	t(181) = 2.19	0.67
Urgent,Bimodal Warm - Low,Thermal Warm	t(192) = 4.67	0.0005*
Urgent,Bimodal Warm - Medium,Thermal Warm	t(191) = 4.27	0.002*
Urgent,Bimodal Warm - Urgent,Thermal Warm	t(181) = 4.13	0.005*
Urgent,Bimodal Warm - Low,Vibration	t(196) = 0.30	1.00
Urgent,Bimodal Warm - Medium,Vibration	t(195) = 0.35	1.00
Urgent, Bimodal Warm - Urgent, Vibration	t(181) = 2.20	0.67

Table A.17: Experiment 5: Part 2 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Time; significant results are bold and marked with an asterisk.

Urgency,Modality - Urgency,Modality	t	р
Low, Thermal Cold - Medium, Thermal Cold	t(151) = 0.53	1.00
Low, Thermal Cold - Urgent, Thermal Cold	t(151) = 1.82	0.89
Low, Thermal Cold - Low, Thermal Warm	t(183) = 0.69	1.00
Low, Thermal Cold - Medium, Thermal Warm	t(191) = 0.41	1.00
Low, Thermal Cold - Urgent, Thermal Warm	t(190) = 0.27	1.00
Low, Thermal Cold - Low, Vibration	t(183) = 4.12	0.005*
Low, Thermal Cold - Medium, Vibration	t(194) = 4.32	0.002*
Low, Thermal Cold - Urgent, Vibration	t(194) = 6.11	<0.0001*
Medium, Thermal Cold - Urgent, Thermal Cold	t(152) = 1.35	0.99
Medium, Thermal Cold - Low, Thermal Warm	t(188) = 1.22	1.00
Medium, Thermal Cold - Medium, Thermal Warm	t(181) = 0.87	1.00
Medium, Thermal Cold - Urgent, Thermal Warm	t(188) = 0.77	1.00
Medium, Thermal Cold - Low, Vibration	t(191) = 3.80	0.02*
Medium, Thermal Cold - Medium, Vibration	t(181) = 3.75	0.02*
Medium, Thermal Cold - Urgent, Vibration	t(193) = 5.64	<0.0001*
Urgent, Thermal Cold - Low, Thermal Warm	t(189) = 2.42	0.50
Urgent, Thermal Cold - Medium, Thermal Warm	t(190) = 2.11	0.73
Urgent, Thermal Cold - Urgent, Thermal Warm	t(181) = 1.90	0.85
Urgent, Thermal Cold - Low, Vibration	t(192) = 2.53	0.42
Urgent, Thermal Cold - Medium, Vibration	t(193) = 2.57	0.40
Urgent, Thermal Cold - Urgent, Vibration	t(181) = 4.29	0.003*
Low, Thermal Warm - Medium, Thermal Warm	t(151) = 0.33	1.00
Low, Thermal Warm - Urgent, Thermal Warm	t(151) = 0.49	1.00
Low, Thermal Warm - Low, Vibration	t(182) = 4.81	0.0003*
Low, Thermal Warm - Medium, Vibration	t(191) = 5.03	0.0001*
Low, Thermal Warm - Urgent, Vibration	t(192) = 6.78	<0.0001*
Medium, Thermal Warm - Urgent, Thermal Warm	t(151) = 0.16	1.00
Medium, Thermal Warm - Low, Vibration	t(191) = 4.66	0.0006*
Medium, Thermal Warm - Medium, Vibration	t(182) = 4.56	0.0009*
Medium, Thermal Warm - Urgent, Vibration	t(191) = 6.49	<0.0001*
Urgent, Thermal Warm - Low, Vibration	t(191) = 4.52	0.001*
Urgent, Thermal Warm - Medium, Vibration	t(190) = 4.58	0.0008*
Urgent, Thermal Warm - Urgent, Vibration	t(181) = 6.20	<0.0001*
Low, Vibration - Medium, Vibration	t(152) = 0.06	1.00
Low, Vibration - Urgent, Vibration	t(153) = 2.06	0.76
Medium, Vibration - Urgent, Vibration	t(153) = 2.01	0.79

Table A.18: Experiment 5: Part 3 of Statistics of *post hoc* tests for the Interaction between Urgency and Modality for Recognition Time; significant results are bold and marked with an asterisk.

A.7 Experiment 6

A.7.1 Experiment 6: Information Sheet

BiModalInfo – Investigating Thermal and Vibrotactile Feedback on the Steering Wheel



BACKGROUND

The goal of this user study is to improve safety in driving situations by providing notifications with different levels of information through thermal feedback with and without vibration on the steering wheel.

TASK

The study will be conducted in a driving simulator and you will be asked to identify notifications consisting of thermal (unimodal) or thermal and vibrotactile (bimodal) feedback. You will perform the tasks sitting in a gaming racing chair. There will be a steering wheel securely attached in front of you and you will be holding the wheel throughout the experiment. Please position your hands on the indicated location of the steering wheel and keep them on the devices. You will be wearing headphones throughout the experiment, playing car sounds.

In the simulated driving you will perform a lane change task while the stimuli are being presented. The simulator requires you to keep a virtual vehicle within the limits of the road. The car will be placed on a five-lane motorway. You will start in the middle lane and should swiftly and precisely change to another lane, when it is indicated by an arrow on one of the bridges and reach the goal lane before reaching that bridge. Please keep then within that lane, until another arrow appears. In between these lane changes you might be presented with notifications, which you should identify as swiftly as possible by reporting the importance (high/low) and the nature (work/private) of the received message to the experimenter.

There will be a training phase before the main driving task begins. The driving task itself will be separated into five blocks of approximately 5 min each. You can break between these blocks, but please let the experimenter know, if you need to rest within a block.

INSTRUCTIONS

- Please place both hands at the indicated spaces on the steering wheel while driving.
- In the driving task, please try to follow the instructions given by the arrows on the bridges to the best of your ability and try to keep inside the lane.

MISCELLANEOUS

- You will be wearing headphones throughout the experiment.
- The experiment is divided into blocks. You can rest between the blocks for as long as you need. If additional breaks are required, please let the experimenter know.
- You can withdraw from the study at any time without penalty. The data recorded until then will be destroyed.

IMPORTANT

The systems you interact with will not ask or store personal information and all data recorded are anonymised. The study will take approximately 60 minutes to complete and a 10£ Amazon voucher will be handed out at the end. If you are a student, no course credits will be awarded for completing this study. To take part in this study, the following criteria must be met:

- I am at least 18 years old
- I hold a full driving licence
- I have no visual impairments (not including wearing glasses or contact lenses)
- I have no neurological disorders (e.g. epilepsy)
- I do not suffer from motion sickness
- I do not have any sensory impairments

You will be given resting sessions between experimental conditions but please do not be afraid to ask for a break if required at any time during the study. Furthermore, if you feel unwell at any point, please let the experimenter know immediately so that medical attention can be given as soon as possible. Should you have any questions about the study or like more information about our research in general, contact details are provided below.

CONTACT DETAILS

Name:	Patrizia Di Campli San Vito (PhD student/Experimenter)
Email:	p.di-campli-san-vito.1@research.gla.ac.uk
Office:	Room F141, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Supervisor:	Professor Stephen Brewster
Email:	stephen.brewster@glasgow.ac.uk
Office:	Room S131, Sir Alwyn Williams Building (Computing Science), Glasgow, G12 8RZ, Scotland
Tel:	+44 (0)141-330-4966

This study has been approved by the Ethics Committee (Reference Number - 300160073).

This research is part funded by the EPSRC and Jaguar Land Rover.

Version No. 1 (June 2019)

A.7.2 Experiment 6: Consent Form

BiModalInfo – Investigating Thermal and Vibrotactile Feedback on the Steering Wheel

CONSENT FORM

I confirm that I have read and unde above study and have had the oppo	rstood the Study Information Sheet provided to me for the ortunity to ask questions.
The study has been explained to me participation will involve.	e, and I understand the explanation given and what my
I understand that during the course instructions concerning the experim only be explained at the end of the	of the study the experimenter will give additional nent and that the purpose of some of the instructions will study.
I understand that my participation i without giving a reason.	s voluntary and that I am free to withdraw at any time,
I understand that the research data that at all times my personal data w guidelines.	may be accessed by researchers working at the GIST, but ill be kept confidential in accordance with data protection
I have initialled the above bo	xes myself and I agree to take part in the study.
FULL NAME:	
I want to be added to the participant's po	
EMAIL:	

Experimenter:	Ms Patrizia Di Campli San Vito
	(p.di-campli-san-vito.1@research.gla.ac.uk)
DATE, SIGNATURE:	

Supervisor details:	Professor Stephen Brewster
	(stephen.brewster.glasgow.gla.ac.uk, 0141-330-4966)

This study has been approved by the Ethics Committee (Reference Number - 300160073).

This research is part funded by the EPSRC and Jaguar Land Rover.



Version No. 1 (March 2019)

A.7.3 Experiment 6: Questionnaires

Microsoft Forms

https://forms.office.com/Pages/DesignPage.aspx?origin=shell#Analysi...

BiModalInfo



"_ "

Responses



7. Please report on how much experience with the technologies you had prior to the experiment:

■ 1 (none) ■ 2 ■ 3 ■ 4 ■ 5 (much)		
Driving Simulators		
Vibrotactile Feedback		
Thermal Feedback		
	100%	0% 100%

- 8. How much did you like the unimodal (only thermal) notifications?
 - 1 Responses



- 9. How much did you like the bimodal (thermal and vibrotactile) notifications?
 - **1** Responses



1.00 Average Rating

10. Please rank the different stimuli on how well you could identify them:



11. Please name reasons for your rating:

1	Latest responses
Responses	"_"

- 12. How well could you differentiate the direction of temperature change (private/work)?
 - 1 Responses

 $\bigstar \And \And \And \And$ 1.00 Average Rating

13. How well could you differentiate the vibration patterns (importance in bimodal stimuli)?

1 Responses

* ☆ ☆ ☆ ☆ 1.00 Average Rating

Microsoft Forms

14. How well could you differentiate the lengths of temperature presentation (importance in thermal stimuli)?

1 Responses

* ☆ ☆ ☆ ☆

1.00 Average Rating

15. Comments:

0 Responses

Latest responses

A.8 Experiment 7

A.8.1 Experiment 7: Information Sheet

Investigating Feedback for Handover Scenarios



INFORMATION SHEET

This study investigates different feedback types for handover scenarios in semiautonomous cars. In a handover scenario the driver transfers the control to the car, i.e. switches from manual to autonomous driving. In this study the driver should activate the traffic jam pilot when it is available, allowing the car to manoeuvre through a traffic jam by itself. This study is conducted in a driving simulator.

Overall you will be asked to activate the pilot four different times, in four different driving blocks. You should activate the traffic jam pilot as soon as it is available. This can be achieved by pressing two buttons on the steering wheel or the paddle shifters for three seconds. Different feedback will support the handover process and may include audio, visual and haptic cues. The specific type of handover and feedback will be explained to you before each block.

The simulator requires you to keep a virtual vehicle within the limits of the road, just like in real life. You will interact with the haptic steering wheel to control the virtual driving task. You will have time to get used to the simulator. You will be asked to fill in questionnaires between the blocks. There will be short breaks between the blocks, but if you need to have a break at any other time, please notify the experimenter.

You should be aware that there is some risk of induced motion sickness when using a driving simulator - if you are particularly susceptible to motion sickness, migraines, epilepsy, dizziness or blurred vision then you should not take part. We advise that you wait for 30 minutes before driving your own car after the end of the study.

INSTRUCTIONS

- Please drive as you would drive in a real car.
- Please try to follow the instructions given by the system to the best of your ability and try to keep inside the lane.

MISCELLANEOUS

- There will be video recordings of your face, your hands and your feet. These recordings will be used for analysis purposes only and will not be shared. You can request at any point for this data to be deleted.
- You can end the experiment at any time without any judgement. The data recorded until then will be destroyed.

IMPORTANT

The systems you interact with will not ask or store personal information and all data recorded is anonymised. You will be filmed during the experiment, but the footage is for internal use only and will not be published directly at any point, however anonymized summaries of driving behaviour might be used for academic publications. Images will not be used without your permission. The study will take approximately 60 minutes to complete. You can withdraw from the study at any time without penalty.

Version No. 1 (November 2018)

Investigating Feedback for Handover Scenarios

To take part in this study, the following criteria must be met:

- I am at least 18 years old
- I hold a full driving licence
- I have no visual impairments (not including wearing glasses or contact lenses)
- I have no neurological disorders (e.g. epilepsy)
- I do not suffer from motion sickness
- I do not have any sensory impairments

You will be given resting sessions between experimental conditions but please do not be afraid to ask for a break, if required, at any time during the study. Furthermore, if you feel unwell at any point, please let the experimenter know immediately so that medical attention can be given as soon as possible. You can withdraw from the study at any time without prejudice and all data recorded to that point will be erased. If you have any questions about the study or would like more information about our research in general, contact details are provided below.

CONTACT DETAILS

Name:	Patrizia Di Campli San Vito (PhD student/Experimenter)						
Email:	p.di-campli-san-vito.1@research.gla.ac.uk, pdicampl@partner.jaguarlandrover.com						
Supervisor JLR:	Eddie Brown						
	ebrown6@jaguarlandrover.com						
Supervisor Uof	Professor Stephen Brewster						
	stephen.brewster@glasgow.ac.uk						

This research is funded by Jaguar Land Rover and EPSRC and has been approved by the Ethics Committee of JLR (number:) and the University of Glasgow (number:).

Version No. 1 (November 2018)

A.8.2 Experiment 7: Consent Form

Investigating Feedback for Handover Scenarios

CONSENT FORM

	I confirm that I have read and understood the Study Information Sheet provided to me for the above study and have had the opportunity to ask questions.								
	The study has been explained to me, and I understand the explanation given and what my participation will involve.								
	I understand that during the study the experimenter will give additional instructions concerning the experiment and that the purpose of some of the instructions will only be explained at the end of the study.								
	I understand that my participation is voluntary and that I am free to withdraw at any time, without giving a reason.								
	I understand that the research data may be accessed by researchers working at the Jaguar Land Rover and the Glasgow Interaction Section, but that at all times my personal data will be kept confidential in accordance with data protection guidelines.								
	I understand ar purposes.	nd agree that during the study video footage will be recorded for analysis							
	l understand ar demonstration	nd consent that parts of the video data may be presented to JLR staff for purposes.							
	I understand th	nat I can ask for my data to be deleted at any time.							
	I have ini	tialled the above boxes myself and I agree to take part in the study.							
FULI	_ NAME:								
SIGN	ATURE:								
DAT	E:								
EMA	NL:								
Exp	erimenter:	Patrizia Di Campli San Vito (p.di-campli-san-vito.1@research.gla.ac.uk, pdicampl@partner.jaguarlandrover.com)							
D 4 T	E, SIGNATURE:								
DAT	,								
Supe	ervisor details:	Eddie Brown (ebrown6@jaguarlandrover.com)							

This research is funded by Jaguar Land Rover and EPSRC and has been approved by the Ethics Committees of JLR and the University of Glasgow.

JAGUAR





Version No. 1 (November 2018)

A.8.3 Experiment 7: Questionnaires

Condition Rating

12/20/2018

Condition Rating

Please rate your experience with the Transfer Control Condition you just experienced

- * Required
 - 1. Participant *

2. Con	dition *	oval									
	Button	ovai.									
		S e with A	udio an	d Vieual	(NoHan	Vie)					
						(Hon)/i	c)				
		s with A	udio on	d Hantio		oVie)	5)				
\square		S WILLI A			s (napri	10 1 15)					
3. Men Mark	tal Dema	nd * oval.									
	1	2	3	4	5	6	7	8	9	10	
Low	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	High
4. Phys Mark	sical Dem only one	and * oval.			_		_				
	1	2	3	4	5	6	7	8	9	10	
Low	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	High
5. Time Mark	e Pressur	e * oval.									
	1	2	3	4	5	6	7	8	9	10	
Low	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	High
6. Effo i Mark	rt * conly one	oval.									
	1	2	3	4	5	6	7	8	9	10	
Low	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	High

https://docs.google.com/forms/d/1YH2oFPq32i0nwdYcd5SC5Tv7knadfNNZ-qZf-lfHT3I/editional field for the formula of the formula
018							Conditio	n Rating				
7	. Perfori Mark o	mance * nly one d	oval.									
		1	2	3	4	5	6	7	8	9	10	
	Poor	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc		\bigcirc (\bigcirc (Good
8	. Frustra Mark o	ation * nly one o	oval.									
		1	2	3	4	5	6	7	8	9	10	
	Low	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc (\bigcirc	\bigcirc	\supset	\supset		ligh
9	. Annoy Mark o	ance * nly one d	oval.									
		1	2	3	4	5	6	7	8	9	10	
	Low	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc (\bigcirc	\bigcirc	\supset	\supset) F	ligh
10	. Compl Mark o	icated * nly one o	oval.									
		1	2	3	4	5	6	7	8	9	10	
	Not At Al	t C				\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Very Much
11	. Pleasa Mark o	nt * nly one d	oval.									
		1	2	3	4	5	6	7	8	9	10	
	Not At	t C				\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Very Much
12	. Disrup Mark o	tive * nly one d	oval.									
		1	2	3	4	5	6	7	8	9	10	
	Not Al	t C				\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Very Much
13	. Comfo Mark o	rtable * nly one o	oval.									
		1	2	3	4	5	6	7	8	9	10	
	Not At Al	t C				\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	Very Much

https://docs.google.com/forms/d/1YH2oFPq32i0nwdYcd5SC5Tv7knadfNNZ-qZf-lfHT3I/edit

12/20/2018

14. When you transferred from manual driving to autonomous driving, when did the car take over control? *

Condition Rating

15. When you transferred from autonomous driving to manual driving, when did you take over control? *

16. Comments



https://docs.google.com/forms/d/1YH2oFPq32i0nwdYcd5SC5Tv7knadfNNZ-qZf-lfHT3I/edit

Questionnaire End

12/20/2018

Questionnaire End

* Required

1. Participant *

2. Please order the conditions from most favourite (1) to least favourite (4) * Mark only one oval per row.

	1 (favourite)	2 3	4 (least favourite)
Buttons	\bigcirc	$\bigcirc\bigcirc$	\bigcirc
Paddles with Audio and Visual (NoHapVis)	\bigcirc	$\bigcirc\bigcirc$	\bigcirc
Paddles with Audio, Visual and Haptics (HapVis)	\bigcirc	$\bigcirc\bigcirc$	\bigcirc
Paddles with Audio and Haptics (HapNoVis)	\bigcirc	$\bigcirc\bigcirc$	\bigcirc

3. How well could you feel the thermal feedback during the transfer of control? Mark only one oval per row.

	1 (not at all)	2	3	4	5 (very much)
With visual progress bar	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Without visual progress bar	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc

4. Did the type of handover influence your trust in the system? *

Mark only one oval.
Yes
No
Maybe
Other:

Please rate your trust in the system during the different transfer conditions:

5.	Buttons * Mark only one	e oval.										
		1	2	3	4	5	6	7	8	9	10	
	Very Trustworthy	\bigcirc	Not Trustworthy									

Mark only one	Audio a	and Vis	ual (Noł	lapVis)	*		ind			
wark only on	1	2	3	4	5	6	7	8	9	10
Very Trustworthy	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
7. Paddles with Mark only one	Audio, e oval.	Visual a	and Hap	otics (Ha	apVis) *					
	1	2	3	4	5	6	7	8	9	10
Very Trustworthy	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
8. Paddles with Mark only one	Audio a e oval.	and Hap	otics (Ha	apNoVis	5) *					
	1	2	3	4	5	6	7	8	9	10
Very	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc	\bigcirc
Demograph 9. Age *	nic Da	ıta								
Pemograph 9. Age * 10. Gender * Mark only one Eremal Male Non-b	nic Da e oval. e inary not to s	ita								
Pemograph 9. Age * 10. Gender * Mark only one Femal Nale Non-b Prefer 11. Which hand Mark only one	e oval. e inary not to sa is your o e oval.	ay domina	nt hand	? *						
Trustworthy Demograph 9. Age * 10. Gender * Mark only one Femal Male Non-b Prefer 11. Which hand Mark only one Left Right Both	iic Da e oval. e inary not to sa is your o e oval.	ay domina	nt hand	?*						
Trustworthy Demograph 9. Age * 10. Gender * Mark only one Femal Male Non-b Prefer 11. Which hand Mark only one Left Right Both 12. What is your	iic Da e oval. e inary not to sa is your o e oval.	ay dominat	nt hand	? *						

https://docs.google.com/forms/d/1PYUAxGnveAWefMXgVRUkUtpj0B0W8iJlkojFGPjPIEI/edit

12/20/2018

Questionnaire End

14. Please rate how familiar you where with the following prior to the experiment: * Mark only one oval per row.

Semi-autonomous Features Driving Simulator		\square	\Box	
Driving Simulator			\leq	
)()) ()
Thermal Feedback	\square		$ \rightarrow $	
Vibrotactile Feedback	\bigcirc		$\overline{)}$	
Comments				
	Vibrotactile Feedback	Vibrotactile Feedback	Vibrotactile Feedback	Vibrotactile Feedback



https://docs.google.com/forms/d/1PYUAxGnveAWefMXgVRUkUtpj0B0W8iJIkojFGPjPIEI/edit

Bibliography

- J. C. Stevens and K. K. Choo, "Temperature sensitivity of the body surface over the life span," *Somatosensory and Motor Research*, vol. 15, no. 1, pp. 13–28, 1998.
 [Online]. Available: https://doi.org/10.1080/08990229870925
- [2] S. G. Klauer, T. a. Dingus, V. L. Neale, J. D. Sudweeks, and D. J. Ramsey, "The Impact of Driver Inattention On Near Crash/Crash Risk: An Analysis Using the 100-Car Naturalistic Driving Study Data," *Analysis*, no. April, p. 226, 2006.
- [3] C. D. Wickens, "Multiple Resources and Mental Workload," *Human Factors*, vol. 50, no. 3, pp. 449–455, 2008.
- [4] K. Suzuki and H. Jansson, "An analysis of driver's steering behaviour during auditory or haptic warnings for the designing of lane departure warning system," *JSAE Review*, vol. 24, no. 1, pp. 65–70, 2003.
- [5] J. B. Cicchino, "Effects of lane departure warning on police-reported crash rates," *Journal of Safety Research*, vol. 66, no. August, pp. 61–70, 2017. [Online]. Available: https://doi.org/10.1016/j.jsr.2018.05.006
- [6] J. B. van Erp and H. A. van Veen, "Vibrotactile in-vehicle navigation system," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 7, no. 4-5, pp. 247–256, 2004.
- [7] T. Nukarinen, R. Raisamo, A. Farooq, G. Evreinov, and V. Surakka, "Effects of directional haptic and non-speech audio cues in a cognitively demanding navigation task," *Proceedings of the NordiCHI 2014: The 8th Nordic Conference on Human-Computer Interaction: Fun, Fast, Foundational*, pp. 61–64, 2014.
- [8] S. Boll, A. Asif, and W. Heuten, "Feel your route: A tactile display for car navigation," *IEEE Pervasive Computing*, vol. 10, no. 3, pp. 35–42, 2011.
- [9] A. Löcken, H. Buhl, W. Heuten, and S. Boll, "TactiCar: Towards Supporting Drivers During Lane Change Using Vibro-Tactile Patterns," *Adjunct Proceedings*

of the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - AutomotiveUI '15, pp. 32–37, 2015. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2809730.2809758

- [10] D. Kern, P. Marshall, E. Hornecker, Y. Rogers, and A. Schmidt, "Enhancing navigation information with tactile output embedded into the steering wheel," *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, vol. 5538 LNCS, pp. 42–58, 2009.
- [11] S. Hwang and J. H. Ryu, "The haptic steering wheel: Vibro-tactile based navigation for the driving environment," 2010 8th IEEE International Conference on Pervasive Computing and Communications Workshops, PERCOM Workshops 2010, pp. 660– 665, 2010.
- [12] E. S. Ege, F. Cetin, and C. Basdogan, "Vibrotactile feedback in steering wheel reduces navigation errors during GPS-guided car driving," 2011 IEEE World Haptics Conference, WHC 2011, pp. 345–348, 2011.
- [13] N. Medeiros-Ward, J. M. Cooper, A. J. Doxon, D. L. Strayer, and W. R. Provancher, "Bypassing the Bottleneck: The Advantage of Fingertip Shear Feedback for Navigational Cues," in *Proceedings of the Human Factors and Ergonomics Society*, 2010, pp. 2043–2047.
- [14] C. J. Ploch, J. H. Bae, C. C. Ploch, W. Ju, and M. R. Cutkosky, "Comparing haptic and audio navigation cues on the road for distracted drivers with a skin stretch steering wheel," 2017 IEEE World Haptics Conference, WHC 2017, no. June, pp. 448–453, 2017.
- [15] D. E. Dass, A. Uyttendaele, and J. Terken, "Haptic in-seat feedback for lane departure warning," *Proceedings of the 5th International Conference on Automotive User Interfaces and Interactive Vehicular Applications AutomotiveUI '13*, pp. 258–261, 2013. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2516540.2516574
- [16] T. L. Brown, J. Gaspar, D. Marshall, and J. D. Lee, "How does lane departure warning effectiveness vary by severity of departure," *Proceedings of the Human Factors and Ergonomics Society*, vol. 2017-Octob, no. 2008, pp. 1929–1933, 2017.
- [17] C. Ho, H. Z. Tan, and C. Spence, "Using spatial vibrotactile cues to direct visual attention in driving scenes," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 8, no. 6, pp. 397–412, 2005.
- [18] I. Politis, S. Brewster, and F. Pollick, "Evaluating Multimodal Driver Displays under Varying Situational Urgency," *Proceedings of the 32nd annual ACM conference on*

Human factors in computing systems - CHI '14, pp. 4067–4076, 2014. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2556288.2556988

- [19] A. Murata, T. Kuroda, and W. Karwowski, "Effects of auditory and tactile warning on response to visual hazards under a noisy environment," *Applied Ergonomics*, vol. 60, pp. 58–67, 2016. [Online]. Available: http://dx.doi.org/10.1016/j.apergo.2016.11.002
- [20] I. Politis, S. Brewster, and F. Pollick, "Language-based multimodal displays for the handover of control in autonomous cars," *Proceedings of the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - AutomotiveUI '15*, no. SEPTEMBER 2015, pp. 3–10, 2015. [Online]. Available: http://eprints.gla.ac.uk/111949/%5Cnhttp://dl.acm.org/citation.cfm?doid= 2799250.2799262
- [21] S. M. Petermeijer, P. Bazilinskyy, K. Bengler, and J. de Winter, "Take-over again: Investigating multimodal and directional TORs to get the driver back into the loop," *Applied Ergonomics*, vol. 62, pp. 204–215, 2017. [Online]. Available: https://doi.org/10.1016/j.apergo.2017.02.023
- [22] K. Salminen and A. Farooq, "Unimodal and Multimodal Signals to Support Control Transitions in Semiautonomous Vehicles," in *Proceedings of the 11th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - Automotive UI'19*, 2019, pp. 308–318. [Online]. Available: https://doi.org/10.1145/3342197.3344522
- [23] C. Ho, N. Reed, and C. Spence, "Multisensory in-Car Warning Signais for Collision Avoidance," *Proceedings of the Human Factors and Ergonomics Society Annual Meeting*, vol. 56, no. 1, pp. 1107–1114, 2006. [Online]. Available: http://hfs.sagepub.com/content/34/5/513%5Cnhttp://hfs.sagepub.com/content/ 34/5/513.full.pdf%5Cnhttp://www.ncbi.nlm.nih.gov/pubmed/1459564%5Cnhttp: //www.scopus.com/inward/record.url?eid=2-s2.0-44349101960&partnerID=40& md5=6e8736d015ac91ab6fc72d6466936a4d%252
- [24] J. Scott and R. Gray, "A Comparison of Tactile, Visual, and Auditory Warnings for Rear-End Collision Prevention in Simulated Driving," *Human Factors: The Journal* of the Human Factors and Ergonomics Society, vol. 50, no. 2, pp. 264–275, 2008.
 [Online]. Available: http://journals.sagepub.com/doi/10.1518/001872008X250674
- [25] A. Murata, M. Kanbayashi, and T. Hayami, "Effectiveness of automotive warning system presented with multiple sensory modalities," *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture*

Notes in Bioinformatics), vol. 8025 LNCS, no. PART 1, pp. 88–97, 2012. [Online]. Available: https://doi.org/10.1007/978-3-642-39173-6_11

- [26] I. Politis, S. Brewster, and F. Pollick, "Evaluating Multimodal Driver Displays of Varying Urgency," *Proceedings of ACM AutoUI 2013*, vol. 13, pp. 92–99, 2013.
 [Online]. Available: https://doi.org/10.1145/2516540.2516543
- [27] M. Ketna and T. Leelanupab, "Evaluating a thermal icon for the enhancement of mobile feedback perception in noisy and bumpy environments," 2017 IEEE World Haptics Conference, WHC 2017, no. June, pp. 418–423, 2017.
- [28] L. A. Jones and M. Berris, "The psychophysics of temperature perception and thermal-interface design," *Proceedings - 10th Symposium on Haptic Interfaces for Virtual Environment and Teleoperator Systems, HAPTICS 2002*, pp. 137–142, 2002.
- [29] G. Wilson, M. Halvey, S. Brewster, and S. A. Hughes, "Some Like it Hot? Thermal Feedback for Mobile Devices," *Proceedings of the 2011 annual conference on Human factors in computing systems - CHI '11*, p. 2555, 2011. [Online]. Available: http://dl.acm.org/citation.cfm?doid=1978942.1979316
- [30] A. Singhal and L. A. Jones, "Perceptual interactions in thermo-tactile displays," 2017 IEEE World Haptics Conference, WHC 2017, no. June, pp. 90–95, 2017.
- [31] C. Ho, N. Reed, and C. Spence, "Assessing the effectiveness of "intuitive" vibrotactile warning signals in preventing front-to-rear-end collisions in a driving simulator," *Accident Analysis and Prevention*, vol. 38, no. 5, pp. 988–996, 2006. [Online]. Available: https://doi.org/10.1016/j.aap.2006.04.002
- [32] T. J. Prescott, E. Ahissar, and E. Izhikevich, *Scholarpedia of Touch*, 2016. [Online]. Available: https://doi.org/10.2991/978-94-6239-133-8_57
- [33] M. Benali-Khoudjal, M. Hafez, J.-M. Alexandre, J. Benachour, and A. Kheddar, "Thermal feedback model for virtual reality," *MHS2003. Proceedings of 2003 International Symposium on Micromechatronics and Human Science (IEEE Cat. No.03TH8717)*, no. November 2003, pp. 153–158, 2003. [Online]. Available: http://ieeexplore.ieee.org/document/1249925/
- [34] S. Cai, P. Ke, T. Narumi, and K. Zhu, "ThermAirGlove: A Pneumatic Glove for Thermal Perception and Material Identification in Virtual Reality," *IEEE Conference* on Virtual Reality and 3D User Interfaces (VR) ThermAirGlove:, pp. 248–257, 2020.
- [35] F. Hülsmann, J. Fröhlich, N. Mattar, and I. Wachsmuth, "Wind and warmth in virtual reality: Implementation and evaluation," ACM International Conference Proceeding Series, vol. 2014-April, 2014.

- [36] N. Ranasinghe, P. Jain, S. Karwita, D. Tolley, and E. Y. L. Do, "Ambiotherm: Enhancing sense of presence in virtual reality by simulating real-world environmental conditions," *Conference on Human Factors in Computing Systems Proceedings*, vol. 2017-May, pp. 1731–1742, 2017.
- [37] S. Günther, O. Elmoghazy, F. Müller, M. Mühlhäuser, D. Schön, and M. Schmitz, "Therminator : Understanding the Interdependency of Visual and On-Body Thermal Feedback in Virtual Reality," in *CHI2020*, 2020.
- [38] G. Gioioso, M. Pozzi, M. Aurilio, B. Peccerillo, G. Spagnoletti, and D. Prattichizzo, "Using Wearable Haptics for Thermal Discrimination in Virtual Reality Scenarios," in *AsiaHaptics*. Springer Singapore, 2018, vol. 1, pp. 144–148. [Online]. Available: http://dx.doi.org/10.1007/978-981-13-3194-7_32
- [39] H. Xu, W. Song, J. Sun, X. Wang, S. Li, X. Wu, W. Yang, and Z. Pan, "Thermal perception method of virtual chemistry experiments," *Virtual Reality* & *Intelligent Hardware*, vol. 2, no. 4, pp. 305–315, 2020. [Online]. Available: https://doi.org/10.1016/j.vrih.2020.07.003
- [40] R. Wettach, C. Behrens, A. Danielsson, and T. Ness, "A thermal information display for mobile applications," *Proceedings of the 9th international conference on Human computer interaction with mobile devices and services - MobileHCI '07*, pp. 182–185, 2007. [Online]. Available: http://dl.acm.org/citation.cfm?id=1377999.1378004
- [41] Z. Li, Y. Wang, W. Wang, W. Chen, T. Hoang, S. Greuter, and F. F. Mueller, "HeatCraft : Designing Playful Experiences with Ingestible Sensors via Localized Thermal Stimuli," *CHI19*, pp. 1–12, 2019.
- [42] M. Halvey, M. Henderson, S. Brewster, G. Wilson, and S. A. Hughes, "Augmenting Media with Thermal Stimulation," *HAID 2012*, 2012.
- [43] G. Wilson, D. Dobrev, and S. Brewster, "Hot Under the Collar: Mapping Thermal Feedback to Dimensional Models of Emotion," *CHI16*, pp. 31–36, 2016. [Online]. Available: http://dx.doi.org/10.1145/2858036.2858205
- [44] G. Wilson, S. Brewster, M. Halvey, and S. Hughes, "Thermal Icons: Evaluating Structured Thermal Feedback for Mobile Interaction," *Proceedings of the* 14th international conference on Human-computer interaction with mobile devices and services - MobileHCI '12, p. 309, 2012. [Online]. Available: http://dl.acm.org/citation.cfm?id=2371574.2371621
- [45] A. Singhal and L. A. Jones, "Creating Thermal Icons A Model-Based Approach," ACM Transactions on Applied Perception, vol. 15, no. 2, 2018.

- [46] M. Akazue, M. Halvey, L. Baillie, and S. Brewster, "The Effect of Thermal Stimuli on the Emotional Perception of Images," *Proceedings of the 2016 CHI Conference on Human Factors in Computing Systems - CHI '16*, pp. 4401–4412, 2016. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2858036.2858307
- [47] K. Suhonen, S. Müller, J. Rantala, K. Väänänen-Vainio-Mattila, R. Raisamo, and V. Lantz, "Haptically augmented remote speech communication," *Proceedings* of the 7th Nordic Conference on Human-Computer Interaction Making Sense Through Design - NordiCHI '12, p. 361, 2012. [Online]. Available: http: //dl.acm.org/citation.cfm?doid=2399016.2399073
- [48] W. Lee and Y.-k. Lim, "Thermo-Message: Exploring the Potential of Heat as a Modality of Peripheral Expression," *Proceedings of the 28th of the international conference extended abstracts on Human factors in computing systems - CHI EA '10*, p. 4231, 2010. [Online]. Available: http://portal.acm.org/citation.cfm?doid=1753846. 1754131
- [49] G. Wilson, H. Maxwell, and M. Just, "Everything's Cool: Extending Security Warnings with Thermal Feedback," in *Proceedings of the 2017 CHI Conference Extended Abstracts on Human Factors in Computing Systems - CHI EA '17*. Denver, Colorado, USA: ACM, 2017, pp. 2232–2239. [Online]. Available: http://dl.acm.org/citation.cfm?doid=3027063.3053127
- [50] I. A. Strigo, F. Carli, and M. C. Bushnell, "Effect of ambient temperature on human pain and temperature perception," *Anesthesiology*, vol. 92, no. 3, pp. 699–707, 2000.
- [51] M. Halvey, G. Wilson, Y. Vazquez-Alvarez, S. Brewster, and S. A. Hughes, "The effect of clothing on thermal feedback perception," *Proceedings of the 13th international conference on multimodal interfaces - ICMI '11*, no. October 2015, p. 217, 2011. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2070481. 2070519
- [52] S. Karjalainen, "Thermal comfort and gender: A literature review," pp. 96–109, 2012. [Online]. Available: https://doi.org/10.1111/j.1600-0668.2011.00747.x
- [53] H. Liu, Y. Wu, D. Lei, and B. Li, "Gender differences in physiological and psychological responses to the thermal environment with varying clothing ensembles," *Building and Environment*, vol. 141, no. May, pp. 45–54, 2018. [Online]. Available: https://doi.org/10.1016/j.buildenv.2018.05.040
- [54] S. Lautenbacher and F. Strian, "Sex differences in pain and thermal sensitivity: The role of body size," *Perception & Psychophysics*, vol. 50, no. 2, pp. 179–183, 1991.

- [55] M. J. Hilz, B. Stemper, F. B. Axelrod, E. H. Kolodny, and B. Neundörfer, "Quantitative thermal perception testing in adults," *Journal of Clinical Neurophysiology*, vol. 16, no. 5, pp. 462–471, 1999.
- [56] G. D. Mower, "Perceived intensity of peripheral thermal stimuli is independent of internal body temperature," *Journal of Comparative and Physiological Psychology*, vol. 90, no. 12, pp. 1152–1155, 1976. [Online]. Available: https://doi.org/10.1037/h0077284
- [57] M. Matsuda-Nakamura, S. Yasuhara, and K. Nagashima, "Effect of menstrual cycle on thermal perception and autonomic thermoregulatory responses during mild cold exposure," *Journal of Physiological Sciences*, vol. 65, no. 4, pp. 339–347, 2015. [Online]. Available: http://dx.doi.org/10.1007/s12576-015-0371-x
- [58] S. A. Prescott and S. Ratté, "Somatosensation and Pain," *Conn's Translational Neuroscience*, pp. 517–539, 2017. [Online]. Available: http://dx.doi.org/10.1016/B978-0-12-802381-5.00037-3
- [59] J. Feher, "Cutaneous Sensory Systems," *Quantitative Human Physiology*, pp. 389–399, 2012. [Online]. Available: https://doi.org/10.1016/B978-0-12-800883-6. 00035-5
- [60] B. G. Green and A. Cruz, "Warmth-insensitive fields': Evidence of sparse and irregular innervation of human skin by the warmth sense," *Somatosensory and Motor Research*, vol. 15, no. 4, pp. 269–275, 1998.
- [61] D. R. Kenshalo, "Somesthetic sensitivity in young and elderly humans," *Journals of Gerontology*, vol. 41, no. 6, pp. 732–742, 1986.
- [62] K. O. Johnson, I. Darian-Smith, and C. LaMotte, "Peripheral neural determinants of temperature discrimination in man: a correlative study of responses to cooling skin." *Journal of neurophysiology*, vol. 36, no. 2, pp. 347–370, 1973. [Online]. Available: https://doi.org/10.1152/jn.1973.36.2.347
- [63] R. S. Dahiya and M. Valle, "Human Tactile Sensing," in *Robotic Tactile Sensing*, 2013, no. 1, p. 32. [Online]. Available: http://link.springer.com/10.1007/978-94-007-0579-1
- [64] B. G. Green and C. Akirav, "Individual differences in temperature perception: Evidence of common processing of sensation intensity of warmth and cold," *Somatosensory and Motor Research*, vol. 24, no. 1-2, pp. 71–84, 2007. [Online]. Available: https://doi.org/10.1080/08990220701388117

- [65] W. M. B. Tiest and A. M. L. Kappers, "Tactile perception of thermal diffusivity." *Attention, perception & psychophysics*, vol. 71, no. 3, pp. 481–489, 2009.
- [66] J. C. Stevens, L. E. Marks, and D. C. Simonson, "Regional sensitivity and spatial summation in the warmth sense," *Physiology and Behavior*, vol. 13, no. 6, pp. 825– 836, 1974.
- [67] J. C. Stevens and L. E. Marks, "Spatial summation of cold," *Physiology and Behavior*, vol. 22, no. 3, pp. 541–547, 1979.
- [68] A. J. Rózsa and D. R. Kenshalo, "Bilateral spatial summation of cooling of symmetrical sites," *Perception & Psychophysics*, vol. 21, no. 5, pp. 455–462, 1977.
- [69] J. C. Stevens and L. E. Marks, "Spatial summation and the dynamics of warmth sensation," *Perception & Psychophysics*, vol. 9, no. 5, pp. 391–398, 1971.
- [70] G.-H. Yang, K. U. Kyung, M. A. Srinivasan, and D. S. Kwon, "Quantitative tactile display device with pin-array type tactile feedback and thermal feedback," *Proceedings -IEEE International Conference on Robotics and Automation*, vol. 2006, no. May, pp. 3917–3922, 2006.
- [71] G.-H. Yang, L. A. Jones, and D.-S. Kwon, "Use of Simulated Thermal Cues for Material Discrimination and Identification with a Multi-Fingered Display," *Presence: Teleoperators and Virtual Environments*, vol. 17, no. 1, pp. 29–42, 2008. [Online]. Available: https://doi.org/10.1162/pres.17.1.29
- [72] E. Eijkman, "Black Box Analysis of the Skin Senses as a Multiple Communication Channel," *Biological Cybernatics*, vol. 61, p. 163 170, 1989.
- [73] G.-H. Yang, D.-S. Kwon, and L. A. Jones, "Spatial acuity and summation on the hand: The role of thermal cues in material discrimination," *Attention, Perception, & Psychophysics*, vol. 71, no. 7, pp. 1439–1459, 2009. [Online]. Available: https://doi.org/10.3758/APP.71.1.156
- [74] D. R. Kenshalo and H. A. Scott, "Temporal course of thermal adaptation," *Science*, vol. 151, no. 3714, pp. 1095–1096, 1966. [Online]. Available: https://doi.org/10.1126/science.151.3714.1095
- [75] D. Yarnitsky and J. L. Ochoa, "Studies of heat pain sensation in man: perception thresholds, rate of stimulus rise and reaction time," *Pain*, vol. 40, no. 1, pp. 85–91, 1990.
- [76] J. L. Harrison and K. D. Davis, "Cold-evoked pain varies with skin type and cooling rate: A psychophysical study in humans," *Pain*, vol. 83, no. 2, pp. 123–135, 1999.

- [77] B. G. Green and C. Akirav, "Threshold and rate sensitivity of low-threshold thermal nociception," *European Journal of Neuroscience*, vol. 31, no. 9, pp. 1637–1645, 2010.
- [78] B. G. Green, "Localization of thermal sensation: An illusion and synthetic heat," *Perception & Psychophysics*, vol. 22, no. 4, pp. 331–337, 1977.
- [79] H.-N. Ho, H. M. Chow, S. Tsunokake, and W. Roseboom, "Thermal-tactile Integration in Object Temperature Perception," *IEEE Transactions on Haptics*, pp. 1–1, 2019.
- [80] A. Y. Leung, M. S. Wallace, G. Schulteis, and T. L. Yaksh, "Qualitative and quantitative characterization of the thermal grill," *Pain*, vol. 116, no. 1-2, pp. 26–32, 2005.
- [81] S. Patwardhan, A. Kawazoe, D. Kerr, M. Nakatani, and Y. Visell, "Dynamics and Perception in the Thermal Grill Illusion," *IEEE Transactions on Haptics*, vol. PP, no. c, pp. 1–1, 2019. [Online]. Available: https://ieeexplore.ieee.org/document/ 8664184/
- [82] A. Singhal and L. A. Jones, "Space-Time Dependencies and Thermal Perception," in *EuroHaptics 2016*, 2016, no. 1. [Online]. Available: https://doi.org/10.1007/ 978-3-319-42321-0_27
- [83] —, "Space-time interactions and the perceived location of cold stimuli," *IEEE Haptics Symposium*, *HAPTICS*, vol. 2016-April, pp. 92–97, 2016. [Online]. Available: https://doi.org/10.1109/HAPTICS.2016.7463161
- [84] F. A. Geldard and C. E. Sherrick, "The Cutaneous "Rabbit": A Perceptual Illusion," *Science*, vol. 178, no. 4057, pp. 178–179, 1972. [Online]. Available: https://science.sciencemag.org/content/178/4057/178
- [85] A. M. L. Kappers and M. A. Plaisier, "Thermal Perception and Thermal Devices Used on Body Parts Other Than Hand or Face," *IEEE Transactions on Haptics*, vol. 12, no. 4, pp. 386–399, 2019.
- [86] S. Akiyama, K. Sato, Y. Makino, and T. Maeno, "Presentation of thermal sensation through preliminary adjustment of adapting skin temperature," *Haptics Symposium* 2012, HAPTICS 2012 - Proceedings, pp. 355–358, 2012.
- [87] K. Sato and T. Maeno, "Presentation of sudden temperature change using spatially divided warm and cool stimuli," *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, vol. 7282 LNCS, no. PART 1, pp. 457–468, 2012.

- [88] M. Halvey, G. Wilson, S. Brewster, and S. Hughes, ""Baby It's Cold Outside": The Influence of Ambient Temperature and Humidity on Thermal Feedback," *CHI12*, vol. 119, no. 25, pp. 5943–5944, 2012. [Online]. Available: http://dl.acm.org/citation.cfm?id=2207779
- [89] G. Wilson, S. Brewster, M. Halvey, and S. Hughes, *Thermal Feedback Identification in a Mobile Environment*, 2013. [Online]. Available: https://www.springer.com/gp/book/9783642410673#
- [90] L. A. Jones and H.-n. Ho, "Warm or Cool, Large or Small? The Challenge of Thermal Displays," *TRANSACTIONS ON HAPTICS*, vol. 1, no. 1, pp. 53–70, 2008.
- [91] S. Kratz and T. Dunnigan, "ThermoTouch : Design of a High Dynamic Temperature Range Thermal Haptic Display," CHI Extended Abstracts on Human Factors in Computing Systems, pp. 1577–1582, 2016.
- [92] M. Nakajim, K. Hasegawa, Y. Makino, and H. Shinoda, "Remotely Displaying Cooling Sensation Using Ultrasound Mist Beam," *AsiaHaptics*, vol. 1, no. Table 1, pp. 196–197, 2019. [Online]. Available: http://dx.doi.org/10.1007/ 978-981-13-3194-7_43
- [93] S. Hirai and N. Miki, "A Thermal Tactile Sensation Display with Controllable Thermal Conductivity," *Micromachines 2019*, 2019.
- [94] J. Brooks, S. Nagels, and P. Lopes, "Trigeminal-based Temperature Illusions," *Chi*, pp. 1–12, 2020.
- [95] A. D. Cheok and K. Karunanayaka, "Thermal Taste Interface," in Virtual Taste and Smell Technologies for Multisensory Internet and Virtual Reality, 2018. [Online]. Available: http://link.springer.com/10.1007/978-3-319-73864-2
- [96] C. B. Zhong and G. J. Leonardelli, "Cold and lonely: Does social exclusion literally feel cold?" *Psychological Science*, vol. 19, no. 9, pp. 838–842, 2008.
- [97] L. E. Williams and J. A. Bargh, "Experiencing physical warmth promotes interpersonal warmth," *Science*, vol. 322, no. 5901, pp. 606–607, 2008.
- [98] H. IJzerman and G. R. Semin, "The Thermometer of Social Relations," *Psychological Science*, vol. 20, no. 10, pp. 1214–1220, 2009. [Online]. Available: http://www3.interscience.wiley.com/cgi-bin/fulltext/122587603/PDFSTART
- [99] K. Salminen, V. Surakka, J. Raisamo, J. Lylykangas, J. Pystynen, R. Raisamo, K. Mäkelä, and T. Ahmaniemi, "Emotional responses to thermal stimuli," *Proceedings*

of the 13th international conference on multimodal interfaces - ICMI '11, p. 193, 2011. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2070481.2070513

- [100] K. Salminen, V. Surakka, J. Raisamo, and J. Lylykangas, "Cold or Hot? How Thermal Stimuli Are Related to Human Emotional System?" *HAID 2013*, pp. 20–29, 2013.
- [101] P. Bergman, H.-n. Ho, A. Koizumi, A. Tajadura-jiménez, P. Bergman, H.-n. Ho, A. Koizumi, and A. Tajadura-jiménez, "The pleasant heat? Evidence for thermalemotional implicit associations occurring with semantic and physical thermal stimulation The pleasant heat? Evidence for thermal-emotional implicit associations occurring with semantic and," *Cognitive Neuroscience*, vol. 6, no. 1, pp. 24–30, 2015. [Online]. Available: http://dx.doi.org/10.1080/17588928.2014.988132
- [102] J. Tewell, J. Bird, and G. R. Buchanan, "The Heat is On: A Temperature Display for Conveying Affective Feedback," *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17*, pp. 1756–1767, 2017. [Online]. Available: http://dl.acm.org/citation.cfm?doid=3025453.3025844
- [103] G. Wilson and S. Brewster, "Multi-Moji: Combining Thermal, Vibrotactile & Visual Stimuli to Expand the Affective Range of Feedback," *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17*, pp. 1743–1755, 2017. [Online]. Available: http://dl.acm.org/citation.cfm?doid=3025453.3025614
- [104] K. Iwasaki, T. Miyaki, and J. Rekimoto, "AffectPhone : A Handset Device to Present User's Emotional State with Warmth / Coolness," *Proceedings* of the 1st InternationalWorkshop on Bio-inspired Human-Machine Interfaces and Healthcare Applications, vol. 1, no. 2010, pp. 83–88, 2010. [Online]. Available: https://www.scitepress.org/Papers/2010/28392/28392.pdf
- [105] D. Gooch and L. Watts, "Communicating Social Presence Through Thermal Hugs," Ubicomp 2010 SISSE Workshop, 2010.
- [106] A. E. Ali, X. Yang, S. Ananthanarayan, T. Röggla, and ..., "ThermalWear: Exploring Wearable On-chest Thermal Displays to Augment Voice Messages with Affect," *CHI2020*, pp. 1–14, 2020. [Online]. Available: http://makingthingsblink. com/publications/CHI2020-ThermalWear.pdf
- [107] K. Suhonen, K. Väänänen-Vainio-Mattila, and K. Mäkelä, "User Experiences and Expectations of Vibrotactile, Thermal and Squeeze Feedback in Interpersonal Communication," 2012.

- [108] S. Song, G. Noh, J. Yoo, I. Oakley, J. Cho, and A. Bianchi, "Hot & tight: Exploring thermo and squeeze cues recognition on wrist wearables," *ISWC 2015 - Proceedings* of the 2015 ACM International Symposium on Wearable Computers, pp. 39–42, 2015.
- [109] Y. Osawa and S. Katsura, "Wearable Thermal Interface for Sharing Palm Heat Conduction," *IECON 2018 - 44th Annual Conference of the IEEE Industrial Electronics Society*, vol. 1, pp. 3310–3315, 2018.
- [110] D. Peña and F. Tanaka, "Validation of the Design of a Robot to Study the Thermo-Emotional Expression," *ICSR 2018: Social Robotics*, pp. 1–10, 2018. [Online]. Available: https://link.springer.com/chapter/10.1007/978-3-030-05204-1_8
- [111] E. Park and J. Lee, "I am a warm robot : the effects of temperature in physical human robot interaction," vol. 32, no. 2014, pp. 133–142, 2018.
- [112] M. Nakashige, M. Kobayashi, Y. Suzuki, H. Tamaki, and S. Higashino, ""Hiya-Atsu" Media: Augmenting Digital Media with Temperature," *Proceedings* of the 27th international conference extended abstracts on Human factors in computing systems - CHI EA '09, p. 3181, 2009. [Online]. Available: http://portal.acm.org/citation.cfm?doid=1520340.1520453
- [113] I. Politis, S. Brewster, and F. Pollick, "To Beep or Not to Beep?: Comparing Abstract versus Language-Based Multimodal Driver Displays," *Proceedings of the* ACM CHI'15 Conference on Human Factors in Computing Systems, vol. 1, no. c, pp. 3971–3980, 2015. [Online]. Available: http://dx.doi.org/10.1145/2702123.2702167
- [114] D. Napoli, S. N. Chaparro, S. Chiasson, and E. Stobert, "Something Doesn't Feel Right: Using Thermal Warnings to Improve User Security Awareness," USENIX Symposium on Usable Privacy and Security (SOUPS), pp. 3–7, 2020.
- [115] R. Poguntke, J. Ilk, A. Schmidt, and Y. Abdelrahman, "Designing Thermal Feedback for Notifying Users About Stress," 2019.
- [116] M. Umair, C. Sas, and M. H. Latif, "Towards Affective Chronometry : Exploring Smart Materials and Actuators for Real - time Representations of Changes in Arousal," *DIS*'19, 2019.
- [117] M. Umair, C. Sas, and M. Alfaras, "ThermoPixels: Toolkit for Personalizing Arousal-based Interfaces through Hybrid Crafting," *Proceedings of the 2020 on Designing Interactive Systems Conference - DIS '20*, pp. 1017–1032, 2020. [Online]. Available: https://doi.org/10.1145/3357236.3395512

- [118] J. Trojan, X. Fuchs, S.-L. Speth, and M. Diers, "The rubber hand illusion induced by visual-thermal stimulation," *Scientific Reports*, vol. 8, no. 1, p. 12417, 2018.
 [Online]. Available: http://www.nature.com/articles/s41598-018-29860-2
- [119] A. D. Blaga, I. Williams, and C. Creed, "Too Hot to Handle : An Evaluation of the Effect of Thermal Visual Representation on User Grasping Interaction in Virtual Reality," pp. 1–16, 2020.
- [120] B. Kleinsman-leusink hill and J. Bronlund, "Thermal and Haptic Interface Design : Adding sensory feedback to VR / AR," 2018 IEEE International Instrumentation and Measurement Technology Conference (I2MTC), pp. 1–6, 2018.
- [121] S. W. Kim, S. H. Kim, C. S. Kim, K. Yi, J. S. Kim, B. J. Cho, and Y. Cha, "Thermal display glove for interacting with virtual reality," *Scientific Reports*, vol. 10, no. 1, pp. 1–12, 2020. [Online]. Available: https://doi.org/10.1038/s41598-020-68362-y
- [122] R. L. Peiris, W. Peng, Z. Chen, L. Chan, and K. Minamizawa, "ThermoVR: Exploring Integrated Thermal Haptic Feedback with Head Mounted Displays," *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17*, pp. 5452–5456, 2017. [Online]. Available: http://dl.acm.org/citation.cfm?doid=3025453. 3025824
- [123] M. Y. Saraiji, R. L. Peiris, L. Shen, K. Minamizawa, and S. Tachi, "Ambient: Facial Thermal Feedback in Remotely Operated Applications," *Extended Abstracts of the 2018 CHI Conference on Human Factors in Computing Systems CHI '18*, pp. 1–4, 2018. [Online]. Available: http://dl.acm.org/citation.cfm?doid=3170427.3186483
- [124] G. Chinazzo, K. Chamilothori, J. Wienold, M. Andersen, and D. Lausanne, "Temperature – Color Interaction : Subjective Indoor Environmental Perception and Physiological Responses in Virtual Reality," 2013.
- [125] N. Ranasinghe, P. Jain, N. T. N. Tram, D. Tolley, Y. Liangkun, C. E. W. Tung, C. C. Yen, E. Y. L. Do, K. C. R. Koh, and K. Shamaiah, "A demonstration of season traveller: Multisensory narration for enhancing the virtual reality experience," *Conference on Human Factors in Computing Systems - Proceedings*, vol. 2018-April, pp. 1–13, 2018.
- [126] B. Deml, A. Mihalyi, and G. Hanning, "Development and Experimental Evaluation of a Thermal Display," *EuroHaptics 2006*, 2006. [Online]. Available: https://athene-forschung.unibw.de/doc/128632/128632.pdf
- [127] X. Guo, Y. Zhang, W. Wei, W. Xu, and D. Wang, "ThermalTex : A Two-Modal Tactile Display for Delivering Surface Texture and Thermal Information,"

EuroHaptics 2020, vol. 1, pp. 288–296, 2020. [Online]. Available: https://doi.org/10.1007/978-3-030-58147-3_32)

- [128] W. Peng, R. L. Peiris, and K. Minamizawa, "Exploring of simulating passing through feeling on the wrist: Using thermal feedback," UIST 2017 Adjunct - Adjunct Publication of the 30th Annual ACM Symposium on User Interface Software and Technology, pp. 187–188, 2017.
- [129] K. Kushiyama, M. Inose, R. Yokomatsu, K. Fujita, T. Kitazawa, M. Tamura, and S. Sasada, "Thermoesthesia: About collaboration of an artist and a scientist," ACM SIGGRAPH 2006 Educators program on - SIGGRAPH '06, p. 23, 2006. [Online]. Available: http://portal.acm.org/citation.cfm?doid=1179295.1179319
- [130] K. Kushiyama, T. Baba, K. Doi, and S. Sasada, "Thermo-Pict neo," *SIGGRAPH 2010*, no. 1, p. 4503, 2010.
- [131] K. Kushiyama, "Thermal Design Display device to use the thermal tactile illusions," SIGGRAPH 2010, no. 1, pp. 4503–4503, 2010.
- [132] K. D. Tetsuaki Baba, Kumiko Kushiyama, "ThermoGame : Video Game Interaction System that Offers Dynamic Temperature Sensation to Users," *Acm Siggraph 2010*, vol. 82, p. 4503, 2010.
- [133] M. Löchtefeld, T. Lappalainen, J. Väyrynen, A. Colley, and J. Häkkilä, "Comparing Thermal and Haptic Feedback Mechanisms for Game Controllers," pp. 1829–1836, 2017.
- [134] J. Tewell, J. Bird, and G. R. Buchanan, "Heat-Nav: Using Temperature Changes as Navigational Cues," *Proceedings of the 2017 CHI Conference on Human Factors in Computing Systems - CHI '17*, pp. 1131–1135, 2017. [Online]. Available: http://dl.acm.org/citation.cfm?doid=3025453.3025965
- [135] T. Maeda, "TherModule : Wearable and Modular Thermal Feedback System based on a Wireless Platform," pp. 1–8, 2018.
- [136] A. Niijima, "ThermalBitDisplay : Haptic Display Providing Thermal Feedback Perceived Differently Depending on Body Parts," pp. 1–8, 2020.
- [137] S. Kim and Y.-k. Row, "Thermal Interaction with a Voice-based Intelligent Agent the prototype system," *Extended Abstracts of the 2018 CHI Conference on Human Factors in Computing Systems*, pp. 1–6, 2018.

- [138] Q. Zhang, Y. Ge, W. Qu, D. A. Hennessy, and K. Zhang, "Effects of anger and collision history on driver space preference," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 63, pp. 108–117, 2019. [Online]. Available: https://doi.org/10.1016/j.trf.2019.04.002
- [139] R. L. Peiris and L. Chan, "ThermalBracelet : Exploring Thermal Haptic Feedback Around the Wrist," *CHI19*, pp. 1–11, 2019.
- [140] T. Narumi, A. Tomohiro, Y. A. Seong, and M. Hirose, "Characterizing the space by thermal feedback through a wearable device," *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics*), vol. 5622 LNCS, pp. 355–364, 2009.
- [141] A. Nasser, K. Zhu, and S. Wiseman, "Thermo-haptic Earable Display for the Hearing and Visually Impaired," *ASSETS*, pp. 630–632, 2019.
- [142] F. Kato, Y. Inoue, and S. Tachi, "Haptic Display Glove Capable of Force / Vibration / Temperature," *IEEE International Symposium on Measurement* and Control in Robotics (ISMCR), pp. 1–5, 2019. [Online]. Available: https: //doi.org/10.1109/ISMCR47492.2019.8955735
- [143] M. Nakatani, K. Sato, K. Sato, Y. Kawana, D. Takai, K. Minamizawa, and S. Tachi,
 "A Novel Multimodal Tactile Module that Can Provide Vibro-Thermal Feedback," pp. 1–15, 2018.
- [144] X. Guo, Y. Zhang, S. Member, W. Wei, W. Xu, D. Wang, and S. Member, "Effect of Temperature on the Absolute and Discrimination Thresholds of Voltage on Electrovibration Tactile Display," *IEEE Transactions on Haptics*, vol. PP, p. 1, 2019.
- [145] L. A. Jones and A. Singhal, "Vibrotactile Pattern Identification in a Multisensory Display," *EuroHaptics 2018*, vol. 10893, pp. 389–400, 2018. [Online]. Available: https://doi.org/10.1007/978-3-319-93445-7_35
- [146] P. Danca, A. Vartires, and A. Dogeanu, "An Overview of Current Methods for Thermal Comfort Assessment in Vehicle Cabin," *Energy Procedia*, vol. 85, no. November 2015, pp. 162–169, 2016. [Online]. Available: https://core.ac.uk/ download/pdf/82036743.pdf
- [147] R. Abou Jaoude, I. Thiagalingam, K. R. El, and G. Crehan, "Berkeley thermal comfort models : Comparison to people votes and indications for user-centric HVAC strategies in car cabins," *Building and Environment*, vol. 180, no. June, p. 107093, 2020. [Online]. Available: https://doi.org/10.1016/j.buildenv.2020.107093

- [148] L. Qi, J. Liu, L. Zhang, and Q. Wu, "Study on local thermal sensation and model applicability in vehicle cabin under different driving states," *Heat and Mass Transfer*, 2020. [Online]. Available: https://doi.org/10.1007/s00231-020-02933-7
- [149] Y. He, J. Yang, J. Ling, Y. Du, and Z. Zhang, "Predictive modeling for overall thermal sensation of vehicle occupants based on local thermal sensation when warming up," *Proc IMechE Part D: J Automobile Engineering*, 2020.
- [150] E. Schmidt and A. C. Bullinger, "Mitigating passive fatigue during monotonous drives with thermal stimuli: Insights into the effect of different stimulation durations," *Accident Analysis and Prevention*, vol. 126, no. December 2017, pp. 115–121, 2017. [Online]. Available: https://www.sciencedirect.com/science/article/ abs/pii/S0001457517304438
- [151] E. Schmidt, R. Decke, R. Rasshofer, and A. C. Bullinger, "Psychophysiological responses to short-term cooling during a simulated monotonous driving task," *Applied Ergonomics*, vol. 62, pp. 9–18, 2017. [Online]. Available: http: //dx.doi.org/10.1016/j.apergo.2017.01.017
- [152] L. L. Di Stasi, E. Gianfranchi, and C. Diaz-Piedra, "Hand-Skin Temperature Response to Driving Fatigue : An Exploratory Study," in *HCII 2020*, vol. 2. Springer International Publishing, 2020, pp. 3–14. [Online]. Available: http://dx.doi.org/10.1007/978-3-030-50537-0_1
- [153] B. C. Lee, B. J. Martin, and K. H. Sienko, "The effects of actuator selection on non-volitional postural responses to torso-based vibrotactile stimulation," *Journal of NeuroEngineering and Rehabilitation*, vol. 10, no. 1, p. 21, 2013. [Online]. Available: JournalofNeuroEngineeringandRehabilitation
- [154] NHTSA, "Driver Distraction Program," U.S. Department of Transportation: National Highway Traffic Safety Administration, pp. 1–36, 2010.
- [155] J. D. Lee, K. Young, and M. Regan, "Defining Driver Distraction," in *Driver Distraction: Theory, Effects, and Mitigation*, 2008, pp. 31–40.
- [156] J. D. Lee, "Dynamics of Driver Distraction: The process of engaging and disengaging." Annals of advances in automotive medicine. Association for the Advancement of Automotive Medicine. Annual Scientific Conference, vol. 58, pp. 24–32, 2014. [Online]. Available: http://www.ncbi.nlm.nih.gov/pubmed/24776224% 0Ahttp://www.pubmedcentral.nih.gov/articlerender.fcgi?artid=PMC4001670

- [157] K. Hurts, L. S. Angell, and M. A. Perez, "The Distracted Driver: Mechanisms, Models, and Measurement," *Reviews of Human Factors and Ergonomics*, vol. 7, no. 1, pp. 3–57, 2011.
- [158] A. Farooq, G. Evreinov, and R. Raisamo, Providing Comprehensive Navigational Cues Through the Driving Seat to Reduce Visual Distraction in Current Generation of Semi-autonomous Vehicles. Springer International Publishing, 2020. [Online]. Available: http://dx.doi.org/10.1007/978-3-030-39512-4_135
- [159] L. Graichen, M. Graichen, and J. F. Krems, "Evaluation of Gesture-Based In-Vehicle Interaction: User Experience and the Potential to Reduce Driver Distraction," *Human Factors: The Journal of the Human Factors and Ergonomics Society*, p. 001872081882425, 2019. [Online]. Available: http://journals.sagepub.com/doi/10. 1177/0018720818824253
- [160] T. a. Dingus, S. G. Klauer, V. L. Neale, A. Petersen, S. E. Lee, J. Sudweeks, M. a. Perez, J. Hankey, D. Ramsey, S. Gupta, C. Bucher, Z. R. Doerzaph, J. Jermeland, and R. Knipling, "The 100-Car naturalistic driving study phase II Results of the 100-Car field experiment," *Dot Hs 810 593*, no. April, pp. No. HS–810 593, 2006.
- [161] M. Pettitt, G. Burnett, and A. Stevens, "Defining driver distraction," Intelligent Transportation Society of America - 12th World Congress on Intelligent Transport Systems 2005, vol. 5, no. 0, pp. 2856–2867, 2009.
- [162] Q. Wu, "An overview of driving distraction measure methods," Proceeding 2009 IEEE 10th International Conference on Computer-Aided Industrial Design and Conceptual Design: E-Business, Creative Design, Manufacturing - CAID and CD'2009, pp. 2391–2394, 2009.
- [163] A. Riener, M. Jeon, I. Alvarez, and A.-K. Frison, "Driver in the Loop: Best Practices in Automotive Sensing and Feedback Mechanisms," in *Automotive User Interfaces*, 2017, pp. 295–323. [Online]. Available: http://link.springer.com/10.1007/ 978-3-319-49448-7
- [164] T. A. Dingus, F. Guo, S. Lee, J. F. Antin, M. Perez, M. Buchanan-King, and J. Hankey, "Driver crash risk factors and prevalence evaluation using naturalistic driving data," *Proceedings of the National Academy of Sciences of the United States of America*, vol. 113, no. 10, pp. 2636–2641, 2016.
- [165] J. K. Caird, S. M. Simmons, K. Wiley, K. A. Johnston, and W. J. Horrey, "Does Talking on a Cell Phone, With a Passenger, or Dialing Affect Driving Performance? An Updated Systematic Review and Meta-Analysis of Experimental Studies," *Human Factors*, vol. 60, no. 1, pp. 101–133, 2018.

- [166] Y. Peng, L. N. Boyle, and S. L. Hallmark, "Driver's lane keeping ability with eyes off road: Insights from a naturalistic study," *Accident Analysis and Prevention*, vol. 50, pp. 628–634, 2013. [Online]. Available: http://dx.doi.org/10.1016/j.aap.2012.06.013
- [167] T. Schaap, A. van der Horst, B. van Arem, and K. Brookhuis, "The Relationship between Driver Fatigue and Driver Distraction," 2008, pp. 383–392.
- [168] N. Schaap, R. van der Horst, B. van Arem, and K. Brookhuis, "The Relationship between Driver Distraction and Mental Workload," in *Driver Distraction and Inattention: Advances in Research and Countermeasures*, 2013.
- [169] N. I. Abd Rahman, S. Z. Md Dawal, and N. Yusoff, "Driving mental workload and performance of ageing drivers," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 69, pp. 265–285, 2020. [Online]. Available: https://doi.org/10.1016/j.trf.2020.01.019
- [170] S. Hart and L. E. Staveland, "Development of NASA- TLX (Task Load Index): Results of Empirical and Theoretical Research," *Human Mental Workload*, pp. 239–250, 1988. [Online]. Available: https://doi.org/10.1016/S0166-4115(08)62386-9
- [171] D. K. McGookin and S. Brewster, "Understanding concurrent earcons," ACM Transactions on Applied Perception, vol. 1, no. 2, pp. 130–155, 2004. [Online]. Available: http://portal.acm.org/citation.cfm?doid=1024083.1024087
- [172] T. S. J. L. Hancock, Peter A. Oron-Gilad, "Elaborations of the Multiple-Resource Theory of Attention," in *Attention: From Theory to Practice*, A. F. K. D. A. W. A. Kirlik, Ed., 2007, ch. 4, pp. 45–56.
- [173] R. W. Mann and S. D. Reimers, "Kinesthetic Sensing for the EMG Controlled "Boston Arm"," *IEEE Transactions on Man-Machine Systems*, vol. 11, no. 1, pp. 110–115, 2000. [Online]. Available: http://citeseerx.ist.psu.edu/viewdoc/summary? doi=10.1.1.133.5356
- [174] K. Majlund Bach, M. G. Jæger, M. B. Skov, and N. G. Thomassen, "Interacting with In-Vehicle Systems: Understanding, Measuring, and Evaluating Attention," vol. 83, no. 8, pp. 3956–67, 2009. [Online]. Available: http://www.pubmedcentral.nih.gov/ articlerender.fcgi?artid=2663281&tool=pmcentrez&rendertype=abstract
- [175] R. McCall, F. McGee, A. Meschtscherjakov, N. Louveton, and T. Engel, "Towards A Taxonomy of Autonomous Vehicle Handover Situations," *Proceedings of the* 8th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - Automotive'UI 16, pp. 193–200, 2016. [Online]. Available: http://dl.acm.org/citation.cfm?doid=3003715.3005456

- [176] A. G. Mirnig, M. Gärtner, A. Laminger, A. Meschtscherjakov, S. Trösterer, M. Tscheligi, R. McCall, and F. McGee, "Control Transition Interfaces in Semiautonomous Vehicles," *Proceedings of the 9th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - AutomotiveUI* '17, pp. 209–220, 2017. [Online]. Available: http://dl.acm.org/citation.cfm?doid= 3122986.3123014
- [177] J. B. van Erp, K.-U. Kyung, S. Kassner, J. Carter, S. Brewster, G. Weber, and I. Andrew, "Setting the Standards for Haptic and Tactile Interactions: ISO's Work," in *Haptics: Generating and Perceiving Tangible Sensations*, 2010.
- [178] M. Steele and R. B. Gillespie, "Shared Control Between Humans and Machine: Using a Haptic Steering Wheel to Aid in Land Vehicle Guidance," *HUMAN FACTORS AND ERGONOMICS SOCIETY 45th ANNUAL MEETING*, pp. 1671–1675, 2001.
- [179] N. Jordan, M. Franck, and H. Jean-Michel, "Lateral control support for car drivers," *Proceedings of the 14th European conference on Cognitive ergonomics invent! explore! - ECCE '07*, no. August, p. 249, 2007. [Online]. Available: http://portal.acm.org/citation.cfm?doid=1362550.1362601
- [180] J. Navarro, F. Mars, J.-F. Forzy, M. El-Jaafari, and J.-M. Hoc, "Objective and subjective assessment of warning and motor priming assistance devices in car driving," *Proceedings of European Conference on Human Centred Design for Intelligent Transport Systems*, pp. 311–320, 2008. [Online]. Available: https://www.hfes-europe.org/wp-content/uploads/2014/06/21-fa.pdf
- [181] F. Beruscha, K. Augsburg, and D. Manstetten, "Haptic warning signals at the steering wheel: a literature survey regarding lane departure warning systems," *Haptics-e*, vol. 4, no. 5, pp. 1–6, 2011. [Online]. Available: http://www.haptics-e.org/Vol_05/he-v5n1.pdf
- [182] M. Itoh, T. Inagaki, and H. Tanaka, "Haptic steering direction guidance for pedestrianvehicle collision avoidance," *Conference Proceedings - IEEE International Conference on Systems, Man and Cybernetics*, pp. 3327–3332, 2012.
- [183] M. Mulder, D. A. Abbink, and E. R. Boer, "Sharing control with haptics: Seamless driver support from manual to automatic control," *Human Factors*, vol. 54, no. 5, pp. 786–798, 2012.
- [184] Z. Huang, Y. Wu, and J. Liu, "Research on effects of pattern, amplitude and frequency of pulse steering torque warnings for lane departure," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 31, pp. 67–76, 2015. [Online]. Available: http://dx.doi.org/10.1016/j.trf.2015.03.008

- [185] M. Johns, B. Mok, D. Sirkin, N. Gowda, C. Smith, W. Talamonti, and W. Ju, "Exploring shared control in automated driving," 2016 11th ACM/IEEE International Conference on Human-Robot Interaction (HRI), pp. 91–98, 2016. [Online]. Available: http://ieeexplore.ieee.org/document/7451738/
- [186] M. Johns, G. Strack, and W. Ju, "Driver Assistance after Handover of Control from Automation," pp. 2104–2110, 2018.
- [187] J. Roozendaal, E. Johansson, J. de Winter, D. Abbink, and S. Petermeijer, "Haptic lane keeping assistance for truck driving: A test track study," *HUMAN FACTORS*, vol. 0, no. 0, pp. 1–18, 2020. [Online]. Available: https://doi.org/10.1177%2F0018720820928622
- [188] Y. Saito and P. Raksincharoensak, "Effect of risk-predictive haptic guidance in one-pedal driving mode," *Cognition, Technology & Work*, no. 0123456789, 2019.
 [Online]. Available: http://link.springer.com/10.1007/s10111-019-00558-3
- [189] J. B. van Erp and H. A. van Veen, "Vibro-Tactile Information Presentation in Automobiles," *EuroHaptics 2001*, no. July, pp. 99–104, 2001. [Online]. Available: http://citeseerx.ist.psu.edu/viewdoc/download?doi=10.1.1.92.319&rep= rep1&type=pdf
- [190] A. Farooq, G. Evreinov, R. Raisamo, E. Mäkinen, T. Nukarinen, and A. A. Majeed, "Developing novel multimodal interaction techniques for touchscreen in-vehicle infotainment systems," *ICOSST 2014 - 2014 International Conference on Open Source Systems and Technologies, Proceedings*, pp. 32–42, 2014.
- [191] K. Harrington, D. R. Large, G. Burnett, and O. Georgiou, "Exploring the Use of Mid-Air Ultrasonic Feedback to Enhance Automotive User Interfaces," *Proceedings* of the 10th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - AutomotiveUI '18, pp. 11–20, 2018. [Online]. Available: https://dl.acm.org/citation.cfm?id=3239089
- [192] J. B. van Erp, "Guidelines for the Use of Vibro-Tactile Displays in Human Computer Interaction," *Proceedings of Eurohaptics*, pp. 18–22, 2002.
- [193] K. S. Hale and K. M. Stanney, "Deriving haptic design guidelines from human physiological, psychophysical, and neurological foundations," *IEEE Computer Graphics* and Applications, vol. 24, no. 2, pp. 33–39, 2004.
- [194] S. Subramanian, C. Gutwin, M. N. M. Sanchez, C. Power, and J. Liu, "Haptic and Tactile Feedback in Directed Movements," *Proceedings of Guidelines On*

Tactile and Haptic Interactions, vol. 2, no. January, pp. 37–42, 2005. [Online]. Available: http://www.cs.usask.ca/research/research_groups/userlab/GOTHI/ CompleteProceedingsofGOTHI-05.pdf#page=38

- [195] J. B. van Erp, "Presenting directions with a vibrotactile torso display," *Ergonomics*, vol. 48, no. 3, pp. 302–313, 2005.
- [196] O. Schneider, K. MacLean, C. Swindells, and K. Booth, "Haptic experience design: What hapticians do and where they need help," *International Journal of Human Computer Studies*, vol. 107, no. August 2016, pp. 5–21, 2017. [Online]. Available: http://dx.doi.org/10.1016/j.ijhcs.2017.04.004
- [197] L. R. Elliott, B. J. P. Mortimer, R. A. Pomranky-hartnett, F. Rapozo, R. A. Pettitt, R. E. Wooldridge, A. Rapozomeran, and G. R. Mort, *Tactile Cues : Taction Characteristics , Salience , Ease of Learning , and Recall*, 2019.
- [198] S. Brewster and L. M. Brown, "Tactons: structured tactile messages for non-visual information display," AUIC'04 Proceedings of the fifth conference on Australasian user interface, pp. 15–23, 2004. [Online]. Available: www.dcs.gla.ac.uk/~stephen
- [199] E. Piateski and L. A. Jones, "Vibrotactile pattern recognition on the arm and torso," Proceedings - 1st Joint Eurohaptics Conference and Symposium on Haptic Interfaces for Virtual Environment and Teleoperator Systems; World Haptics Conference, WHC 2005, vol. 38, no. 1, pp. 90–95, 2005.
- [200] L. M. Brown, S. Brewster, and H. C. Purchase, "Multidimensional Tactons for Non-Visual Information Presentation in Mobile Devices," *Proceedings of the 8th conference on Human-computer interaction with mobile devices and services -MobileHCI '06*, vol. 159, no. September, pp. 231–238, 2006. [Online]. Available: http://portal.acm.org/citation.cfm?doid=1152215.1152265
- [201] H. Seifi and K. E. Maclean, "A first look at individuals' affective ratings of vibrations," 2013 World Haptics Conference, WHC 2013, pp. 605–610, 2013.
- [202] E. T. Chancey, J. C. Brill, A. Sitz, U. Schmuntzsch, and J. P. Bliss, "Vibrotactile stimuli parameters on detection reaction times," *Proceedings of the Human Factors and Ergonomics Society*, vol. 2014-Janua, no. 1, pp. 1701–1705, 2014. [Online]. Available: http://pro.sagepub.com/lookup/doi/10.1177/1541931214581355
- [203] A. Riener, Sensor-Actuator Supported Implicit Interaction in Driver Assistance Systems, 2011. [Online]. Available: https://books.google.de/books?id= wHEGcYXGonAC

- [204] S. Trento, A. de Götzen, and S. Serafi, "CARess, a Gentle Touch Informs the Driver," in *EuroHaptics 2014*, 2014, vol. 8618, pp. 10–17. [Online]. Available: http://link.springer.com/10.1007/978-3-662-44193-0
- [205] O. Halabi, M. Ba Hameish, L. Al-Naimi, and A. Al-Kaabi, "A study on the design and effectiveness of tactile feedback in driving simulator," *Proceedings of IEEE/ACS International Conference on Computer Systems and Applications, AICCSA*, vol. 2014, pp. 742–746, 2014.
- [206] M. Gong, B. Zhao, and X. Wang, "Vibration Information Acquisition and Representation Control Strategy for Construction Vehicle Driver Seat," vol. 2019, 2019.
- [207] K. Tsukada and M. Yasumura, "ActiveBelt: Belt-Type Wearable Tactile Display for Directional Navigation," *UbiComp 2004*, pp. 384–399, 2004. [Online]. Available: http://link.springer.com/10.1007/978-3-540-30119-6_23
- [208] R. Gray, C. Ho, and C. Spence, "A comparison of different informative vibrotactile forward collision warnings: Does the warning need to be linked to the collision event?" *PLoS ONE*, vol. 9, no. 1, 2014.
- [209] F. Meng, R. Gray, C. Ho, M. Ahtamad, and C. Spence, "Dynamic vibrotactile signals for forward collision avoidance warning systems," *Human Factors*, vol. 57, no. 2, pp. 329–346, 2015. [Online]. Available: https://doi.org/10.1177%2F0018720814542651
- [210] G. Shakeri, A. Ng, J. H. Williamson, and S. Brewster, "Evaluation of Haptic Patterns on a Steering Wheel," *Proceedings of the 8th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - Automotive'UI* 16, pp. 129–136, 2016. [Online]. Available: http://dl.acm.org/citation.cfm?doid= 3003715.3005417
- [211] A. Meschtscherjakov, "The Steering Wheel: A Design Space Exploration," *Automotive User Interfaces*, vol. 54, no. 4, pp. 155–156, 2017.
- [212] R. Grimm and A. Meyer-Damcke, "Steering wheel, the central interface between driver and vehicle in case of autonomous driving," 7th International Munich Chassis Symposium 2016, pp. 527–543, 2017. [Online]. Available: http://link.springer.com/10.1007/978-3-658-18459-9
- [213] D. A. Abbink, E. R. Boer, and M. Mulder, "Motivation for continuous haptic gas pedal feedback to support car following," *IEEE Intelligent Vehicles Symposium, Proceedings*, pp. 283–290, 2008.

- [214] M. Mulder, M. Mulder, M. M. van Paassen, and D. A. Abbink, "Haptic gas pedal feedback," *Ergonomics*, vol. 51, no. 11, pp. 1710–1720, 2008.
- [215] S. A. Birrell, M. S. Young, and A. M. Weldon, "Delivering smart driving feedback through a haptic pedal," *Contemporary Ergonomics and Human Factors 2010*, pp. 431–439, 2010. [Online]. Available: http://www.scopus.com/inward/record.url?eid= 2-s2.0-84859575810&partnerID=40&md5=1794cea8bbe05b7a92e8674b6c7d19cc
- [216] C. Geitner, S. Birrell, L. Skrypchuk, C. Krehl, A. Mouzakitis, and P. Jennings, "Good Vibrations – Driving with a Haptic Pedal," *Adjunct Proceedings of the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - AutomotiveUI '15*, pp. 100–105, 2015. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2809730.2809742
- [217] S. M. Petermeijer, D. A. Abbink, M. Mulder, and J. C. De Winter, "The Effect of Haptic Support Systems on Driver Performance: A Literature Survey," *IEEE Transactions* on *Haptics*, vol. 8, no. 4, pp. 467–479, 2015.
- [218] A. Lécuyer, "The Use of Haptic and Tactile information in the Car to improve Driving Safety : A Review of Current Technologies," *Frontiers in ICT*, vol. 5, no. March, pp. 1–11, 2018.
- [219] S. J. Breitschaft, S. Clarke, and C.-c. Carbon, "A Theoretical Framework of Haptic Processing in Automotive User Interfaces and Its Implications on Design and Engineering," *Frontiers in Psychology*, vol. 10, no. July, 2019.
- [220] J. B. van Erp, L. C. M. Kroon, T. Mioch, and K. I. Paul, "Obstacle Detection Display for Visually Impaired: Coding of Direction, Distance, and Height on a Vibrotactile Waist Band," *Frontiers in ICT*, vol. 4, no. October, pp. 1–19, 2017. [Online]. Available: http://journal.frontiersin.org/article/10.3389/fict.2017.00023/full
- [221] S. Cœugnet, A. Dommes, S. Panëels, A. Chevalier, F. Vienne, N. T. Dang, and M. Anastassova, "Helping older pedestrians navigate unknown environments through vibrotactile guidance instructions," *Transportation Research Part F: Traffic Psychol*ogy and Behaviour, vol. 58, pp. 816–830, 2018.
- [222] A. Montuwy, B. Cahour, and A. Dommes, "Using Sensory Wearable Devices to Navigate the City: Effectiveness and User Experience in Older Pedestrians," *Multimodal Technologies and Interaction*, vol. 3, no. 1, p. 17, 2019. [Online]. Available: https://www.mdpi.com/2414-4088/3/1/17

- [223] A. Bajpai, J. C. Powell, A. J. Young, and A. Mazumdar, "Enhancing Physical Human Evasion of Moving Threats Using Tactile Cues," *IEEE Transactions on Haptics*, vol. PP, no. c, p. 1, 2019.
- [224] M. Arikawa, S. Konomi, and K. Ohnishi, "NaviTime: Supporting Pedestrian Navigation in the Real World," *Context*, pp. 21–29, 2007.
- [225] M. Pielot, B. Poppinga, and S. Boll, "PocketNavigator: Vibro-Tactile Waypoint Navigation for Everyday Mobile Devices," *Proceedings of the 12th international conference on Human computer interaction with mobile devices and services -MobileHCI '10*, p. 423, 2010. [Online]. Available: http://portal.acm.org/citation.cfm? doid=1851600.1851696
- [226] S. Robinson, M. Jones, P. Eslambolchilar, R. Murray-Smith, and M. Lindborg, "I did it my way: moving away from the tyranny of turn-by-turn pedestrian navigation," *Proceedings of Mobile HCI*, pp. 1–4, 2010. [Online]. Available: http://dl.acm.org/citation.cfm?id=1851660
- [227] M. Prathilothamai, P. R. Nair, R. A. P. Singh, and P. N. Aditya, "Offline navigation: Gps based location assisting system," *Indian Journal of Science* and Technology, vol. 9, no. 45, pp. 1–6, 2016. [Online]. Available: http: //www.indjst.org/index.php/indjst/article/view/99707
- [228] S. Fagernes and T.-m. Grønli, "Navigation for Visually Impaired Using Haptic Feedback," in *Human-Computer Interaction. Interaction in Context.* Springer International Publishing, 2018, vol. 10902, pp. 347–356. [Online]. Available: http://link.springer.com/10.1007/978-3-319-91244-8
- [229] F. Kiss, R. Boldt, B. Pfleging, and S. Schneegass, "Navigation Systems for Motorcyclists : Exploring Wearable Tactile Feedback for Route Guidance in the Real World," *Proceedings of the SIGCHI Conference on Human Factors in Computing Systems -CHI*'18, pp. 1–7, 2018.
- [230] J. B. van Erp, "Vibrotactile waypoint navigation at sea and in the air: two case studies," *Proceedings of ...*, pp. 166–173, 2004. [Online]. Available: http://www.eurohaptics.vision.ee.ethz.ch/2004/15f.pdf
- [231] J. B. van Erp, H. a. H. C. van Veen, T. Dobbins, and C. Jansen, "Waypoint navigation with a vibrotactile waist belt," *ACM Transactions on Applied Perception*, vol. 2, no. 2, pp. 106–117, 2005.

- [232] Y. Salzer, T. Oron-Gilad, and A. Ronen, "Vibrotactor-Belt on the Thigh–Directions in the Vertical Plane," *Proceedings of EuroHaptics 2010*, pp. 359–364, 2010. [Online]. Available: http://www.springerlink.com/index/P3017547606167LV.pdf
- [233] A. Asif and S. Boll, "Where to turn my car?: comparison of a tactile display and a conventional car navigation system under high load condition," *Proceedings of the International Conference on Automotive User Interfaces and Interactive Vehicular Applications (AutomotiveUI '10)*, no. AutomotiveUI, pp. 64–71, 2010. [Online]. Available: http://portal.acm.org/citation.cfm?doid=1969773.1969786
- [234] F. Beruscha, L. Wang, K. Augsburg, and H. Wandke, "Do drivers steer towards or away from lateral directional vibrations at the steering wheel?" *Proc. 2nd European Conference on Human Centred Design for Intelligent Transport Systems*, pp. 227–236, 2010.
- [235] S. Kim, J. H. Hong, K. A. Li, J. Forlizzi, and A. K. Dey, "Route guidance modality for elder driver navigation," *Lecture Notes in Computer Science (including subseries Lecture Notes in Artificial Intelligence and Lecture Notes in Bioinformatics)*, vol. 7319 LNCS, pp. 179–196, 2012.
- [236] C. J. Ploch, J. H. Bae, W. Ju, and M. Cutkosky, "Haptic Skin Stretch on a Steering Wheel for Displaying Preview Information in Autonomous Cars," *International Conference on Intelligent Robots and Systems (IROS)*, 2016.
- [237] A. Asif, W. Heuten, and S. Boll, "Exploring distance encodings with a tactile display to convey turn by turn information in automobiles," *Proceedings of the 6th Nordic Conference on Human-Computer Interaction Extending Boundaries NordiCHI '10*, p. 32, 2010. [Online]. Available: http://portal.acm.org/citation.cfm?doid=1868914. 1868923
- [238] J. B. van Erp and P. Werkhoven, "Validation of principles for tactile navigation displays," *Proceedings of the Human Factors and Ergonomics Society*, pp. 1687–1691, 2006.
- [239] T. Grah, F. Epp, M. Wuchse, A. Meschtscherjakov, F. Gabler, A. Steinmetz, and M. Tscheligi, "Dorsal Haptic Display: A Shape-changing Car Seat for Sensory Augmentation of Rear Obstacles," in *AutomotiveUI15*, 2015, pp. 305–312. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2799250.2799281
- [240] A. Riener, A. Ferscha, P. Frech, M. Hackl, and M. Kaltenberger, "Subliminal vibrotactile based notification of CO 2 economy while driving," no. AutomotiveUI, p. 92, 2010.

- [241] R. Mohebbi, R. Gray, and H. Z. Tan, "Driver Reaction Time to Tactile and Auditory Rear-End Collision Warnings While Talking on a Cell Phone," *Human Factors, Vol. 51, No. 1, February*, vol. 51, no. 1, p. 8700, 2009. [Online]. Available: https://doi.org/10.1177/0018720809333517
- [242] C. Ho, V. Santangelo, and C. Spence, "Multisensory warning signals: When spatial correspondence matters," *Experimental Brain Research*, vol. 195, no. 2, pp. 261–272, 2009.
- [243] J. Lylykangas, V. Surakka, K. Salminen, A. Farooq, and R. Raisamo, "Responses to visual, tactile and visual-tactile forward collision warnings while gaze on and off the road," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 40, pp. 68–77, 2016. [Online]. Available: http://dx.doi.org/10.1016/j.trf.2016.04.010
- [244] K. Reinmueller and M. Steinhauser, "Adaptive forward collision warnings: The impact of imperfect technology on behavioral adaptation, warning effectiveness and acceptance," *Accident Analysis and Prevention*, vol. 128, no. December 2018, pp. 217–229, 2019. [Online]. Available: https://doi.org/10.1016/j.aap.2019.04.012
- [245] M. Enriquez, O. Afonin, B. Yager, and K. Maclean, "A Pneumatic Tactile Alerting System for the Driving Environment," *Proceedings of the 2001 workshop* on Percetive user interfaces - PUI '01, pp. 1–7, 2001. [Online]. Available: http://portal.acm.org/citation.cfm?doid=971478.971506
- [246] G. Shakeri, S. Brewster, J. Williamson, and A. Ng, "Evaluating Haptic Feedback on a Steering Wheel in a Simulated Driving Scenario," *Proceedings of the* 2016 CHI Conference Extended Abstracts on Human Factors in Computing Systems - CHI EA '16, pp. 1744–1751, 2016. [Online]. Available: http: //dl.acm.org/citation.cfm?doid=2851581.2892497
- [247] G. Shakeri, J. H. Williamson, and S. Brewster, "Novel Multimodal Feedback Techniques for In-Car Mid-Air Gesture Interaction," *Proceedings of the 9th International Conference on Automotive User Interfaces and Interactive Vehicular Applications -AutomotiveUI* '17, pp. 84–93, 2017.
- [248] B. A. Lewis and C. L. Baldwin, "Equating perceived urgency across auditory, visual, and tactile signals," *Proceedings of the Human Factors and Ergonomics Society*, pp. 1307–1311, 2012.
- [249] C. L. Baldwin, J. L. Eisert, A. Garcia, B. Lewis, S. M. Pratt, and C. Gonzalez, "Multimodal urgency coding: Auditory, visual, and tactile parameters and their impact on perceived urgency," *Work*, vol. 41, no. SUPPL.1, pp. 3586–3591, 2012.

- [250] M. Martens and W. Van Winsum, "Effects of speech versus tactile driver support messages on driving behavior and workload," SAE technical papers, pp. 6–80, 2001. [Online]. Available: https://pdfs.semanticscholar.org/d162/ f70339ecaaddea423dae839b18c19c3a7efc.pdf
- [251] S. M. Petermeijer, F. Doubek, and J. de Winter, "Driver response times to auditory, visual, and tactile take-over requests: A simulator study with 101 participants," 2017 IEEE International Conference on Systems, Man, and Cybernetics, SMC 2017, vol. 2017-Janua, pp. 1505–1510, 2017.
- [252] I. Politis, S. Brewster, and F. Pollick, "Using multimodal displays to signify critical handovers of control to distracted autonomous car drivers," *International Journal of Mobile Human Computer Interaction*, vol. 9, no. 3, pp. 1–16, 2017.
- [253] C. Geitner, F. Biondi, L. Skrypchuk, P. Jennings, and S. Birrell, "The comparison of auditory, tactile, and multimodal warnings for the effective communication of unexpected events during an automated driving scenario," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 65, pp. 23–33, 2019. [Online]. Available: https://linkinghub.elsevier.com/retrieve/pii/S1369847818305473
- [254] S. H. Yoon, Y. W. Kim, and Y. G. Ji, "The effects of takeover request modalities on highly automated car control transitions," *Accident Analysis and Prevention*, vol. 123, no. September 2017, pp. 150–158, 2019. [Online]. Available: https://doi.org/10.1016/j.aap.2018.11.018
- [255] M. Schwalk, N. Kalogerakis, and T. Maier, "Driver Support by a Vibrotactile Seat Matrix – Recognition, Adequacy and Workload of Tactile Patterns in Take-over Scenarios During Automated Driving," *Procedia Manufacturing*, vol. 3, no. Ahfe, pp. 2466–2473, 2015. [Online]. Available: http://dx.doi.org/10.1016/j.promfg.2015. 07.507
- [256] A. Telpaz, B. Rhindress, I. Zelman, and O. Tsimhoni, "Haptic Seat for Automated Driving: Preparing the Driver to Take Control Effectively," *Proceedings of the 7th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - AutomotiveUI '15*, pp. 23–30, 2015. [Online]. Available: http://dl.acm.org/citation.cfm?doid=2799250.2799267
- [257] S. Sadeghian Borojeni, L. Chuang, W. Heuten, and S. Boll, "Assisting Drivers with Ambient Take-Over Requests in Highly Automated Driving," *Proceedings of* the 8th International Conference on Automotive User Interfaces and Interactive Vehicular Applications - Automotive'UI 16, pp. 237–244, 2016. [Online]. Available: http://dl.acm.org/citation.cfm?doid=3003715.3005409

- [258] S. M. Petermeijer, S. Cieler, and J. C. de Winter, "Comparing spatially static and dynamic vibrotactile take-over requests in the driver seat," *Accident Analysis and Prevention*, vol. 99, pp. 218–227, 2016. [Online]. Available: http://dx.doi.org/10.1016/j.aap.2016.12.001
- [259] A. P. van den Beukel, M. C. van der Voort, and A. O. Eger, "Supporting the changing driver's task: Exploration of interface designs for supervision and intervention in automated driving," *Transportation Research Part F: Traffic Psychology and Behaviour*, vol. 43, pp. 279–301, 2016. [Online]. Available: http://dx.doi.org/10.1016/j.trf.2016.09.009
- [260] S. M. Petermeijer, S. Cieler, and J. C. de Winter, "Comparing spatially static and dynamic vibrotactile take-over requests in the driver seat," *Accident Analysis and Prevention*, vol. 99, pp. 218–227, 2017. [Online]. Available: http://dx.doi.org/10.1016/j.aap.2016.12.001
- [261] N. Merat, A. H. Jamson, F. C. H. Lai, and O. Carsten, "Highly automated driving, secondary task performance, and driver state," *Human Factors*, vol. 54, no. 5, pp. 762–771, 2012.
- [262] C. Gold, M. Körber, C. Hohenberger, D. Lechner, and K. Bengler, "Trust in Automation – Before and After the Experience of Take-over Scenarios in a Highly Automated Vehicle," *Procedia Manufacturing*, vol. 3, no. Ahfe, pp. 3025–3032, 2015. [Online]. Available: http://dx.doi.org/10.1016/j.promfg.2015.07.847
- [263] S. Stockert, N. T. Richardson, and M. Lienkamp, "Driving in an Increasingly Automated World – Approaches to Improve the Driver-automation Interaction," *Procedia Manufacturing*, vol. 3, pp. 2889–2896, 2015.
- [264] K. Zeeb, A. Buchner, and M. Schrauf, "What determines the take-over time? An integrated model approach of driver take-over after automated driving," *Accident Analysis* and Prevention, vol. 78, pp. 212–221, 2015.
- [265] T. J. Wright, S. Samuel, A. Borowsky, S. Zilberstein, and D. L. Fisher, "Experienced drivers are quicker to achieve situation awareness than inexperienced drivers in situations of transfer of control within a level 3 autonomous environment," *Proceedings of the Human Factors and Ergonomics Society*, pp. 270–273, 2016.
- [266] M. Albert, A. Lange, A. Schmidt, M. Wimmer, and K. Bengler, "Automated Driving – Assessment of Interaction Concepts Under Real Driving Conditions," *Procedia Manufacturing*, vol. 3, no. Ahfe, pp. 2832–2839, 2015. [Online]. Available: http://dx.doi.org/10.1016/j.promfg.2015.07.767

- [267] P. Kerschbaum, L. Lorenz, and K. Bengler, "A transforming steering wheel for highly automated cars," *IEEE Intelligent Vehicles Symposium, Proceedings*, vol. 2015-Augus, no. Iv, pp. 1287–1292, 2015.
- [268] G. Wilson, S. Brewster, M. Halvey, and S. Hughes, "Thermal feedback identification in a mobile environment," 8th International Workshop on Haptic and Audio Interaction Design, 2013. [Online]. Available: https://doi.org/10.1007/ 978-3-642-41068-0_2